Development of Innovative Pretreatments for Simultaneous Saccharification and Citric Acid Production from Banana Pseudostem: Bioprocess Optimization and Kinetic Assessment

By

Milesh Laltha

Student number: 216011104 BSc. Hons. Microbiology

Submitted in fulfilment of the academic requirements for the degree of Master of Science

In Microbiology



School of Life Sciences

College of Agriculture, Engineering and Science

University of KwaZulu-Natal

Pietermaritzburg

South Africa

19 July 2022

PREFACE

The research contained in this thesis/dissertation was completed by the candidate (Milesh Laltha, 216011104) while based in the Discipline of Microbiology, School of Life Sciences, of the College of Agriculture, Engineering and Science, University of KwaZulu-Natal, Pietermaritzburg Campus, South Africa. The research was financially supported by the National Research Foundation (Grant number: 122341).

The contents of this work have not been submitted in any form to another university, and except where the work of others is acknowledged in the text, the results reported are due to investigations by the candidate.

Signed: Professor Evariste Bosco Gueguim Kana (Supervisor)

Date:

Date:

Signed: Doctor Yeshona Sewsynker-Sukai (Co-supervisor)

I, Milesh Laltha, declare that:

(i) the research reported in this dissertation, except where otherwise indicated or acknowledged, is my original work;

(ii) this dissertation has not been submitted in full or in part for any degree or examination to any other university;

(iii) this dissertation does not contain other persons' data, pictures, graphs or other information, unless specifically acknowledged as being sourced from other persons;

(iv) this dissertation does not contain other persons' writing, unless specifically acknowledged as being sourced from other researchers. Where other written sources have been quoted, then:

a) their words have been re-written but the general information attributed to them has been referenced;

b) where their exact words have been used, their writing has been placed inside quotation marks, and referenced;

(v) where I have used material for which publications followed, I have indicated in detail my role in the work;

(vi) this dissertation is primarily a collection of material, prepared by myself, published as journal articles or presented as a poster and oral presentations at conferences. In some cases, additional material has been included;

(vii) this dissertation does not contain text, graphics or tables copied and pasted from the Internet, unless specifically acknowledged, and the source being detailed in the dissertation and in the References sections.



Signed: Milesh Laltha Date: 06 October 2022

DECLARATION 2: PUBLICATIONS AND MANUSCRIPTS

This thesis represents a compilation of a published paper and two manuscripts (under review). Details of contributions to a publication that form part and/or include research presented in this thesis are provided within each publication and manuscript under the Credit Author Statement section. The student (Milesh Laltha) contributed towards experimental work, data collection and manuscript preparation, under the guidance of Professor E.B. Gueguim Kana (Supervisor) and Doctor Y. Sewsynker-Sukai (Co-supervisor).

1. Laltha, M., Sewsynker-Sukai, Y., Meyer, E.L., Gueguim Kana, E.B. 2022. Banana pseudostem as a renewable feedstock for microbial-based value-added products. Bioresource Technology. (Chapter 2 - Under review)

 Laltha, M., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2022. Innovative microwave-assisted iodized table salt or paper wastewater pretreatments for enhanced sugar recovery from banana pseudostem. *Biomass Conversion and Biorefinery*, 1-15.
DOI: <u>https://doi.org/10.1007/s13399-022-02630-x</u> (Chapter 3)

3. Laltha, M., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2022. Simultaneous saccharification and citric acid production from paper wastewater pretreated banana pseudostem: Optimization of fermentation medium formulation and kinetic assessment. *Bioresource Technology*, 1-10. DOI: <u>https://doi.org/10.1016/j.biortech.2022.127700</u> (Chapter 4)



Signed: Milesh Laltha Date: 06 October 2022 1. Laltha, M. Valorization of lignocellulosic waste for itaconic acid and bioethanol production. College of Agriculture, Engineering and Science. Postgraduate research and innovation symposium (PRIS). 11 November 2020. University of KwaZulu-Natal, South Africa. Oral Presentation.

ABSTRACT

The bioconversion of renewable lignocellulosic biomass into value-added products such as citric acid (CA) has become a research hotspot towards a sustainable economy. Approximately 2 million tons of CA are produced annually with widespread use in the food, chemical, pharmaceutical and textile industries. Nevertheless, the use of lignocellulosic-derived feedstocks for CA production requires expensive, energy- and resource intensive pretreatment and fermentation processes, that result in low product yields. Therefore, the present study explores the valorization of banana pseudostem (BP) for CA production using the simultaneous saccharification and fermentation (SSF) process to alleviate these challenges.

A review on the potential of BP as a feedstock for microbial production of value-added products is discussed together with the pretreatment and bioprocess challenges in addition to the possibilities to overcome these bottlenecks. Subsequently, two novel pretreatments consisting of a: (1) microwave-assisted-iodized table salt (M-ITS) and (2) microwave-assisted-paper wastewater (M-PWW) were modelled and optimized using the response surface methodology (RSM) for the enhancement of sugar recovery from BP. Then, the SSF process with pretreated BP was modelled and optimized using RSM for CA production. Thereafter, the logistic and modified Gompertz models were used to assess the kinetics of *Aspergillus brasiliensis* growth and CA production, respectively.

The pretreatment input parameters for the M-ITS model comprised of salt concentration (1-5%, w/v), microwave power intensity (100-900 W) and pretreatment time (2-10 min). On the other hand, the M-PWW model consisted of substrate solid loading (10-30% w/v), microwave power intensity (100-900 W) and pretreatment time (2-10 min). The output responses were

reducing sugar yield (RSY) and glucose yield (GY) for both models. The optimal pretreatment conditions predicted by the M-ITS model were 2.48% ITS concentration, 800 W (power intensity) and 10 minutes (pretreatment time), corresponding to an experimental RSY (0.515 g/g) and GY (0.433 g/g). Pretreatment and optimization revealed a slightly higher RSY and GY for the M-ITS strategy when compared to the M-PWW method (RSY = 0.498 g/g and GY = 0.413 g/g). The M-PWW model predicted optimal pretreatment conditions of 30% solid loading, 800 W for 8 minutes. Despite the higher sugar yields observed for the M-ITS pretreatment, the M-PWW pretreatment represents a more suitable strategy owing to its complete waste-based approach that generates three times the quantity of pretreated substrate required for subsequent fermentation.

SSF optimization included nitrogen (ammonium nitrate) concentration (0.5-5 w/v), desorbent (acetone) concentration (1-5% v/v) and temperature (30-40 °C) as inputs with CA concentration as the output (g/L). The optimized SSF conditions (0.5% w/v ammonium nitrate, 1% v/v acetone and 35 °C) (SSF_{optimizedFW}) gave a maximum CA concentration of 14.408 g/L. For the kinetic experiments, three bioprocesses consisting of the: (1) optimized SSF process with freshwater (SSF_{optimizedFW}), (2) SSF_{optimizedFW} process conditions while substituting dairy wastewater in place of freshwater (SSF_{DWW}), and (3) Sabouraud Dextrose Emmon's medium modified by substituting glucose with BP (SSF_{SDEmodified}), were assessed. While all three bioprocesses gave the same maximum specific growth rate (μ_{max}) of 0.05 h⁻¹, the SSF_{SDEmodified} process resulted in the highest maximum potential CA concentration (P_m) (13.991 g/L) compared to the SSF_{DWW} (P_m= 13.095 g/L) and SSF_{optimizedFW} (P_m= 12.967 g/L) systems.

The developed pretreatment strategies demonstrated glucose yields >0.40 g/g, shown to be higher than previously established chemical pretreatments. Moreover, the SSF_{DWW} process

displayed a slightly lower P_m value compared to the SSF_{SDEmodified} strategy, and interestingly a higher P_m value than the SSF_{optimizedFW} bioprocess. Major findings from this study paves the way for lignocellulosic bioprocesses by potentially negating the use of costly pretreatment chemicals, fermentation medium constituents and/or fresh water, while achieving a "waste treating waste" approach. Thus, contributing to the water-energy-food nexus in line with global sustainable development goals towards a circular bioeconomy.

Keywords: Banana pseudostem; pretreatment strategies; fermentation; citric acid; kinetics.

I am grateful that God almighty has bestowed virtuous health, physical and mental strength, and a strong ability to overcome this chapter in my life. I am blessed by my ancestors, who have guided me on a righteous path, and I will ensure that their legacy lives on.

I would like to extend sincere gratitude to my supervisor Prof. Kana for providing me with the opportunity, research environment and moral support. Thank you for enlightening my understanding of concepts and reminding me 'to think outside of the box.' It was a great privilege to be your student.

To my co-supervisor, Yeshona, you have played a significant role in making this research project a reality. Your aptitude, guidance and passion as a research scientist has shaped me into a proud microbiologist. Thank you for inspiring me to be conscientious while providing valuable feedback to get through difficult situations. I will never forget your invaluable teachings.

To my dearest colleagues of Lab 117, thank you all for your support, kindness and motivation. Isaac, thank you for always being there, especially on those late nights and early mornings when nobody else was around. Your intellect, generosity and selflessness have taught me to be more appreciative of life. To my parents, I am who I am because of you. I am thankful for the unending love, support and strength that you have showered upon me. Your ability, belief and dedication to 'working smarter and not harder' has brushed upon me to do the same. I will never forget your wise words, "Knowledge is the greatest power that nobody can take away from you."

To my sister Sam, thank you for being the best sibling one could ask for. I am appreciative of your kindness, love and support. I will always look up to you. To my brother-in-law, Ashveer, thank you for being a respectable and kind-hearted individual. Your addition to our family has brought tremendous happiness to our lives.

To my other half Tejal, thank you for all that you do, for motivating and encouraging me to further my studies. It was not easy, but we made it together. You bring out the best version I could possibly be, and I cannot wait to see where life takes us, knowing that we shall make many more memorable moments together. I also give my gratitude to the Bhagwandeen family for your love and support.

I would like to acknowledge the National Research Foundation (NRF) for financially supporting this research and the University of KwaZulu-Natal for providing laboratory facilities and equipment to facilitate this research.

To all, even those that have not been mentioned, thank you for sharing your time and spirit with me on this lifelong journey. In reminiscence, I cherish the valuable memories that we have shared together.

TABLE OF CONTENTS

Section

Preface	i
Declaration 1: Plagiarism	ii
Declaration 2: Publications	iii
Conference contribution	iv
Abstract	v
Acknowledgments	viii
Table of contents	x
List of tables	xii
List of figures	xiv
List of abbreviations	xvi
CHAPTER 1: General Introduction	1
1.1. The importance for renewable biobased products	2
1.2. The significance of lignocellulosic substrates in biorefineries	2
1.3. The requirement for lignocellulosic pretreatment regimes	4
1.4. Bioprocessing of valuable renewable organic acids	5
1.5. The food, energy and water nexus approach	7
1.6. Development of innovative bioprocess strategies utilizing industrial wastes	9
1.7. Bioprocess optimization and kinetic modelling	11
1.8. Research motivation/Problem statement	13
1.9. Aims and objectives	14
2. Outline of thesis structure	16
References	17
CHAPTER 2: Banana pseudostem as a renewable feedstock for microbial-base	d value-
added products	25
1. Introduction	27
2. Lignocellulosic biomass	
2.1. Banana pseudostem as a suitable feedstock for valuable biofuels and biochemic	cals31
2.2. Limitations of lignocellulosic biomass	
2.3. The significance of lignocellulosic pretreatment	
3. Lignocellulosic pretreatment regimes on Banana pseudostem	
3.1. Physical pretreatment	

3.2. Chemical pretreatment	
3.3. Physicochemical pretreatments	46
3.4. Biological pretreatment	49
3.5. Combined pretreatment strategies	
4. Lignocellulosic bioprocess types	
4.1. Development of valuable bio-based products	59
5. Current challenges and future prospectives	60
5.1. Lignocellulosic biomass selection	60
5.2. Development of cost-effective and/or waste-based pretreatments	61
5.3. Lignocellulosic bioprocess optimization	
6. Conclusion	65
References	66
Supplementary material	
CHAPTER 3: Innovative microwave-assisted iodized table salt or pap	er wastewater
pretreatments for enhanced sugar recovery from banana pseudostem	
1. Introduction	
2. Materials and methods	
3. Results and discussion	91
4. Conclusion	100
References	100
Supplementary material	
CHAPTER 4: Simultaneous saccharification and citric acid productio	n from paper
wastewater pretreated banana pseudostem: Optimization of ferments	ation medium
formulation and kinetic assessment	112
1. Introduction	114
2. Materials and methods	119
3. Results and discussion	
4. Conclusion	144
References	145
Supplementary material	151
CHAPTER 5: Conclusions and recommendations for further research	158
5.1. Conclusions	158
5.2. Recommendations for future studies	
Appendix A	

ppendix B

LIST OF TABLES

Table	Page no.

CHAPTER 2:

Table 1. Composition of various lignocellulosic biomass feedstocks
Table 2. Studies investigating the potentiality of banana pseudostem for bioproducts
corresponding to the various input process conditions
Table 3. The mode of action of various pretreatment strategies in combination with their
advantages and disadvantages
Table 4. The application of diverse pretreatment strategies on banana pseudostem and similar
derivatives of the banana plant
Supplementary material:
Table S1. Optimal fermentation conditions and concentration of citric acid, itaconic acid and
lactic acid production from various studies
Table S2. Pretreatment applications of waste by-products obtained from the Kraft paper and
pulp industry

CHAPTER 3:

Table 1. Reducing sugar and glucose yields observed for the M-ITS pretreatments91
Table 2. Reducing sugar and glucose yields observed for the M-PWW pretreatments92
Table 3. RSM polynomial model equations relating the input parameters to the output
reducing sugar and glucose yields for the M-ITS and M-PWW pretreatments93
Table 4. Summary of analysis of variance (ANOVA) for the developed pretreatment models

Supplementary material:

Table S1. Sample analysis of the major constituents present in paper wastewater102
Table S2. Preliminary screening of M-NaCl, M-ITS and M-PWW to determine their
pretreatment effectiveness on banana pseudostem103
Table S3. Analysis of variance of the developed ITS_{RSY} pretreatment model on banana
pseudostem104
Table S4. Analysis of variance of the developed ITSGY pretreatment model on banana
pseudostem

Table S5. Analysis of variance of the developed PWW _{RSY} pretreatment model on banana
pseudostem106
Table S6. Analysis of variance of the developed PWW_{GY} pretreatment model on banana
pseudostem107
Table S7. Validation of optimized conditions for the M-ITS and M-PWW pretreatment on
banana pseudostem
Table S8. Structural composition of the control (untreated) and optimally pretreated banana
pseudostem108
Table S9. Comparison of the developed pretreatment strategies on various lignocellulosic
substrates

CHAPTER 4:

Table 1. Citric acid concentration determined for the SSF process on inputs of acetone,
ammonium nitrate concentration and temperature123
Supplementary material:
Table S1. Preliminary screening of various desorbents and nitrogen sources to determine their
effectiveness for citric acid production151
Table S2. Citric acid concentration determined for the SSF process using the Megazyme
assay kit152
Table S3. Analysis of variance (ANOVA) of the developed SSF model for citric acid
production153
Table S4. Comparative assessment of citric acid derived from lignocellulosic and dairy
wastes utilizing Aspergillus sp154
Table S5. Kinetic parameters of the microbial growth and citric acid production for the
SSF _{optimizedFW} , SSF _{DWW} and SSF _{SDEmodified} processes

LIST OF FIGURES

Figure

CHAPTER 1:

Fig. 1. Structure of lignocellulosic biomass and its biopolymeric constituents4
Fig. 2. Schematic representation of the main metabolic reactions involved in the production
of citric acid. $PFK = phosphofructokinase$, $PC = pyruvate carboxylase$, $ACO = aconitase7$
Fig. 3. The interactions and synergy of the food, energy and water nexus
Fig. 4. Projected water scarcity by 2025, indicating "physical water scarcity" occurring in
South Africa, and in the equatorial plane9
Fig. 5. Circular economy illustrating the bioconversion of agro-industrial wastes to value-
added products

CHAPTER 2:

Fig. 1. A banana tree consisting of its several parts including the banana pseudostem 36

CHAPTER 3:

Supplementary material:

Fig. S1. Response surface graphs illustrating the interactive effects of the various input pretreatment parameters on the output reducing sugar and glucose yields: (A) pretreatment time and salt concentration (M-ITS_{RSY}); (B) pretreatment time and salt concentration (M-

ITS_{GY} ; (C) pretreatment time and power intensity (M-PWW _{RSY}); (D) pretreatment time and
power intensity (M-PWW _{GY})11
$Fig. \ S2. \ FTIR \ spectra \ of \ the \ untreated, \ M-ITS_{optimized} \ and \ M-PWW_{optimized} \ banana \ pseudoster \ and \ banana \ pseudoster \ and \ banana \ banana \ pseudoster \ and \ banana \ banana \ banana \ pseudoster \ banana \ ba$

CHAPTER 4:

Fig. 1. Three-dimensional response surface graphs showing the interactive effects of the (A)
Ammonium nitrate concentration (%) and acetone concentration (%), (B) Temperature (°C)
and acetone concentration (%), and (C) Temperature (°C) and ammonium nitrate
concentration (%) on the citric acid concentration135
Fig. 2. Aspergillus brasiliensis ATCC16404 (A) biomass concentration, (B) glucose
$utilization \ and \ (C) \ citric \ acid \ concentration, \ for \ the \ SSF_{optimizedFw}, \ SSF_{Dww} \ and \ SSF_{SDEmodified}$
processes

LIST OF ABBREVIATIONS

AFEX	Ammonia fibre explosion
ANOVA	Analysis of Variance
ATCC	American Type Culture Collection
BP	Banana pseudostem
CA	Citric acid
СМС	Carboxymethyl cellulose
CSL	Corn steep liquor
DNA	Deoxyribonucleic acid
DNS	3,5-dinitrosalicyclic acid
DWW	Dairy wastewater
Eq.	Equation
FPU	Filter paper unit
FTIR	Fourier-Transform Infrared spectroscopy
F-value	Fisher-Snedecor distribution value
GLD	Green liquor dregs
GY	Glucose yield
IL	Ionic liquid
ITS	Iodized table salt
LCB	Lignocellulosic biomass
LHW	Liquid hot water
M-	Microwave-assisted
МНР	Microwave heating processes
MI	Microwave irradiation
NAD/H	Nicotinamide adenine dinucleotide/hydrogen
рН	Potential of hydrogen
Pm	Potential maximum citric acid concentration
РРО	Polyphenol oxidase
PSSF	Pre-hydrolysis SSF
PWW	Paper wastewater
p-value	Probability value
r p,m	Maximum citric acid production rate

R ²	Coefficient of determination
RSM	Response surface methodology
RSY	Reducing sugar yield
S/L	Solid to liquid ratio
SDE	Sabouraud Dextrose Emmon's
SEM	Scanning electron microscopy
SHF	Separate hydrolysis and fermentation
sp.	Species
SSF	Simultaneous saccharification and fermentation
t	Time
TCA	The citric acid cycle
tL	Lag time
USD	United States dollar
v	Volume
w	Weight
X ₀	Initial cell concentration
X _{max}	Maximum cell concentration
β	Beta
$\mu_{\rm max}$	Maximum specific growth rate

CHAPTER 1

General Introduction

This chapter provides a general outline of the current research and states the aims and objectives to be carried out. Additionally, it elaborates on the development of novel bioprocess strategies, the suitability of utilizing industrial wastes, the generation of citric acid, and optimization and kinetic modelling. Some of these aspects have not been covered in the literature review (Chapter 2). The literature review has been submitted to Bioresource Technology and focuses on innovative pretreatment strategies applied to renewable banana pseudostem for microbial-based value-added products.

1.1. The importance for renewable biobased products

The depletion of sustainable resources has been a fundamental worldwide concern for commercial industries. Over the last three decades, the demand for natural resources has continuously increased due to the exponential population and economic growth rate (Mabhaudhi et al., 2018). The current process for producing versatile commodities such as organic acids and fossil fuels (coal, oil and gas) from finite non-renewable resources are no longer sustainable due to the exhaustion of these sources (Sharma et al., 2021). In addition, these processes result in the accumulation of non-biodegradable and hazardous wastes (toxins and leachates) that are harmful both to human health and to the environment (landfills and water bodies) resulting in pollution and subsequent global warming (Raj et al., 2022). To resolve these problems, carbon-neutral biorefineries are emerging as a promising alternative to support the sustainable production of biobased industrial products such as polymers, chemicals and fuels from renewable lignocellulosic biomass (Singh et al., 2022). Furthermore, for the transition to a complete circular economy, raw materials and wastes from industrial processes are considered ideal for producing essential biobased materials (Ashokkumar et al., 2022). The commercial scale lignocellulosic bioconversion to organic acids is not yet fully established due to several factors that requires optimization to reduce the bioprocessing costs during lignocellulosic pretreatment, enzymatic saccharification and microbial fermentation. To achieve low-cost bioprocesses, efficient microbial consortia capable of producing metabolites in large quantities are employed for the utilization of renewable and cheap lignocellulosic biomass-derived sugars (Ashokkumar et al., 2022).

1.2. The significance of lignocellulosic substrates in biorefineries

In recent times, agro-industrial and crop residual wastes have garnered substantial attention as sustainable raw materials for the generation of marketable products such as organic acids, without posing ethical issues to food security (Kostas et al., 2021). Renewable lignocellulosic biomass is one of the most abundant and economical resources available with an annual production rate estimated at 181.5 billion tons, while only 8.2 billion tons of biomass is currently utilized (Singh et al., 2022). Lignocellulosic feedstocks comprise of polysaccharides that include cellulose and hemicellulose, which are rich in hexose (glucose and fructose) and pentose (xylose and arabinose), intertwined by fractions of lignin moieties (Hernández Beltrán et al., 2019) (Fig. 1). The valorization and bioconversion of crop residues to value-added organic acids are carried out through a series of established processes that encompass pretreatment, saccharification and fermentation. Pretreatment is implemented to overcome the inherent high degree of recalcitrance, followed by saccharification to fermentable sugars that can be consumed by the microorganisms during the fermentation process (Shimizu et al., 2018). Recently, studies have reported on the utilization of lignocellulosic waste residues and include bamboo (Laltha et al., 2021), corn cobs (David et al., 2020), banana pseudostem (Legodi et al., 2021), sugarcane bagasse (Jugwanth et al., 2020), potato peel waste (Sanusi et al., 2019) and Napier grass (Mohammed et al., 2019), among others. More specifically, banana pseudostem represents an attractive, nonedible, cellulosic feedstock that has shown to be suitable for the production of biofuels and biochemicals. The chemical composition of banana pseudostem is estimated to consist of a high cellulose (42-60%), moderate hemicellulose (19-23%) and a low lignin content (15-17%) (Legodi et al., 2021, Islam et al., 2019). The global annual production of banana pseudostem amounts to approximately 300 million tons (Subagyo and Chafidz, 2018). Post harvesting of the banana fruit leaves behind banana pseudostem that consists of about 75% of the total waste that is generated. This waste is either left to rot at the local dumpsite prior to incineration resulting in the release of unpleasant greenhouse gases such as hydrogen sulphide and ammonia gas in addition to the spread of Panama disease caused by the soil-borne fungus Fusarium oxysporum (Costa et al., 2018). Therefore, the utilization of banana pseudostem will lead to the production of valuable commodities such as citric acid and also overcome problems that are associated with its disposal.



Fig. 1. Structure of lignocellulosic biomass and its biopolymeric constituents (Adapted from Hernández Beltrán et al. 2019).

1.3. The requirement for lignocellulosic pretreatment

Pretreatment is a rate-limiting step that is vital for altering the structure of LCB. By reducing the lignin content, pretreatment enables the access of cellulose and hemicellulose to saccharifying enzymes during enzymatic hydrolysis for enhanced sugar recovery that are required during microbial bioprocesses (Hernández Beltrán et al., 2019). Common pretreatment strategies that have previously been assessed include alkali, acid, microwave, inorganic salts, steam explosion and ammonia fibre explosion, among others (Yu et al., 2019). From the aforementioned methods, alkaline pretreatments have been sought after attributable to the favourable reaction conditions, low reaction pressures and temperatures, less corrosive chemicals, ability to recover leftover alkali and the use of simplified reactors (Phitsuwan et al., 2016). Nevertheless, these methods are limited by expensive chemical catalysts, requirement for specialized equipment and the utilization of vast amounts of fresh water during pretreatment steps and in between wash cycles (Badiei et al., 2014). Therefore, pretreatment strategies that minimize energy and fresh water usage, do not pose a threat to the environment, utilize costeffective chemicals that are non-corrosive, minimize the formation of inhibitor compounds and enhance the availability of fermentable sugars for microbial fermentation are imperative (Hernández Beltrán et al., 2019, Baruah et al., 2018). In this regard, our recent study investigated two different alkaline (NaOH and Na₃PO₄.12H₂O) pretreatments on bamboo and corn cobs and determined their potential recyclability (Laltha et al., 2021) (Appendix A). Findings from the aforementioned study indicated that the alkaline spent hydrolysate could be recycled up to three times, thereby minimizing the pretreatment costs by reducing the consumption of expensive alkaline chemicals and fresh water. However, sequential pretreatment processes are still faced with the initial chemical and water costs and/or usage, which is considered an imperative bottleneck in the present time (Laltha et al., 2021).

1.4. Bioprocessing of valuable renewable organic acids

Biobased chemicals with three-carbon backbones that include citric acid, lactic acid and itaconic acid have garnered significant attention in academia and industry (Song et al., 2022, Son et al., 2022). This is due to their wide range of applications in the pharmaceutical, manufacturing, agrochemical and textile industries (Behera et al., 2021). Citric acid (CA) also

known as 2-hydroxy-1,2,3-propane tricarboxylic acid is a weak organic acid that has emerged as an important organic acid owing to its palatability, pleasant taste and low toxicity (Ayeni et al., 2019). Moreover, CA is also used as a chelating agent and detergent for metal finishing, cleaning, lubrication, animal feed and as a plasticizer (Show et al., 2015). Approximately 2 million tons of CA is generated on an annual basis while the rate of demand is expected to rise by 5% each year (Ozdal et al., 2019). The global market for CA is expected to exceed \$3.2 billion USD by 2023 (Behera et al., 2021). Additionally, the cost of CA production via chemical synthesis is approximately ~\$10 USD per kg, whereas for fungal fermentation it has been estimated to be ~50% lower (Ozdal et al., 2019). In industry, CA is mainly produced by the filamentous fungus, Aspergillus niger (Adeoye and Lateef, 2022). In addition, bacteria (Bacillus sp., Brevibacterium sp., and Pseudomonas sp.) and yeasts (Candida sp., and Yarrowia sp.) are capable of producing citric acid (Tong et al., 2019). CA is produced via the citric acid cycle (TCA) through various metabolic pathways (Fig. 2). CA biosynthesis is initiated by the uptake of glucose which undergoes glycolysis to produce two molecules of pyruvate. Pyruvate molecules are oxidized and combined with coenzyme A to form CO₂, Acetyl-CoA and NADH. Acetyl-co-A combines with oxaloacetate to produce citric acid, yielding two molecules of CO₂, Adenosine Triphosphate (ATP) and four-carbon oxaloacetate again (Behera et al., 2021). The expensive and inefficient conversion technologies of biobased organic acid production hinder the commercial establishment of biobased products (Guragain and Vadlani, 2021). In this regard, biobased organic acids can be viewed as economically competitive only if their production costs are made more affordable by utilizing agro-industrial processes (Chan et al., 2022).



Fig. 2. Schematic representation of the main metabolic reactions involved in the production of citric acid. PFK = phosphofructokinase, PC = pyruvate carboxylase, ACO = aconitase (Adapted from Behera et al., 2021).

1.5. The food, energy and water nexus approach for a circular bioeconomy

Water, energy and food are three of the key pillars upon which humanity exists (Mabhaudhi et al., 2018). Recently, the food, energy and water nexus has attracted significant attention by the political, social, and scientific community (Harwood, 2018). The food, energy and water nexus approach offer several opportunities for improving security while contributing to a green economy (Fernández-Ríos et al., 2021). The approach considers the interactions, synergies and opportunities of the linked securities, to optimize the use of these resources, whereby usage within one sector influences the use and availability in the adjacent sectors (Fernández-Ríos et al., 2021) (Fig. 3).



Fig. 3. The interactions and synergy of the food, energy and water nexus (Adapted from Fernández-Ríos et al., 2021).

Water scarcity has been forecasted to intensify within the next few years, resulting in a 60% increase in water demand by 2025 (Fig. 4) (Ghazal and Kazmi, 2014). Water is essential for agricultural and industrial practices, contributing to one of the main costs incurred. Additionally, most forms of energy production are made possible by the incorporation of water. For example, water is required to support livelihoods, fisheries, irrigation, wastewater treatment and food production. Global food production accounts for an estimated 70% of fresh water use, while primary energy production and power generation amount to 10% of water utilization (Fernández-Ríos et al., 2021). Hence, there is a need for efficient alternatives to conventional fresh water.



Fig. 4. Projected water scarcity by 2025, indicating "physical water scarcity" occurring in South Africa and in countries coinciding in the equatorial plane (Adapted from Ghazal and Kazmi, 2014).

1.6. Development of innovative bioprocess strategies utilizing industrial wastewater

The Kraft pulp and paper industry is one of the most important industrial sectors in the world (Manasa and Mehta, 2020). Nevertheless, pulp and paper mills are faced with challenges due to energy intensive processes that utilize vast amounts of fresh water in turn generating alkaline paper wastewater (PWW) that has adverse effects on the aquatic environment (Kamali and Khodaparast, 2015). The characteristics of these waste by-products depend on various parameters that include, type of raw material employed, processing technique and property of the desired end-product (Mäkelä et al., 2016). The application of PWW as a chemical and water source during lignocellulosic pretreatment processes may drastically reduce costs and could potentially generate profits that can be channelled towards less invasive biohazardous management, thereby lowering the downstream costs for Kraft industries. Kraft PWW exhibits strong alkali chemicals that lead to the degradation of ester and glycosidic bonds between

hemicellulose and ferulic acid in the cell wall of lignocellulosic biomass, resulting in the swelling of cellulose, hemicellulose solvation, and lignin removal (Costa et al., 2018). PWW serve as a dual-purpose commodity serving as an alternative to fresh water and also for the replacement of expensive lab-derived alkaline reagents (NaOH). In the same vein, the exploitation of cheaper alternatives (iodized table salt) to conventional lab-based chemicals (NaCl) would be economically beneficial for pretreatment processes thereby reducing expenditure for lignocellulosic biorefineries. Recently, inorganic salts have gained significant attention due to their ability to increase cellulose and hemicellulose conversion rates and enhance enzymatic hydrolysis yields (Liu et al., 2009). Inorganic salts function by dissociating Lewis acids in an aqueous solvent to form hydrated complex cations surrounded by water molecules. These water molecules further act as nucleophiles, thus depolymerizing hemicellulose into monosaccharides (Moodley et al., 2020).

In addition to PWW as a pretreatment alternative, another promising waste effluent that can be used in bioprocessing is dairy wastewater (DWW). Water plays a significant role in the dairy industry and is mainly used for cleaning, washing, disinfection, heating and cooling (Slavov, 2017). In recent years, the global production of the dairy industry has increased, resulting in large amounts of discharged wastewater effluent. DWW comprises of abundant nutrients, rich in lactose, proteins, fats, vitamins, and minerals, which can lead to severe eutrophication if disposed of without prior treatment (Kusmayadi et al., 2022). Eutrophication leads to many water quality problems including increased purification costs and the possible sublethal effects of algal toxins on animals and humans (Prazeres et al., 2020). Hence, the application of DWW in bioprocesses could alleviate water security while avoiding its detrimental effects on ecosystems. DWW like PWW serves a dual-purpose in biorefinery systems since this waste can be used as a replacement of fresh water in fermentation reactors and simultaneously

provide nutrients for growth and metabolism of the fermentative microorganism (Son et al., 2022). Furthermore, the utilization of DWW will enable the dairy industry to avoid treatment costs, thus synergistically saving time and capital expenses for both industries. Therefore, the concept of utilizing waste effluent from industry for value-added products will contribute to reducing production costs while alleviating the current water crisis.

1.7. Bioprocess optimization and kinetic modelling

The success of bioconversion processes is dependent on a systemic process that incorporates pretreatment, enzymatic saccharification and fermentation (Fig. 5). The expenditure endured at each of these steps affects the viability of the overall end-product (Devi et al., 2022). Despite the abundance and widespread availability of lignocellulosic biomass, expensive and inefficient bioconversion technologies hinder the commercial establishment of biobased products (Guragain and Vadlani, 2021). Therefore, process integration and optimization for the development of cost-effective and green technologies that valorize lignocellulosic biomass to biofuels and biochemicals is vital for the transition of the current fossil-fuel economy to a sustainable and circular bioeconomy (Guragain and Vadlani, 2021). Modern mathematical techniques are sought after for the optimization of fermentation processes due to their effectiveness, efficiency and robustness. For instance, response surface methodology (RSM) is a collection of statistical and mathematical tools used for the design of experiments (DOE) and optimization of process variables. RSM offers several benefits, for example it reduces the number of experimental trials and takes into consideration the interaction of process parameters (Kumari and Gupta, 2019). RSM has been employed for the optimization of various process parameters that comprise of temperature (Baskar et al., 2018), catalyst concentration (Manmai et al., 2020), solid loading (David et al., 2020), time and microwave power intensity (Laltha et al., 2021). Additional parameters that have been optimized include desorbent (Baskar et al.,

2018) and nitrogen concentration (da Silva et al., 2012). These studies achieved maximum product concentrations for the response variables by optimizing the aforementioned parameters.

Various bioprocess types have emerged that comprise of (1) separate hydrolysis and fermentation (SHF), (2) simultaneous saccharification and fermentation (SSF) (Dahnum et al., 2015) and (3) SSF with pre-hydrolysis (PSSF) (Giang et al., 2019). Recently, the simultaneous saccharification and fermentation (SSF) strategy has attracted significant research among these, owing to its beneficial capabilities that it offers for saving processing time, energy and resources. The SSF method involves the degradation of cellulose by cellulolytic enzymes in combination with the conversion of glucose to fermentable products (Devi et al., 2022). The SSF process is a simpler approach that occurs in a single reactor, thereby saving resources, reducing the risk for contamination and inhibitor compounds.

On the other hand, kinetic models can be utilized to predict microbial growth and product output by quantifying experimental data. The logistic model elaborates on the growth of the microorganism to the rate-limiting substrate over time (Germec et al., 2019). Conversely, the modified Gompertz model estimates the maximum production rate, maximum concentration and the lag phase (Mozhiarasi et al., 2020). Both mathematical and kinetic models maximize product output while reducing chemical expenditure, inhibitor formation and time. Furthermore, the revenue that is sequestered can be channelled towards other bioprocessing steps to enhance productivity and promote a circular bioeconomy.



Fig. 5. Circular bioeconomy illustrating the bioconversion of agro-industrial wastes to valueadded products (Adapted from Son et al., 2022).

1.8. Research motivation/Problem statement

Establishing a sustainable bioeconomy necessitates overcoming several key challenges encountered at various steps of the lignocellulosic biorefinery value chain. Biorefining is a capital-intensive endeavour that aims to valorize lignocellulosic waste to value-added products (Usmani et al., 2021). However, the bioprocess conversion remains a technical and economic challenge due to various factors that include but is not limited to (1) lignocellulosic substrate availability, structure, and properties, (2) complex pretreatment and saccharification technologies, and (3) high capital and operating expenditures for large-scale industrial implementation (Raj et al., 2022). Citric acid is extensively used in the food and pharmaceutical industry and is one of the largest biotechnological industries for organic acid production. However, a key bottleneck endured during bioprocessing is the availability of fermentable sugars (Massadeh et al., 2022). Therefore, there has been an increasing trend towards the efficient utilization of lignocellulosic biomass as a cost-effective and abundant resource of polysaccharides. A necessary step towards the effective utilization of lignocellulosic biomass is a pretreatment step that eases the fermentation process subsequently enhancing the downstream product yield (Ashokkumar et al., 2022). However, most conventional pretreatment techniques possess various drawbacks that include high-energy and cost consumption, intermediates and by-product formation that diminish the overall attractiveness of biomass valorization. In addition, the pretreatment step has been estimated to account for nearly 40% of the overall processing cost (Kucharska et al., 2018). These processing costs are mainly attributed to the utilization of freshwater and expensive chemicals in lignocellulosic bioprocesses. Hence, the development of novel, efficient and low-cost pretreatment and fermentation strategies that result in optimum product formation are required. Optimization of key process parameters are essential to determine the interactive effects on the product output yield which saves resources, energy, and time (Odu et al., 2020). On the other hand, employing mathematical kinetic models to assess the microbial growth and product formation is vital for understanding metabolic regulations that occur under varying conditions (Augustine et al., 2015). These findings may contribute to improving the product quality and yield thus enabling industrial-scale implementation. Furthermore, the utilization of agro-industrial wastewater and lignocellulosic biomass in biorefining mitigates environmental pollution and greenhouse gas emissions thus creating a green environment, consequently improving economic growth, and contributing to a sustainable energy system.

1.9. Aims and objectives

The present study aimed to: (1) model and optimize novel and cost-effective pretreatment strategies for the enhancement of reducing sugar from banana pseudostem, (2) optimize key parameters for microbial-based citric acid production by using the simultaneous saccharification and fermentation approach and (3) assess the kinetics of *Aspergillus brasiliensis* growth and product formation under different medium conditions.

To achieve the aforementioned aims, this study is guided by the following objectives:

i. Review banana pseudostem (BP) as a potential lignocellulosic substrate for the generation of microbial-based value-added products. Discuss pretreatment regimes on BP together with their advantages and limitations. Outline the development and utilization of lignocellulosic bioproducts, bioprocess types, kinetic models and present challenges and future prospects.

ii. Model and optimize two novel pretreatments consisting of a: (1) microwave-assisted-iodized table salt (M-ITS) and (2) microwave-assisted-paper wastewater (M-PWW) using the response surface methodology (RSM) for the enhancement of sugar recovery from banana pseudostem.

iii. Evaluate the effectiveness and versatility of the developed M-ITS and M-PWW pretreatment techniques on several lignocellulosic substrates (corn cobs, sorghum leaves, sugarcane bagasse and bamboo).

iv. Model and optimize the simultaneous saccharification and fermentation (SSF) process using the RSM for citric acid production from optimally pretreated BP derived in objective (ii).

v. Utilize the logistic and modified Gompertz models to assess the kinetics of *Aspergillus brasiliensis* growth and citric acid production, respectively, under different medium conditions using the SSF bioprocess type.

2. Outline of thesis structure

This thesis is based on five chapters, the general introduction chapter, literature review chapter and two experimental chapters that are presented in research paper format and a concluding chapter. More specifically, it represents a compilation of one published paper and two manuscripts that have been submitted and are under review, whereby each chapter is an individual entity prepared as per the journals' specifications. Consequently, some repetition between chapters has been unavoidable. The thesis format is based on the template outline provided by the College of Agriculture, Engineering and Science (CAES) of the University of KwaZulu-Natal, South Africa. Each experimental chapter is self-contained, comprising of an introduction, materials and methods, results and discussion, conclusion and references. The representation, evaluation and application of novel, agro-industrial wastes (wastewater and banana pseudostem) for microbial-based value-added products featuring pretreatment and fermentation are pivotal to all chapters.

Chapter 1 provides a general background of the current research and highlights the aims and objectives that were undertaken.

Chapter 2 presents an overview of banana pseudostem as a potential substrate for bioconversion to microbial-based value-added products. Additionally, previous pretreatment regimes and novel strategies applied to BP for the enhancement of sugar recovery and optimum product formation are discussed aligned with their advantages and limitations. The current challenges of lignocellulosic bioprocesses and future innovations for energy efficient and cost-competitive technologies during pretreatment, enzymatic saccharification and fermentation are presented. Chapter 3 focuses on the modelling and optimization of two novel pretreatments that include: (1) microwave-assisted-iodized table salt (M-ITS) and (2) microwave-assisted-paper wastewater (M-PWW) strategy for the enhancement of sugar recovery from BP. In addition, the efficiency and adaptability of the developed pretreatment models were assessed on various lignocellulosic substrates (corn cobs, sorghum leaves, sugarcane bagasse and bamboo).

Chapter 4 encompasses modelling and optimization of key input parameters such as acetone concentration, ammonium nitrate concentration and temperature in the SSF process for optimal citric acid (CA) production from pretreated BP. Thereafter, kinetic assessment of *Aspergillus brasiliensis* growth and CA production were determined for the optimum conditions using fresh water (SSF_{optimizedFW}) or dairy wastewater (SSF_{DWW}) and compared to Sabouraud Dextrose Emmon's medium modified with BP (SSF_{SDEmodified}).

Chapter 5 concludes and integrates the key findings from the experimental chapters, highlights the significant conclusions and provides recommendations for future research on lignocellulosic bioprocessing.

References

- Adeoye, A.O., Lateef, A. 2022. Improving the Yield of Citric Acid Through Valorization of Cashew Apple Juice by Aspergillus niger: Mutation, Nanoparticles Supplementation and Taguchi Technique. Waste and Biomass Valorization, 1-12.
- Ashokkumar, V., Venkatkarthick, R., Jayashree, S., Chuetor, S., Dharmaraj, S., Kumar, G., Chen, W.-H., Ngamcharussrivichai, C. 2022. Recent advances in lignocellulosic biomass for biofuels and value-added bioproducts - A critical review. Bioresource Technology, 344, 126195.

- Augustine, A., Imelda, J., Paulraj, R., David, N.S. 2015. Growth kinetic profiles of Aspergillus niger S14 a mangrove isolate and Aspergillus oryzae NCIM 1212 in solid state fermentation. Indian Journal of Fisheries, 62, 100-106.
- Ayeni, A., Daramola, M.O., Taiwo, O., Olanrewaju, O.I., Oyekunle, D.T., Sekoai, P.T., Elehinafe, F.B. 2019. Production of Citric Acid from the Fermentation of Pineapple Waste by. The Open Chemical Engineering Journal, 13.
- Badiei, M., Asim, N., Jahim, J.M., Sopian, K. 2014. Comparison of chemical pretreatment methods for cellulosic biomass. APCBEE procedia, 9, 170-174.
- Baruah, J., Nath, B.K., Sharma, R., Kumar, S., Deka, R.C., Baruah, D.C., Kalita, E. 2018. Recent trends in the pretreatment of lignocellulosic biomass for value-added products. Frontiers in Energy Research, 6, 141.
- Baskar, G., Aberna Ebenezer Selvakumari, I., Aiswarya, R. 2018. Biodiesel production from castor oil using heterogeneous Ni doped ZnO nanocatalyst. Bioresource Technology, 250, 793-798.
- 8. Behera, B.C., Mishra, R., Mohapatra, S. 2021. Microbial citric acid: Production, properties, application, and future perspectives. Food Frontiers, 2, 62-76.
- Chan, S.S., Khoo, K.S., Chew, K.W., Ling, T.C., Show, P.L. 2022. Recent advances biodegradation and biosorption of organic compounds from wastewater: Microalgae-bacteria consortium-A review. Bioresource Technology, 344, 126159.
- Costa, S., Rugiero, I., Larenas Uria, C., Pedrini, P., Tamburini, E. 2018. Lignin degradation efficiency of chemical pre-treatments on banana rachis destined to bioethanol production. Biomolecules, 8, 141.
- 11. Da Silva, L.V., Tavares, C.B., Amaral, P., Coelho, M.A.Z. 2012. Production of citric acid by Yarrowia lipolytica in different crude glycerol concentrations and in different nitrogen sources. Chemical Engineering Transactions, 27.
- 12. Dahnum, D., Tasum, S.O., Triwahyuni, E., Nurdin, M., Abimanyu, H. 2015. Comparison of SHF and SSF processes using enzyme and dry yeast for optimization of bioethanol production from empty fruit bunch. Energy Procedia, 68, 107-116.
- 13. David, A., Sewsynker-Sukai, Y., Sithole, B., Gueguim Kana, E.B. 2020. Development of a green liquor dregs pretreatment for enhanced glucose recovery from corn cobs and kinetic assessment on various bioethanol fermentation types. 274, 117797.
- Devi, A., Bajar, S., Kour, H., Kothari, R., Pant, D., Singh, A. 2022. Lignocellulosic biomass valorization for bioethanol production: a circular bioeconomy approach. Bioenergy Research, 1-22.
- 15. Fernández-Ríos, A., Laso, J., Campos, C., Ruiz-Salmón, I., Hoehn, D., Cristóbal, J., Batlle-Bayer, L., Bala, A., Fullana-I-Palmer, P., Puig, R. 2021. Towards a Water-Energy-Food (WEF) nexus index: A review of nutrient profile models as a fundamental pillar of food and nutrition security. Sci. Total Environ., 789, 147936.
- 16. Germec, M., Turhan, I., Karhan, M., Demirci, A. 2019. Kinetic Modeling and Technoeconomic Feasibility of Ethanol Production From Carob Extract Based Medium in Biofilm Reactor. Applied Sciences, 9.
- Ghazal, L., Kazmi, S.J.H. 2014. Monitoring and measuring surface water in semi-arid environment using satellite data: A case study of Karachi. Journal of Basic & Applied Sciences, 10, 484.
- Giang, T.T., Lunprom, S., Liao, Q., Reungsang, A., Salakkam, A. 2019. Enhancing hydrogen production from Chlorella sp. biomass by pre-hydrolysis with simultaneous saccharification and fermentation (PSSF). Energies, 12, 908.
- 19. Guragain, Y.N., Vadlani, P.V. 2021. Renewable biomass utilization: a way forward to establish sustainable chemical and processing industries. Clean Technologies, 3, 243-259.
- 20. Harwood, S.A. 2018. In search of a (WEF) nexus approach. Environ. Sci. Policy, 83, 79-85.

- 21. Hernández Beltrán, J., Hernandez De Lira, I., Cruz-Santos, M., Saucedo-Luevanos, A., Hernández-Terán, F., Balagurusamy, N. 2019. Insight into Pretreatment Methods of Lignocellulosic Biomass to Increase Biogas Yield: Current State, Challenges, and Opportunities. Applied Sciences, 9, 3721.
- 22. Islam, M., Asad, M., Hossain, M., Paul, S., Sujan, S. 2019. Bioethanol production from banana pseudostem by using separate and cocultures of cellulase enzyme with Saccharomyces cerevisiae. Journal of Environmental Science Technology, 12, 157-163.
- 23. Jugwanth, Y., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2020. Valorization of sugarcane bagasse for bioethanol production through simultaneous saccharification and fermentation: Optimization and kinetic studies. Fuel, 262, 1-10.
- 24. Kamali, M., Khodaparast, Z. 2015. Review on recent developments on pulp and paper mill wastewater treatment. Ecotoxicology and environmental safety, 114, 326-342.
- 25. Kostas, E.T., Adams, J.M., Ruiz, H.A., Durán-Jiménez, G., Lye, G.J. 2021. Macroalgal biorefinery concepts for the circular bioeconomy: A review on biotechnological developments and future perspectives. Renewable and Sustainable Energy Reviews, 151, 111553.
- 26. Kucharska, K., Rybarczyk, P., Hołowacz, I., Łukajtis, R., Glinka, M., Kamiński, M. 2018. Pretreatment of lignocellulosic materials as substrates for fermentation processes. Molecules, 23, 2937.
- 27. Kumari, M., Gupta, S.K. 2019. Response surface methodological (RSM) approach for optimizing the removal of trihalomethanes (THMs) and its precursor's by surfactant modified magnetic nanoadsorbents (sMNP)-An endeavor to diminish probable cancer risk. Sci. Rep., 9, 1-11.
- 28. Kusmayadi, A., Lu, P.-H., Huang, C.-Y., Leong, Y.K., Yen, H.-W., Chang, J.-S. 2022. Integrating anaerobic digestion and microalgae cultivation for dairy wastewater treatment and

potential biochemicals production from the harvested microalgal biomass. Chemosphere, 291, 133057.

- 29. Laltha, M., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2021. Development of microwaveassisted alkaline pretreatment methods for enhanced sugar recovery from bamboo and corn cobs: Process optimization, chemical recyclability and kinetics of bioethanol production. Industrial Crops and Products, 174, 114166.
- 30. Legodi, L.M., Lagrange, D.C., Jansen Van Rensburg, E.L., Ncube, I. 2021. Enzymatic hydrolysis and fermentation of banana pseudostem hydrolysate to produce bioethanol. Int. J. Microbiol., 2021.
- 31. Liu, L., Sun, J., Cai, C., Wang, S., Pei, H., Zhang, J. 2009. Corn stover pretreatment by inorganic salts and its effects on hemicellulose and cellulose degradation. Bioresource technology, 100, 65-71.
- 32. Mabhaudhi, T., Simpson, G., Badenhorst, J., Mohammed, M., Motongera, T., Senzanje, A., Jewitt, A., Naidoo, D., Mpandeli, S. 2018. Assessing the state of the water-energy-food (WEF) nexus in South Africa. Water Research Commission (WRC), Pretoria.
- 33. Mäkelä, M., Geladi, P., Grimm, A., Dahl, O., Pietiläinen, A., Larsson, S.H. 2016. Cyclone processing of green liquor dregs (GLD) with results measured and interpreted by ICP-OES and NIR spectroscopy. Chemical Engineering Journal, 304, 448-453.
- Manasa, R.L., Mehta, A. 2020. Wastewater: Sources of Pollutants and Its Remediation. Environmental Biotechnology, 2, 197-219.
- 35. Manmai, N., Unpaprom, Y., Ponnusamy, V.K., Ramaraj, R. 2020. Bioethanol production from the comparison between optimization of sorghum stalk and sugarcane leaf for sugar production by chemical pretreatment and enzymatic degradation. Fuel, 278, 1-14.

- 36. Massadeh, M.I., Fandi, K., Al-Abeid, H., Alsharafat, O., Abu-Elteen, K. 2022. Production of Citric Acid by Aspergillus niger Cultivated in Olive Mill Wastewater Using a Two-Stage Packed Column Bioreactor. Fermentation, 8, 153.
- 37. Mohammed, I.Y., Abakr, Y.A., Mokaya, R. 2019. Biofuel and valuable products recovery from Napier grass pre-processing: Process design and economic analysis. Journal of Environmental Chemical Engineering, 7, 102962.
- 38. Moodley, P., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2020. Progress in the development of alkali and metal salt catalysed lignocellulosic pretreatment regimes: Potential for bioethanol production. Bioresource Technology, 1-49.
- 39. Mozhiarasi, V., Weichgrebe, D., Srinivasan, S.V. 2020. Enhancement of Methane Production from Vegetable, Fruit and Flower Market Wastes Using Extrusion as Pretreatment and Kinetic Modeling. Water Air Soil Pollut., 231, 126.
- 40. Odu, N., Uzah, G., Akani, N. 2020. Optimization of Citric Acid Production by Aspergillus niger and Candida tropicalis for Solid State Fermentation Using Banana Peel Substrate. Journal of Life and Bio Sciences Research, 1, 51-60.
- 41. Ozdal, M., Kurbanoglu, E.B. 2019. Citric acid production by Aspergillus niger from agroindustrial by-products: Molasses and chicken feather peptone. Waste and Biomass Valorization, 10, 631-640.
- Phitsuwan, P., Sakka, K., Ratanakhanokchai, K. 2016. Structural changes and enzymatic response of Napier grass (Pennisetum purpureum) stem induced by alkaline pretreatment. Bioresource technology, 218, 247-256.
- 43. Prazeres, A.R., Luz, S., Fernandes, F., Jerónimo, E.J. 2020. Cheese wastewater treatment by acid and basic precipitation: Application of H2SO4, HNO3, HCl, Ca (OH) 2 and NaOH. Journal of environmental management, 8, 103556.

- 44. Raj, T., Chandrasekhar, K., Kumar, A.N., Banu, J.R., Yoon, J.-J., Bhatia, S.K., Yang, Y.-H., Varjani, S., Kim, S.-H. 2022. Recent advances in commercial biorefineries for lignocellulosic ethanol production: Current status, challenges and future perspectives. Bioresource technology, 344, 126292.
- 45. Sanusi, I.A., Faloye, F.D., Gueguim Kana, E.B. 2019. Impact of Various Metallic Oxide Nanoparticles on Ethanol Production by Saccharomyces cerevisiae BY4743: Screening, Kinetic Study and Validation on Potato Waste. Catalysis Letters, 149, 2015-2031.
- 46. Sharma, N.K., Govindan, K., Lai, K.K., Chen, W.K., Kumar, V. 2021. The transition from linear economy to circular economy for sustainability among SMEs: A study on prospects, impediments, and prerequisites. Business Strategy and the Environment, 30, 1803-1822.
- 47. Shimizu, F.L., Monteiro, P.Q., Ghiraldi, P.H.C., Melati, R.B., Pagnocca, F.C., Souza, W.D., Sant'anna, C., Brienzo, M. 2018. Acid, alkali and peroxide pretreatments increase the cellulose accessibility and glucose yield of banana pseudostem. Industrial Crops and Products, 115, 62-68.
- Show, P.L., Oladele, K.O., Siew, Q.Y., Aziz Zakry, F.A., Lan, J.C.-W., Ling, T.C. 2015.
 Overview of citric acid production from Aspergillus niger. Frontiers in Life Science, 8, 271-283.
- 49. Singh, B., Korstad, J., Guldhe, A., Kothari, R. 2022. Emerging feedstocks & clean technologies for lignocellulosic biofuel. Frontiers in Energy Research, 734.
- 50. Singh, N., Singhania, R.R., Nigam, P.S., Dong, C.-D., Patel, A.K., Puri, M. 2022. Global status of lignocellulosic biorefinery: Challenges and perspectives. Bioresource Technology, 344, 126415.
- 51. Slavov, A.K. 2017. Dairy wastewaters–general characteristics and treatment possibilities–a review. Food Technology and Biotechnology, 55, 14-28.

- 52. Son, J., Baritugo, K.-A., Lim, S.H., Lim, H.J., Jeong, S., Lee, J.Y., Choi, J.-I., Joo, J.C., Na, J.-G., Park, S.J. 2022. Microbial cell factories for the production of three-carbon backbone organic acids from agro-industrial wastes. Bioresource Technology, 126797.
- 53. Song, L., Yang, D., Liu, R., Liu, S., Dai, L., Dai, X. 2022. Microbial production of lactic acid from food waste: latest advances, limits, and perspectives. Bioresource Technology, 345, 126052.
- 54. Subagyo, A., Chafidz, A. 2018. Banana pseudo-stem fiber: Preparation, characteristics, and applications. Intech Open, 1-20.
- 55. Tong, Z., Zheng, X., Tong, Y., Shi, Y.-C., Sun, J. 2019. Systems metabolic engineering for citric acid production by Aspergillus niger in the post-genomic era. Microbial cell factories, 18, 1-15.
- 56. Usmani, Z., Sharma, M., Awasthi, A.K., Lukk, T., Tuohy, M.G., Gong, L., Nguyen-Tri, P., Goddard, A.D., Bill, R.M., Nayak, S.C. 2021. Lignocellulosic biorefineries: The current state of challenges and strategies for efficient commercialization. Renewable and sustainable energy reviews, 148, 111258.
- 57. Yu, J., Xu, Z., Liu, L., Chen, S., Wang, S., Jin, M. 2019. Process integration for ethanol production from corn and corn stover as mixed substrates. Bioresource Technology, 279, 10-16.

CHAPTER 2

Banana pseudostem as a renewable feedstock for microbial-based value-added products

This chapter has been submitted to the journal *Bioresource Technology* entitled: Banana pseudostem as a renewable feedstock for microbial-based value-added products. (Under review). The manuscript and supplementary material are presented in the following pages.

Highlights

- Reviews the potential of banana pseudostem for microbial-based value-added products.
- Highlights current pretreatment regimes together with their advantages and limitations.
- Investigates novel and cost-effective alternatives to fresh water utilization.
- Comparatively assesses various bioprocess types for optimum product formation.
- Discusses the key challenges and future prospectives of lignocellulosic bioprocesses.



Graphical Abstract

Banana pseudostem as a renewable feedstock for microbial-based value-added products Milesh Laltha^a, Y. Sewsynker-Sukai^b, E.L. Meyer^b and ^{*}E.B. Gueguim Kana^a ^aUniversity of KwaZulu-Natal, School of Life Sciences, Pietermaritzburg, South Africa ^bUniversity of Fort Hare, Fort Hare Institute of Technology, Private Bag X1314, Alice 5700, South Africa

*Correspondence to: E.B. Gueguim Kana (kanag@ukzn.ac.za)

Abstract

Bananas have been touted as a vital food crop typically cultivated under tropical climates. Of particular interest is the banana pseudostem (BP) from a biorefinery standpoint, since it is renewable, abundant and a low-cost carbon source for generating a spectrum of high-value biobased products globally. The production and utilization of lignocellulosic bio-based products offer several advantages to the commercial sector; however, the relatively high cost of production has restricted its mainstream adoption. In light of this, the present review assesses BP as a potential lignocellulosic substrate for the generation of microbial-based value-added products. Previous pretreatment regimes for the enhancement of sugar recovery and optimum product formation are discussed together with their advantages and limitations. Furthermore, the development of different lignocellulosic bioproducts, bioprocess types and the utilization of mathematical kinetic models are outlined. Finally, present challenges and prospects to overcome the bottlenecks associated with the lignocellulosic biorefinery platform are presented.

Keywords: Banana pseudostem; pretreatment; fermentation; bio-based products.

1. Introduction

Lignocellulosic based bioprocesses have emerged as a sustainable approach for the production of various industrial bioproducts (de Paula et al., 2019). Lignocellulosic biomass (LCB) is a naturally occurring complex composite that originates from plant dry matter (Moodley et al., 2020) and are characterized as waste materials. Although considered a waste, LCB is viewed as an ideal feedstock for the production of renewable biofuels and bioproducts, because it does not pose a risk to socio-economic concerns such as food security, deforestation and water shortages. Moreover, LCB is abundant and considered to be cost-effective compared to foodbased feedstocks (Sekoai et al., 2019). Current technological advancements focus on exploiting lignocellulosic substrates to produce microbial-based renewable biofuels and other commercial value-added bioproducts (de Paula et al., 2019). The application of LCB aims to offset the fossil fuel demand by replenishing fuel reserves and generating renewable high value bioproducts. LCB represents the world's most economical and highly renewable natural resource with an estimated annual production of 200 billion tons (Hernández Beltrán et al., 2019, Lu et al., 2022). Lignocellulosic feedstocks are primarily composed of three major polymers, namely cellulose (30 to 50%), hemicellulose (15 to 35%) and lignin (10 to 20%) as well as other minor components including heterogeneous phenolic polymers (proteins, extractives and inorganic minerals) (Mussatto and Dragone, 2016, Limayem et al., 2012). Corn cobs (David et al., 2020), banana pseudostem (Islam et al., 2019), sugarcane leaves (Moodley and Gueguim Kana, 2019), and bamboo (Laltha et al., 2021) are some examples of LCB that have previously been assessed. More specifically, banana pseudostem (BP) has emerged as an attractive and non-edible feedstock that exhibits high tensile strength and biodegradability (Shimizu et al., 2018). Additionally, BP has a high cellulosic and low lignin content, thus making it suitable for the production of biofuels and biochemicals. Although LCB such as BP offers several advantages, its recalcitrant nature hinders direct microbial utilization and represents a major bottleneck that must be overcome. More specifically, the crystallinity of cellulose, encapsulation of cellulose by the lignin-hemicellulose matrix and hydrophobicity of lignin all contribute to the ability of lignocellulose to resist degradation (Kumar et al., 2020, Tayyab et al., 2018). Thus, effective fractionation of the biomass into its main constituents (cellulose, hemicellulose and lignin) is crucial and can be achieved by subjecting the feedstock to a pretreatment stage (Mussatto and Dragone, 2016). Pretreatment is an important step for efficiently converting LCB to biofuel and value-added products (Mussatto and Dragone, 2016). Additionally, this stage is necessary to produce highly digestible solids to enhance the sugar yields obtained during enzymatic hydrolysis (Mussatto and Dragone, 2016). Numerous strategies for biomass pretreatment have been investigated and developed over the years. These include alkali, acid, microwave, inorganic salts, steam explosion and ammonia fibre expansion, among others (Yu et al., 2019). Pretreatment is one of the most energy intensive steps in a lignocellulosic biorefinery and accounts for up to 40% of the total costs of biomass processing towards biofuels or value-added products (Kucharska et al., 2018). This cost can be attributed to the chemicals, water and energy utilized during the pretreatment stage. Furthermore, several studies have reported on the low sugar yields in addition to the degradation of sugars and subsequent formation of inhibitory compounds that hinder the enzymatic and fermentation process (Mussatto and Dragone, 2016, Saha et al., 2019). The development of an economically robust and reliable bioprocess is essential to effectively utilize lignocellulosic materials (Mussatto and Dragone, 2016). Over the last decade, lignocellulosic bioprocessing that incorporates pretreatment and fermentation has become a research hotspot with thousands of studies generated on an annual basis (Usmani et al., 2021). Nevertheless, limitations that continue to plague these systems include the high costs and energy, long processing times, low product yields and detrimental effects on the environment (Baruah et al., 2018). To overcome the aforementioned bottlenecks, it is necessary to tap into various aspects pertaining to

lignocellulosic bioprocesses. For instance, suitable LCB substrates with lower fractions of resistant structures (lignin and hemicellulose), thus inferring a higher digestibility with increased potential to be completely valorized is highly sought and need to be explored (Zhang et al., 2019). Moreover, the development of pretreatment and fermentation strategies that are environmentally friendly, cost-effective and result in high product yields is imperative. Therefore, this review assesses banana pseudostem (BP) as a potential lignocellulosic substrate for the production of microbial-based value-added products. A compilation of previous pretreatment and fermentation studies on BP as a feedstock are presented. In addition, previous pretreatment regimes for improving sugar recovery and optimum product formation are reviewed, along with their benefits and drawbacks. Furthermore, the development of lignocellulosic bioproducts, the use of mathematical kinetic models, and bioprocess types are discussed. Finally, the current challenges and potential prospects for overcoming bottlenecks associated with the lignocellulosic bioprocessing platform are highlighted.

2. Lignocellulosic biomass

Lignocellulosic biomass is an innate macromolecule that is synthesized from sunlight, nutrients and CO₂ sequestration by plants (Fatma et al., 2018). It is regarded as a carbon-neutral resource, since the quantity of CO₂ released during combustion is merely identical to that which the plant absorbs during photosynthesis (Ibarra-Gonzalez and Rong, 2019). There are various groups of raw materials that have been distinguished by their composition, origin and structure such as agricultural residues, woody feedstocks and municipal solid wastes (Limayem et al., 2012). Lignocellulosic substrates are defined as heterogeneous matrices that contain hemicellulose and cellulose bound together by lignin that comprise of cellulose (C₆ sugars) and hemicellulose (C₅ and C₆ sugars) bound together by layers of recalcitrant lignin (Moodley et al., 2020). The lignocellulosic composition varies between substrates and is dependent on the physical properties and source of the biomass as shown in Table 1. These differentiations may also be attributed to the type of plant species, weather and soil fertility implications (Limayem et al., 2012). Cellulose is the most abundant organic molecule available and is a principal constituent of plants (Wang et al., 2017). The structure of cellulose has been described as a linear homopolymer of glucose (C₆H₁₂O₆) units that are linked together in the form of D-anhydroglucopyranose units through β -(1,4)-glycosidic bonds (Dashtban et al., 2009). Adjacent cellulosic chains are coupled with hydrogen bonds, hydrophobic interactions and Van der Waal's forces. Hemicellulose is the second most abundant component of LCB. Hemicellulose is an amorphous, single chain, branched polysaccharide made up of heterogenous polymers and includes pentose (L-arabinose and D-xylose) as well as hexose (D-glucose, D-mannose and D-galactose) (Kang et al., 2014). The backbone chain of hemicellulose comprises of xylan linkages, namely L-arabinose (10%) and α -xylose (90%) linked together via hydrogen bonding (Kang et al., 2014). The resistant lignin polymer is an amorphous phenolic compound that is made up of three monomeric groups: coniferyl, sinapyl alcohols and *p*-coumaryl (Hernández Beltrán et al., 2019). These alcohols are linked together by ether and carbon-carbon bonds (Stöcker, 2008). Lignin is distributed between the exterior layers of fibres, therefore keeping the polysaccharide fibres together, enhancing the strength and rigidity of the cell wall (Stöcker, 2008). This aromatic biopolymer is covalently bonded to hemicellulose and cellulose, providing the impermeable and recalcitrant characteristic to plant cell walls (Kang et al., 2014, Moodley et al., 2020). The lignin content of LCB is gaining considerable attention due to its inhibitory effect on the hydrolysis and fermentation processes (Kang et al., 2014). Therefore, a pretreatment step is integral to reducing LCB recalcitrance to improve its cellulose digestibility.

Feedstock	Cellulose (%)	Hemicellulose	Lignin (%)	References
		(%)		
Banana pseudostem	42-60	19-23	15-17	(Islam et al., 2019, Shimizu
				et al., 2018)
Napier grass	35.0-39.4	19.2-23.4	15.3-19.3	Narinthorn et al. (2019)
Switch grass	40-45	30-35	12	Kang et al. (2014)
Corn cobs	42-45	35-39	14-15	Rabemanolontsoa and Saka
				(2013)
Bamboo	39.80	19.49	20.81	Rabemanolontsoa and Saka
				(2013)
Wheat straw	35-39	22-30	12-16	De Bhowmick et al. (2018)
Barley	36-43	24-33	6.3-9.8	De Bhowmick et al. (2018)
Rice husk	25-35	18-21	26-31	Rabemanolontsoa and Saka
				(2013)
Eucalyptus	52.07	24.51	25.2	Xu et al. (2017)
Sugarcane bagasse	46.1	20.1	20.3	Nosratpour et al. (2018)

Table 1. Composition of various lignocellulosic biomass feedstocks.

2.1. Banana pseudostem as a suitable feedstock for valuable biofuels and biochemicals

Banana (*Musa acuminata*) is a monocotyledonous annual herbaceous plant that belongs to the family Musaceae (Shimizu et al., 2018). It is mainly cultivated in the tropical and subtropical regions of Asia, South America and Africa (Islam et al., 2019). Banana is one of the most essential, perennial fruit crops globally, with an overall yearly production of more than 100 million tons (Costa et al., 2018). In South Africa, approximately 450 000 tons are produced yearly (Panigrahi et al., 2021). After the fruit is harvested, banana lignocellulosic biomass including rachis, foliage, and stems remain (Chen et al., 2017b). This waste is usually crushed and left to decompose on farms to replenish nutrients in the soil (Chen et al., 2017b). However, this disposal method leads to environmental pollution in ground and water bodies resulting in the release of unpleasant greenhouse gases (hydrogen sulphide and ammonia gas) and the

spread of Panama disease caused by the fungus Fusarium oxysporum (Chen et al., 2017b, Costa et al., 2018). Therefore, utilizing this lignocellulosic biomass will minimize the aforementioned problems associated with it. For every ton of banana that is harvested, approximately 4 tons of agricultural residues (leaf, pseudostem, rachis, rotten fruit, peel, rhizome) are generated, of which 75% is the pseudostem (Islam et al., 2019). The pseudostem is the cylindrical part of the banana plant, made up of overlapping leaf sheaths that are 20 to 50 cm in diameter (Islam et al., 2019). To date, banana pseudostem has been proposed as a useful biomass for paper production, animal feed production, organic fertilizer, renewable materials, wastewater purification and biofuel production (Chen et al., 2017b). Banana pseudostem contains on average 42-60% cellulose, 19-23% hemicellulose and 15-17% lignin (Islam et al., 2019, Shimizu et al., 2018). Additionally, banana pseudostem also contains approximately 8-10% ash and 3-3.1% extractives (Idrees et al., 2013, da Silva et al., 2019, Begum and Chittibabu, 2015). The fibre of banana pseudostem has a low density and strong moisture absorption ability enabling it to be flexible and increasing the transfer of heat (Li et al., 2010, Subagyo and Chafidz, 2018). Moreover, this agricultural waste displays satisfactory physical properties, including relatively high tensile stiffness and strength, light weight, and biodegradability (Shimizu et al., 2018). The high cellulosic and hemicellulosic content makes it an ideal feedstock for renewable energy production (Narinthorn et al., 2019). Furthermore, it has been used previously to produce valuable bio-based chemicals such as sodium carboxymethyl cellulose (CMC) and polyphenol oxidase (PPO) (Shimizu et al., 2018). Due to the low lignin content, it could serve as a potential substrate for biogas production (Islam et al., 2019). The pseudostem of the banana plant illustrated in Fig. 1. provides and transports nutrients from the soil to the fruits. The pseudostem is the part of the banana plant that resembles a trunk consisting of a soft central core and tightly wrapped by up to 25 leaf sheaths (Subagyo and Chafidz, 2018). These leaf sheaths unwrap from the stem and transform into recognizable

banana leaves after they have matured. Moreover, the height of the banana plant can reach approximately 7.5 metres (Subagyo and Chafidz, 2018). The characteristics of banana pseudostem demonstrated by its high availability, structural composition and previous use in bioprocesses elicits its application for the production of biofuels and other commercial microbial-based value-added products. Recent reports on the utilization of banana pseudostem for the generation of various bioproducts and the respective pretreatment and/or bioprocess parameters are outlined in Table 2. Islam et al. (2019) reported on the production of bioethanol from banana pseudostem using Saccharomyces cerevisiae. This author pretreated BP using dilute acid (H₂SO₄) to obtain 1.45 g/L dextrose fermented to 8.28 g/L bioethanol. In the same vein, Mustofa (2018) attained 8.51 g/L of bioethanol from BP using a similar pretreatment. Other authors have also reported on the production of bioethanol from BP using other forms of pretreatment. For instance, Sukmaningtyas (2018), Ingale et al. (2014) and Thakur et al. (2013) pretreated BP using alkali (NaOH) pretreatment and observed 4.32, 17.1 and 3.8 g/L of bioethanol respectively using Saccharomyces cerevisiae. Interestingly, in addition to bioethanol, some studies have reported on the production of other valuable bioproducts from BP. For example, Chen et al. (2017b) and Sivanarutselvi et al. (2019) pretreated BP using 8 and 2% w/v NaOH for the attainment of 41.6% lignin removal and 92% delignification, respectively. Bioconversion of these pretreated substrates to biomethane (239.9 mL/g) and biobutanol (10.12 g/L) was achieved using anaerobic digestion and Clostridium sp., respectively. Furthermore, Idrees et al. (2013) pretreated BP using sulfuric acid and obtained 38 g/L reducing sugar. These sugars were converted by Lactobacillus sp. thus achieving a 92% lactic acid conversion yield. In many of these studies, BP has been pretreated using several pretreatment strategies resulting in various fermentative value-added products. Hence, BP is a versatile lignocellulosic substrate that confers multiple advantages that necessitate exploration more decisively.

Fermentative microorganism	Pretreatment parameters	Pretreatment output	Bioprocess parameters	Product output	References
Saccharomyces	Dilute H ₂ SO ₄ (0.25%	1.45 g/L dextrose	10 g of pretreated substrate, 48	8.28 g/L	Islam et al. (2019)
cerevisiae	v/v), 121°C, 30 min and		hr, 37 °C, 5 units/mL enzyme,	bioethanol	
	1 hr		200 mL of 0.1 M citrate buffer,		
			pH 5, 100 rpm		
Lactobacillus	1% H ₂ SO ₄ , 130 °C, 2.18	38 g/L reducing sugar	Inoculum to solution ratio of	92% lactic acid	Idrees et al. (2013)
acidophilus	hr		1:20, 10 g/L yeast extract, 2 g/L	conversion	
			(NH4)2HPO4, 0.1 g/L MgSO4,	yield	
			72 hr		
Anaerobic	8% NaOH (w/v), 50 °C,	41.6% lignin removal,	-	239.9 mL/g	Chen et al. (2017b)
digest	24 hr	23.6% hemicellulose		biomethane	
		removal			
Clostridium	2% NaOH (w/v), S/L	92% delignification and	10% v/v inoculum, 37 °C, 72 hr,	10.12 g/L	Sivanarutselvi et al. (2019)
sporogenes	ratio 1:10, 30 °C, 24 hr	524 mg/g glucose	pH 7	biobutanol	
Saccharomyces	2 N NaOH, 121 °C, 72 hr	29.8 g/L glucose yield	2 N NaOH , 120 hr, cellulase	4.32 g/L	Sukmaningtyas (2018)
cerevisiae			enzyme 20 FPU/g, 10% v/v	bioethanol	
			inoculum, 37.5 °C		
Aspergillus	2% w/v H ₂ SO ₄ , 121 °C,	18% delignification, 15%	10% w/v substrate, pH 5, 30 °C,	8.51 g/L	Mustofa (2018)
niger	20 min	cellulose increase	150 rpm, 6 d	bioethanol	

Table 2. Studies investigating the potentiality of banana pseudostem for bioproducts corresponding to the various input process conditions.

Table 2. Continued.

Fermentative	Pretreatment parameters	Pretreatment output	Bioprocess parameters	Product output	References
microorganism					
Saccharomyces	1 N NaOH, 5g substrate	833 mg/mL reducing	4.1 g sugar, 5% inoculum, 0.5	17.1 g/L	Ingale et al. (2014)
cerevisiae	solid loading, room	sugar	g/L (NH4)2SO4, 0.5 g/L	bioethanol	
	temperature, 18 hr		KH ₂ PO ₄ , 2.5 g/L yeast extract,		
			pH 5.5, 30 °C, 72 hr		
Saccharomyces	1 N NaOH, 50 °C, 48 hr	15.3 g/L reducing sugar	2% v/v inoculum, 0.5 g/L	3.8 g/L	Thakur et al. (2013)
cerevisiae			(NH4)2SO4, 0.5 g/L KH2PO4,	bioethanol,	
			2.5 g/L yeast extract, pH 5.5, 30	0.35 g/g	
			°C, 48 hr	bioethanol	



Fig. 1. A banana tree consisting of its several parts including the banana pseudostem (Subagyo and Chafidz, 2018; modified).

2.2. Limitations of lignocellulosic biomass

Even with the immense benefits of LCB substrates, challenges that presently exist render these feedstocks non-viable for large scale application. For one, pretreatment of LCB is necessary but the high: (1) energy consumed, (2) chemical costs and (3) volume of fresh water used during this stage significantly impacts the overall process efficiency and has become the rate-limiting step (Mohammed et al., 2019). This is attributed to the recalcitrance of LCB, which resists degradation into simple sugars such as glucose that are mandatory for microbial consumption towards the generation of biofuel and bio-based products (Narinthorn et al., 2019). Natural factors that contribute to biomass recalcitrance include degree of lignification, the arrangement and density of the vascular bundles, quantity of thick-walled tissue and structural heterogeneity and complexity (Himmel et al., 2007). Additionally, the adjacent

structures of cellulosic and hemicellulosic microfibrils are bound to lignin moieties via hydrogen and covalent bonds, rendering them highly recalcitrant to depolymerization (Zhao et al., 2011). Furthermore, the activity of enzymes that catalyze the substrate into simpler molecules are inhibited, thereby raising production costs (Narinthorn et al., 2019). Therefore, several concerns need to be addressed in order to enhance the digestibility of cellulose. These comprise of the accessible surface region, cellulose crystallinity, lignin resistance and the various characteristics of biomass particles (Zabed et al., 2016). Most often, a pretreatment step involving either physical, chemical, biological or a combination of these are implemented to reduce biomass recalcitrance and enhance lignocellulosic degradation towards enzymatic hydrolysis.

2.3. The significance of lignocellulosic pretreatment

The pretreatment process is an imperative stage that is employed for altering the structure of LCB, making cellulose and hemicellulose available for bioconversion. The main purpose of pretreatment is to disrupt the physical (macrostructure) and chemical (microstructure) barriers of the plant cell wall, to enhance the surface-to-volume ratio, depolymerize and reduce the crystallinity of cellulose (Hernández Beltrán et al., 2019, Tayyab et al., 2018). Pretreatment also reduces the lignin content, thereby enabling the access of cellulose to hydrolytic enzymes. Hence, pretreatment processes inevitably enhance the saccharification efficiency, while lowering production costs (Liu et al., 2017). This stage is regarded as being crucial, both technically and economically, since operating and capital costs could amount to more than 40% of the total processing cost (Bhutto et al., 2017, Mupondwa et al., 2017). Effective pretreatment techniques should aim to minimize energy and fresh water usage, not to pose a threat to the environment by lowering its carbon footprint, utilize cost-effective chemicals that are non-corrosive, minimize the formation of enzymatic and fermentation inhibitor compounds and

enhance the availability of fermentable sugars for microbial fermentation (Brodeur et al., 2011, Chen et al., 2017a).

3. Lignocellulosic pretreatment regimes on Banana pseudostem

The physicochemical and structural complexity of the LCB matrix differs for each substrate (Mussatto and Dragone, 2016). Consequently, the selected pretreatment method is not as equally effective for each type of biomass, therefore it is essential to identify the most efficient pretreatment technology for a particular substrate. The major pretreatment methods that have been studied in recent years comprise of (i) physical pretreatment (mechanical, microwave irradiation and ultrasound), (ii) chemical pretreatment (acid, alkaline, ionic liquids and deep eutectic solvents and inorganic salt), (iii) physicochemical (steam explosion, ammonia fibre explosion, CO₂ explosion and liquid hot water), (iv) biological pretreatment and (v) combination of these strategies. The benefits and limitations of these pretreatment technologies are discussed and highlighted in Table 3. Additionally, the primary pretreatment methods that have been studied on banana pseudostem and similar derivatives of the banana plant in recent years are presented in table 4. Following the pretreatment stage, an enzymatic saccharification step is necessary to depolymerize the cellulose carbohydrate polymer to fermentable sugar (glucose) (Zabed et al., 2016). Cellulase is an enzyme cocktail that comprises mainly of three enzymes that include endoglucanase, cellobiohydrolase and β -glucosidase. At the onset of enzymatic hydrolysis, endoglucanase hydrolyzes amorphous cellulose to a reducing end group. Subsequently, cellobiohydrolase cleaves the glycosidic bonds to form cellobiose. Lastly, cellobiose is hydrolyzed by β -glucosidase to yield glucose in a liquid phase. Cellulase enzymes are highly effective for catalytically converting cellulose to glucose, however, this can only be successfully achieved through efficient pretreatment regimes (Shrotri et al., 2017).

3.1. Physical pretreatment

Physical pretreatment causes damage to the lignocellulosic structure with minimal to no chemical alterations to the different cell wall components (Chundawat et al., 2010). This pretreatment affects the substrate particle size, surface area, crystallinity index and degree of polymerization (Kumar et al., 2020). Physical pretreatment processes circumvent the usage of chemicals and is deemed safe for the environment, since they generate minimal to no toxic compounds. Nevertheless, implementing particle size reduction as an individual pretreatment is not adequate to significantly improve enzymatic digestibility due to the high cost. Despite the cost factor, minimal particle size reduction is required preceding most pretreatments to enhance material handling during processing (Chundawat et al., 2010). Some examples of commonly employed physical pretreatment techniques include mechanical, microwave, and ultrasound and are briefly discussed below.

3.1.1. Mechanical

Various mechanical pretreatment procedures (shredding, chipping, coarse size reduction and milling) have been employed to improve the digestibility of LCB (Tayyab et al., 2018). Mechanical pretreatment decreases cellulose crystallinity, assists in the removal of lignin and hemicellulose, increases the bulk density and surface area, and alters the degrees of polymerization (Tayyab et al., 2018). Additionally, the smaller particle size and larger surface area eradicates the mass and heat transfer limitations. Jiang et al. (2022) applied a high-pressure CO₂-hydrothermal pretreatment to banana pseudostem for value-added porous carbons. Prior to pretreatment, the BP was chopped into 2×5 cm pieces and air dried at 30 °C for 6 h to avoid spoilage. Following this, the BP was ground into powder and sieved through a 60-mesh screen, and thereafter dried in an oven at 105 °C for 24 h. These authors observed an improvement in activated carbon yield of 1340%. Likewise, Zhang et al. (2013) applied mechanical

pretreatment to banana stem. The substrate was cut into 1 cm sections, air dried, and stored at -20 °C. After pretreatment, it was determined that banana stem contained 30.07% of cellulose, 10.65% hemicellulose and 17.10% lignin. This pretreatment method has been utilized to facilitate and enhance the efficacy of subsequent biorefinery procedures (Chen et al., 2017a). Furthermore, mechanical pretreatment increases the digestion rate and product yield obtained from LCB (Narinthorn et al., 2019). These strategies produce fewer emissions, but intense energy and capital are required, thus increasing production costs (Chen et al., 2017a).

3.1.2. Microwave irradiation (MI)

Microwave heating technology has garnered immense consideration as a valuable source of power for rapid volumetric heating (Zhu and Chen, 2014). Many studies have extensively used this pretreatment technology for different lignocellulosic materials (Puligundla et al., 2016, Sewsynker-Sukai and Gueguim Kana, 2018, Moodley and Gueguim Kana, 2019). Microwave irradiation (MI) involves two principal mechanisms, ionic conduction and dipolar polarization (Zhu and Chen, 2014). This form of pretreatment functions by heating a material that contains electrical charges such as polar molecules in a liquid or solvent (Zhu and Chen, 2014). During microwave heating, polar water molecules change their orientation due to the rapidly alternating electromagnetic fields (Zhu and Chen, 2014, Singh et al., 2014). Therefore, heat is generated by the friction, rotation and collision of these molecules (Zhu and Chen, 2014). Moreover, MI displays a strong heating capacity, rendering it possible to dissolve the lignin and improve enzymatic susceptibility (Singh et al., 2014). MI exhibits several advantages over the conventional steam heating liquid hot water method. These advantages include: (1) short pretreatment times, (2) lower cost and energy, (3) minimum production of inhibitor compounds, (4) is considered environmentally benign, and (5) employs direct heating that prevents overheating of surfaces (Tayyab et al., 2018, Sewsynker-Sukai and Gueguim Kana,

2018). Furthermore, MI is environmentally safe and can also be conveniently coupled with chemical reactions to speed up the reaction rate, thus reducing the reaction time and energy usage (Nagahata and Takeuchi, 2019). Previous reports on microwave pretreatment have shown enhancements in the cellulose digestibility and/or sugar recovery. For instance, Gabhane et al. (2014) investigated microwave-assisted acid pretreatment on banana waste (pseudostem, leaf and pith) and demonstrated a maximum recovery of reducing sugar of 36.84% under 25 min of reaction time at 700 W. Egwim and Shittu (2015) evaluated the effects of microwave-assisted alkaline pretreatment on banana pseudostem for bioethanol production. A bioethanol yield of 300 mL was attained at an alkaline concentration (2% NaOH) and microwave irradiation (170 W, 10 minutes). Pretreatment studies have indicated that although microwave irradiation is efficient on lab-scale, this heating method has been limited by the disproportionate power distribution, lack of uniformity and reduced penetration efficiency in larger samples, which restricts its industrial scale application (Aguilar-Reynosa et al., 2017).

3.1.3. Ultrasound

Ultrasonic radiation degrades the lignocellulosic material by creating shear forces that disrupt the intricate network structures of LCB. Advantages of ultrasonic pretreatment include the (1) short reaction time, (2) low temperatures, and (3) low concentration of chemicals that are used in combination or for further pretreatment processing. Numerous factors have shown to impact on the efficiency of ultrasound pretreatment and include the sonication power, temperature, duration and ultrasound frequency (Baruah et al., 2018). For instance, Gabhane et al. (2014) investigated the influence of acid ultrasonication for the pretreatment of banana pseudostem. The authors revealed that pretreatment yielded a reducing sugar of 41.98% at a temperature of 50 °C for 60 min. Nevertheless, ultrasonication methods are energy intensive and further

experimentation is imperative to determine the optimum conditions for large scale application (Baruah et al., 2018)

3.2. Chemical pretreatments

Chemical pretreatment utilizes chemical reactions to alter the resistant lignocellulose. Frequently used chemical methods consist of acids, bases, inorganic salts, ionic liquids and organosolvents. The mechanisms of biomass pretreatment and selectivity of lignocellulosic structures differs depending on the chemical that is employed.

3.2.1. Acid

Acid pretreatment encompasses both concentrated and dilute acid that could be used to break down the rigid structure of LCB (Fatma et al., 2018). Sulfuric acid (H₂SO₄), hydrochloric acid (HCl), phosphoric acid (H₃PO₄) and nitric acid (HNO₃) are commonly used acids for pretreatment (Fatma et al., 2018). Concentrated acid (above 30% v/v) targets the polysaccharide chains, mainly hemicelluloses and celluloses (Fatma et al., 2018). Flexible substrate choice and high fermentable sugar yields at low temperatures are the main advantages of acid-catalysed pretreatments (Kumar et al., 2020). Regardless of the low temperature requirement, acid recovery is still a major challenge that is encountered during commercial-scale applications (Fatma et al., 2018). Furthermore, high acid concentrations poses dangers to operating personnel and may result in the corrosion of the bioreactor vessel resulting in high maintenance costs (Hernández Beltrán et al., 2019). Moreover, concentrated acid pretreatment generates inhibitor compounds at a high rate, thereby increasing purification costs (Hernández Beltrán et al., 2019).

Dilute acid pretreatment has been shown to successfully hydrolyze hemicellulose and disrupt the structure of a variety of lignocellulosic feedstocks. Low acid concentrations (0.2 to 2.5% v/v) and temperatures (between 130 °C and 210 °C) are typically employed in industrial applications (Hernández Beltrán et al., 2019). Frequently exploited inorganic acids include (sulfuric, nitric, phosphoric and hydrochloric acid) and organic acids (carboxylic and sulfonic acid) (Chen et al., 2017a). Shimizu et al. (2018) studied the effect of different acid (H₂SO₄) concentrations (5, 10, 15, 20, 25, 30, 35 and 40%) on the chemical composition, cellulose accessibility, and enzymatic digestibility of banana pseudostem. Under the most severe acid pretreatment, hemicellulose was removed entirely, and its percentage decreased to approximately 4.38%. In addition, a glucose yield of 60 to 80% was achieved at a high acid concentration indicating the suitability of banana pseudostem for biofuel production. In the recent time, Legodi et al. (2021) investigated 5% (v/v) H₂SO₄ pretreatment on BP and achieved 48% cellulose amounting to 21.8 g/L glucose after enzymatic hydrolysis (10 FPU/g, 50 °C). Dilute acid pretreatments minimize the amount of degradation products formed while lowering the rate of corrosion in comparison to concentrated acids (Fatma et al., 2018). Additionally, dilute acid improve the pretreatment process efficiency and are feasible for potential largescale implementation, and can be quickly neutralized by treating with ammonium or lime (Fatma et al., 2018). Nevertheless, the generation of calcium sulfate gypsum creates a disposal problem. Acid recovery is a non-feasible procedure that may result in environmental problems caused by the disposal of it into waste streams (Peral, 2016).

3.2.2. Alkali

Lignin removal from the lignocellulosic biomass is crucial to enhance the reactivity of the remaining polysaccharides. Pretreatment with bases is the most extensively used chemical technique that has shown to be highly efficient for lignin removal from LCB (Kim et al., 2020,

Sewsynker-Sukai et al., 2020). Alkaline pretreatment entails the solvation of lignocellulosic units and the hydrolytic decomposition of lignocellulose, termed saponification (Kucharska et al., 2018). As a result of alkaline pretreatment, cellulose crystallinity decreases, and the maximum specific area of the biomass increases (Kucharska et al., 2018). Consequently, the lignin structure is altered and undergoes unfolding. Popular alkaline reagents include calcium hydroxide (Ca(OH)₂), sodium hydroxide (NaOH), ammonia (NH₃), and hydrogen peroxide (H₂O₂) (Phitsuwan et al., 2016). Alkaline reagents interact with lignin by splitting the intermolecular ester bonds between carbohydrate and lignin complexes, thereby modifying the composition of LCB and contributes to the solubilization of lignin (Phitsuwan et al., 2016). Zhang et al. (2013) studied the impact of 6% alkali NaOH pretreatment on banana stem for biogas generation. After pretreatment, it was ascertained that 30.07% of cellulose, 10.65% hemicellulose and 17.10% lignin content was present in banana stem. These authors obtained a maximum biomethane yield of 232.4 mL/g amounting to 21.4%. Likewise, Sukmaningtyas (2018) investigated alkali NaOH (0.5 to 2 N) pretreatment on banana pseudostem in the autoclave for 90 min. The highest sugar content attained after enzymatic hydrolysis (72 h) was 29.8 g/L (2N NaOH solution). The variation of the pretreatment process by increasing NaOH concentration led to a decline of lignin content while increasing the amount of cellulose. The lowest lignin and the highest cellulose content achieved was 11.44% and 51.66%, respectively. Advantages of alkaline-based pretreatment strategies include favourable reaction conditions, low reaction pressures and temperatures, less corrosive chemicals, the potential to recover leftover alkali and the use of simplified reactors (Phitsuwan et al., 2016). Moreover, alkalibased pretreatments have shown to be more effective when used on LCB with low lignin content (herbaceous crops and agricultural residues) compared to hardwood substrates (Baruah et al., 2018). Despite its efficiency, alkaline methods employ chemical catalysts that are expensive and utilize vast amounts of fresh water during pretreatment steps and in between wash cycles (Badiei et al., 2014).

3.2.3. Ionic liquids and Deep Eutectic Solvents

Ionic liquids (IL) have been described as salts with melting points below 100 °C and are composed of cations and anions (Stark, 2007). Some examples of IL that have been assessed for lignocellulosic pretreatment include imidazolium and pyrrolidinium. The pretreatment mechanism of ionic liquids is thought to be as a result of the dissolution of cellulose (Stark, 2007). ILs possess high solvating properties and can selectively cleave glycosidic bonds in polysaccharides (Xu et al., 2017). Ai et al. (2021) explored the potential of cellulose derived from banana stem as a substitute for polyethylene (PE) film for the long-term preservation of mangos. Cellulose film (23.1 µm) was synthesized from banana stem via pretreatment (1 g substrate, 80 to 90 °C, 200 rpm, 3 to 4 h) with 19 mL of ionic liquid known as 1-Allyl-3-methylimidazolium chloride. Even with the benefits of IL pretreatments, their industrial scale applicability is challenged by the release of inhibitor compounds and high costs (Kucharska et al., 2018). Deep eutectic solvents (DES) that display similar features to ionic liquids, have emerged as low cost and easily synthesized alternatives. Other striking benefits of DES include their low volatility, broad liquid range, low toxicity, high biodegradability and enzyme compatibility.

3.2.4. Inorganic salt

Recently, the utilization of inorganic salt has emerged as an attractive form of pretreatment (Kang et al., 2013). Inorganic salts are gaining attention due to their ability to increase cellulose and hemicellulose conversion rates and hydrolysis yields (Liu et al., 2009). Inorganic salts may be categorized into three different classes and comprise of alkaline earth metal chlorides (CaCl₂

and MgCl₂), alkaline metal chlorides (KCl and NaCl) and transition metal chlorides (CuCl₂, FeCl₃ and FeCl₃) (Kang et al., 2013). The primary mechanism by which inorganic salts function is the dissociation of Lewis acids in an aqueous solvent to form complex cations. Glycosidic linkages present in LCB are cleaved by the formation of metal cations. The water molecules from the hydrated cation act as nucleophiles that depolymerize hemicellulose into monosaccharides (Moodley et al., 2020). Furthermore, inorganic salts undergo a hydrolysis reaction that releases hydrogen ions into solution, providing a more stable reactive environment (Liu et al., 2009). Madhu et al. (2014) synthesized activated carbon using banana stem. To activate the process, banana stem powder was heated at 700 °C. To determine the nitrite interference, inorganic salts (KCl, ZnCl₂) were added to the samples with no significant response. A high porous surface area of 1465 m²g⁻¹ was observed. Advantages of inorganic salts include their cost-effectiveness, recyclability and minimal release of inhibitor compounds (Moodley and Gueguim Kana, 2017, Moodley et al., 2020).

3.3. Physicochemical pretreatments

Physicochemical methods couple physical changes and chemical reactions. The LCB is pretreated under elevated temperature and/or pressure conditions with chemicals, resulting in degradation of the resistant components. Consequently, the easily degradable LCB structures are accessible for subsequent enzymatic saccharification and microbial fermentation processes. Physicochemical strategies consist of steam explosion, ammonia fibre explosion, CO₂ explosion and liquid hot water, which are highlighted below.

3.3.1 Steam explosion

Steam explosion is a frequently used pretreatment method. Temperature and pressure conditions of 160 to 260°C and 0.69 to 4.83 MPa, respectively, are implemented allowing water

molecules to effectively infiltrate the lignocellulosic biomass structure (Baruah et al., 2018). A sudden reduction in the pressure releases water and causes explosive decompression. More specifically, the elevated temperature and pressure conditions degrade glycosidic bonds present within cellulose and hemicellulose and cleaves bonds between hemicellulose and lignin (Cavalaglio et al., 2016). The steam explosion method boasts several advantages for industrial scale application including minimal effects on the environment, short pretreatment time, low chemical usage and reduced energy consumption, making it cost efficient. Santa-Maria et al. (2013) evaluated steam explosion on banana stem and rachis and demonstrated glucan conversion yields of 93 and 77% for stem and rachis, respectively. Steam explosion was carried out in a 100 L reactor with 150 g solids at 190 to 200 °C. Guerrero et al. (2017) optimized acidcatalysed steam explosion of banana pseudostem for fermentable sugar production. A high glucose yield of 91% was attained using 177 °C, 5 min and 2.2% H₂SO₄ (v/v). Nonetheless, longer pretreatment times at elevated temperature may lead to sugar degradation products such as inhibitor compounds that reduces the downstream process efficiency (Carvalho et al., 2018). While steam explosion is an attractive pretreatment technique, limitations that hinder its advancement include low lignin removal and the formation of inhibitory compounds that usually occur under harsh conditions (higher temperatures and/or longer pretreatment times) (Bhutto et al., 2017).

3.3.2. Ammonia fibre explosion

The ammonia fibre explosion (AFEX) technique employs a liquid ammonia to LCB (1:1) ratio at elevated temperature (60-100 °C) and pressure between 0.7 and 2.7 MPa for a specified period of time (5-30 min) with a subsequent rapid release of pressure (Jędrzejczyk et al., 2019). The processing conditions lead to LCB swelling, which increases the substrate surface area, the hemicellulose conversion to sugars, and alteration of the lignin structure. Advantages of AFEX methods include the high selectivity of ammonia for reaction with lignin, costeffectiveness of ammonia, which is easily recoverable and recycled, enhancing the industrial feasibility. Numerous factors affect AFEX pretreatment including temperature, pressure, water and ammonia loading. Isibika et al. (2019) evaluated the pretreatment efficiency of concentrated ammonia (24.5% NH₃) and heat (120 °C) on banana peel waste and obtained a biomass conversion ratio (% VS) of 9.6% and 3.5%, respectively. AFEX methods have emerged as efficient techniques for the pretreatment of biomass, however, bioreactor corrosion and highly concentrated ammonia are costly and therefore reduce the economic feasibility of this pretreatment process towards industrial scale (Jędrzejczyk et al., 2019).

3.3.3. CO₂ explosion

Supercritical CO₂ exhibits mass transfer characteristics with both gas-like and liquid-like solvating power, which permits its diffusion across interphases (Lachos-Perez et al., 2017). Supercritical CO₂ infiltrates LCB and fragments hemicellulose and lignin structures upon pressure release and enhancing cellulose accessibility (Capolupo and Faraco, 2016). Benefits of CO₂ explosion pretreatment include the low-cost of CO₂, minimal effect on the environment, lack of inhibitor formation and ease of recovery that justifies its application at an industrial scale. Jiang et al. (2022) investigated a high-pressure CO₂-hydrothermal pretreatment to banana pseudostem for value-added carbons. The pretreatment process resulted in a 1340% yield of activated carbon using a temperature in the range of 120 to 280 °C at 5 MPa, 700 rpm for 5 min.

3.3.4. Liquid hot water (LHW)

Similar to steam explosion, liquid hot water pretreatment employs water at high temperature (140-230°C) and pressure (up to 5 MPa) conditions but in the liquid phase (Baruah et al., 2018,

Jędrzejczyk et al., 2019). Liquid hot water causes partial hemicellulose solubilisation, dissolution of lignin and reduces the biomass structure (Jędrzejczyk et al., 2019). More specifically, liquid hot water cleaves acetyl groups in hemicellulose moieties and also causes the removal of lignin, which exposes the cellulose fibres to enzymatic hydrolysis. Advantages of liquid hot water pretreatment include the formation of little to no toxic compounds and the non-requirement of chemical catalysts. Also, the biomass size does not affect the process because the particles are fragmented using the pretreatment (Bhutto et al., 2017). In a previous report on a liquid hot water pretreatment of banana pseudostem, a cellulose availability of 25% was obtained resulting in 21.8 g/L of glucose (Legodi et al., 2021). This was achieved using a solid loading of 15% w/v, heated using an autoclave at 121 °C, 15 psi for 1 h. Similarly, the study by Santa-Maria et al. (2013) assessed LHW pretreatment on banana rachis and observed an 80% glucose yield in 44 h using 27 FPU/g glucan supplemented with β -glucosidase and xylanase. Even though liquid hot water offers several advantages, it is an energy intensive process requiring large quantities of water (Baruah et al., 2018).

3.4. Biological pretreatment

Biological pretreatments are usually performed by either isolating enzymes from microbes or cultivating the cells directly on the lignocellulosic substrates. Generally, biological pretreatment methods have been shown to be selective, do not require the use of auxiliary chemicals and the energy input is relatively low. Microbial-based or whole cell pretreatment utilizes bacteria or fungi containing enzymes for biological pretreatment. Two different extracellular enzyme types are present within microbes, known as hydrolytic and ligninolytic systems (Wagner et al., 2018). The hydrolytic enzyme system disrupts cellulose and hemicellulose, whereas the ligninolytic type removes lignin macromolecules. White-rot fungi are most commonly used, due to the higher lignin degradation efficiency and enzymatic

digestibility compared to the latter two. Examples of hydrolytic enzymes produced by whiterot fungi include cellulases and xylanases that hydrolyse cellulose and hemicellulose, respectively. Contrary to fungal pretreatment, limited studies are available using bacteria, since only few bacterial strains have demonstrated lignin-degrading ability. Bacteria metabolize and depolymerize lignin releasing high molecular weight metabolites such as organic acids (Chen and Nielsen, 2016). Baig (2005) treated banana pseudostem using a wild strain of Trichoderma lignorum to determine the amounts of enzymes produced. The three enzymes that were produced comprised of 0.18 U/mL FPase, 0.35 U/mL CMCase and 0.21 U/mL β-glucosidase. Reaction conditions included an optimum pH that ranged between 5.6 to 5.8 and an optimal temperature of 45 °C. More recently, Islamiyati and Asriany (2020) investigated the nutritive value of banana stem treated using Trichoderma sp. The banana stem was inoculated with 2.5, 5 and 7.5% inoculum for 2 weeks. The authors concluded that the higher levels of Trichoderma sp. increased the content of crude protein from 4.11 to 4.93% (Islamiyati and Asriany, 2020). Even with the advantage of selectivity when using biological pretreatments, major bottlenecks include the comparatively low sugar recovery due to the low hydrolysis rate and the long incubation times that range between 7 and 60 days (Maurya et al., 2015). Other limitations in fungi and bacterial lignin degradation occur as a result of: (1) strong and stable carbon-carbon and ether bonds that exist within the large lignin polymers that require oxidative systems for degradation and (2) simultaneously degrading lignin and sugar monomers, which leads to carbohydrates losses (Martinez et al., 2004).

3.5. Combined pretreatment strategies

Pretreatment methods have advanced drastically over time, with many studies employing combined methods that incorporate physical, chemical and/or biological strategies for the enhancement of lignocellulosic degradation towards enzymatic hydrolysis. Combined

pretreatment methods are able to effectively penetrate the LCB structure with several different effects on the biomass structure. Costa et al. (2018) determined the potential effect of a combined (acid-organosolv) pretreatment on banana rachis for fermentable sugar production. Glacial acetic acid and acetone were combined and applied to the substrate for 30 min at boiling temperature for the attainment of 16.6% glucose. In addition, the same authors investigated combined electro chemical activated solution using hypochlorous acid (electrolyzed 5 g/L NaCl) and ethanol (96%) and observed a higher glucose content of 48.7% when compared to the previous combined pretreatment. Combined pretreatments are rapidly researched due to the various advantages of effective degradation of the lignocellulosic structures. Nevertheless, the high cost, release of high concentration of inhibitor compounds, energy consumption, and complexity of downstream processing significantly reduce their feasibility (Sewsynker-Sukai and Gueguim Kana, 2018, Moodley et al., 2020). The development of waste-based combined pretreatment methods is highly sought after to improve the industrial feasibility (Sewsynker-Sukai et al., 2020).

Pretreatment	Pretreatment	Mechanism of action	Advantages	Disadvantages	References
classification	method				
Physical	Mechanical	Reduces substrate particle size and increases the substrate surface area and bulk density	Minimum inhibitor formation Eliminates mass and heat transfer limitations Increased surface area No use of chemicals Manage large biomass quantities	High energy requirement High maintenance costs Requires an additional pretreatment stage Not adequate as a single pretreatment	Yusuf and Inambao (2019)
	Microwave irradiation	Uses dielectric polarization that causes molecular collisions to generate thermal energy, leading to breakage of the complex lignocellulosic structures	Short residence time Rapid heat transfer Easy operation and controlled heating Environmentally friendly process Minimum formation of inhibitors Low-cost	Non-uniform heating Low penetration of irradiation energy in bulk materials, restricts its industrial scale application.	Aguilar-Reynosa et al. (2017)
	Ultrasound	Degrades the lignocellulosic material by creating shear forces that disrupt the intricate network structures of LCB	Short reaction time Low temperatures Low concentration of chemicals that are used	Energy intensive Further research is required to determine the optimum conditions for industrial scale application	Baruah et al. (2018), (Contreras-Hernández et al., 2018)
Chemical	Acid	Hydrolyses hemicellulose, thus making cellulose more accessible to enzymatic treatment	Removal of lignin and hemicellulose Dilute acid is more eco-friendly than concentrated acid, generating less inhibitors Flexible substrate choice	Expensive chemicals High quantities of inhibitor compounds High corrosivity of equipment	Hernández Beltrán et al. (2019), Fatma et al. (2018)

Table 3. The mode of action of various pretreatment strategies in combination with their advantages and disadvantages.

Pretreatment	Pretreatment	Mechanism of action	Advantages	Disadvantages	References
classification	method				
			High fermentable sugar yield at low temperatures	Requires specialized equipment	
	Alkali	Intermolecular saponification of ester bonds present within lignocellulosic biomass, improvement of the internal substrate surface area, reduces cellulose crystallinity and the degree of polymerization	Solubilizes hemicellulose and lignin Decreases polymerization and crystallinity High sugar yield High pretreatment efficiency Low energy usage Minimal inhibitor compound generation	Chemicals are expensive Large amount of fresh water required for washing Downstream recovery of the chemicals is troublesome Further experimentation is imperative	Kucharska et al. (2018), Qing et al. (2016)
	Solvent- dependent	Causes hemicellulose solubilization and lignin removal	Separates three different fractions consisting of dry lignin, an aqueous hemicellulose stream, and pure cellulose fraction. Generates high-quality lignin	Solvent may act as an inhibitor during hydrolysis reactions May be involved in side- reactions, thus, making its removal difficult prior to enzymatic hydrolysis	Kim and Yoo (2021), Hou et al. (2017)
	Ionic liquids	Pretreatment mechanism of ionic liquids is thought to be because of the dissolution of cellulose. The mechanism of action of ionic liquids is not entirely understood	High solvating properties Selectively cleave glycosidic bonds in polysaccharides	Industrial scale applicability is challenged Release of inhibitor compounds High costs	Raj et al. (2018), Xu et al. (2017), Kucharska et al. (2018)

Table 3. Continued.

Table 3. Continued.

Pretreatment	Pretreatment	Mechanism of action	Advantages	Disadvantages	References
classification	method				
	Inorganic salt	Generate metal cations that act as Lewis acids and cleave glycosidic linkages within lignocellulosic structures	Solubilizes hemicellulose Less corrosive Efficient at mild temperatures Low production of inhibitors Cost-effectiveness Recyclability	Expensive lab-based chemicals Partial disruption of lignocellulosic structures Low sugar yield	Loow et al. (2015), Liu et al. (2009)
Physico-	Steam	Degrades glycosidic	Minimal effects on the environment	Low lignin removal	Bhutto et al. (2017),
chemical	explosion	bonds within cellulose and hemicellulose and cleaves bonds between hemicellulose and lignin	Short pretreatment time Low chemical usage Reduced energy consumption	The release of inhibitory compounds Occurs under harsh conditions (high temperatures and/or long pretreatment times)	Carvalho et al. (2018)
	Ammonia	Leads to LCB swelling,	High selectivity of ammonia for	Corrosive reaction requires	Jędrzejczyk et al.
fibre explosion	which increases the	reaction with lignin spe	specialized equipment	(2019), Flores-Gómez	
	the hemicellulose	Easily recovered and recycled	Recycling processes is	et al. (2018)	
	(AFEX)	conversion to sugars, and alteration of the lignin structure	Enhances the industrial feasibility	energy intensive costly Ammonia poses safety and environmental concerns	
	CO_2	Fragments hemicellulose	Low-cost of CO ₂	High capital investment due	Yin et al. (2014),
explosion	explosion	and lignin structures	Minimal effect on the environment	to high-pressure conditions (13.8 MPa)	Capolupo and Faraco
		upon pressure release, and enhancing cellulose accessibility	Lack of inhibitor formation Ease of recovery enhances industrial scale potential		(2016)
Table 3. Continued.

Pretreatment	Pretreatment	Mechanism of action	Advantages	Disadvantages	References
classification	method				
	Liquid hot water	Cleaves acetyl groups in hemicellulose moieties and also causes the removal of lignin, which exposes the cellulose fibres to enzymatic hydrolysis	Little to no toxic compounds Does not require auxiliary catalysts Biomass size does not affect the process because the particles are fragmented using the pretreatment	Energy intensive process attributable to the excessive quantities of water that are required	Bhutto et al. (2017), Baruah et al. (2018)
Biological	Microbial or whole cell	The hydrolytic enzyme system within microorganisms disrupts cellulose and hemicellulose, whereas the ligninolytic type degrades lignin macromolecules	Selective Does not require the use of chemicals Energy input is relatively low No release of inhibitor compounds Easy downstream processing due to the lack of chemical use	Low sugar recovery and long incubation times White-rot fungi presents limitations in lignin degradation Simultaneous degradation of lignin and sugar monomers lead to carbohydrates losses	Wagner et al. (2018), Martinez et al. (2004)
Combined	Physical, chemical and/or biological strategies combined	Effectively penetrate the LCB structure with several different effects on the biomass structure	High pretreatment efficiency High sugar release	High cost Release of high concentration of inhibitor compounds Energy intensive Difficult downstream processing	Sewsynker-Sukai and Gueguim Kana (2018), Moodley et al. (2020)

Pretreatment method	Lignocellulosic substrate	Pretreatment parameters	Output	References
Mechanical	Banana pseudostem	Chopped and dried at 30 °C for 6 h, sieved through 60-mesh screen and oven dried at 105 °C for 24 h	1340% yield of activated carbon	Jiang et al. (2022)
	Banana stem	Cut into 1 cm sections, air dried and stored at -20 °C.	30.07% of cellulose, 10.65% hemicellulose and 17.10% lignin	Zhang et al. (2013)
Microwave irradiation	Banana waste	5% H ₂ SO ₄ , 25 min at 700 W	36.84% reducing sugar	Gabhane et al. (2014)
	Banana pseudostem	2% NaOH, 170 W, 10 minutes	300 mL bioethanol yield	Egwim and Shittu (2015)
Ultrasound	Banana pseudostem	50 °C for 60 min	41.98% reducing sugar	Gabhane et al. (2014)
Acid	Banana pseudostem	H ₂ SO ₄ concentrations (5, 10, 15, 20, 25, 30, 35 and 40%)	60 to 80% glucose yield	Shimizu et al. (2018)
	Banana pseudostem	5% (v/v) H ₂ SO ₄ , 10 FPU/g, 50 °C	21.8 g/L glucose	Legodi et al. (2021)
Alkali	Banana stem	6% NaOH, 36 °C	232.4 mL/g biomethane yield	Zhang et al. (2013)
	Banana pseudostem	0.5 to 2 N NaOH, 90 min, 121 °C	29.8 g/L sugar yield	Sukmaningtyas (2018)
Ionic liquid	Banana stem	1 g substrate, 19 mL 1-Allyl-3- methylimidazolium chloride, 80 to 90 °C, 200 rpm, 3 to 4 h	23.1 µm translucent cellulose film	Ai et al. (2021)

Table 4. The application of diverse pretreatment strategies on banana pseudostem and similar derivatives of the banana plant.

Table 4. Continued.

Pretreatment method	Lignocellulosic	Pretreatment parameters	Output	References
	substrate			
Inorganic salt	Banana stem	10 g substrate, 0.1 M KCl, ZnCl ₂ , 700 °C, 2 h	High porous surface area of 1465 m ² g ⁻¹	Madhu et al. (2014)
Steam explosion	Banana stem	100 L reactor, 150 g solids, 190 to 200 °C	93% glucan conversion yield	Santa-Maria et al. (2013)
	Banana pseudostem	2.2% H ₂ SO ₄ , 5 min, 177 °C	91% glucose yield	Guerrero et al. (2017)
Ammonia fibre explosion	Banana peel waste	24.5% NH ₃ , 120 °C	9.6% and 3.5% biomass conversion yield	Isibika et al. (2019)
CO ₂ explosion	Banana pseudostem	120 to 280 °C at 5 MPa, 700 rpm for 5 min	1340% yield of activated carbon	Jiang et al. (2022)
Liquid hot water	Banana pseudostem	Solid loading of 15% w/v, 121 °C, 15 psi for 1 h	21.8 g/L glucose	Legodi et al. (2021)
	Banana rachis	44 h, 27 FPU/g glucan supplemented with β -glucosidase and xylanase	80% glucose yield	Santa-Maria et al. (2013)
Biological pretreatment	Banana pseudostem	<i>Trichoderma lignorum,</i> pH 5.6 to 5.8, 45 °C	0.18 U/mL FPase, 0.35 U/mL CMCase and 0.21 U/mL β- glucosidase	Baig (2005)
	Banana stem	<i>Trichoderma sp.</i> , 2.5, 5 and 7.5% inoculum, 2 weeks	4.93% crude protein	Islamiyati and Asriany (2020)
Combined pretreatment	Banana rachis	Glacial acetic acid, acetone, 30 min	16.6% glucose	Costa et al. (2018)
	Banana rachis	Hypochlorous acid (electrolyzed 5 g/L NaCl) and ethanol (96%)	48.7% glucose	Costa et al. (2018)

4. Lignocellulosic bioprocess types

A lignocellulosic bioprocess is based on the utilization of lignocellulosic biomass to produce marketable products such as bio-based fuels and chemicals via chemical and biochemical routes (Konwar et al., 2018). Pretreatment is one step that is necessary to improve the accessibility of polysaccharides, making it amenable to enzymatic hydrolysis for subsequent bioprocessing. The fermentation of LCB to valuable bio-based products can be achieved by using different bioprocess types that include separate hydrolysis and fermentation (SHF), simultaneous saccharification and fermentation (SSF) or SSF with a short separate hydrolysis step (PSSF) (Dashtban et al., 2009). In SHF processes, enzymatic hydrolysis and fermentation are performed in separate vessels under specific reaction conditions (Fatma et al., 2018). An advantage of this approach is that specific processes can operate at their optimum setpoints (i.e., optimal temperature ranges between 45 to 50 °C for hydrolysis, whereas it is 30-37°C for fermentation) (Dashtban et al., 2009). However, the main problems associated with SHF are end-product inhibition of enzymes, increased costs, and higher risk of contamination (Dashtban et al., 2009). Major drawbacks of the SHF process may be overcome by the development of the SSF approach in which hydrolysis and fermentation both occur simultaneously in a single vessel (da Silva et al., 2018). The integration of enzymatic hydrolysis and fermentation decreases the number of unit operations, thus saving processing costs and production time. Additionally, the direct conversion of sugars to fermentation products eliminates issues such as sugar accumulation, enzyme inhibition, and the risk for contamination (Öhgren et al., 2007, Mithra et al., 2018). A variation of the SSF configuration is the pre-hydrolysis and simultaneous saccharification and fermentation (PSSF) process. In the PSSF process, the substrate is subjected to a pre-hydrolysis step for short time at an optimum temperature for cellulolytic enzymes (50 °C), following this, the temperature of the bioreactor is lowered for inoculation of the microorganism (Paschos et al., 2020, Carrillo-Nieves et al., 2017). The main advantage of PSSF over SHF is that it simplifies the process by avoiding the separation of the slurry before fermentation, thus saving processing time and energy. When compared to the SSF configuration, the rate of enzymatic hydrolysis is not reduced by the suboptimal temperature and the production rate is not limited by the low concentration of carbon source (Morales-Martínez et al., 2017). In addition, PSSF is performed at its optimum temperature stimulating the release of sugar and lowering the substrate viscosity (Carrillo-Nieves et al., 2017). In the recent years, the application of SSF processes has been recommended owing to its versatility and ease of applicability in large scale biorefineries. Furthermore, in the SSF process, the enzyme and microorganism are in a synergy in a single step resulting in the immediate conversion of lignocellulosic biomass to valuable products.

4.1. Development of valuable bio-based products

The global demand for organic acid has exceeded the natural supply, hence sustainable biotechnological fermentation processes have become imperative (Odu et al., 2020). Industrially important organic acids include citric acid, lactic acid and itaconic acid (Chen and Nielsen, 2016). Citric acid is one of the most important organic acids with 2 million tons commercially produced annually (Aboyeji et al., 2020). In comparison, approximately 335 000 tons of lactic acid and 80 000 tons of itaconic acid are generated each year (Saha et al., 2019). Citric acid is an essential building block of cellular respiration that is incorporated into the food, beverage, pharmaceutical, chemical, cosmetic and textile industries (Sarkar and Das, 2017, Sawant, 2018). In addition, lactic acid is a precursor for commercially important polylactic acid, polylactide (biopolymer), and ethyl lactate, a bio-based solvent (Konwar et al., 2018). Itaconic acid can be used as a co-monomer of thermoplastics, detergents, acrylic polymers, surfactants and polyester resins (El-Imam and Du, 2014). Within industry, citric acid, lactic acid and itaconic acid are mainly produced by microorganisms such as *Aspergillus*

niger, Lactobacillus sp., and *Aspergillus terreus*, respectively (Aboyeji et al., 2020, Idrees et al., 2013, Kolláth et al., 2019). The production pathway also known as the tricarboxylic acid cycle (TCA) generates these microbial metabolites. Several studies have demonstrated the production of these bio-based products from various lignocellulosic substrates (See supplementary material), however, banana pseudostem is an abundant, perennial and renewable lignocellulosic waste, rich in cellulose and low lignin content that renders it feasible for the bioconversion to value-added products (Narinthorn et al., 2019, Shimizu et al., 2018). Additionally, its physicochemical properties allow for low-cost and innovative pretreatment and fermentation technologies that may provide financial sustenance for adjacent sectors. Furthermore, the utilization of BP will combat the implications that are associated with its disposal (airborne pathogens and environmental pollution).

Even with the availability of these studies, a number of experimental conditions must be considered to improve organic acid production by modelling, optimization and kinetic assessment. Due to the complex dynamics of the biorefinery process, there is a dearth of knowledge for understanding the bioprocess and its product output. Mathematical kinetic models are useful for providing knowledge on the metabolic aspects of a fermentation process (Germec et al., 2019). In addition, kinetic models play an essential role in simulating, optimizing, monitoring and predicting the performance of a process under varying parameter conditions (Pramanik et al., 2019). Furthermore, kinetic modelling allows for increased product yield and productivity, to ensure high product quality while reducing the formation of unwanted by-products and saving processing time and costs.

5. Current challenges and future prospectives

5.1. Lignocellulosic biomass selection

The commercial and industrial sectors are seeking novel, cost-effective, and economical biorefinery technologies to keep up with the current adverse market conditions (Raj et al., 2022). In light of this, a cost reduction can be achieved by utilizing less expensive feedstocks such as agro-industrial substrates (lignocellulosic biomass) and their derivatives. However, the biomass recalcitrance of lignocellulosic biomass renders it to be chemically complex in nature and to resist the release of fermentable sugars, thus necessitating pretreatment to overcome these rheological properties and ensure adequate cellulose and hemicellulose availability (Chen and Nielsen, 2016). In addition, the usage of food-based agricultural feedstocks threatens global food security and the economy. Hence, the application of non-food-based feedstocks are highly sought after. Large quantities of banana pseudostem waste end up in land and water bodies, thus occupying space and contributing to adverse environmental impacts when it is incinerated. Banana pseudostem does not coincide with food security and its utilization will minimize the release of detrimental greenhouse gases into the environment. The high cellulosic (42-60%) and hemicellulosic (19-23%) content of banana pseudostem makes it a suitable feedstock for the generation of valuable microbial bio-based products (Islam et al., 2019, Shimizu et al., 2018). Bananas are cultivated in over 135 countries throughout tropical and subtropical climates (Magdama et al., 2020). Furthermore, unlike other LCB substrates from seasonal-dependant crops, bananas are available all year round (Subagyo and Chafidz, 2018). Therefore, the cultivation and harvesting of bananas (along with the BP) is steady with a yearround supply, that is crucial for effective functioning of lignocellulosic bioprocessing operations.

5.2. Development of cost-effective and/or waste-based pretreatments

A severe challenge of lignocellulosic bioconversion is the pretreatment costs incurred due to the energy, chemicals and fresh water consumption. Numerous pretreatment technologies require fresh water and organic solvents in at least one of the stages. Conventional pretreatment methods utilize expensive solvents that are volatile and potentially toxic resulting in low sugar yields. Moreover, the disposal of spent pretreatment chemicals is problematic and costly, while commonly used acidic chemicals tend to corrode the bioreactor (Hernández Beltrán et al., 2019). On the other hand, the Kraft paper and pulp industry produces high quantities of waste products that include, mesa lime, green liquor dregs (GLD), paper wastewater (PWW), fibre sludge and fly ash (Mäkitalo et al., 2014), that have emerged as a potential solution to chemical-based pretreatments. Some of these by-products have been employed for lignocellulosic pretreatment and are summarized (See supplementary material). More specifically, the Kraft paper and pulp industry is one of the third largest producers of paper wastewater (PWW) after primary metals and chemical industries (Ashrafi et al., 2015). PWW contains effluent solids, sediments, absorbable organic halides and chlorinated organic compounds that contribute majorly to its high pH and thus alkaline nature (Tobin et al., 2020). Main challenges encountered by the Kraft industry include paper wastewater accumulation and treatment costs that are incurred approximating to a chemical cost of \$2.09 USD per cubic meter of wastewater treated. In addition, activated-carbon treatment costs range between \$1 to \$3 USD/m³ for the treatment of pulp and paper mill wastewater (Mehmood et al., 2019). Recently, Kraft paper wastewater and table salt have garnered significant attention as effective pretreatment technologies instead of commonly used alkaline and inorganic salt, respectively. Unlike lab-based NaCl, iodized table salt is a cheaper alternative. This salt is an ionic complex composed of sodium and chloride ions with the chemical composition of NaCl (Carapeto et al., 2018). Inorganic salts function by dissociating Lewis acids in an aqueous solvent to form hydrated complex cations surrounded by water molecules. These water molecules further act as nucleophiles, thus depolymerizing hemicellulose into monosaccharides (Moodley et al., 2020). In the same vein, the utilization of Kraft paper wastewater to treat lignocellulosic waste instead of alkaline chemicals such as NaOH provides an economically viable pretreatment option. The alkaline properties of PWW allow it to cleave the ester linkages of lignin and the cross-linkages between hemicellulose and lignin (Saratale et al., 2016). Furthermore, the degradation and removal of lignin moieties increases the solubility of LCB and therefore improves the accessibility of cellulose to enzymatic hydrolysis (Saratale et al., 2016, Gu et al., 2012). The properties of paper wastewater enable it to function as a dual-purpose commodity in pretreatment processes as it exhibits characteristics as an attractive replacement for: (1) fresh water usage within lignocellulosic bioprocessing systems and (2) common and expensive alkaline reagents. Reusing PWW for lignocellulosic pretreatment could potentially generate profits that may then be utilized for biohazardous treatment and management, ultimately reducing the cost of downstream processing for Kraft industries.

Research has shown that exposure of pretreated lignocellulosic polysaccharides to cellulases for enzymatic saccharification processes at a reduced cost is of a great importance (Phitsuwan et al., 2016). However, this can only be effectively achieved through pretreatment strategies that increase the biomass polymers' reactive surface area and porosity for penetration by enzymes. Furthermore, reduced cellulose crystallinity and low lignin content are desirable (Phitsuwan et al., 2016, Raj et al., 2022).

5.3. Lignocellulosic bioprocess optimization

The formation of a valuable product necessitates a specific set of parameter ranges for the optimization of the bioprocess. The addition of a desorbent, enzyme, inoculum, and substrate solid loading may be utilized in the bioprocess but require optimization for optimal fermentation and product formation. For example, in citric acid production, nitrogen is a key requirement, therefore it is essential to determine an alternative to expensive lab-based derivatives by using low-cost nitrogen sources. In addition to nitrogen, desorbents have been

shown to enhance citric acid production by improving the cell permeability to citrate which enables the cell to respond to the diminished intracellular level by increasing citric acid production (Sekoai et al., 2018). Temperature is one of the most influential factors in lignocellulosic citric acid production. This is attributed to the presence of the cellulase-based enzyme and fungal culture. It is therefore advantageous to employ the SSF process to ensure the simultaneous conversion of glucose to a valuable microbial based bioproduct. This process is currently preferred over conventional SHF and PSSF due to the minimization of contamination, energy, time, resources and processing costs. Additionally, the substitution of wastewater as a water source during the SSF bioprocess could prove valuable from an environmental, financial, and social perspective. The dairy sector contributes to a financially stable economy that generates significant amounts of wastewater and may be channelled towards lignocellulosic bioprocesses. Furthermore, mathematical models play a vital role in optimizing, predicting, simulating, and monitoring the performance of a process under varying process conditions (Pramanik et al., 2019). The interactive effects of input parameters and their corresponding output are analyzed using the response surface methodology model. Despite advancements in the process optimization, there needs to be further research attempted to enhance the ability to recover and reuse spent chemicals (paper wastewater) and by-products (dairy wastewater) from the Kraft and dairy industry, respectively. These would be a valuable substitute for large amounts of finite fresh water to eliminate its usage from lignocellulosic bioprocesses and improve the overall cost-efficiency. Moreover, the implementation of mathematical models provides a strong foundation for process design, control and optimization, which may reduce the challenges faced during lignocellulosic bioprocessing (Linville et al., 2013).

6. Conclusion

This review highlights banana pseudostem (BP) as a potential renewable resource for producing valuable microbial by-products. The different pretreatments on BP with its advantages and limitations are discussed. Furthermore, the various bioprocess types and use of mathematical kinetic models are outlined. The challenges faced by lignocellulosic bioprocesses and future perspective on energy efficient and cost-competitive technologies during pretreatment, enzymatic hydrolysis and microbial fermentation has been presented. The development of these technologies could potentially transform the face of these systems, aligned with global sustainable goals regarding the water-food-energy nexus for a circular bioeconomy approach.

Acknowledgements

This work is established on research funded partially by the National Research Foundation (NRF) of South Africa (Grant number 122341). Opinions conveyed and conclusions derived, are of the authors and not ascribed to the NRF.

Appendix A. Supplementary data

Supplementary data associated with this article can be found, in the online version.

Credit authorship contribution statement

Milesh Laltha: Conceptualization, Investigation, Writing - original draft, Writing - review & editing, Visualization, Project administration. **Y. Sewsynker-Sukai:** Conceptualization, Writing – review & editing, Project administration, Supervision. **E.L. Meyer:** Writing – review & editing. **E.B. Gueguim Kana:** Writing – review & editing, Supervision.

References

- Aboyeji, O., Oloke, J., Arinkoola, A., Oke, M., Ishola, M.M. 2020. Optimization of media components and fermentation conditions for citric acid production from sweet potato peel starch hydrolysate by Aspergillus niger. Scientific African, 10, e00554.
- Aguilar-Reynosa, A., Romaní, A., Ma. Rodríguez-Jasso, R., Aguilar, C.N., Garrote, G., Ruiz, H.A. 2017. Microwave heating processing as alternative of pretreatment in second-generation biorefinery: An overview. Energy Conversion and Management, 136, 50-65.
- Ai, B., Zheng, L., Li, W., Zheng, X., Yang, Y., Xiao, D., Shi, J., Sheng, Z. 2021. Biodegradable cellulose film prepared from banana pseudo-stem using an ionic liquid for mango preservation. Frontiers in plant science, 12, 625878.
- 4. Ashrafi, O., Yerushalmi, L., Haghighat, F. 2015. Wastewater treatment in the pulp-and-paper industry: A review of treatment processes and the associated greenhouse gas emission. Journal of Environmental Management, 158, 146-157.
- 5. Badiei, M., Asim, N., Jahim, J.M., Sopian, K. 2014. Comparison of chemical pretreatment methods for cellulosic biomass. APCBEE procedia, 9, 170-174.
- 6. Baig, M. 2005. Cellulolytic enzymes of Trichoderma lignorum produced on banana agrowaste: Optimisation of culture medium and conditions, 64, 57-60.
- Baruah, J., Nath, B.K., Sharma, R., Kumar, S., Deka, R.C., Baruah, D.C., Kalita, E. 2018. Recent trends in the pretreatment of lignocellulosic biomass for value-added products. Frontiers in Energy Research, 6, 141.
- 8. Begum, A., Chittibabu, S. 2015. Microwave assisted lime pretreatment of banana pseudostem for ethanol production. International Journal of ChemTech Research, 7, 2287-2295.
- Bhutto, A.W., Qureshi, K., Harijan, K., Abro, R., Abbas, T., Bazmi, A.A., Karim, S., Yu, G.
 2017. Insight into progress in pre-treatment of lignocellulosic biomass. Energy, 122, 724-745.

- Brodeur, G., Yau, E., Badal, K., Collier, J., Ramachandran, K.B., Ramakrishnan, S. 2011. Chemical and Physicochemical Pretreatment of Lignocellulosic Biomass: A Review. Enzyme Research, 2011, 1-17.
- Capolupo, L., Faraco, V. 2016. Green methods of lignocellulose pretreatment for biorefinery development. Applied microbiology and biotechnology, 100, 9451-9467.
- Carapeto, C., Brum, S., Rocha, M. 2018. Which Table Salt to Choose? Journal of Nutrition & Food Sciences, 08.
- Carrillo-Nieves, D., Ruiz, H.A., Aguilar, C.N., Ilyina, A., Parra-Saldivar, R., Torres, J.A., Hernández, J.L.M. 2017. Process alternatives for bioethanol production from mango stem bark residues. Bioresource technology, 239, 430-436.
- 14. Carvalho, A.F.A., Marcondes, W.F., De Oliva Neto, P., Pastore, G.M., Saddler, J.N., Arantes,
 V. 2018. The potential of tailoring the conditions of steam explosion to produce xylooligosaccharides from sugarcane bagasse. Bioresource technology, 250, 221-229.
- 15. Cavalaglio, G., Gelosia, M., Ingles, D., Pompili, E., D'antonio, S., Cotana, F. 2016. Response surface methodology for the optimization of cellulosic ethanol production from Phragmites australis through pre-saccharification and simultaneous saccharification and fermentation. Industrial Crops and Products, 83, 431-437.
- Chen, H., Liu, J., Chang, X., Chen, D., Xue, Y., Liu, P., Lin, H., Han, S. 2017a. A review on the pretreatment of lignocellulose for high-value chemicals. Fuel Processing Technology, 160, 196-206.
- Chen, X., Kuhn, E., Nagle, N., Nelson, R., Tao, L., Crawford, N., Tucker, M. 2018. Recycling of dilute deacetylation black liquor to enable efficient recovery and reuse of spent chemicals and biomass pretreatment waste. Frontiers in Energy Research, 6, 51.
- Chen, X.L., Y.X., Han, Zhang, C.M., Feng, G.Q., Zhao, M.X., Yue, R.X., Li, Y.F., Jiang, L.,
 Zhang, L., Li, J.H. 2017b. Alkaline Pretreatment of Banana Stems for Methane Generation:

Effects of Temperature and Physicochemical Changes. Bioresource Technology, 12, 5601-5616.

- Chen, Y., Nielsen, J. 2016. Biobased organic acids production by metabolically engineered microorganisms. Curr. Opin. Biotechnol., 37, 165-172.
- 20. Chundawat, S., Balan, V., Da Costa Sousa, L., Dale, B. 2010. Thermochemical pretreatment of lignocellulosic biomass, Bioalcohol production. Woodhead Publishing, 24-72.
- 21. Contreras-Hernández, M., Ochoa-Martínez, L., Rutiaga-Quiñones, J., Rocha-Guzmán, N., Lara-Ceniceros, T., Contreras-Esquivel, J., Barragán, L.P., Rutiaga-Quiñones, O. 2018. Effect of ultrasound pre-treatment on the physicochemical composition of Agave durangensis leaves and potential enzyme production. Bioresource technology, 249, 439-446.
- Costa, S., Rugiero, I., Larenas Uria, C., Pedrini, P., Tamburini, E. 2018. Lignin degradation efficiency of chemical pre-treatments on banana rachis destined to bioethanol production. Biomolecules, 8, 141.
- 23. Da Silva, F.L., De Oliveira Campos, A., Dos Santos, D.A., Batista Magalhães, E.R., De Macedo, G.R., Dos Santos, E.S. 2018. Valorization of an agroextractive residue—Carnauba straw—for the production of bioethanol by simultaneous saccharification and fermentation (SSF). Renewable Energy, 127, 661-669.
- 24. Da Silva, I.F., De Queiroz, J.H., Da Luz, J.M.R., De Faria, N.R.B., Colodette, J.L., Machado, S.S., Santos, F.A., Farias, L.M. 2019. Optimization Of Hydrothermal Pretreatment For Enzymatic Hydrolysis Of Banana Pseudo Stem Using Response Surface Methodology. Fungal Territory, 4, 32-38.
- Dashtban, M., Schraft, H., Qin, W. 2009. Fungal bioconversion of lignocellulosic residues; opportunities & perspectives. International journal of biological sciences, 5, 578-595.

- 26. David, A., Sewsynker-Sukai, Y., Sithole, B., Gueguim Kana, E.B. 2020. Development of a green liquor dregs pretreatment for enhanced glucose recovery from corn cobs and kinetic assessment on various bioethanol fermentation types. Fuel, 274.
- De Bhowmick, G., Sarmah, A.K., Sen, R. 2018. Lignocellulosic biorefinery as a model for sustainable development of biofuels and value added products. Bioresource technology, 247, 1144-1154.
- De Paula, R.G., Antoniêto, A.C.C., Ribeiro, L.F.C., Srivastava, N., O'donovan, A., Mishra,
 P.K., Gupta, V.K., Silva, R.N. 2019. Engineered microbial host selection for value-added bioproducts from lignocellulose. Biotechnol. Adv., 37, 107347.
- Egwim, E., Shittu, O. 2015. Microwave-Alkaline Assisted Pretreatment of Banana Trunk for Bioethanol Production. Journal of Energy and Power Engineering, 9, 1-88.
- El-Imam, A. A., Du, C. 2014. Fermentative itaconic acid production. J Biodivers Biopros Dev, 1, 1-8.
- Fatma, S., Hameed, A., Noman, M., Sohail, I., Shahid, M., Tariq, M., Tabassum, R. 2018.
 Lignocellulosic Biomass: A Sustainable Bioenergy Source for Future. Protein & Peptide Letters, 25.
- Flores-Gómez, C.A., Escamilla Silva, E.M., Zhong, C., Dale, B.E., Da Costa Sousa, L., Balan,
 V. 2018. Conversion of lignocellulosic agave residues into liquid biofuels using an AFEXTM based biorefinery. Biotechnology for biofuels, 11, 1-18.
- 33. Gabhane, J., Prince William, S.P.M., Gadhe, A., Rath, R., Vaidya, A.N., Wate, S. 2014. Pretreatment of banana agricultural waste for bio-ethanol production: Individual and interactive effects of acid and alkali pretreatments with autoclaving, microwave heating and ultrasonication. Waste Management, 34, 498-503.

- Germec, M., Turhan, I., Karhan, M., Demirci, A.J.a.S. 2019. Kinetic Modeling and Technoeconomic Feasibility of Ethanol Production From Carob Extract Based Medium in Biofilm Reactor. 9, 2121.
- Gu, F., Yang, L., Jin, Y., Han, Q., Chang, H.-M., Jameel, H., Phillips, R. 2012. Green liquor pretreatment for improving enzymatic hydrolysis of corn stover. Bioresource Technology, 124, 299-305.
- Guerrero, A.B., Ballesteros, I., Ballesteros, M. 2017. Optimal conditions of acid-catalysed steam explosion pretreatment of banana lignocellulosic biomass for fermentable sugar production. J. Chem. Technol. Biotechnol., 92, 2351-2359.
- 37. Hernández Beltrán, J., Hernandez De Lira, I., Cruz-Santos, M., Saucedo-Luevanos, A., Hernández-Terán, F., Balagurusamy, N. 2019. Insight into Pretreatment Methods of Lignocellulosic Biomass to Increase Biogas Yield: Current State, Challenges, and Opportunities. Applied Sciences, 9, 3721.
- Himmel, M.E., Ding, S.-Y., Johnson, D.K., Adney, W.S., Nimlos, M.R., Brady, J.W., Foust, T.D.J.S. 2007. Biomass recalcitrance: engineering plants and enzymes for biofuels production. 315, 804-807.
- Hou, Q., Ju, M., Li, W., Liu, L., Chen, Y., Yang, Q. 2017. Pretreatment of lignocellulosic biomass with ionic liquids and ionic liquid-based solvent systems. Molecules, 22, 490.
- 40. Ibarra-Gonzalez, P., Rong, B.-G. 2019. A review of the current state of biofuels production from lignocellulosic biomass using thermochemical conversion routes. Chinese Journal of Chemical Engineering, 27, 1523-1535.
- 41. Idrees, M., Adnan, A., Malik, F., Qureshi, F.A. 2013. Enzymatic saccharification and lactic acid production from banana pseudo-stem through optimized pretreatment at lowest catalyst concentration. EXCLI journal, 12, 269.

- 42. Ingale, S., Joshi, S.J., Gupte, A. 2014. Production of bioethanol using agricultural waste: banana pseudo stem. Braz. J. Microbiol., 45, 885-892.
- 43. Isibika, A., Vinnerås, B., Kibazohi, O., Zurbrügg, C., Lalander, C. 2019. Pre-treatment of banana peel to improve composting by black soldier fly (Hermetia illucens (L.), Diptera: Stratiomyidae) larvae. Waste Management, 100, 151-160.
- 44. Islam, M., Asad, M., Hossain, M., Paul, S., Sujan, S. 2019. Bioethanol production from banana pseudostem by using separate and cocultures of cellulase enzyme with Saccharomyces cerevisiae. Journal of Environmental Science Technology, 12, 157-163.
- Islamiyati, R., Asriany, A. 2020. Nutritive value of banana stem with applicative technology Trichoderma sp. as ruminant feed. IOP Conference Series: Earth and Environmental Science, IOP Publishing, 012016.
- 46. Jędrzejczyk, M., Soszka, E., Czapnik, M., Ruppert, A.M., Grams, J. 2019. Physical and chemical pretreatment of lignocellulosic biomass. Second and third generation of feedstocks, 143-196.
- Jiang, F., Cao, D., Hu, S., Wang, Y., Zhang, Y., Huang, X., Zhao, H., Wu, C., Li, J., Ding, Y.,
 Liu, K. 2022. High-pressure carbon dioxide-hydrothermal enhance yield and methylene blue adsorption performance of banana pseudo-stem activated carbon. Bioresource Technology, 354, 127137.
- Kang, K.E., Park, D.-H., Jeong, G.-T. 2013. Effects of inorganic salts on pretreatment of Miscanthus straw. Bioresource technology, 132, 160-165.
- 49. Kang, Q., Appels, L., Tan, T., Dewil, R. 2014. Bioethanol from Lignocellulosic Biomass:Current Findings Determine Research Priorities. The Scientific World Journal, 2014, 1-13.
- 50. Kim, K.H., Yoo, C.G. 2021. Challenges and Perspective of Recent Biomass Pretreatment Solvents. Frontiers in Chemical Engineering, 61.

- 51. Kim, S., Kim, S.-D., Sohn, S.Y. 2020. Evaluation of the wastewater generated during alkaline pretreatment of biomass for feasibility of recycling and reusing. Renewable energy, 155, 1156-1164.
- 52. Kolláth, I.S., Molnár, Á.P., Soós, Á., Fekete, E., Sándor, E., Kovács, B., Kubicek, C.P., Karaffa, L. 2019. Manganese Deficiency Is Required for High Itaconic Acid Production From D-Xylose in Aspergillus terreus. Frontiers in Microbiology, 10.
- 53. Konwar, L.J., Mikkola, J.-P., Bordoloi, N., Saikia, R., Chutia, R.S., Kataki, R. 2018.
 Sidestreams from bioenergy and biorefinery complexes as a resource for circular bioeconomy.
 Waste Biorefinery, 85-125.
- Kucharska, K., Rybarczyk, P., Hołowacz, I., Łukajtis, R., Glinka, M., Kamiński, M. 2018.
 Pretreatment of lignocellulosic materials as substrates for fermentation processes. Molecules, 23, 2937.
- 55. Kumar, B., Bhardwaj, N., Agrawal, K., Chaturvedi, V., Verma, P. 2020. Current perspective on pretreatment technologies using lignocellulosic biomass: An emerging biorefinery concept. Fuel Process. Technol., 199, 106244.
- 56. Lachos-Perez, D., Brown, A., Mudhoo, A., Martinez, J., Timko, M., Rostagno, M., Forster-Carneiro, T. 2017. Applications of subcritical and supercritical water conditions for extraction, hydrolysis, gasification, and carbonization of biomass: a critical review. Biofuel Research Journal, 4, 611.
- 57. Laltha, M., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2021. Development of microwaveassisted alkaline pretreatment methods for enhanced sugar recovery from bamboo and corn cobs: Process optimization, chemical recyclability and kinetics of bioethanol production. Industrial Crops and Products, 174, 114166.

- 58. Legodi, L.M., Lagrange, D.C., Jansen Van Rensburg, E.L., Ncube, I. 2021. Enzymatic hydrolysis and fermentation of banana pseudostem hydrolysate to produce bioethanol. Int. J. Microbiol, 2021, 1-14.
- 59. Li, K., Fu, S., Zhan, H., Zhan, Y., Lucia, L.J.B. 2010. Analysis of the chemical composition and morphological structure of banana pseudo-stem. Bioresources, 5, 576-585.
- Limayem, A., Ricke, S.C. 2012. Lignocellulosic biomass for bioethanol production: current perspectives, potential issues and future prospects. Progress in energy and combustion science, 38, 449-467.
- 61. Liu, L., Sun, J., Cai, C., Wang, S., Pei, H., Zhang, J. 2009. Corn stover pretreatment by inorganic salts and its effects on hemicellulose and cellulose degradation. Bioresource technology, 100, 65-71.
- Liu, Y.K., Chen, W.C., Huang, Y.C., Chang, Y.K., Chu, I.M., Tsai, S.L., Wei, Y.H. 2017. Production of bioethanol from Napier grass via simultaneous saccharification and cofermentation in a modified bioreactor. Journal of Bioscience and Bioengineering, 124, 184-188.
- 63. Loow, Y.-L., Wu, T.Y., Tan, K.A., Lim, Y.S., Siow, L.F., Md. Jahim, J., Mohammad, A.W., Teoh, W.H. 2015. Recent Advances in the Application of Inorganic Salt Pretreatment for Transforming Lignocellulosic Biomass into Reducing Sugars. Journal of Agricultural and Food Chemistry, 63, 8349-8363.
- Lu, H., Yadav, V., Bilal, M., Iqbal, H.M.N. 2022. Bioprospecting microbial hosts to valorize lignocellulose biomass – Environmental perspectives and value-added bioproducts. Chemosphere, 288, 132574.
- 65. Madhu, R., Veeramani, V., Chen, S.-M. 2014. Heteroatom-enriched and renewable bananastem-derived porous carbon for the electrochemical determination of nitrite in various water samples. Sci. Rep., 4, 1-8.

- Magdama, F., Monserrate-Maggi, L., Serrano, L., García Onofre, J., Jiménez-Gasco, M.D.M.
 2020. Genetic Diversity of Fusarium oxysporum f. sp. cubense, the Fusarium Wilt Pathogen of Banana, in Ecuador. Plants, 9, 1133.
- 67. Mäkitalo, M., Maurice, C., Jia, Y., Öhlander, B. 2014. Characterization of green liquor dregs, potentially useful for prevention of the formation of acid rock drainage. Minerals, 4, 330-344.
- Martinez, D., Larrondo, L.F., Putnam, N., Gelpke, M.D.S., Huang, K., Chapman, J., Helfenbein, K.G., Ramaiya, P., Detter, J.C., Larimer, F. 2004. Genome sequence of the lignocellulose degrading fungus Phanerochaete chrysosporium strain RP78. Nat. Biotechnol., 22, 695-700.
- 69. Maurya, D.P., Singla, A., Negi, S. 2015. An overview of key pretreatment processes for biological conversion of lignocellulosic biomass to bioethanol. Biotechnology, *5*, 597-609.
- Mehmood, K., Rehman, S.K.U., Wang, J., Farooq, F., Mahmood, Q., Jadoon, A.M., Javed, M.F., Ahmad, I. 2019. Treatment of pulp and paper industrial effluent using physicochemical process for recycling. Water, 11, 2393.
- 71. Mithra, M.G., Jeeva, M.L., Sajeev, M.S., Padmaja, G. 2018. Comparison of ethanol yield from pretreated lignocellulo-starch biomass under fed-batch SHF or SSF modes. Heliyon, 4, e00885.
- 72. Mohammed, I.Y., Abakr, Y.A., Mokaya, R. 2019. Biofuel and valuable products recovery from Napier grass pre-processing: Process design and economic analysis. Journal of Environmental Chemical Engineering, 7, 102962.
- 73. Moodley, P., Gueguim Kana, E.B. 2017. Development of a steam or microwave-assisted sequential salt-alkali pretreatment for lignocellulosic waste: Effect on delignification and enzymatic hydrolysis. Energy Conversion and Management, 148, 801-808.
- Moodley, P., Gueguim Kana, E.B. 2019. Bioethanol production from sugarcane leaf waste:
 Effect of various optimized pretreatments and fermentation conditions on process kinetics.
 Biotechnology Reports, 22, e00329.

- 75. Moodley, P., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2020. Progress in the development of alkali and metal salt catalysed lignocellulosic pretreatment regimes: Potential for bioethanol production. Bioresource Technology, 310, 123372.
- 76. Morales-Martínez, T.K., Díaz-Blanco, D.I., Rodríguez-De La Garza, J.A., Morlett-Chávez, J., Castro-Montoya, A.J., Quintero, J., Aroca, G., Rios-González, L.J. 2017. Assessment of different saccharification and fermentation configurations for ethanol production from Agave lechuguilla. BioResources, 12, 8093-8105.
- 77. Mupondwa, E., Li, X., Tabil, L., Sokhansanj, S., Adapa, P. 2017. Status of Canada's lignocellulosic ethanol: Part I: Pretreatment technologies. Renewable and Sustainable Energy Reviews, 72, 178-190.
- 78. Mussatto, S.I., Dragone, G. 2016. Biomass pretreatment, biorefineries, and potential products for a bioeconomy development. Biomass fractionation technologies for a lignocellulosic feedstock based biorefinery, 1-22.
- 79. Mustofa, A. 2018. Bioethanol production from banana stem by using simultaneous saccharification and fermentation (SSF). IOP Publishing, 358, 012004.
- Nagahata, R., Takeuchi, K. 2019. Encouragements for the Use of Microwaves in Industrial Chemistry. The Chemical Record, 19, 51-64.
- 81. Narinthorn, R., Choorit, W., Chisti, Y. 2019. Alkaline and fungal pretreatments for improving methane potential of Napier grass. Biomass and Bioenergy, 127, 105262.
- Nosratpour, M.J., Karimi, K., Sadeghi, M. 2018. Improvement of ethanol and biogas production from sugarcane bagasse using sodium alkaline pretreatments. Journal of Environmental Management, 226, 329-339.
- 83. Odu, N., Uzah, G., Akani, N. 2020. Optimization of Citric Acid Production by Aspergillus niger and Candida tropicalis for Solid State Fermentation Using Banana Peel Substrate. Journal of Life and Bio Sciences Research, 1, 51-60.

- 84. Öhgren, K., Bura, R., Lesnicki, G., Saddler, J., Zacchi, G. 2007. A comparison between simultaneous saccharification and fermentation and separate hydrolysis and fermentation using steam-pretreated corn stover. Process Biochemistry, 42, 834-839.
- 85. Panigrahi, N., Thompson, A.J., Zubelzu, S. and Knox, J.W. 2021. Identifying opportunities to improve management of water stress in banana production. Scientia Horticulturae, 276, 1-9.
- 86. Paschos, T., Louloudi, A., Papayannakos, N., Kekos, D., Mamma, D. 2020. Potential of barley straw for high titer bioethanol production applying pre-hydrolysis and simultaneous saccharification and fermentation at high solid loading. Biofuels, 1-7.
- 87. Peral, C. 2016. Biomass pretreatment strategies (technologies, environmental performance, economic considerations, industrial implementation). Biotransformation of agricultural waste and by-products, 125-160.
- Phitsuwan, P., Sakka, K., Ratanakhanokchai, K. 2016. Structural changes and enzymatic response of Napier grass (Pennisetum purpureum) stem induced by alkaline pretreatment. Bioresource technology, 218, 247-256.
- 89. Pramanik, S.K., Suja, F.B., Porhemmat, M., Pramanik, B.K. 2019. Performance and Kinetic Model of a Single-Stage Anaerobic Digestion System Operated at Different Successive Operating Stages for the Treatment of Food Waste. Processes, 7, 600.
- 90. Puligundla, P., Oh, S.-E., Mok, C. 2016. Microwave-assisted pretreatment technologies for the conversion of lignocellulosic biomass to sugars and ethanol: a review. Carbon letters, 17, 1-10.
- 91. Qing, Q., Zhou, L., Guo, Q., Huang, M., He, Y., Wang, L., Zhang, Y. 2016. A combined sodium phosphate and sodium sulfide pretreatment for enhanced enzymatic digestibility and delignification of corn stover. Bioresource Technology, 218, 209-216.
- Rabemanolontsoa, H., Saka, S. 2013. Comparative study on chemical composition of various biomass species. RSC advances, 3, 3946-3956.

- 93. Raj, T., Chandrasekhar, K., Kumar, A.N., Banu, J.R., Yoon, J.-J., Bhatia, S.K., Yang, Y.-H., Varjani, S., Kim, S.-H. 2022. Recent advances in commercial biorefineries for lignocellulosic ethanol production: Current status, challenges and future perspectives. Bioresource technology, 344, 126292.
- Raj, T., Gaur, R., Lamba, B.Y., Singh, N., Gupta, R.P., Kumar, R., Puri, S.K., Ramakumar, S.
 2018. Characterization of ionic liquid pretreated plant cell wall for improved enzymatic digestibility. Bioresource technology, 249, 139-145.
- 95. Saha, B.C., Kennedy, G.J. 2019. Phosphate limitation alleviates the inhibitory effect of manganese on itaconic acid production by Aspergillus terreus. Biocatalysis and Agricultural Biotechnology, 18, 10101618.
- 96. Santa-Maria, M., Ruiz-Colorado, A., Cruz, G., Jeoh, T. 2013. Assessing the feasibility of biofuel production from lignocellulosic banana waste in rural agricultural communities in Peru and Colombia. Bioenergy Research, 6, 1000-1011.
- 97. Saratale, G.D., Jung, M.-Y., Oh, M.-K. 2016. Reutilization of green liquor chemicals for pretreatment of whole rice waste biomass and its application to 2,3-butanediol production. Bioresource technology, 205, 90-96.
- Sarkar, D., Das, K. 2017. Optimization Of Citric Acid Production From Aspergillus Niger Using Pineapple Waste As Feedstock In Submerged Fermentation. World J. Pharm. Res, 6, 810-818.
- 99. Sawant, O., Mahale, S., Ramchandran, V., Nagaraj, G., Bankar, A. 2018. Fungal citric acid production using waste materials: a mini-review. Journal of microbiology, biotechnology and food sciences, 8, 821-828.
- 100. Sekoai, P.T., Ayeni, A.O., Daramola, M.O. 2018. Parametric optimization of citric acid production from apple pomace and corn steep liquor by a wild type strain of Aspergillus niger:

A Response surface methodology approach. International Journal of Engineering Research in Africa. Trans Tech Publ, 98-113.

- Sekoai, P.T., Ouma, C.N.M., Du Preez, S.P., Modisha, P., Engelbrecht, N., Bessarabov, D.G.,Ghimire, A. 2019. Application of nanoparticles in biofuels: an overview. Fuel, 237, 380-397.
- 102. Sewsynker-Sukai, Y., David, A.N., Gueguim Kana, E.B. 2020. Recent developments in the application of kraft pulping alkaline chemicals for lignocellulosic pretreatment: Potential beneficiation of green liquor dregs waste. Bioresource Technology, 306, 123225.
- 103. Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2018. Microwave-assisted alkalic salt pretreatment of corn cob wastes: Process optimization for improved sugar recovery. Industrial Crops and Products, 125, 284-292.
- Shimizu, F.L., Monteiro, P.Q., Ghiraldi, P.H.C., Melati, R.B., Pagnocca, F.C., Souza, W.D., Sant'anna, C., Brienzo, M. 2018. Acid, alkali and peroxide pretreatments increase the cellulose accessibility and glucose yield of banana pseudostem. Industrial Crops and Products, 115, 62-68.
- 105. Shrotri, A., Kobayashi, H., Fukuoka, A. 2017. Catalytic conversion of structural carbohydrates and lignin to chemicals. Advances in catalysis, 60, 59-123.
- 106. Singh, R., Tiwari, S., Srivastava, M., Shukla, A. 2014. Microwave Assisted Alkali Pretreatment of Rice Straw for Enhancing Enzymatic Digestibility. Energy, 1-7.
- 107. Sivanarutselvi, S., Poornima, P., Muthukumar, K., Velan, M. 2019. Studies on effect of alkali pretreatment of banana pseudostem for fermentable sugar production for biobutanol production. Journal of Environmental Biology, 40, 393-399.
- Stark, A. 2007. KR Seddon in Kirk-Othmer Encyclopedia of Chemical Technology, Wiley-Interscience, New York, 26.

- Stöcker, M. 2008. Biofuels and biomass-to-liquid fuels in the biorefinery: Catalytic conversion of lignocellulosic biomass using porous materials. Angewandte Chemie International Edition, 47, 9200-9211.
- 110. Subagyo, A., Chafidz, A. 2018. Banana pseudo-stem fiber: Preparation, characteristics, and applications. Intech Open, 1-20.
- 111. Sukmaningtyas, R.P. Pretreated of banana pseudo-stem as raw material for enzymatic hydrolysis and bioethanol production. MATEC Web of Conferences, 2018. EDP Sciences, 01035.
- Tayyab, M., Noman, A., Islam, W., Waheed, S., Arafat, Y., Ali, F., Zaynab, M., Lin, S., Zhang,
 H., Lin, W. 2018. Bioethanol production from lignocellulosic biomass by environment-friendly
 pretreatment methods: A review. Appl. Ecol. Environ. Res, 16, 225-249.
- 113. Thakur, S., Shrivastava, B., Ingale, S., Kuhad, R.C., Gupte, A. 2013. Degradation and selective ligninolysis of wheat straw and banana stem for an efficient bioethanol production using fungal and chemical pretreatment. Biotech, 3, 365-372.
- 114. Tobin, T., Gustafson, R., Bura, R., Gough, H.L. 2020. Integration of wastewater treatment into process design of lignocellulosic biorefineries for improved economic viability. Biotechnology for Biofuels, 13, 24.
- 115. Usmani, Z., Sharma, M., Awasthi, A.K., Lukk, T., Tuohy, M.G., Gong, L., Nguyen-Tri, P., Goddard, A.D., Bill, R.M., Nayak, S.C. 2021. Lignocellulosic biorefineries: The current state of challenges and strategies for efficient commercialization. Renewable and sustainable energy reviews, 148, 111258.
- 116. Vassilev, N., Medina, A., Eichler-Lobermann, B., Flor-Peregrin, E., Vassileva, M. 2012.
 Animal bone char solubilization with itaconic acid produced by free and immobilized
 Aspergillus terreus grown on glycerol-based medium. Appl Biochem Biotechnol, 168, 13118.

- 117. Wagner, A.O., Lackner, N., Mutschlechner, M., Prem, E.M., Markt, R., Illmer, P. 2018.
 Biological pretreatment strategies for second-generation lignocellulosic resources to enhance biogas production. Energies, 11, 1797.
- Wang, S., Dai, G., Yang, H., Luo, Z. 2017. Lignocellulosic biomass pyrolysis mechanism: A state-of-the-art review. Progress in Energy and Combustion Science, 62, 33-86.
- Xu, J., Liu, B., Hou, H., Hu, J. 2017. Pretreatment of eucalyptus with recycled ionic liquids for low-cost biorefinery. Bioresource Technology, 234, 406-414.
- 120. Yin, J., Hao, L., Yu, W., Wang, E., Zhao, M., Xu, Q., Liu, Y. 2014. Enzymatic hydrolysis enhancement of corn lignocellulose by supercritical CO2 combined with ultrasound pretreatment. Chinese journal of catalysis, 35, 763-769.
- Yu, J., Xu, Z., Liu, L., Chen, S., Wang, S., Jin, M. 2019. Process integration for ethanol production from corn and corn stover as mixed substrates. Bioresource Technology, 279, 10-16.
- 122. Yusuf, A., Inambao, F. 2019. Bioethanol Production Techniques from Lignocellulosic Biomass as Alternative Fuel: A Review. International Journal of Mechanical Engineering and Technology, 10, 34-71.
- 123. Zabed, H., Sahu, J., Boyce, A.N., Faruq, G. 2016. Fuel ethanol production from lignocellulosic biomass: an overview on feedstocks and technological approaches. Renewable and Sustainable Energy Reviews, 66, 751-774.
- 124. Zhang, C., Li, J., Liu, C., Liu, X., Wang, J., Li, S., Fan, G., Zhang, L. 2013. Alkaline pretreatment for enhancement of biogas production from banana stem and swine manure by anaerobic codigestion. Bioresource Technology, 149, 353-358.
- 125. Zhang, Y., Huang, M., Su, J., Hu, H., Yang, M., Huang, Z., Chen, D., Wu, J., Feng, Z. 2019. Overcoming biomass recalcitrance by synergistic pretreatment of mechanical activation and

metal salt for enhancing enzymatic conversion of lignocellulose. Biotechnology for Biofuels, 12, 1-12.

- 126. Zhao, X.-Q., Zi, L.-H., Bai, F.-W., Lin, H.-L., Hao, X.-M., Yue, G.-J., Ho, N.W.Y. 2011. Bioethanol from Lignocellulosic Biomass. Biotechnology in China III: Biofuels and Bioenergy, 1, 25-51.
- 127. Zhou, X., Wang, M., Fang, S., Liu, X., Liu, P. 2019. Effect of alkaline black liquor recycling on alkali combined with ozone pretreatment of corn stalk. Molecules, 24, 2836.
- Zhu, Y.-J., Chen, F. 2014. Microwave-Assisted Preparation of Inorganic Nanostructures in Liquid Phase. Chemical Reviews, 114, 6462-6555.

Chapter 2 - Supplementary material

Organic acid	Substrate	Fermentative microorganism	Optimal Parameters	Concentration	References
Citric	Banana peel	A. niger	30 °C, pH 4, 5% glucose (w/v), 2% zinc (w/v), 0.5% ammonium	97.6 g/L	Odu et al. (2020)
acid			chloride (w/v), 3% methanol (v/v), 10 days		
	Pineapple waste	A. niger	30 °C, 15% sucrose (w/v), 0.25% ammonium nitrate (w/v), 4%	44.1 g/L	Sarkar and Das
			methanol (v/v), 7 days		(2017)
	Sweet potato	A. niger	$25\ ^\circ C$, 97.25% carbon source, 1.25% nitrogen (w/v), 7 days, pH 6.5	4.36 g/L	Aboyeji et al. (2020)
	peel waste				
	Apple pomace	A. niger	32.88 °C, 33.81 g/L Apple pomace, 42.5 g/L of corn steep liquor	62-68 g/L	Sekoai et al. (2018)
			(CSL), 2.05% methanol (v/v), 33 h, pH 4.54		
Itaconic	Jatropha seed	A. terreus	~30 °C, pH 1.5, 5 mL inoculum, 50% substrate, 400 rpm, 9 days	48.7 g/L	Amina et al. (2013)
acid	cake				
	Glucose media	A. terreus	100 g/L substrate conc., pH 4.5, 120 hr	30.2 g/L	Vassilev et al.
					(2012)
	Olive and beet	A. terreus	30 °C, 2 mL inoculum, pH 5.5, 70% humidity	44 g/L	Nikolay et al. (2013)
	waste				
	Jatropha cake	A. terreus	32 °C, 10% inoculum, pH 3, 200 rpm, 120 hr	24.45 g/L	Rao et al. (2006)

Table S1. Optimal fermentation conditions and concentration of citric acid, itaconic acid and lactic acid production from various studies.

Table S1. Continued.

Organic	Substrate	Fermentative	Optimal Parameters	Concentration	References
acid		microorganism			
Lactic	Corn stover	Bacillus	10 g/L sodium acetate, 5 g/L levulinic acid, 2 g/L monophenol, 30%	35.8 g/L	Zou et al. (2021)
acid		coagulans	solid loading, 24 hr		
	Sugarcane	Lactobacillus	37 °C, 10% inoculum (v/v), 35 rpm, 120 hr	21.3 g/L	Oonkhanond et al.
	bagasse	casei			(2017)
	Sugarcane	Clostridium	35 °C, 10% inoculum (v/v), 35 rpm, 96 hr, pH 6	13 g/L	Vieira et al. (2021)
	bagasse	beijerinckii			
	Corn cobs	Lactobacillus	37 °C, 25 mL DWW citrate buffer, 10% pretreated substrate, 25 g/L	11.15 g/L	(David et al., 2022)
		plantarum	CSL, 2 mL/L Tween 80, 10% inoculum (v/v)		

Lignocellulosic substrate	Waste by- product	Pretreatment parameters	Output	References
Oil palm	Black liquor	500 g substrate, 4 bar pressure, 2.5 L pretreatment	137.9 g/L glucose	Triwahyuni et al. (2015)
		solution, 60 g NaOH, 160 °C, 40 min		
Corn stover	Black liquor	$80\ ^{\circ}\text{C}, 2\ \text{h}, 30\ \text{kg}$ black liquor with 0.165 kg NaOH, 3	68-78% glucose yield, 60 g/L ethanol	Chen et al. (2018)
		kg substrate		
Corn stover	Black liquor	2 g substrate, 30 mL of 2% NaOH, 150 rpm, 80 °C, 2	81.53% cellulolytic conversion, 35.41%	Zhou et al. (2019)
		h	hemicellulose conversion	
Corn stover	Green liquor	8% Alkali, 40% sulfidity, 140 °C	70% pulp yield, 45% lignin removal	Gu et al. (2012)
Poplar	Green liquor	Liquor ratio 1:4, 1.5L/kg green liquor, 100 °C, 90	13.81% pentosan extracted	Li et al. (2013)
		min		
Rice waste	Green liquor	Liquor ratio 1:1 (10%), 100 °C, 6 h	58.2% delignification, 88% glucan yield,	Saratale et al. (2016)
			545 mg/g total sugar	
Bamboo	Green liquor	20% titratable alkali, 25% sulfidity, 4 ml/g ratio, 150	48.5% maximum sugar yield	Wang et al. (2014)
		°C, 1 h		
Corn cobs	Green liquor	80% GLD, 20% S/L, ~50 min	0.42 g/g glucose yield	David et al. (2020)
	dregs		23.69 g/L ethanol	
Paper mill	Green liquor	56% GLD, 44% substrate, 4.5% Tween 80, 60 min,	16.38 g/L reducing sugar, 3.72 mL/g	Rorke et al. (2021)
sludge	dregs	and S:L of 9.5%, 120 °C	hydrogen	

Table S2. Pretreatment applications of waste by-products obtained from the Kraft paper and pulp industry.

Table S2. Continued.

Lignocellulosic	Waste by-	Pretreatment parameters	Output	References
substrate	product			
Corn cobs	Green liquor	49.89% GLD, 118 °C, 5 min	1.53 g/g reducing sugar yield, 0.85 g/g	David et al. (2021)
	dregs + paper		glucose yield	
	wastewater			

CHAPTER 3

Innovative microwave-assisted iodized table salt or paper wastewater pretreatments for enhanced sugar recovery from banana pseudostem

This chapter has been published in the journal Biomass Conversion and Biorefinery entitled: Innovative microwave-assisted iodized table salt or paper wastewater pretreatments for enhanced sugar recovery from banana pseudostem. *Biomass Conversion and Biorefinery*. DOI: <u>https://doi.org/10.1007/s13399-022-02630-x</u>. The published paper and supplementary material are presented in the following pages.

Highlights

- First report on table salt or paper wastewater for lignocellulosic pretreatment.
- Salt and paper wastewater revealed comparable sugar yields to expensive chemicals.
- Applicability on various lignocellulosic substrates was demonstrated.
- Paves the way towards negating chemicals and/or fresh water in pretreatment.



Graphical Abstract

ORIGINAL ARTICLE



Innovative microwave-assisted iodized table salt or paper wastewater pretreatments for enhanced sugar recovery from banana pseudostem

Milesh Laltha¹ · Y. Sewsynker-Sukai² · E. B. Gueguim Kana¹

Received: 5 January 2022 / Revised: 21 March 2022 / Accepted: 22 March 2022 © The Author(s), under exclusive licence to Springer-Verlag GmbH Germany, part of Springer Nature 2022

Abstract

Iodized table salt and Kraft paper and pulp wastewater represent potential, cost-effective lignocellulosic pretreatment catalysts. Therefore, to determine their pretreatment applicability, the present study modelled and optimized two novel methods, namely (1) microwave-assisted-iodized table salt (M-ITS) and (2) microwave-assisted-paper wastewater (M-PWW) strategy for the enhancement of sugar recovery from banana pseudostem. Pretreatment optimization revealed a slightly higher reducing sugar (0.515 g/g) and glucose yield (0.433 g/g) for the M-ITS strategy compared to the M-PWW method (reducing sugar = 0.498 g/g and glucose = 0.413 g/g). Although both methods displayed comparable sugar yields to previous pretreatment studies that used expensive chemicals, the M-PWW strategy demonstrated a higher efficiency, since it is completely waste-based and requires fewer pretreatment cycles. Major findings from this study paves the way for lignocellulosic pretreatment by potentially negating the use of costly chemicals and/or fresh water, while achieving a "waste treating waste" approach.

Keywords Pretreatment · Lignocellulosic biomass · Iodized table salt · Paper wastewater · Banana pseudostem

1 Introduction

Lignocellulosic biomass (LCB) represents an attractive, abundant, and cost-effective feedstock for the production of bio-based products [1]. These substrates are characterized as a plant waste that may be harnessed for microbial fermentation, since it is a carbon–neutral and an inedible feedstock that does not interfere with food and feed supplies [2]. LCB is a heterogeneous polymer mainly comprised of 30 to 50% cellulose (C₆ sugars), 15 to 35% hemicellulose (C₅ and C₆ sugars), and 10 to 20% lignin, that are bound by covalent and hydrogen bonds, forming lignin-carbohydrate complexes [3]. Other constituents of LCB that are present in lower quantities include organic extractives (phenols, proteins) and inorganic minerals (ash). Some examples of LCB substrates include banana pseudostem [4], corn cobs [5], bamboo [6], and sugarcane leaves [7]. The banana pseudostem (*Musa sp.*) is a lignocellulosic waste that has recently been recognized for its potential as a renewable and sustainable resource, due to its extraordinary growth rate, high productivity yields, and abundance. Approximately 300 million tonnes of banana pseudostem waste is generated annually after harvesting of the banana fruit [8]. In addition, the high cellulosic (14-31%)and low lignin (6-15%) content makes it a potential feedstock for the microbial production of valuable products [8, 9]. Despite the advantages that LCB offers, hemicellulose and cellulose are strongly held together by the recalcitrant lignin structure via hydrogen and covalent bonding [3]. Furthermore, LCB consists of lignin-carbohydrate bonds such as benzyl ethers, esters, phenyl glycosides, and hemiacetal linkages [3]. Depolymerization of these interlinked constituents has proven to be problematic, since it prevents access of cellulase enzymes to the cellulose polymer and subsequently hinders the enzymatic release of fermentable sugars that can be available for microbial fermentation [10]. Pretreatment of LCB is therefore necessary for the disruption of the resistant network, which ultimately removes lignin and reduces the crystalline structure of cellulose and hemicellulose [8]. Following pretreatment, cellulose is enzymatically hydrolyzed to fermentable sugars such as glucose that are subsequently converted to high-value chemical commodities during

E. B. Gueguim Kana kanag@ukzn.ac.za

¹ School of Life Sciences, University of KwaZulu-Natal, Pietermaritzburg, South Africa

² Fort Hare Institute of Technology, University of Fort Hare, Private Bag X1314, Alice 5700, South Africa

microbial fermentation [1]. Pretreatment methods comprise physical, chemical, physicochemical, biological, or a combination of these approaches [11]. Examples of common pretreatment methods include solvents or catalysts such as acid and alkali [1] and inorganic salt [7]. Effective pretreatment techniques are required to have low energy requirements, use cost-effective chemicals, minimize the formation of enzymatic and fermentation inhibitor compounds, and achieve high fermentable sugar yields [1]. The development of a pretreatment regime that meets these criteria and overcomes the technoeconomic barriers associated with these systems is imperative for the enhancement of its industrial-scale implementation. Recently, numerous studies have utilized inorganic salt as an effective pretreatment approach [12-14], due to their high catalytic activity, thereby increasing cellulosic and hemicellulosic content, for improved hydrolysis towards sugar release [13, 15]. Additionally, inorganic salts are environmentally friendly and do not corrode pretreatment reactors, thereby avoiding the use of highly specialized systems [7]. In a recent report, Suwannabun et al. [14] compared the effectiveness of six different salts (NaCl, MgCl₂, NaNO₃, KCl, KNO₃, and Mg(NO₃)₂) for the pretreatment of rice straw and observed that 0.9% NaCl resulted in the maximum reducing sugar concentration (16.60 mg/ml). Similarly, Moodley and Gueguim Kana [13] assessed three microwave-assisted inorganic salts, namely FeCl₃, ZnCl₂, and NaCl on sugarcane leaves and obtained maximum reducing sugar yields of 0.406 g/g, 0.241 g/g, and 0.179 g/g, respectively. Likewise, Kang et al. [12] evaluated five different inorganic salts (NaCl, KCl, CaCl₂, ZnCl₂, and FeCl₃) for the pretreatment of Miscanthus straw and observed that FeCl₃ resulted in the highest xylan removal of 100% and enzymatic digestibly of 63.4% compared to the other salts.

The primary mechanism by which inorganic salts function is the dissociation of Lewis acids in an aqueous solvent to form complex cations. The Lewis acid accepts an electron from the Lewis base to create a Lewis acid-base adduct. Subsequently, six water molecules are formed around the central metal cation with co-ordinate covalent bonds. Glycosidic linkages present in LCB are then cleaved by the formation of metal cations. Metal ions further assist with bond rupture, as a result of their positive charge and thus pair with negatively charged electrons. The water molecules from the hydrated cation then act as nucleophiles that depolymerize hemicellulose into monosaccharides [16]. Although inorganic salts have emerged as efficient chemical catalysts for lignocellulosic pretreatment, the relatively high costs of laboratory-grade salts coupled with the low sugar recovered, remain major limitations of these processes. To overcome these limitations, cheaper alternative chemicals are currently being sought. Sodium chloride (NaCl) has shown to be a promising, cost-effective chemical pretreatment that is available in two forms: (1) laboratory-grade NaCl and

(2) food-grade NaCl. The chemical composition and manufacturing process differentiates these salts. Within industry, food-grade or commonly known as table salt, is mined from inland (underground natural brine streams), coastal saltpans (solar and artificial evaporation), and seawater (salt refining) [17]. Most table salts are supplemented with an essential nutrient such as iodine. Conversely, laboratory-grade salt (halite) is present in the Earth's surface and can be extracted through deep-shaft mining. The mined rock salt is thereafter crushed and used for industrial and other non-food applications that include chemical processes, anti-icing (ice prevention), and de-icing (ice removal) applications on roadways [18]. To date, only the application of laboratory-grade NaCl for lignocellulosic pretreatment has been reported [12, 14], and although it has been deemed cost-effective when compared to other salts (FeCl₃, KCl and MgCl₂), it is an expensive, purified chemical that is not readily available and is usually supplied in small quantities. On the other hand, its impurified, food-grade equivalent, also known as iodized table salt (ITS), is abundantly available and commonly used as a condiment and preservative of food sources, but has not yet been evaluated for lignocellulosic pretreatment. Interestingly, ITS costs approximately 2% of purified, laboratorygrade NaCl. The average cost per kg of laboratory-grade NaCl ranges between \$76.06 and \$79.49 USD (Merck, South Africa) compared to less than \$1.5 (~\$1.24 to \$1.38) for the ITS version (local supermarket, South Africa).

While salt pretreatments such as ITS target hemicellulose solubilization, alkaline-based methods have been deemed the gold standard on the pretreatment front and cause disruptions to the recalcitrant aromatic polymer lignin. Alkaline pretreatment methods utilize chemical catalysts that significantly improve sugar yields, use lower energy input, and have been known to generate a low concentration of inhibitor compounds [19]. Examples of alkaline methods include ammonia recycle percolation, ammonia fiber expansion, aqueous and liquid ammonia, strong alkali (NaOH), and lime $(Ca(OH)_2)$ that are expensive and therefore not feasible for use in lignocellulosic pretreatment processes [20]. Recently, cheaper alkaline methods such as Na₃PO₄.12H₂O have been assessed in comparison to NaOH [5, 21]. For instance, our recent study investigated NaOH and Na₃PO₄.12H₂O pretreatments and compared the chemical recyclability of these methods to potentially reduce chemicals and water usage to circumvent costs [21]. Although the alkaline spent liquor hydrolysate from both NaOH and Na₃PO₄.12H₂O could be recycled up to three times, the reduced pretreatment efficiency after successive treatments in addition to the initial water and chemical cost still plague this process [21].

Paper wastewater (PPW), an alkaline industrial waste generated in large quantities by the Kraft paper and pulp industry has emerged as a prospective and economically viable pretreatment approach. Kraft paper and pulp mill industries discharge an estimated 75 to 225 m³ effluent (constituting PWW) per ton of paper produced [22]. Although the Kraft industry has a major impact on the global economy, it consumes large amounts of fresh water, a finite resource during the paper-making process [23]. Industrially produced wastewater requires energy and cost intensive treatments prior to its expulsion into the environment, since it is detrimental to the ecosystem and poses a severe threat to plant and aquatic life [23]. Striking characteristics of PWW include high organic loading, in addition to the presence of effluent solids, sediments, absorbable organic halides, and chlorinated organic compounds that contribute majorly to its high pH and thus alkaline nature [24]. Hence, PWW can function as a dual-purpose commodity in pretreatment processes as it exhibits characteristics as an attractive replacement for (1) common and expensive alkaline reagents and (2) freshwater usage within lignocellulosic biorefinery systems. Repurposing PWW for lignocellulosic pretreatment could potentially generate profits that may then be utilized for less invasive biohazardous treatment and management, thereby reducing the downstream processing costs for Kraft industries. Both ITS and PWW display significantly cheaper alternatives to their lab-derived counterparts such as NaCl and NaOH, respectively.

In addition, microwave heating processes (MHP) are emerging as an energy-efficient alternative to conventional heating methods such as autoclave and steam [25]. MHP can be used for rapid volumetric heating and exhibits a shorter reaction time and a higher reaction rate compared to steam [26] and autoclave [3] heat treatment. Despite the significant advancements in lignocellulosic pretreatment, technoeconomic challenges associated with chemicals, water, and energy are still major obstacles hindering its commercialization. Conventional lignocellulosic pretreatments with alkaline chemicals are expensive compared to PWW, an alkaline industrial waste that is freely available in large quantities and may potentially achieve a "waste treating waste" approach. Furthermore, food-grade NaCl or ITS is easily accessible, abundant, and a cheaper alternative to lab-grade NaCl. The development of these methods could unlock the potential of lignocellulosic pretreatment by releasing high fermentable sugar that can be channelled towards valuable microbial biobased compounds.

Therefore, the objectives of this study were to (1) model and optimize two novel pretreatments that include a microwave-assisted-iodized table salt (M-ITS) method and microwave-assisted-paper wastewater (M-PWW) strategy for enhanced sugar recovery from banana pseudostem, (2) determine the individual and interactive effects of (i) salt concentration, microwave power intensity, and pretreatment time (M-ITS) and (ii) substrate solid loading, microwave power intensity, and pretreatment time (M-PWW) on the reducing sugar and glucose yields, and (3) evaluate the

effectiveness and versatility of the developed M-ITS and M-PWW pretreatments by applying the optimized conditions on several lignocellulosic substrates (corn cobs, sorghum leaves, sugarcane bagasse, and bamboo) with varying structural compositions.

2 Materials and methods

2.1 Materials

The banana pseudostem substrate used in this study was obtained from Bakerville Heights (Pietermaritzburg, South Africa) (29.5418° S, 30.4077° E). The substrate was dried at 75 °C for 48 h in an oven (Scientific series 2000, South Africa) to reduce the remaining moisture content and was thereafter milled to a particle size that ranged from 1 to 2 mm. The powdered banana pseudostem substrate was stored in an airtight container at room temperature. The lignocellulosic composition (cellulose, hemicellulose, and lignin) of the untreated (native) and optimally pretreated banana pseudostem was analyzed using the detergent fiber technique described by Van Soest and McQueen [27]. The chemical reagents were purchased from Merck (South Africa) where applicable. The iodized table salt (ITS) (Cerebos, South Africa) used in this study was obtained from a local supermarket, while the paper wastewater (PWW) with its chemical composition (Table S1) was kindly provided by Mondi (Richards Bay Mill, South Africa). The Cellic CTec 2 enzyme (160 FPU/mL) that was used for the saccharification stage was generously donated by Novozymes (Novozymes A/S, Denmark). Other lignocellulosic wastes that include corn cobs, sorghum leaves, sugarcane bagasse, and bamboo were procured from local agricultural industries. These substrates were dried, milled, and stored as previously stated above.

2.2 Preliminary screening

For the preliminary screening, laboratory-grade sodium chloride (NaCl) salt and food-grade iodized table salt (ITS) were comparatively assessed for hemicellulose solubilization. Conversely, paper wastewater (PWW) generated from the Kraft paper and pulp industry served as the alkaline waste-based pretreatment method that was used for targeting lignin. Lab-grade NaCl, food-grade ITS, or PWW was coupled with microwave-assisted heating and were used individually in a single-stage to compare their effectiveness and were designated as M-NaCl_{preliminary}, M-ITS_{preliminary}, and M-PWW_{preliminary} respectively. Additionally, two-stage pretreatments consisting of PWW first, followed by either NaCl or ITS were denoted M-PWW-NaCl_{preliminary} and M-PWW-ITS_{preliminary} respectively. The pretreatment

Deringer

parameter ranges were selected based on an extensive literature review of related pretreatment studies [12–14]. The preliminary screening experiments were carried out with a standard substrate solid loading of 10% (w/v). The M-NaCl_{preliminary} and M-ITS_{preliminary} pretreatments were performed with a salt concentration of 3% (w/v), microwave power intensity of 500 watts (W), and pretreatment time of 6 min. In addition, the same power intensity and time were employed for the M-PWW_{preliminary} pretreatment. The twostage pretreatments (M-PWW-NaCl_{preliminary} and M-PWW-ITS_{preliminary}) were executed at 500 W in the microwave for 3 min per stage, whereby a salt concentration of 3% (w/v) was employed for the second stage. The pretreated banana pseudostem samples were filtered using a domestic sieve (<1 mm) and washed several times with deionized water. The washed biomass was subsequently oven-dried at 55 °C overnight prior to the enzymatic hydrolysis stage. Control experiments encompassed (1) distilled water only pretreated banana pseudostem (BP_{water}) at 500 W for 6 min, and (2) untreated/native banana pseudostem (BPuntreated) by omitting pretreatment.

2.3 Modelling and optimization of pretreatment parameters

The response surface methodology (RSM) was modelled using a three-level Box-Behnken design (Design Expert 7.0, Stat Ease Inc, USA) and generated two separate experimental designs (M-ITS and M-PWW) consisting of 17 pretreatments each. The pretreatment input parameters for the M-ITS model comprised of salt concentration (1-5%), w/v), microwave power intensity (100-900 W), and pretreatment time (2–10 min). On the other hand, the M-PWW model consisted of substrate solid loading (10-30% w/v), microwave power intensity (100-900 W), and pretreatment time (2-10 min). The output responses were reducing sugar yield (RSY) and glucose yield (GY) for both models. The pretreatment parameter ranges for the M-ITS and M-PWW experimental designs were selected according to an extensive literature survey [12–14] in conjunction with data obtained from the preliminary screening. Therefore, four RSM pretreatment models were designated as follows: (1) banana pseudostem pretreated with ITS-reducing sugar yield (M-ITS_{RSY}), (2) banana pseudostem pretreated with ITS-glucose yield (M-ITS_{GY}), (3) banana pseudostem pretreated with PWW-reducing sugar yield (M-PWW_{RSY}), and (4) banana pseudostem pretreated with PWW-glucose yield (M-PWW_{GY}).

2.4 Enzymatic saccharification

The M-ITS or M-PWW pretreated banana pseudostem residue was submerged in sodium citrate buffer (0.05 M, pH

4.8) with a working volume of 10 mL. In addition, standard enzyme loading (10 FPU/g) and solid loading (10%, w/v) were used. Enzymatic hydrolysis was carried out at an agitation and temperature of 120 rpm and 50 °C, respectively, for 72 h in an orbital shaker incubator (FMH Instruments, South Africa). Subsequently, the hydrolysate samples were centrifuged at 10 000 rpm (Microspin 12, Grant-bio) for 5 min to remove any unhydrolyzed biomass. The reducing sugar yield (g/g) and glucose yield (g/g) released after enzymatic saccharification was analyzed using the 3,5-dinitrosalicyclic acid (DNS) method [28] and Megazyme GOPOD assay kits (@Megazyme, Wicklow, Ireland) respectively. For the detection of reducing sugar, the samples collected at the end of the enzymatic hydrolysis stage were centrifuged at 10 000 rpm for 5 min. Samples were thereafter diluted accordingly with distilled water. The diluted sample (3 mL) was pipetted into test tubes prior to the addition of 3 mL of DNS reagent (10 g/L DNS, 2 g/L phenol, 0.5 g/L sodium sulfite, and 10 g/L NaOH). The samples were incubated in a preheated water bath at 90 °C for 15 min. Thereafter, 1 mL of 40% (w/v) sodium potassium tartrate (Rochelle salts) was added to each tube and mixed thoroughly. The absorbance values were recorded at a wavelength of 575 nm. Parallel to this, a glucose (reducing sugar) standard curve (0, 0.2, 0.4, 0.6, 0.8, and 1 g/L glucose) was set up using the above protocol. The absorbance values recorded for the known glucose concentrations were plotted against the glucose concentrations to produce a standard curve. The absorbance values obtained for the experimental runs were substituted into the glucose standard curve equation to determine the reducing sugar concentration (g/L). The reducing sugar yield (g/g)was computed by dividing the reducing sugar concentration by 100 to determine the quantity of reducing sugar produced per gram of substrate.

Glucose was determined using the Megazyme GOPOD assay kit protocols provided by the manufacturer (Megazyme, Ireland). Centrifuged enzymatic hydrolyzed samples were diluted according to the manufacturer's instructions. Diluted sample aliquots (0.1 mL) were thereafter added to test tubes containing 3 mL of GOPOD reagent and mixed thoroughly. These test tubes were incubated at 45 °C for 20 min in a preheated water bath. The glucose standard solution (1.0 g/L) supplied by the manufacturer, and blank (distilled water) were also performed as per the specified protocols. The absorbance values were recorded at a wavelength of 510 nm. The glucose concentration (g/L) was extrapolated by dividing the absorbance values for the experimental samples by the absorbance value obtained for the glucose standard and thereafter multiplying by the dilution factor. The glucose yield (g/g)was computed similar to the reducing sugar yield. The glucose concentration (g/L) obtained was divided by 100 to extrapolate the quantity of glucose produced per gram
of substrate. All experiments were carried out in duplicate according to the RSM design for the iodized table salt (Table 1) and paper wastewater (Table 2) pretreatment models.

2.5 Validation of the optimized pretreatment conditions and assessment on different types of lignocellulosic wastes

The optimized pretreatment conditions for each developed model (M-ITS and M-PWW) were applied on banana pseudostem as validation experiments and designated as M-ITS_{optimized} and M-PWW_{optimized} respectively. In addition, other lignocellulosic wastes that include corn cobs, sorghum leaves, sugarcane bagasse, and bamboo were subjected to the optimized M-ITS and M-PWW pretreatment conditions. Each feedstock was then pretreated and enzymatically hydrolyzed following the protocols outlined previously in Sections 2.3 and 2.4 above.

2.6 Analytical techniques

Table 1Reducing sugar andglucose yields observed forthe M-ITS pretreatments

2.6.1 Scanning electron microscopy (SEM)

The banana pseudostem before (untreated) and after (optimized M-ITS or M-PWW) pretreatments were visually analyzed using scanning electron microscopy (SEM). Samples were initially mounted onto aluminum stubs and then sputter coated with four layers of a thin gold film for electron conductivity (Quorum Q150 R ES). Subsequently, the samples were examined at an accelerating voltage of 20 kV with a magnification of ~ $500 \times$ (ZEISS EVO LS 15).

2.6.2 Fourier-Transform Infrared (FTIR) spectroscopy

The chemical structure and functional group changes of the native (untreated) and optimally pretreated (M-ITS or M-PWW) substrate were analyzed by Fourier transform infrared (FTIR) spectroscopy using an Agilent Cary 630 spectrometer. Sample preparation was carried out by mixing the dried samples with spectroscopic grade KBr that was further pressed into pellets for analysis. Data from the analysis were collected using Agilent MicroLab PC 5.1.22 and processed using ResolutionPro 5.0.0.395 software. The resulting FTIR spectra was recorded in the range of 650 and 4000 cm⁻¹ at a resolution of 4 cm⁻¹ and 30 scans per sample.

3 Results and discussion

3.1 Preliminary screening

Among the various preliminary screening methods evaluated on banana pseudostem, the M-ITS_{preliminary}

	Input parameter	s		Output parameters		
Run	Salt concentration (%)	Power intensity (W)	Pretreatment time (min)	Reducing sugar yield (g/g)	Glucose yield (g/g)	
1	1	100	6	0.107 ± 0.001	0.068	
2	3	100	10	0.113 ± 0.001	0.072	
3	3	900	10	0.450 ± 0.001	0.389 ± 0.010	
4	3	500	6	0.428 ± 0.001	0.331 ± 0.010	
5	3	500	6	0.407	0.309	
6	1	500	10	0.436 ± 0.001	0.324 ± 0.020	
7	5	100	6	0.089 ± 0.006	0.057	
8	1	500	2	0.386 ± 0.005	0.256 ± 0.010	
9	1	900	6	0.418 ± 0.001	0.334	
10	3	100	2	0.119 ± 0.010	0.057	
11	5	500	10	0.432 ± 0.003	0.283 ± 0.010	
12	3	500	6	0.431 ± 0.003	0.315 ± 0.020	
13	3	500	6	0.412	0.295	
14	5	900	6	0.433 ± 0.016	0.338 ± 0.030	
15	3	900	2	0.411 ± 0.003	0.258 ± 0.020	
16	5	500	2	0.371 ± 0.001	0.228	
17	3	500	6	0.419 ± 0.007	0.291 ± 0.010	

g/g, gram-reducing sugar/gram dry weight substrate or gram glucose/gram dry weight substrate; $\pm =$ standard deviation

🙆 Springer

Table 2 Reducing sugar and
glucose yields observed for
the M-PWW pretreatments

	Input paramete	rs		Output parameters			
Run	Solid loading (%)	PowerPretreatmentintensitytime(W)(min)		Reducing sugar yield (g/g)	Glucose yield (g/g)		
1	10	100	6	0.112 ± 0.003	0.079 ± 0.003		
2	20	100	10	0.116 ± 0.001	0.075 ± 0.003		
3	20	900	10	0.438 ± 0.015	0.295 ± 0.017		
4	20	500	6	0.516 ± 0.004	0.231 ± 0.012		
5	20	500	6	0.457 ± 0.007	0.286 ± 0.009		
6	10	500	10	0.478 ± 0.016	0.292 ± 0.011		
7	30	100	6	0.127	0.055 ± 0.001		
8	10	500	2	0.435 ± 0.006	0.206 ± 0.018		
9	10	900	6	0.524 ± 0.001	0.273 ± 0.011		
10	20	100	2	0.135 ± 0.004	0.078 ± 0.006		
11	30	500	10	0.469 ± 0.009	0.307 ± 0.013		
12	20	500	6	0.477 ± 0.009	0.333 ± 0.008		
13	20	500	6	0.473	0.334 ± 0.028		
14	30	900	6	0.534 ± 0.013	0.426 ± 0.003		
15	20	900	2	0.455 ± 0.008	0.308 ± 0.002		
16	30	500	2	0.342 ± 0.003	0.174 ± 0.011		
17	20	500	6	0.454 ± 0.012	0.339 ± 0.005		

g/g, gram-reducing sugar/gram dry weight substrate or gram glucose/gram dry weight substrate; $\pm =$ standard deviation

resulted in the highest reducing sugar yield (RSY) and glucose yield (GY) of 0.424 and 0.298 g/g respectively, followed by the M-NaCl_{preliminary} method (RSY = 0.383 g/g and GY = 0.260 g/g) (Table S2). The M-PWW_{preliminary} pretreatment led to a RSY and GY of 0.366 and 0.266 g/g respectively. On the other hand, the two-stage pretreatment processes demonstrated similar reducing sugar (M-PWW-NaCl_{preliminary} = 0.404 g/gand M-PWW-ITS_{preliminary} = 0.402 g/g) and glucose $(M-PWW-NaCl_{preliminary} = 0.245 \text{ g/g} \text{ and } M-PWW ITS_{preliminary} = 0.272 \text{ g/g}$ yields to the single-stage systems. The control experiments displayed significantly lower reducing sugar (BPwater = 0.286 g/g and BPuntreated = 0.187 g/g) and glucose yields (BP_{water}=0.182 g/g and BP_{untreated}=0.107 g/g) compared to the single- and two-stage preliminary pretreatments. Major observations from the preliminary screening revealed that (1) single-stage and two-stage systems exhibited similar RS and GY, and (2) ITS (food-grade NaCl) resulted in a higher RS and GY compared to the commonly used lab-based NaCl for pretreatment. Hence, ITS and PWW pretreatments in single-stage systems were selected for subsequent modelling and optimization using the response surface methodology (RSM) model.

3.2 Effect of the pretreatment input parameters on the output reducing sugar and glucose yields

The output reducing sugar yield (RSY) and glucose yields (GY) obtained for the M-ITS and M-PWW experimental runs are illustrated in Tables 1 and 2 respectively. The M-ITS pretreatment displayed maximum RSY and GY of 0.450 and 0.389 g/g respectively, with 3% salt concentration, 900 W, and 10 min (run 3). The median pretreatment conditions of 3% salt concentration, 500 W, and 6 min exhibited similar RSY and GY as shown for experimental run 4 (RSY = 0.428 g/g and GY = 0.331 g/g), run 5 (RSY = 0.407 g/g and GY = 0.309 g/g), run 12 (RSY = 0.431 g/g and GY = 0.315 g/g), run 13 (RSY = 0.412 g/g and GY = 0.295 g/g), and run 17(RSY = 0.419 g/g and GY = 0.291 g/g). The lowest RSY and GY for the M-ITS pretreatment were obtained with 5% salt concentration, 100 W, and 6 min (run 7: RSY = 0.089 g/gand GY = 0.057 g/g).

On the other hand, the highest RSY and GY of 0.534 and 0.426 g/g were achieved for the M-PWW pretreatment respectively, with a solid loading of 30%, heated at 900 W for 6 min

(run 14). Additionally, similar RSY and GY were obtained under the median pretreatment conditions (20% solid loading, 500 W, and 6 min) for experimental run 4 (RSY=0.516 g/g and GY=0.231 g/g), run 5 (RSY=0.457 g/g and GY=0.286 g/g), run 12 (RSY=0.477 g/g and GY=0.333 g/g), run 13 (RSY=0.473 g/g and GY=0.334 g/g), and run 17 (RSY=0.454 g/g and GY=0.339 g/g). Alternatively, experimental run 1 (10% solid loading, 100 W, and 6 min) led to the lowest RSY (run 1=0.112 g/g) whereas run 7 (30% solid loading, 100 W for 6 min) resulted in the lowest GY (run 7=0.055 g/g).

3.2.1 Interactive effects of the input parameters on the sugar yields for the M-ITS pretreatment

The pretreatment experimental data (Table 1 and 2) were used to develop four polynomial model equations shown in Table 3 that relate the output reducing sugar and glucose yields to the pretreatment input variables. Analysis of variance (ANOVA) was used to assess the fitness of the RSM models (Table 4). The interactions of the pretreatment input parameters on the output reducing sugar and glucose yields are depicted in Fig. 1A-D. The interactive effects of microwave power intensity and salt concentration on the RSY and GY, while pretreatment time was maintained at its median value, are illustrated in Fig. 1A and B for the M-ITS_{RSY} and M-ITS_{GY} models respectively. A high F value for the $M-ITS_{RSY} = 182.87 \text{ model and a low } p \text{ value } (< 0.05) \text{ indi-}$ cated that this model was statistically significant (Table S3). A simultaneous increase in power intensity and salt concentration from 100 to 700 W and 1 to 3% respectively, increased the RSY from 0.12 to 0.463 g/g (Fig. 1A). Similarly, for the M-ITS_{GY} model, a high F value and low p value of (92.26 and < 0.05) respectively displayed significance of this model (Table S4). The same incremental variations in the power intensity (100-700 W) and salt concentration (1 to 3%) led to an increase in the GY (0.08–0.352 g/g) (Fig. 1B). High R^2 values were achieved for the M-ITS_{RSY} (0.9958) and M-ITS_{GY} (0.9916) model, indicating that these models could account for 99.58% (ITS_{RSY}) and 99.16% (ITS_{GY}) of the variation in the observed data. From the pretreatment parameters that were investigated, microwave power intensity and pretreatment time were shown to have the most significant effect for the M-ITS_{RSY} (p < 0.0001 and p = 0.0071) and M-ITS_{GY} (p < 0.0001 and p = 0.0005) models (Table S3) and Table S4 respectively). The interaction of power intensity and pretreatment time was also shown to have a significant effect (p = 0.0072) for the M-ITS_{GY} model (Table S4). This indicates that exposure of the substrate to high microwave power intensities and salt concentration increases the degradation of the recalcitrant lignin constituents [29]. During the microwave heating processes, polar molecules or side-chains of macromolecules rapidly oscillate, due to the incoming electromagnetic waves [30]. Microwave energy is converted to heat by the internal resistance of particles, thus resulting in bond disintegration and reorientation [30]. In addition to microwave power intensity, inorganic salt disrupts the hydrogen bonds and cleaves glycosidic linkages [31]. This reduces cellulose crystallinity, while increasing substrate porosity [15]. Furthermore, water molecules from the hydrated salt participates as nucleophiles, yielding glucose from the substrate [12]. Moreover, during pretreatment, inorganic salt undergoes hydrolysis and releases hydrogen ions as a weak electrolyte into solution, therefore providing a steadier reactive environment [15]. Similar trends on sugarcane leaf waste and Miscanthus straw were reported by Moodley and Gueguim Kana [13] and Kang et al. [12] respectively, using inorganic salt pretreatment. Moodley and Gueguim Kana [13] observed a maximum reducing sugar yield of 0.179 g/g after pretreatment with 1.5 M NaCl and 700 W for 8 min. Likewise, Kang et al. [12] observed a high glucan recovery (86.5%) and improved enzymatic digestibility (39.8%) following salt pretreatment (5% NaCl, 200 °C, 15 min). Although the power intensity and salt concentration enhanced the sugar recovery, a further increase in these parameters (> 700 W and > 3%) led to a decrease in the RSY (0.453-0.420 g/g). A similar trend was observed for the GY (0.355–0.33 g/g) when power intensity and salt concentration were simultaneously increased above 700 W and 3% respectively. Water plays a significant role during

Table 3RSM polynomialmodel equations relatingthe input parameters tothe output reducing sugarand glucose yields for theM-ITS and M-PWW pre-treatments

Model	Equation	Equation number
M-ITS _{RSV}	$-0.02 + 1.23 \times 10^{-3}B + 6.56 \times 10^{-4}C - 9.07 \times 10^{-7}B^{2}$	(1)
M-ITS _{GY}	$-0.07 + 7.98 \times 10^{-4}B + 0.02C + 1.80 \times 10^{-5}BC - 5.87 \times 10^{-7}B^2 - 1.26 \times 10^{-3}C^2$	(2)
M-PWW _{RSY}	$-0.03 + 1.39 \times 10^{-3} B - 9.25 \times 10^{-7} B^2$	(3)
M-PWW _{GY}	$-0.07 + 5.82 \times 10^{-4} B - 4.76 \times 10^{-7} B^2$	(4)

M- ITS_{RSY} , microwave-assisted iodized table salt pretreatment reducing sugar yield; M- ITS_{GY} , microwaveassisted iodized table salt pretreatment glucose yield; M- PWW_{RSY} , microwave-assisted paper wastewater pretreatment reducing sugar yield; M- PPW_{GY} , microwave-assisted paper wastewater pretreatment glucose yield; B, power intensity; C, pretreatment time

Springer

Table 4Summary of analysisof variance (ANOVA) for thedeveloped pretreatment models

Model	Sum of squares	df	Mean squares	F value	p value	R^2	
M-ITS _{RSY}	0.30	9	0.033	182.87	< 0.0001	0.9958	_
M-ITS _{GY}	0.20	9	0.022	92.26	< 0.0001	0.9916	
M-PWW _{RSY}	0.37	9	0.042	21.27	0.0003	0.9647	
M-PWW _{GY}	0.18	9	0.020	7.01	0.0088	0.9002	

df, degrees of freedom; F value, Fisher-Snedecor distribution value; p value, probability value; R^2 , coefficient of determination

pretreatment by facilitating heat and irradiation energy. The water content not only affects the dielectric characteristics of LCB, but also determines the efficiency of pretreatment [32]. Furthermore, elevated temperatures that occur under higher power intensities may result in low molecular weight sugars and amino acids that polymerize to form recalcitrant organic compounds via Maillard reactions, also commonly known

as caramelization [30], that may have negative implications for further biological reactions such as enzymatic hydrolysis. This finding was in accordance with the study by Sewsynker-Sukai and Gueguim Kana [5], whereby an increase in power intensity above 700 W also showed a 30.8% decrease in the reducing sugar yield from corn cobs. High salt concentrations (> 3%) attract water molecules around their core metal



Fig. 1 Response surface graphs illustrating the interactive effects of the various input pretreatment parameters on the output reducing sugar and glucose yields: (A) power intensity and salt concentration

(M-ITS_{RSY}); (**B**) power intensity and salt concentration (M-ITS_{GY}); (**C**) pretreatment time and power intensity (M-ITS_{RSY}); (**D**) pretreatment time and power intensity (M-ITS_{GY})

🖄 Springer

cations. This reduces the amount of water that is available to interact with the substrate, which decreases the amount of cellulose in solution, thereby resulting in a low sugar recovery [14]. In a recent study, Suwannabun et al. [14] also observed a decrease in the reducing sugar concentration (3.486 to 2.890 mg/mL) from rice straw when the NaCl concentration was increased (>5%). The influence of pretreatment time and power intensity on the RSY and GY while the salt concentration was kept at its center point is depicted in Fig. 1C (M-ITS_{RSY}) and D (M-ITS_{GY}) respectively. A simultaneous increment in pretreatment time (2-10 min) and power intensity (100-700 W) demonstrated an increase in the RSY from 0.11 to 0.484 g/g (M-ITS_{RSY}). Similarly, incremental variations in pretreatment time (2-10 min) and power intensity (100-900 W) resulted in an increase in the GY (0.06–0.39 g/g) (Fig. 1D). During microwave heating, polarization influences the dipole-dipole molecules to align in the radiation field. This ultimately leads to the rapid heating of the hydrolysate, thus enhancing the solubility of the substrate [32]. Microwave irradiation has a fast heating transfer rate, thereby reducing pretreatment time and energy consumption in comparison to conventional autoclave heating [32]. The high power intensities and thus high temperatures rapidly generate pressure within the absorbent biopolymers [32], while an increase in exposure time facilitates the accumulation of heat within the solution, leading to the effective disruption of lignin moieties [30]. A similar trend in the reducing sugar yield (0.135-0.232 g/g) was observed by Moodley and Gueguim Kana [13] by simultaneously increasing the ZnCl₂ concentration (0-2 M) and power intensity (0-800 W). Nevertheless, further increases in power intensity (>700 W) led to a decrease in the RSY (0.484 to 0.41 g/g) (Fig. 1C). The lower RSY observed at higher power intensities (>700 W) may be attributed to heat accumulation, which may result in the evaporation of the solvent (water), thereby limiting the ability of the solute molecules (ITS) to interact and diffuse into the substrate, which reduces the pretreatment efficiency.

3.2.2 Interactive effects of input parameters on the sugar yields for the M-PWW pretreatment

The pairwise effects of the pretreatment input parameters on the output reducing sugar and glucose yields are depicted in Fig. 2A–D. The interaction between power intensity and solid loading on the RSY and GY, while pretreatment time was maintained at its midpoint in the presence of the PWW, are illustrated in Fig. 2A and B for the M-PWW_{RSY} and M-PWW_{GY} models, respectively. A high *F* value for the M-PWW_{RSY} = 21.27 and a low *p* value (<0.05) indicated that this model was statistically significant (Table S5). An increase in the power intensity (100–900 W), while the solid loading was maintained at 10%, increased the RSY from 0.13 to 0.538 g/g). Likewise, for the M-PWW_{GY} model, a high F value and low p value of 7.01 and < 0.05, respectively displayed significance of this model (Table S6). On the other hand, simultaneous increases in power intensity (100-900 W) and solid loading (10-30%) resulted in an increase in the GY from 0.12 to 0.39 g/g (Fig. 2B). High R^2 values were achieved for the M-PWW_{RSY} (0.9647) and M-PWW_{GY} (0.9002) model, indicating that these models could account for 96.47% (PWW_{RSY}) and 90.02% (PWW_{GY}) of the variation in the observed data. Moreover, microwave power intensity displayed the highest significant effect for the M-PWW_{RSY} (p < 0.0001) and M-PWW_{GY} (p = 0.0003) models. Improvements in the RSY and GY may be attributed to the selected pretreatment parameters. For instance, alkaline chemicals present in the PWW such as chloride, sulfate, and chromium originate from sodium hypochlorite, sodium sulfate, and sodium chromate respectively from the Kraft paper and pulp industrial process [22, 23]. These alkaline constituents lead to the expansion of the substrate surface area and further alter and modify the physical properties (crystallinity and porosity) of lignocellulosic biomass. This results in the saponification of intermolecular ester bonds and the solubilization of the amorphous regions (lignin and hemicellulose) [20]. Additionally, high microwave power intensities is associated with rapid collision of anions and cations further enhancing the breakage of hydrogen bonds within the lignocellulosic structural chain [32]. Alagu et al. [33] investigated the pretreatment effect of NaOH concentration (2-4% w/v) on cassava stem and noted an optimum reducing sugar concentration of 43.60 µg/ml with 3.21% NaOH. However, further increments in the solid loading (>10%) led to a slight decrease in the RSY (0.538–0.505 g/g) (Fig. 2A). The decrease in RSY can be ascribed to the increase in total solid content above its optimal value, since high viscosity of the reaction mixture results in mixing barriers that limit heat and mass transfer efficiencies [34]. Pensri et al. [34] also observed a decrease in the reducing sugar yield (768-612 mg/g) when the Napier grass solid loading was increased from 10 to 20%. The combined effects of pretreatment time and solid loading on the RSY and GY when the power intensity was maintained at its median value is illustrated in Fig. 2C and D respectively. Simultaneous increases in the pretreatment time from 2 to 6 min and solid loading from 10 to 20% increased the RSY from 0.446 to 0.482 g/g (Fig. 2C). Figure 2D displayed a similar trend, whereby incremental variations in these same parameters (pretreatment time: 2 to 6 min and solid loading: 10 to 20%) led to an increase in the GY (0.22 to 0.31 g/g). The higher RSY and GY recorded under pretreatment times of 2 to 6 min may be ascribed to a high heating efficiency of the microwave oven. Short pretreatment times are generally preferred, due to the lower energy usage, reduced inhibitor formation, and increase in number of pretreatment cycles [32]. Moodley and

🖉 Springer





Fig.2 Response surface graphs illustrating the interactive effects of the various input pretreatment parameters on the output reducing sugar and glucose yields. A Power intensity and solid loading

Gueguim Kana [13] also noted that an increase in pretreatment time (2-8 min) increased the RSY (0.10 to 0.185 g/g) from NaCl pretreated sugarcane leaf waste. Apart from the effects of microwave power intensity, pretreatment time, and the biomass solid loading, the high sugar yields can also be attributed to the characteristics of the PWW, that enhance the degradation of lignocellulosic structures. For one, a high pH is vital for delignification by facilitating the oxidation of ethylene double bonds in the side chains of lignin structures, thus enhancing the exposure of cellulose and hemicellulose to enzymes during saccharification [35]. In addition, the presence of strong alkali chemicals such as chloride, sulfate, and chromium within PWW has shown to break down glycosidic and ester bonds between ferulic acid and hemicellulose in the extracellular matrix resulting in cellulose swelling, hemicellulose solvation. and lignin degradation using the lab-derived versions [36]. Despite the positive effect of the aforementioned input parameters, further increases in the pretreatment time (>6 min) and solid loading (>20%)

(M-PWW_{RSY}). **B** Power intensity and solid loading (M-PWW_{GY}). **C** Pretreatment time and solid loading (M-PWW_{RSY}). **D** Pretreatment time and solid loading (M-PWW_{GY})

negatively influenced the RSY (0.482–0.438 g/g) (Fig. 2C) and GY (0.31-0.25 g/g) (Fig. 2D). Generally, long exposure times to microwave heating results in high evaporation rates and can account for the reduction in the RSY and GY, since water plays a pivotal role by facilitating heat and mass transfer during chemical processes that occur within the pretreatment reactor [32]. A similar trend was observed by Sewsynker-Sukai and Gueguim Kana [37] after investigating the interactive effects of solid loading (7.5-15%)and alkali-salt concentration (Na₃PO₄.12H₂O) (5-7.5%) on corn cobs. A high solid loading and low Na₃PO₄.12H₂O concentration decreased the RSY (9.64%). Both pretreatments (M-ITS and M-PWW) demonstrated that microwave power intensity had the most significant pretreatment effect on the output reducing sugar and glucose yields obtained from banana pseudostem. All four pretreatment models displayed a strong fit regarding the correlation between the input pretreatment parameters and the output reducing sugar and glucose yield.

3.3 Experimental validation of the developed pretreatment models

The developed pretreatment models were thereafter validated to achieve maximum RSY and GY. The optimal pretreatment conditions predicted by the M-ITS model were 2.48% ITS concentration, 800 W (power intensity), and 10 min (pretreatment time), corresponding to a RSY and GY of 0.48 g/g and 0.39 g/g respectively (M-ITS_{optimized}). Experimental validation of these pretreatment conditions displayed RSY and GY of 0.515 g/g and 0.433 g/g respectively. Deviations of 7.04% and 10.45% were observed between the M-ITS model predicted and experimental yields for RSY and GY respectively. Conversely, the M-PWW model predicted optimal pretreatment conditions of 30% solid loading, 800 W for 8 min that coincided with a RSY and GY of 0.520 g/g and 0.390 g/g respectively (M-PWW_{optimized}). Comparable RSY (0.498 g/g) and GY (0.413 g/g) were observed for the experimentally validated pretreatment conditions. Therefore, minor deviations (RSY = 4.32%and GY = 5.73%) were observed between the model predicted and observed yields. The slight variations between the predicted and experimental sugar yields illustrate the high predictability, accuracy, and validity of the developed pretreatment models. Moreover, optimized pretreatment strategies increased the RSY (M-ITS_{optimized}=2.75-fold and M-PWW_{optimized}=2.66-fold) and GY (M-ITS_{optimized}=4.05fold and \dot{M} -PWW_{optimized} = 3.86-fold) by several fold compared to BP_{untreated}.

3.4 Effects of the M-ITS and M-PWW pretreatments on the lignocellulosic structure of banana pseudostem

3.4.1 Compositional analysis

Changes in lignocellulosic biomass structure may be demonstrated by various analyses that include composition, scanning electron microscopy (SEM), and Fourier transform infrared (FTIR) spectroscopy. The untreated banana pseudostem composition consisted of 5.98% cellulose, 17.54% hemicellulose, and 14.08% lignin. Banana pseudostem pretreated under M-ITS_{optimized} conditions comprised 13.55% cellulose, 16.72% hemicellulose, and 9.73% lignin. This corresponds to a 126.59% cellulose improvement, 4.68% hemicellulose solubilization, and 30.89% lignin removal. The changes in structural composition following the M-ITS_{optimized} pretreatment are due to the dissociation of inorganic salt into complex ions in water that leads to the rupturing of glycosidic linkages and disruption of hydrogen bonds [31]. Similar observations were noted on inorganic salt pretreatment of corn stover [15] and sugarcane leaf waste [13]. In an earlier report, Liu et al. [15] pretreated

corn stover with 0.1 M FeCl₃ at 140 °C for 5–30 min and observed a hemicellulose removal and cellulose recovery of 91% and 30% respectively. More recently, Moodley and Gueguim Kana [13] reported an 8.9, 48.6, and 71.5% hemicellulose solubilization observed for the optimized microwave-assisted-NaCl, -ZnCl₂, and -FeCl₃ pretreatments respectively.

Likewise, the M-PWW_{optimized} banana pseudostem exhibited 17.06% cellulose, 13.04% hemicellulose, and 5.9% lignin. The structural changes corresponded to a 185.28% cellulose improvement, 25.66% hemicellulose solubilization, and 58.1% lignin removal. Variation in the biomass structure can be accounted for by the PWW chemical components (chloride, sulfate, and chromium) coupled with microwave heating (power intensity of 800 W). As previously noted, the strong alkali chemicals present in PWW leads to the degradation of ester and glycosidic bonds between hemicellulose and ferulic acid in the cell wall resulting in the cellulose swelling, hemicellulose solvation, and lignin removal [36]. In addition, hemicellulose solubilization and lignin removal lead to an increase in the pore size and accessible surface area of banana pseudostem, thus exposing cellulose for enzymatic saccharification. Shimizu et al. [19] pretreated banana pseudostem with NaOH and recorded a 75.48% increase in cellulose content, 4.38% hemicellulose solubilization, and 7.65% lignin removal. Therefore, the M-ITS_{optimized} and M-PWW_{optimized} pretreatment methods were effective for disruption of the lignocellulosic structures and enhanced the sugar recovery. The M-PWW_{optimized} method resulted in a 25.9%, 22%, and 39.36% higher cellulose improvement, hemicellulose solubilization, and lignin removal respectively compared to the M-ITS_{optimized} pretreatment. Compositional analysis data is further substantiated with SEM and FTIR analyses.

3.4.2 Visual observations using scanning electron microscopy (SEM)

SEM analysis was used to visually evaluate changes that occurred in the cell wall surface morphology of banana pseudostem before and after pretreatment (Fig. 3). It was observed that pretreatment had induced significant physical changes in the biomass. Native (untreated) banana pseudostem exhibited a smooth and rigid surface area with low porosity and an intact substrate structure. The M-ITS_{optimized} and M-PWW_{optimized} samples both demonstrated the disintegration of the lignocellulosic structures. Damaged cell walls with large pores were observed in the optimally pretreated samples. Furthermore, a highly disorganized morphology was characterized by the exposure of fibers and loosening of the fibrous network in these samples (M-ITS_{optimized} and M-PWW_{optimized}). Therefore, the banana pseudostem was prone to high levels of fragmentation when pretreated with ITS and PWW. During pretreatment,

🖄 Springer



Fig. 3 SEM micrographs (~500×). A untreated, B M-ITS $_{\rm optimized}$ and C M-PWW $_{\rm optimized}$ banana pseudostem

hydrogen bonds are damaged, resulting in the leakage of lignin moieties [31]. Furthermore, an increase in surface area and porosity enhances the availability of cellulose and hemicellulose structures to enzymatic attack during saccharification [29]. Particularly, the use of salts such as ITS lead to hemicellulose solubilization, due to the cleavage of glycosidic linkages, whereas PWW causes disruptions to the recalcitrant aromatic polymer lignin. The SEM results were in accordance with the compositional analysis in addition to previous studies on corn stover [15] and banana pseudostem [4] pretreated with inorganic salt and alkali (NaOH) respectively.

3.4.3 Fourier transform infrared (FTIR) analysis

FTIR spectral analysis was utilized to study the physicochemical properties and conformational variations introduced by the optimized pretreatments on banana pseudostem. The chemical and structural changes of cellulose, hemicellulose, and lignin are represented in the wavelength range of 650 to 4000 cm⁻¹. The absorption spectrum of native and optimized (M-ITS_{optimized} and M-PWW_{optimized}) samples displayed distinct peaks at various wavelengths with strong, banding patterns. More specifically, chemical modifications introduced by pretreatment for the M-ITS_{optimized} and M-PWW_{optimized} displayed higher absorption peaks and intensities compared to the native banana pseudostem substrate. For instance, the absorptive peak at 840 cm^{-1} arises from the C-H out of plane bending and implies that the benzene rings of lignin are 1,3,5-trisubstituted [38]. Conversely, the absorptive band at 1000 cm⁻¹ is ascribed to the aromatic C-H and C-O distortion and stretching in (alcohols, ethers, or carboxylic acids) of cellulose and hemicellulose [38, 39]. The absorption peak at 1170 cm^{-1} represents the C-O-C asymmetric stretching and vibrations in cellulose and hemicellulose associated with the pyranose ring skeleton [39]. Two absorption peaks were detected at wavelengths of 1315 cm⁻¹ and 1490 cm⁻¹ and occur in the range between 1300 and 1500 cm⁻¹, illustrating the C-H-bending vibrations of cellulose [40]. On the other hand, the band observed at 1615 cm⁻¹ demonstrates in-plane C = C (carbonyl and/or carboxylic) aromatic vibration in the aromatic ring within lignin [40]. A faint band occurred at 2145 cm^{-1} and is indicative of stretching of the triple $-C \equiv C$ -bond in the alkyne compounds present within lignin. Additionally, C-H stretching and -OH stretching vibrations of intramolecular hydrogen within cellulose were represented by absorptive peaks at 2905 cm⁻¹ and 3272 cm⁻¹, respectively [40]. Increases in the intensity at these band positions characteristically indicate the recovery of cellulose from lignocellulosic biomass after pretreatment. Spectral data from the FTIR analysis indicated that the M-ITS_{optimized} and M-PWW_{optimized} pretreatments effectively unravelled lignin and hemicellulose, while altering crystalline cellulose to its amorphous form. The FTIR data was in agreement with the compositional and SEM analysis, in addition to previous reports on inorganic salt (NaClO) [36] and alkali (NaOH) [9] pretreatments.

3.5 Comparative assessment of the optimized M-ITS and M-PWW pretreatment conditions on different lignocellulosic substrates

The variation in the structural composition of different lignocellulosic substrates remains high due to the distribution and proportion of cellulose, hemicellulose, and lignin content. Additionally, the recalcitrant nature of a particular substrate hinders the application of different pretreatment strategies rendering it a research hotspot, since a "one-sizefits-all" approach results in low hydrolysis rates and subsequently negligible fermentation product yields. Therefore, to determine the versatility of the developed pretreatment methods, different lignocellulosic substrates such as corn cobs, sorghum leaves, sugarcane bagasse, and bamboo were subjected to the M-ITS_{optimized} and M-PWW_{optimized} process conditions. Following pretreatment of the different substrates (corn cobs, sorghum leaves, sugarcane bagasse, and bamboo), RSY and GY from these experiments together with the optimized banana pseudostem data were compared to previous pretreatment reports for the corresponding lignocellulosic substrate. The M-ITS_{optimized} and M-PWW_{optimized} pretreatments led to higher sugar yields when compared to previous studies using banana pseudostem. For instance, Idrees et al. [41] utilized an autoclave-assisted dilute acid pretreatment and observed a lower RS concentration of 38 g/L (corresponding to 380 mg/g) in comparison to this study (M-ITS_{optimized}). Similarly, Chen et al. [42] employed a strong alkali (8% NaOH) pretreatment on banana pseudostem and observed a lower RSY of 0.34 g/g (340 mg/g) when compared to this study (M-PWW_{optimized}). In recent studies, Ayeni et al. [26] and Dolmani and Livleen [43] employed an inorganic salt (ZnCl₂) and alkali (NaOH) pretreatment respectively on corn cobs and noted a lower reducing sugar yield of 0.075 g/g (75.54 mg/g) and 0.071 g/g (71.7 mg/g) respectively, in comparison to the present study. Deshavath et al. [44] pretreated sorghum leaves with dilute acid (0.1 M H_2SO_4) and observed a lower GY of 0.052 g/g (52.54 mg/g) compared to the present study (M-ITS_{optimized}). Likewise, Manmai et al. [45] pretreated sorghum leaves with strong alkali (2% NaOH) and observed a lower RSY 0.176 g/g (176 mg/g) compared to the present study (M-PWW_{optimized}). Interestingly, Tiwari et al. [46] investigated H₂SO₄ and NaOH pretreatments using sugarcane bagasse and observed a higher reducing sugar yield of 0.251 g/g (251 mg/g) and 0.290 g/g (290 mg/g) respectively. Similarly, Wang et al. [47] and Leenakul and Tippayawong [48] pretreated bamboo and reported lower reducing sugar yields of 0.121 (121 mg/g) and 0.085 g/g (85 mg/g) respectively, compared to the $M\text{-}ITS_{\text{optimized}}$ and $M\text{-}PWW_{\text{optimized}}$ pretreatment of bamboo. Even though the aforementioned studies [46–48] demonstrated higher RSY compared to the present corresponding substrate, the components that constitute reducing sugar include monosaccharides, disaccharides, and polysaccharides, which are not a direct indication of high fermentable sugars (such as glucose). Besides the reducing sugar yield not being representative of actual fermentable sugars, the use of pure chemicals such as acids and/or alkali in the previous reports by Tiwari et al. [46], Wang et al. [47], and Leenakul and Tippayawong [48] adds to the process costs and negatively impacts on the environment, thereby rendering these unfeasible for lignocellulosic pretreatment.

On the other hand, ITS is cost-effective, abundant, and readily available when compared to its lab-based counterpart (NaCl). The higher RSY and GY achieved in the present study may be attributed to the nucleophilic ability of ITS to depolymerize hemicellulose into monosaccharides, while cleaving glycosidic linkages and hydrogen bonds thereby reducing the crystalline structure of cellulose [12, 16]. Additionally, PWW exhibits a strong alkalic effect that targets the recalcitrant aromatic polymer lignin and facilitates in the saponification of intermolecular ester bonds and oxidation of ethylene double bonds [19, 20, 35]. The present study provides valuable insights for lignocellulosic pretreatment processes, since both ITS and PWW demonstrate high effectiveness compared to their lab-grade equivalents. Despite both pretreatments demonstrating comparable reducing sugar and glucose yields compared to previous pretreatment reports for the corresponding lignocellulosic substrate, the M-PWW_{optimized} is a completely waste-based method that outweighs the M-ITS_{optimized} pretreatment in several aspects. For instance, the $M-\dot{P}WW_{optimized}$ conditions gave a 30% solid loading, which led to a threefold higher quantity of pretreated banana pseudostem substrate compared to the M-ITS_{optimized} strategy (10% solid loading). Therefore, the M-PWW_{optimized} reduces the number of pretreatment cycles required, by up to 3 times in contrast to the M-ITS_{optimized} pretreatment. This implies that a threefold lower energy and shorter time will be required when using the M-PWW_{optimized} pretreatment instead of the M-ITS_{optimized} method. In addition, repurposing PWW for lignocellulosic pretreatment has the potential to drastically reduce expenditure and negative environmental effects by (1) eliminating hazardous chemicals, (2) reducing wastewater treatment costs, and (3) negating the use of finite resources such as fresh water in lignocellulosic pretreatment systems. Moreover, collaborative efforts between the agricultural, Kraft paper and pulp, and biotechnology sectors may increase profits, while reducing wastes, since the exploitation of PWW and LCB residues as an additional revenue stream can be used to (1) finance downstream wastewater treatment processes for Kraft industries and (2) prevent the burning and/or disposal of LCB that ends up in land and water bodies, thereby minimizing the release of harmful greenhouse gases into the environment and curbing climate change.

🖄 Springer

4 Conclusion

Two novel microwave-assisted pretreatments using iodized table salt (M-ITS) and paper wastewater (M-PWW) were optimized on banana pseudostem. Although the M-PWW_{optimized} regime displayed a slightly lower RSY (3.3%) and GY (4.62%) compared to the M-ITS_{optimized} strategy, it has been concluded to be more effective, since it is a complete waste-based method and requires fewer pre-treatment cycles. Furthermore, assessment of the optimized conditions on different lignocellulosic substrates validated the suitability of the developed pretreatment models. This study provides major insights for eliminating chemicals and/ or fresh water application in lignocellulosic biorefineries, while generating additional revenue streams for the industrial sector.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s13399-022-02630-x.

Funding This research is financially supported by the National Research Foundation (NRF) of South Africa (Grant Numbers 122341 and 129556).

Declarations

Competing interests The authors declare no competing interests.

Disclosure Opinions expressed and conclusions arrived at, are those of the author and are not necessarily attributed to the NRF.

References

- Malik K, Salama ES, Kim TH, Li X (2020) Enhanced ethanol production by Saccharomyces cerevisiae fermentation post acidic and alkali chemical pretreatments of cotton stalk lignocellulose. Int Biodeterior 147:104869
- Wi SG, Cho EJ, Lee DS, Lee SJ, Lee YJ, Bae HJ (2015) Lignocellulose conversion for biofuel: a new pretreatment greatly improves downstream biocatalytic hydrolysis of various lignocellulosic materials. Biotechnol Biofuels 8:1–11
- Deshavath NN, Mukherjee G, Goud VV, Veeranki VD, Sastri CV (2020) Pitfalls in the 3, 5-dinitrosalicylic acid (DNS) assay for the reducing sugars: interference of furfural and 5-hydroxymethylfurfural. Int J Biol 156:180–185
- Sivanarutselvi S, Poornima P, Muthukumar K, Velan M (2019) Studies on effect of alkali pretreatment of banana pseudostem for fermentable sugar production for biobutanol production. J Environ Biol 40:393–399
- Sewsynker-Sukai Y, Gueguim Kana EB (2018) Microwaveassisted alkalic salt pretreatment of corn cob wastes: process optimization for improved sugar recovery. Ind Crops Prod 125:284–292
- Yang H, Shi Z, Xu G, Qin Y, Deng J, Yang J (2019) Bioethanol production from bamboo with alkali-catalyzed liquid hot water pretreatment. Bioresour Technol 274:261–266

- Moodley P, Gueguim Kana EB (2019) Bioethanol production from sugarcane leaf waste: effect of various optimized pretreatments and fermentation conditions on process kinetics. Biotechnol Rep 22:1–8
- Islam M, Asad M, Hossain M, Paul S, Sujan S (2019) Bioethanol production from banana pseudostem by using separate and cocultures of cellulase enzyme with Saccharomyces cerevisiae. Environ Sci Technol 12:157–163
- 9. Subagyo A, Chafidz A (2018) Banana pseudo-stem fiber: preparation, characteristics, and applications. INTECH 1:1–20
- Gu T, Wang B, Zhang Z, Wang Z, Chong G, Ma C, Tang YJ, He Y (2019) Sequential pretreatment of bamboo shoot shell and biosynthesis of ethyl (R)-4-chloro-3-hydroxybutanoate in aqueous-butyl acetate media. Process Biochem 80:112–118
- Maurya DP, Singla A, Negi S (2015) An overview of key pretreatment processes for biological conversion of lignocellulosic biomass to bioethanol. Biotechnol J 5:597–609
- Kang KE, Park DH, Jeong GT (2013) Effects of inorganic salts on pretreatment of Miscanthus straw. Bioresour Technol 132:160–165
- Moodley P, Gueguim Kana EB (2017) Microwave-assisted inorganic salt pretreatment of sugarcane leaf waste: effect on physiochemical structure and enzymatic saccharification. Bioresour Technol 235:35–42
- Suwannabun P, Cheenkachorn K, Prongjit M, Tawai A, Sriariyanun M (2019) Pretreatment of rice straw by inorganic salts and 1-Ethyl-3-methylimdazolium acetate for biofuel production. ACEEE 1:12–15
- Liu L, Sun J, Cai C, Wang S, Pei H, Zhang J (2009) Corn stover pretreatment by inorganic salts and its effects on hemicellulose and cellulose degradation. Bioresour Technol 100:65–71
- Moodley P, Sewsynker-Sukai Y, Gueguim Kana EB (2020) Progress in the development of alkali and metal salt catalysed lignocellulosic pretreatment regimes: potential for bioethanol production. Bioresour Technol 310:1–49
- Gorjian S, Jamshidian FJ, Hosseinqolilou B (2019) Feasible solar applications for brines disposal in desalination plants. Desalination 1:25–48
- Haake DM, Knouft JH (2019) Comparison of contributions to chloride in urban stormwater from winter brine and rock salt application. Environ Sci Technol 53:11888–11895
- Shimizu FL, Monteiro PQ, Ghiraldi PHC, Melati RB, Pagnocca FC, Souza WD, Sant'anna C, Brienzo M (2018) Acid, alkali and peroxide pretreatments increase the cellulose accessibility and glucose yield of banana pseudostem. Ind Crops Prod 115:62–68
- Kim JS, Lee Y, Kim TH (2016) A review on alkaline pretreatment technology for bioconversion of lignocellulosic biomass. Bioresour Technol 199:42–48
- 21. Laltha M, Sewsynker-Sukai Y, Gueguim Kana EB (2021) Development of microwave-assisted alkaline pretreatment methods for enhanced sugar recovery from bamboo and corn cobs: process optimization, chemical recyclability and kinetics of bioethanol production. Ind Crops Prod 174:114166
- 22. Wagle D, Lin CJ, Nawaz T, Shipley H (2020) Evaluation and optimization of electrocoagulation for treating Kraft paper mill wastewater. J Environ Chem Eng 8:1–9
- Ashrafi O, Yerushalmi L, Haghighat F (2015) Wastewater treatment in the pulp-and-paper industry: a review of treatment processes and the associated greenhouse gas emission. J Environ Manage 158:146–157
- 24. Manasa RL, Mehta A (2020) Wastewater: sources of pollutants and its remediation. Environ Biotechnol 2:197–219
- Zhu YJ, Chen F (2014) Microwave-assisted preparation of inorganic nanostructures in liquid phase. Chem Rev 114:6462–6555
- Ayeni AO, Daramola MO, Agboola O, Ayoola AA, Babalola R, Oni BA, Omodara JO, Dick DT (2020) A comparative evaluation

D Springer

of fermentable sugars production from oxidative, alkaline, alkaline peroxide oxidation, dilute acid, and molten hydrate salt pretreatments of corn cob biomass. AIMS Energy 9:15–28

- Van Soest PJ, McQueen RW (1973) The chemistry and estimation of fibre. Proc Nutr Soc 32:123–130
- Miller GL (1959) Use of dinitrosalicylic acid reagent for determination of reducing sugar. Anal Chem 31:426–428
- Qing Q, Zhou L, Huang M, Guo Q, He Y, Wang L, Zhang Y (2016) Improving enzymatic saccharification of bamboo shoot shell by alkalic salt pretreatment with H2O2. Bioresour Technol 201:230–236
- Bozkurt YC, Apul OG (2020) Critical review for microwave pretreatment of waste-activated sludge prior to anaerobic digestion. Curr Opin Environ Sci 14:1–9
- 31. Zhang Y, Huang M, Su J, Hu H, Yang M, Huang Z, Chen D, Wu J, Feng Z (2019) Overcoming biomass recalcitrance by synergistic pretreatment of mechanical activation and metal salt for enhancing enzymatic conversion of lignocellulose. Biotechnol Biofuels 12:1–12
- 32. Aguilar-Reynosa A, Romaní A, Rodríguez-Jasso RM, Aguilar CN, Garrote G, Ruiz HA (2017) Microwave heating processing as alternative of pretreatment in second-generation biorefinery: an overview. Energy Convers Manag 136:50–65
- 33. Alagu K, Muthusamy SR, Muthusamy B, Pugazhendhi A (2018) Optimization of sugar recovery efficiency using microwave assisted alkaline pretreatment of cassava stem using response surface methodology and its structural characterization. J Mol Liq 254:55–63
- Pensri B, Aggarangsi P, Chaiyaso T, Chandet N (2016) potential of fermentable sugar production from Napier cv. Pakchong 1 grass residue as a substrate to produce bioethanol. Energy Procedia 89:428–436
- 35. Geng W, Huang T, Jin Y, Song J, Chang HM, Jameel H (2014) Comparison of sodium carbonate–oxygen and sodium hydroxide– oxygen pretreatments on the chemical composition and enzymatic saccharification of wheat straw. Bioresour Technol 161:63–68
- 36. Costa S, Rugiero I, Larenas Uria C, Pedrini P, Tamburini E (2018) Lignin degradation efficiency of chemical pre-treatments on banana rachis destined to bioethanol production. Biomolecules 8:141
- Sewsynker-Sukai Y, Gueguim Kana EB (2017) Optimization of a novel sequential alkalic and metal salt pretreatment for enhanced delignification and enzymatic saccharification of corn cobs. Bioresour Technol 243:785–792
- Shinde SD, Meng X, Kumar R, Ragauskas A (2018) Recent advances in understanding the pseudo-lignin formation in a lignocellulosic biorefinery. Green Chem 20:2192–2205

- Boukir A, Fellak S, Doumenq P (2019) Structural characterization of Argania spinosa Moroccan wooden artifacts during natural degradation progress using infrared spectroscopy (ATR-FTIR) and X-ray diffraction (XRD). Heliyon 5:1–9
- 40. Li X, Wei Y, Xu J, Xu N, He Y (2018) Quantitative visualization of lignocellulose components in transverse sections of moso bamboo based on FTIR macro-and micro-spectroscopy coupled with chemometrics. Biotechnol 11:1–16
- Idrees M, Adnan A, Malik F, Qureshi FA (2013) Enzymatic saccharification and lactic acid production from banana pseudo-stem through optimized pretreatment at lowest catalyst concentration. EXCLI J 12:269
- 42. Chen XL, Han YX, Zhang CM, Feng GQ, Zhao MX, Yue RX, Li YF, Jiang L, Zhang L, Li JH (2017) Alkaline pretreatment of banana stems for methane generation: effects of temperature and physicochemical changes. Bioresour Technol 12:5601–5616
- Dolmani A, Livleen S (2021) Evaluation of different pretreatment methods for xylooligosaccharides production. Int J Curr Microbiol Appl Sci 1:517–527
- 44. Deshavath NN, Mahanta S, Goud VV, Dasu VV (2018) Chemical composition analysis of various genetically modified sorghum traits: pretreatment process optimization and bioethanol production from hemicellulosic hydrolyzates without detoxification. J Environ Chem Eng 6:5625–5634
- 45. Manmai N, Unpaprom Y, Ponnusamy VK, Ramaraj R (2020) Bioethanol production from the comparison between optimization of sorghum stalk and sugarcane leaf for sugar production by chemical pretreatment and enzymatic degradation. J Fuels 278:1–14
- 46. Tiwari S, Yadav J, Gaur R, Yadav JS (2020) Evaluation of pretreatment methods (acid and alkali) in improving the enzymatic saccharification of sugarcane baggase: structural and chemical analysis. Res Sq 1:1–27
- 47. Wang BQ, Peng B, Sun JF, Wang C, Jiang K, Ma JH, Zhang PQ, He YC (2019) Pretreatment of bamboo shoot shell with surfactant OP-10 in the acidified media for enhancing the biosynthesis of ethyl (S)-4-chloro-3-hydroxybutanoate. Bioresour Technol Rep 5:74–79
- Leenakul W, Tippayawong N (2010) Dilute acid pretreatment of bamboo for fermentable sugar production. Int J Sustain Energy 1:117–120

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Chapter 3 - Supplementary material

Chemical substance	LC (mg/L)
Arsenic	<0.0025
Boron	0.215
Barium	0.175
Cadmium	<0.0005
Cobalt	<0.002
Chromium	0.5351
Hexavalent Chromium	<0.006
Copper	<0.007
Mercury	<0.001
Manganese	1
Molybdenum	<0.002
Nickel	0.009
Lead	<0.005
Antimony	0.007
Selenium	<0.003
Vanadium	0.0144
Zinc	0.031
Total Dissolved Solids	7522
Chloride	1114.1
Sulphate	448.5
Nitrate	<0.05
Fluoride	<0.3
Carbon and Nitrogen total	<0.06

Table S1. Sample analysis of the major constituents present in paper wastewater

Footnote: LC = The leachable concentration of a particular element or chemical substance in a waste (mg/L). The chemical substance LC was below the threshold toxic level limit (<LCT1) and classified the paper wastewater as a Type 3 (low-risk) waste (Government Gazette 36784, 2013).

Screening method	Input parameters	Reducing sugar	Glucose yield (g/g)
		yield (g/g)	
M-NaCl _{preliminary}	3% w/v NaCl, 500 W, 6 min	0.383	0.260
M-ITS _{preliminary}	3% w/v ITS, 500 W, 6 min	0.424	0.298
M-PWW _{preliminary}	10% S/L, 500 W, 6 min	0.366	0.266
M-PWW-NaClpreliminary	10% S/L ^a , 500 W ^a , 3 min ^a ,	0.404	0.245
	3% w/v NaCl ^b , 500 W ^b , 3 min ^b		
M-PWW-ITS _{preliminary}	10% S/L ^a , 500 W ^a , 3 min ^a ,	0.402	0.272
	3% w/v ITS ^b , 500 W ^b , 3 min ^b		
BP _{water}	500 W, 6 min	0.286	0.182
BPuntreated	-	0.187	0.107

Table S2. Preliminary screening of M-NaCl, M-ITS and M-PWW to determine their pretreatment effectiveness on banana pseudostem

Footnote: M-NaCl = Microwave-assisted-sodium chloride, M-ITS = Microwave-assistediodized table salt, M-PWW = Microwave-assisted-paper wastewater, BP_{water} = distilled water pretreated banana pseudostem, $BP_{untreated}$ = untreated/native banana pseudostem, S/L = solid loading, ^a = first-stage pretreatment, ^b = second-stage pretreatment.

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.3004	9	3.34×10 ⁻²	182.87	<0.0001 Significant
A-Salt concentration	1×10 ⁻⁴	1	1×10 ⁻⁴	0.3372	0.5796
B-Power intensity	2.063×10 ⁻¹	1	2.063×10 ⁻¹	1130.27	< 0.0001
C-Pretreatment time	2.6×10 ⁻³	1	2.6×10 ⁻³	14.08	0.0071
AB	3×10 ⁻⁴	1	3×10 ⁻⁴	1.46	0.2656
AC	0	1	0	0.1376	0.7217
BC	5×10 ⁻⁴	1	5×10 ⁻⁴	2.87	0.1338
A^2	6×10 ⁻⁴	1	6×10 ⁻⁴	3.47	0.1049
B^2	8.886×10 ⁻²	1	8.886×10 ⁻²	485.32	< 0.0001
C^2	3.6×10 ⁻⁶	1	3.6×10 ⁻⁶	0.0197	0.8923
Residual	1.3×10 ⁻³	7	2×10 ⁻⁴		
Lack of fit	9×10 ⁻⁴	3	3×10 ⁻⁴	2.93	0.1631 Not significant

Table S3.	Analysis	of	variance	of	the	developed	ITSRSY	pretreatment	model	on	banana
pseudosten	n										

Footnote: df = degrees of freedom, F-value = Fisher-Snedecor distribution value, P-value = probability value.

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.1967	9	2.19×10 ⁻²	92.26	< 0.0001 Significant
A-Salt concentration	7×10 ⁻⁴	1	7×10 ⁻⁴	3.04	0.1246
B-Power intensity	0.1417	1	0.1417	598.37	< 0.0001
C-Pretreatment time	9×10 ⁻³	1	9×10 ⁻³	38.21	0.0005
AB	1×10 ⁻⁴	1	1×10 ⁻⁴	0.2255	0.6493
AC	0	1	0	0.15	0.71
BC	3.3×10 ⁻³	1	3.3×10 ⁻³	14.02	0.0072
A^2	1×10 ⁻³	1	1×10 ⁻³	4.11	0.0822
B^2	3.71×10 ⁻²	1	3.71×10 ⁻²	156.75	< 0.0001
C^2	1.7×10 ⁻³	1	1.7×10 ⁻³	7.26	0.0309
Residual	1.7×10 ⁻³	7	2×10 ⁻⁴		
Lack of fit	7×10 ⁻⁴	3	2×10 ⁻⁴	0.8769	0.5240 Not significant

Table S4. Analysis of variance of the developed ITS_{GY} pretreatment model on banana pseudostem

Footnote: df = degrees of freedom, F-value = Fisher-Snedecor distribution value, P-value = probability value.

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.3747	9	4.16×10 ⁻²	21.27	0.0003 Significant
A-Solid loading	7×10 ⁻⁴	1	7×10 ⁻⁴	0.3724	0.5610
B-Power intensity	0.2669	1	0.2669	136.34	< 0.0001
C-Pretreatment time	2.2×10 ⁻³	1	2.2×10 ⁻³	1.14	0.3214
AB	4.008×10 ⁻⁶	1	4.008×10 ⁻⁶	2×10 ⁻³	0.9652
AC	1.8×10 ⁻³	1	1.8×10 ⁻³	0.9053	0.3731
BC	2.416×10 ⁻⁷	1	2.416×10 ⁻⁷	1×10 ⁻⁴	0.9914
A^2	0	1	0	1.81×10 ⁻²	0.8969
B^2	9.22×10 ⁻²	1	9.22×10 ⁻²	47.09	0.0002
C^2	7.2×10 ⁻³	1	7.2×10 ⁻³	3.67	0.0971
Residual	1.37×10 ⁻²	7	2×10 ⁻³		
Lack of fit	1.12×10 ⁻²	3	3.7×10 ⁻³	6	0.0580 Not significant

Table S5. Analysis of variance of the developed PWW_{RSY} pretreatment model on banana pseudostem

Footnote: df = degrees of freedom, F-value = Fisher-Snedecor distribution value, P-value = probability value.

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.1794	9	1.99×10 ⁻²	7.01	0.0088 Significant
A-Solid loading	1.6×10 ⁻³	1	1.6×10 ⁻³	0.5472	0.4835
B-Power intensity	0.1289	1	0.1289	45.35	0.0003
C-Pretreatment time	5.2×10 ⁻³	1	5.2×10 ⁻³	1.83	0.2186
AB	7.8×10 ⁻³	1	7.8×10 ⁻³	2.74	0.1418
AC	5×10 ⁻⁴	1	5×10 ⁻⁴	0.1929	0.6737
BC	0	1	0	9.7×10 ⁻³	0.9243
A^2	1.8×10 ⁻³	1	1.8×10 ⁻³	0.6189	0.4572
B^2	2.44×10 ⁻²	1	2.44×10 ⁻²	8.58	0.0220
C^2	6.5×10 ⁻³	1	6.5×10 ⁻³	2.29	0.1742
Residual	1.99×10 ⁻²	7	2.8×10 ⁻³		
Lack of fit	1.12×10 ⁻²	3	3.7×10 ⁻³	1.73	0.2985 Not significant

Table S6. Analysis of variance of the developed PWW_{GY} pretreatment model on banana pseudostem

Footnote: df = degrees of freedom, F-value = Fisher-Snedecor distribution value, P-value = probability value

Model		Input parameters			Reducing	sugar yield	Glucose yield (g/g)		
					(g	/g)			
	Salt	Solid	Power	Pretreatment	Predicted	Observed	Predicted	Observed	
	conc.	loading	intensity	time (min)					
	(%)	(%)	(W)						
M-ITS	2.48	-	800	10	0.48	0.515	0.39	0.433	
M-PWW	-	30	800	8	0.52	0.498	0.39	0.413	

Table S7. Validation of optimized conditions for the M-ITS and M-PWW pretreatment on banana pseudostem

Table S8. Structural composition of the control (untreated) and optimally pretreated banana pseudostem.

Sample	Cellulose (%)	Hemicellulose (%)	Lignin (%)
Native (untreated)	5.98	17.54	14.08
M-ITS _{optimized}	13.55	16.72	9.73
	17.06	12.04	5.0
IVI-P VV VV optimized	17.00	15.04	5.9

Footnote: M-ITS_{optimized} = Optimized microwave-assisted-iodized table salt, M-PWW_{optimized} = Optimized microwave-assisted-paper wastewater.

Substrate	Heating	Pretreatment parameters	Reducing sugar yield	Glucose yield (g/g)	References
	mechanism		(g/g)		
Banana pseudostem	Microwave	2.48% ITS, 800 W, 10 min	0.515	0.433	Present study
Banana pseudostem	Microwave	30% solid loading, 800 W, 8 min	0.498	0.413	Present study
Banana pseudostem	Autoclave	1% H ₂ SO ₄ , 130 °C, 2 h	0.380	ND	Idrees et al. (2013)
Banana pseudostem	Autoclave	8% NaOH (w/w), 100 °C, 24 h	0.340	ND	Chen et al. (2017)
Corn cobs	Microwave	2.48% ITS, 800 W, 10 min	0.174	0.086	Present study
Corn cobs	Microwave	30% solid loading, 800 W, 8 min	0.238	0.124	Present study
Corn cobs	Water bath	0.1 g ZnCl ₂ .4H ₂ O/g substrate, 45 °C,	0.093	ND	Ayeni et al. (2020)
		96 h			
Corn cobs	Incubator	2% NaOH, 20 °C, 6 h	0.072	ND	Dolmani and Livleen (2021)
Sorghum leaves	Microwave	2.48% ITS, 800 W, 10 min	0.154	0.096	Present study
Sorghum leaves	Microwave	30% solid loading, 800 W, 8 min	0.211	0.123	Present study
Sorghum leaves	Autoclave	0.1 M H ₂ SO ₄ , 121 °C, 30 min	ND	0.038	Deshavath et al. (2018)
Sorghum leaves	Water bath	2% NaOH, 40 °C, 3 d	0.032	ND	Manmai et al. (2020)

Table S9. Comparison of the developed pretreatment strategies on various lignocellulosic substrates

Table S9. Continued

Substrate	Heating	Pretreatment parameters	Reducing sugar yield	Glucose yield (g/g)	References
	mechanism		(g/g)		
Sugarcane bagasse	Microwave	2.48% ITS, 800 W, 10 min	0.093	0.046	Present study
Sugarcane bagasse	Microwave	30% solid loading, 800 W, 8 min	0.117	0.051	Present study
Sugarcane bagasse	Autoclave	5% H ₂ SO ₄ , 121 °C, 30 h	0.251	ND	Tiwari et al. (2020)
Sugarcane bagasse	Autoclave	5% NaOH, 121 °C, 30 h	0.290	ND	Tiwari et al. (2020)
Bamboo	Microwave	2.48% ITS, 800 W, 10 min	0.058	0.027	Present study
Bamboo	Microwave	30% solid loading, 800 W, 8 min	0.065	0.029	Present study
Bamboo	Autoclave	0.5% OP-10 emulsifier, pH 4, 120	0.121	0.088	Wang et al. (2019)
		°C, 40 min			
Bamboo	Autoclave	1.2% H ₂ SO ₄ , 120 °C, 1 h	0.084	ND	Leenakul and Tippayawong
					(2010)

Footnote: ND = Not determined.



Fig. S1. Response surface graphs illustrating the interactive effects of the various input pretreatment parameters on the output reducing sugar and glucose yields: (A) pretreatment time and salt concentration (M-ITS_{RSY}); (B) pretreatment time and salt concentration (M-ITS_{GY}); (C) pretreatment time and power intensity (M-PWW_{RSY}); (D) pretreatment time and power intensity (M-PWW_{GY}).



Fig. S2. FTIR spectra of the untreated, M-ITS_{optimized} and M-PWW_{optimized} banana pseudostem.

CHAPTER 4

Simultaneous saccharification and citric acid production from paper wastewater pretreated banana pseudostem: Optimization of fermentation medium formulation and kinetic assessment

This chapter has been submitted to the journal *Bioresource Technology* with the title: Simultaneous saccharification and citric acid production from paper wastewater pretreated banana pseudostem: Optimization of fermentation medium formulation and kinetic assessment. (Under review). The manuscript and supplementary material are presented in the following pages.

Highlights

- First report on the optimization of citric acid using banana pseudostem
- The optimized process gave a maximum citric acid concentration of 14.408 g/L
- Kinetics were determined on fresh water, wastewater and standard medium
- The wastewater-based bioprocess demonstrated only a 9.11% lower citric acid concentration
- Findings facilitate the development of waste-based lignocellulosic bioprocesses

Graphical Abstract



Simultaneous saccharification and citric acid production from paper wastewater pretreated banana pseudostem: Optimization of fermentation medium formulation and kinetic assessment

Milesh Laltha^a, Y. Sewsynker-Sukai^b and ^{*}E.B. Gueguim Kana^a

^aUniversity of KwaZulu-Natal, School of Life Sciences, Pietermaritzburg, South Africa ^bUniversity of Fort Hare, Fort Hare Institute of Technology, Private Bag X1314, Alice 5700, South Africa

*Correspondence to: E.B. Gueguim Kana (kanag@ukzn.ac.za)

Abstract

This study optimized the simultaneous saccharification and citric acid (CA) production from banana pseudostem (BP). Thereafter, kinetic assessment of *Aspergillus brasiliensis* growth and CA production were determined for the optimum conditions using fresh water (SSF_{optimizedFw}) or dairy wastewater (SSF_{Dww}) and compared to Sabouraud Dextrose Emmon's medium modified with BP (SSF_{SDEmodified}). The optimized conditions (0.5% ammonium nitrate, 1% acetone and 35 °C) gave a CA concentration of 14.408 g/L. Kinetic assessment revealed the same maximum specific growth rates (μ_{max}) (0.05 h⁻¹) for all three bioprocesses, while the SSF_{SDEmodified} process resulted in the highest maximum potential CA concentration (Pm) (13.991 g/L) in comparison to the SSF_{DWW} (Pm= 13.095 g/L) and SSF_{optimizedFW} (Pm= 12.967 g/L) systems. Findings from this study facilitates the implementation of waste-based lignocellulosic bioprocesses that may eradicate the use of expensive pretreatment chemicals, fermentation medium constituents, and resources, in keeping with the water, energy and food nexus towards developing a circular bioeconomy.

Keywords: citric acid; dairy wastewater; banana pseudostem; simultaneous saccharification and fermentation.

1. Introduction

The bioconversion of renewable lignocellulosic biomass to valuable products such as citric acid (CA) has become a research hotspot towards achieving sustainable global economic development (Yadav et al., 2020). Citric acid is a valuable commercial bioproduct with an existing market and widespread use in the chemical, food, pharmaceutical, beverage, electroplating and cosmetic industries (Sekoai et al., 2018). Numerous microorganisms have been shown to produce CA and comprise Absidia sp., Botrytis, Acremonium, Eupenicillium, Aspergillus and Penicillium sp. However, approximately 80% of the world's CA is generated mainly by Aspergillus niger. A. niger has typically been used, due to its exclusive physiological characteristics for industrial CA fermentation. This microbe exhibits polymer degrading enzymes that hydrolyze polymeric substrates, enabling rapid growth and fermentation on several low-cost substrates. In addition, it tolerates extreme acidic environments and outcompetes other microorganisms, thereby reducing the risk for contamination (Behera et al., 2021). A. niger shares a phylogenetic relationship with several Aspergillus genomes that include Aspergillus tubingensis, Aspergillus brasiliensis and Aspergillus japonicus (Varga et al., 2007). More specifically, A. brasiliensis isolates produce tensidol A and B, naphthogamma-pyrones and share this with A. tubingensis and A. niger. On the contrary, there are also various other unique compounds that A. brasiliensis produces, justifying its separation into a separate species from A. niger (Varga et al., 2007), while still demonstrating high potential for CA production. Citric acid fermentation occurs via the citric acid cycle (TCA) in microorganisms. During the bioconversion process, the production of organic acids helps to regulate cytoplasmic pH (3 to 3.5) for optimal cell growth (Odu et al., 2020). Approximately 2 million tons of CA is generated annually and its rate of demand is expected to rise by 5% each year (Ozdal et al., 2019, Sekoai et al., 2018). The CA market is constantly under pressure due to expensive substrates and low production yields (Sekoai et al., 2018). Additionally, the cost of CA production via chemical synthesis is approximately ~\$10 USD per kg, whereas for fungal fermentation it has been estimated to be ~50% lower (Ozdal et al., 2019). The global market for CA is expected to surpass \$3.2 billion USD by 2023 (Behera et al., 2021). More importantly, the CA global demand has surpassed the natural supply, therefore the incorporation of lignocellulosic-based biotechnological processes that are novel, low-cost, efficient, and environmentally friendly are urgently required. Lignocellulosic biomass is a naturally occurring complex composite that originates from plant dry matter (Moodley et al., 2020), and has emerged as an ideal feedstock for generating renewable bioproducts such as CA. This is attributed to its abundance and cost-effectiveness. In addition, LCB is considered a waste and therefore does not pose any risk to socio-economic concerns such as deforestation, food security and water shortages (Sekoai et al., 2019). Recently, CA has been derived from lignocellulosic substrates that include banana peels (Odu et al., 2020), sweet potato peels (Aboyeji et al., 2020), and sugarcane bagasse (Campanhol et al., 2019). Banana pseudostem with a composition of cellulose (42.35%), hemicellulose (22.63%), lignin (15.36%) represents a novel, cost-effective and abundant substrate for generating CA (Islam et al., 2019). However, lignocellulosic CA production at an industrial scale is prone to several drawbacks that include lack of suitable feedstocks, expensive chemicals and/or nutrients as well as energy intensive processes that often result in low product yields. Therefore, overcoming the aforementioned bottlenecks is imperative to enhance lignocellulosic CA production. For one, finding an alternative to fresh water within lignocellulosic bioprocesses could substantially reduce the burden placed on this finite resource and costs incurred. Taking this into consideration, untreated wastewater from the Kraft and dairy industries represents a sustainable and costeffective option for the replacement of fresh water and/or chemicals within lignocellulosic bioprocessing systems. The emergence of waste-based technologies for lignocellulosic CA production combined with simultaneous saccharification and fermentation (SSF) processes could reduce the time, expenditure for operation, energy input, risk of contamination and enhance the overall productivity and product yields. For instance, a Kraft paper wastewater pretreatment has been optimized on BP and demonstrated a high glucose yield (0.413 g/g)(Laltha et al., 2022). Similarly, fermentation medium constituents that are required to satisfy the nutritional needs of microbes are costly. More specifically, the microbial growth of Aspergillus sp. typically occurs in the Sabouraud Dextrose Emmon's modification (SDE) media. The media constituents of SDE mainly include peptone and glucose that function as nitrogen and carbon sources, respectively for the microorganism. While the glucose may be supplied by the lignocellulosic substrate, peptone is an expensive constituent estimated at \$151 USD per kg, contributing majorly to the entire bioprocessing cost. Therefore, the investigation of a cost-effective nitrogen source is imperative. Ammonium nitrate has an estimated cost of \$108 USD per kg and may be employed as a cheaper alternative to peptone, translating to a cost reduction of ~28%. Odu et al. (2020), Sarkar and Das (2017) and Sekoai et al. (2018) previously utilized ammonium-based salts (ammonium nitrate or ammonium sulfate) in the range of 0.1 to 8% w/v and demonstrated its adequacy as a nitrogen source in CA production. Other sources of nitrogen such as urea, corn steep liquor (CSL) and yeast extract are also considered significantly cheaper than peptone and have previously been successfully employed for CA production (Sekoai et al., 2018). In addition, several studies have reported on the improvement of CA yields by the addition of desorbents such as methanol and acetone in the range of 1 to 4% (Sarkar and Das, 2017, Odu et al., 2020). Desorbents have been shown to enhance CA production by improving the cell permeability (Sekoai et al., 2018). In addition to these nutritional components, temperature has a major influence on the cellulase-based enzyme and fungal culture in lignocellulosic CA production. Fungi have been shown to thrive under various temperatures ranging from 25 to 45 °C (Aboyeji et al., 2020), while cellulase-based enzymes require temperatures between 45 to 50 °C for optimal functioning. Exploring a suitable, intermediate temperature that can provide optimal conditions for both the enzyme and microbial culture functioning is crucial in lignocellulosic bioprocessing. Therefore, optimizing input parameters that comprise of temperature, nitrogen and desorbent concentration may prove valuable for the improvement of CA production. The response surface methodology (RSM) is a mathematical model utilized for the optimization process and determines the interactive effects of input parameters and their corresponding output (Gaitonde et al., 2017). Apart from bioprocess optimization of key parameters, kinetic modelling provides an in-depth understanding of the metabolic nature of a fermentation process (Germec et al., 2019).

Kinetic models play an essential role in predicting, monitoring, optimizing and simulating the performance of a process under varying process conditions (Pramanik et al., 2019). For instance, the logistic model is used to describe the kinetics of microbial biomass growth to the rate-limiting substrate over a certain time period (Germec et al., 2019). Conversely, the modified Gompertz model improves the understanding of kinetic processes relative to the logistic model (Pramanik et al., 2019). The modified Gompertz model estimates the maximum production rate, maximum concentration and the lag phase (Srimachai et al., 2015). Although SDE media has shown to be effective for the cultivation of Aspergillus sp., this medium consists of expensive constituents as highlighted above and uses fresh water, which is a finite resource globally. Ideally, the application of industrial wastes as a water replacement in SSF bioprocesses could alleviate water security. Interestingly, the dairy industry produces high quantities of wastewater $(1.575 \times 10^{12} \text{ L})$ that may be harnessed as a substitute for fresh water bioprocesses (Slavov, 2017). Dairy wastewater typically undergoes expensive treatment processes to reduce contaminants before it is expelled into the environment. From an industrial perspective, these treatments are considered an economical burden, due to the high costs and energy as well as the long processing times. Adding to this, is the release of greenhouse gas emissions that has severe negative implications on the environment (Gogoi et al., 2021).

Therefore, using DWW to replace fresh water in lignocellulosic bioprocesses also presents opportunities for commercial industries regarding wastewater disposal and treatment costs. Another striking characteristic of DWW is the presence of trace amounts of carbohydrates, soluble organic compounds, protein and nitrogen that could contribute to promoting the growth of the microbial culture (Kaur, 2021). Even though several studies have reported on the production of CA from different lignocellulosic substrates (Odu et al., 2020, Sarkar and Das, 2017, Aboyeji et al., 2020), there has been a scarcity of information on the effects of temperature, nitrogen concentration (ammonium nitrate) and desorbent concentration (acetone) on CA production using BP. Furthermore, the existing knowledge gap on kinetic data of A. brasiliensis growth and CA production with fresh water, wastewater and standard SDE media modified with pretreated banana pseudostem has attracted interest. Therefore, the present study aimed to model and optimize the SSF process using the RSM for CA production from pretreated BP. For the RSM optimization, the effect of acetone concentration, ammonium nitrate concentration and temperature on the CA production was determined. Subsequently, the modified Gompertz and logistic models were employed to ascertain the kinetics of A. brasiliensis CA production and growth. For the kinetic experiments, three bioprocesses consisting of the: (1) optimized SSF process with fresh water (SSF_{optimizedFW}), (2) SSF_{optimizedFW} process conditions while substituting dairy wastewater in place of fresh water (SSF_{DWW}), and (3) Sabouraud Dextrose Emmon's medium adapted by substituting glucose with BP (SSF_{SDEmodified}) were assessed.

2. Materials and methods

2.1. Materials

The banana pseudostem substrate was procured, dried and milled according to Laltha et al. (2022). The chemical reagents utilized were acquired from Merck (South Africa), where suitable. The Cellic CTec2 enzyme (160 FPU/mL) harnessed for the SSF processes was supplied by Novozymes (Novozymes A/S, Denmark). The paper wastewater (pH ~8) together with the safety data sheet containing its chemical composition (Laltha et al., 2022) was provided by a Kraft paper and pulp mill (Mondi, Richards Bay, South Africa). The dairy wastewater employed was obtained from a regional dairy (Fairfield, Howick, South Africa) and its composition has been reported by David et al. (2022).

2.2. Revival and microbial development

The fungal strain employed in this study, *Aspergillus brasiliensis* ATCC16404 was obtained from ATCC® type cultures originating from Microbiologics Incorporated. The Microbiologics KWIK-STIK pack contained a protocol, ampoule of hydrating fluid, qualitative lyophilized microorganism pellet and an inoculating swab. The hydrating fluid was applied to the lyophilized pellet and aseptically swabbed onto SDE (Sabouraud Dextrose Emmon's) agar by using the three-way streak plating technique to facilitate colony isolation. The SDE agar medium formulation comprised of 20 g/L glucose, 20 g/L agar and 10 g/L peptone autoclaved at 121 °C for 15 minutes. The fungal strain was then incubated at 30 °C for 7 days. Preparation of a fungal spore suspension was carried out by the addition of 10 mL of sterilized distilled water that contained non-ionic surfactant (0.1% v/v Tween 80) by covering the fungal growth and using a spatula to scrape the spores into 100 mL of SDE broth (20 g/L glucose and 10 g/L peptone) incubated at 30 °C and 150 rpm for 7 days.

2.3. Short-term and long-term storage of culture

To ensure continuous longevity of the microorganism, the fungal strain was subcultured every 7 days by transferring a piece of inoculated agar containing mature fungal hyphae and spores onto SDE agar, which was thereafter subjected to incubation at 30 °C for 7 days. The previous growth plate was stored at 4 °C in the refrigerator. For long-term storage, *A. brasiliensis* was inoculated in glycerol stocks (10%, 20% and 50% v/v) and agar slants (double-strength medium) and stored at -80 °C and 4 °C, respectively.

2.4. Pretreatment of banana pseudostem

The powdered BP substrate was subjected to pretreatment using a previously optimized microwave-assisted paper wastewater method (Laltha et al., 2022). The BP substrate at a 30% (w/v) solid loading was submerged in 100 mL of paper wastewater and exposed to microwave heating at 800 W for 8 minutes. The pretreated BP biomass was filtered, washed and dried as stipulated by Laltha et al. (2022), prior to being used in the SSF bioprocess. The composition of the optimally pretreated banana pseudostem was previously determined to contain cellulose (17.06%), hemicellulose (13.04%) and lignin (5.9%) (Laltha et al., 2022).

2.5. Simultaneous saccharification and fermentation (SSF) process

2.5.1. Inoculum development

A. brasiliensis was grown on SDE agar as previously described in section 2.2. The grown suspension was diluted to a final spore concentration of 1×10^6 spores/mL. A Neubauer counting chamber (Neubauer, Germany) was used to determine the spore concentration

2.5.2. Sodium citrate buffer preparation

Sodium citrate buffer (0.05 M) was made up by combining 0.1 M citric acid and 0.1 M sodium citrate in distilled water. The pH was adjusted to 4.8 prior to autoclaving at 121 °C for 15 min.

2.5.3. Preliminary screening

Preliminary screening of three desorbents (3% v/v) (acetone, chloroform and propan-2-ol) and four nitrogen sources (2% w/v) (corn steep liquor, ammonium nitrate, yeast extract and peptone) were utilized to determine their individual effects on CA production. The SSF parameter ranges were designated based on a substantial literature review of related lignocellulosic fermentation studies (Sarkar and Das, 2017, Odu et al., 2020). The pretreated BP with a solid loading of 10% w/v, an inoculum size of 10% (v/v), and 10 FPU/g enzyme loading was added to an Erlenmeyer flask and brought up to a final volume of 50 mL using 0.05 M sodium citrate buffer. The desorbents (3% v/v) (acetone, chloroform or propan-2-ol) and nitrogen sources (2% w/v) (corn steep liquor, ammonium nitrate, yeast extract or peptone) were also added accordingly (See supplementary material). The SSF flasks constituted a working volume of 50 mL, incubated at 35 °C and 150 rpm (Ho and Hood, 2014) for a period of 144 h. Samples (1 mL) were aseptically collected every 24 h and centrifuged at 10 000 rpm for 5 min prior to CA quantification using an acid-base titration method (Ayeni et al., 2019).

2.5.4. RSM experimental design of the SSF process

The response surface methodology (RSM) model (Box-Behnken design) (Design Expert 7.0, Stat Ease Inc, USA) was employed for the optimization. The experimental design consisted of seventeen (17) SSF processes. The fermentation input parameters comprised of desorbent (acetone) concentration (1-5% v/v), nitrogen (ammonium nitrate) concentration (0.5-5% w/v) and temperature (30-40 °C), while the output was CA concentration (g/L). The SSF input parameters with its ranges were chosen based on an extensive literature survey (Sarkar and

Das, 2017, Sekoai et al., 2018, Odu et al., 2020) in combination with data curated from the screening of preliminary experiments. The resultant experimental CA concentration data fitted the RSM polynomial equations to assess the combined input parameter effects for the enhancement and generation of CA.

2.5.5. Citric acid production

The CA experiments were performed in 100 mL Erlenmeyer flasks comprising of a 50 mL working volume. A solid loading of 10% (w/v) of paper wastewater pretreated BP was added to each flask. In addition, 10 FPU/g Cellic CTec2 enzyme, 0.05 M sodium citrate buffer, and 10% v/v inoculum containing 1×10^6 spores/mL *A. brasiliensis* were aseptically added. The respective amounts of acetone (1-5% v/v) and ammonium nitrate (0.5-5% w/v) were added to each fermentation flask and incubated at the desired temperature (30-40 °C) as specified in the RSM experimental design (Table 1). The SSF processes were agitated at 150 rpm for a period of 96 h. Using aseptic techniques, the aliquots (1 mL) were sampled from the SSF flasks every 12-h and stored at -20 °C for further analysis.

2.6. Validation of the optimized SSF bioprocess and kinetic modelling

The variable conditions for optimal CA generation were determined and carried out to validate the developed RSM model. The SSF reaction working volume of 50 mL contained: 10% (w/v) pretreated BP, 1% (v/v) acetone, 0.5% (w/v) ammonium nitrate, with an inoculum size of 10% (v/v) and citrate buffer. This process was designated as SSF_{optimizedFW}. In addition, the SSF_{optimizedFW} process was substituted with dairy wastewater in place of fresh water (SSF_{DWW}). Table 1. Citric acid concentration determined for the SSF process on inputs of acetone,

		Input variables		Output
Run	Acetone (v/v)	Ammonium nitrate	Temperature (°C)	Citric acid concentration (g/L)
		(w/v)		
1	3	0.50	30	1.857 ± 0.121
2	1	0.50	35	14.408 ± 0.594
3	5	2.75	30	0.640 ± 0.005
4	3	2.75	35	9.093 ± 0.126
5	3	2.75	35	9.413 ± 0.162
6	3	0.50	40	1.153 ± 0.052
7	5	2.75	40	0.704 ± 0.033
8	1	2.75	30	2.433 ± 0.128
9	3	5.00	40	1.217 ± 0.063
10	5	0.50	35	6.275 ± 0.178
11	3	2.75	35	8.324 ± 0.059
12	3	2.75	35	5.571 ± 0.115
13	3	2.75	35	9.413 ± 0.231
14	5	5.00	35	6.852 ± 0.220
15	3	5.00	30	2.305 ± 0.098
16	1	5.00	35	12.679 ± 0.106
17	1	2.75	40	0.832 ± 0.143

ammonium nitrate concentration and temperature

The SSF_{Dww} process contained: 10% (w/v) pretreated BP, 1% (v/v) acetone, 0.5% (w/v) ammonium nitrate, with an inoculum size of 10% (v/v) and citrate buffer formulated using DWW. The DWW formulated citrate buffer was the same as described in section 2.5.2., with the substitution of DWW instead of fresh water. For comparison, the standard SDE medium was modified by substituting glucose with BP (SSF_{SDEmodified}) and consisted of 10% (w/v) pretreated BP, 1% (w/v) peptone, with an inoculum size of 10% (v/v) and citrate buffer. Each SSF process was therefore characterized as follows: (1) optimized SSF process with fresh water in place of fresh water (SSF_{SDEmodified}). All SSF experiments were incubated at 150 rpm, 35 °C for 96 h. The

experimental controls were performed parallel to the SSF_{optimizedFw}, SSF_{Dww} and SSF_{SDEmodified} and constituted the same conditions with the exception of inoculum, which was omitted. The aforementioned uninoculated controls were used as a reference to determine the initial sugar concentration at a certain time period corresponding to the test SSF experiments (SSF_{optimizedFw}, SSF_{Dww} and SSF_{SDEmodified}). All of the experiments were carried out in duplicate. Aseptically, sample aliquots (1 mL) were removed from the experimental reaction flask and its equivalent control every 12 h for 96 h. The samples were thereafter analyzed for CA, glucose concentrations and biomass as explained below.

2.7. Analytical methods

2.7.1. Citric acid concentration

SSF processes are dynamic, biological systems and by-products generated during the fermentation may sometimes cause interference when using methods encompassing titration. In the present study, CA concentration was primarily analyzed using the acid-base titration method employed by Ayeni et al. (2019) and thereafter verified by the Megazyme citric acid assay kit, product code (K-CITR) (©Megazyme, Wicklow, Ireland) (See supplementary material). In the acid-base titration analysis, the SSF samples collected from each time point were first centrifuged for 5 min at 10 000 rpm. Subsequently, 100 μ L of the supernatant fluid was transferred to a flask containing phenolphthalein indicator (2 to 3 drops). A glass burette was rinsed with distilled water and then filled with the 0.01 M NaOH titrant. The burette was opened to allow few drops of the NaOH (base) titrant to react with the sample (acid) and shaken continuously to observe a change in colour (from colourless to light pink). Excess titrant was washed of the sides of the flask with distilled water. The endpoint of the reaction was determined when the colour change had occurred. Following this, the volume of NaOH utilized was read directly from the burette and substituted into the formula described by Ayeni et al.

(2019) to ascertain the amount of CA present. The corresponding number of moles of alkali NaOH required to neutralize CA, C₃H₅O(COOH)₃, was then determined based on the volume and concentration of NaOH solution utilized during the acid-base titration (See supplementary material) (Ayeni et al., 2019). For verification of the acid-base titration method described above, the CA concentration was measured by means of the Megazyme citric acid assay kit, product code (K-CITR) (©Megazyme, Wicklow, Ireland) corresponding to the specific instructions in the manufacturers protocol. The Megazyme citric acid assay kit is an enzymebased analytical method that is specific. The CA content was analyzed spectrophotometrically using a SpectraMax ABS microplate reader (Molecular Devices, California, USA).

2.7.2. Glucose concentration

The glucose concentration of the control (uninoculated) and experimental kinetic SSF experiments were analyzed using the Megazyme glucose kit (K-GLUC) (©Megazyme, Wicklow, Ireland) and protocols detailed by the manufacturer. To determine the glucose utilization (%), the final glucose concentration (inoculated) was subtracted from the initial glucose concentration as per Eq. (1).

Glucose utilization (%) =
$$\frac{\text{Initial glucose concentration } (g/L) - \text{Final gluose concentration } (g/L)}{\text{Initial glucose concentration } (g/L)} \times 100$$

(1)

Where the final glucose concentration (g/L) is derived from the experimental SSF flask while the initial glucose concentration (g/L) is attained from the uninoculated control flask at the exact moment in time.

2.7.3. Biomass concentration

The biomass concentration of *A. brasiliensis* was analyzed using the dry cell weight (g/L) as a function of the total spore count (spores/mL). The fungal isolate was grown for 7 days in SDE broth to reach exponential phase (10^6 to 10^7 spores/mL) and diluted accordingly (Vrabl et al., 2019). A 15 mL volume per diluted sample (1, 1/2 ,1/4, 1/8, 1/16 and 1/32) was centrifuged for 5 min at 10 000 rpm. All dilutions were performed in duplicate. Succeeding centrifugation involved discarding of the supernatant and drying of the biomass pellet at 60 °C until the mass was constant. The standard curve was developed by plotting the total spore counts (spores/mL) to the corresponding dry cell weights (biomass concentration in g/L) for each dilution. The spore counts (spores/mL) attained for the kinetic experiments were computed in the equation and extrapolated to ascertain the biomass concentration (g/L).

2.8. Kinetic modelling

2.8.1. The logistic model for A. brasiliensis growth

The logistic model's differential formula, stated in Eq. (2), was integrated to obtain Eq (3). Equation (3) considers the exponential and stationary growth phases. The logistic model Eq. (3) relates biomass (X) to the initial cell concentration (X₀), maximum cell concentration (X_{max}) and maximum specific growth rate (μ_{max}) during the exponential and stationary growth stages of *A. brasiliensis*.

$$\frac{dX}{dt} = \mu_{max} \left(1 - \frac{X}{X_{max}} \right) X \tag{2}$$

$$X = \frac{X_0 \exp(\mu_{max}t)}{1 - \left[\left(\frac{X_0}{X_{max}}\right)(1 - \exp(\mu_{max}t))\right]}$$
(3)
2.8.2. The modified Gompertz model for citric acid production

Using the least squares approach, the CA data from the SSF processes (SSF_{optimizedFw}, SSF_{Dww} and SSF_{SDEmodified}) were fitted to the modified Gompertz model(CurveExpert V1.5.5, MyBiosource, Inc., USA). As indicated in Eq. (4), the modified Gompertz model depicts the link between the CA concentration (P), maximum CA production rate ($r_{p,m}$), potential maximum CA concentration (P_m) and lag time (t_L) from the start of fermentation to the exponential CA generation.

$$P = P_m \cdot exp \left\{ -exp \left[\frac{r_{p,m} \cdot exp(1)}{P_m} \right] \cdot (t_L - t) + 1 \right\}$$

$$\tag{4}$$

3. Results and discussion

3.1. Preliminary screening

Preliminary screening data revealed that the desorbent that gave the highest concentration of 10.886 g/L CA was acetone, followed by propan-2-ol (10.245 g/L) and chloroform (7.044 g/L) (See supplementary material). For the nitrogen source, ammonium nitrate resulted in the highest CA concentration (12.807 g/L), while peptone, corn steep liquor and yeast extract resulted in CA concentrations of 9.605 g/L, 8.324 g/L and 7.684 g/L, respectively. Therefore, acetone and ammonium nitrate were selected as the desorbent and nitrogen source, respectively, for the SSF optimization.

3.2. Development of the RSM model

To assess the fitness of the developed RSM model, Analysis of Variance (ANOVA) was used (See supplementary material). P-values < 0.05 indicate parameter and model significance, whereas high F-values demonstrate that the response trends could be explained using the regression equations (Chaganti et al., 2012). The model had a low P-value of 0.0063, which was < 0.05, indicating model significance. In addition, a high F-value (7.89) was attained for

the SSF model, thus illustrating statistical significance of the model. Of the three key variables that were investigated, acetone concentration (p = 0.0282) demonstrated the most significant effect on the CA concentration compared to temperature (p = 0.5812) and ammonium nitrate concentration (p = 0.9146). Acetone serves as the desorbent and enhances cell permeability of the microorganism for nutrient sequestration and may also repress the enzyme 2-oxoglutarate dehydrogenase, thus increasing CA concentration (Sekoai et al., 2018, Maddox et al., 1986). This desorbent (acetone) is also a ketone that plays a vital role as an alternative metabolic fuel source for eukaryotes (Puchalska and Crawford, 2017). On the other hand, temperature is a key parameter for the optimum functioning of cellulolytic enzymes for the effective saccharification of cellulose to glucose that will be consumed by the microbe. Additionally, temperature also plays a significant role for regulating the internal temperature, growth, secondary metabolite production and activity of fungi such as A. brasiliensis (Tudzynski, 2014). The presence of nitrogen such as ammonium nitrate in the fermentation medium is essential for fungal growth since it facilitates cell adaptation to different niches and survival of adverse environmental conditions such as nutrient limitation (Tudzynski, 2014). The coefficient of determination (R^2) value is a statistic that measures the proportion of variation in the output values that can be attributed to the input variables. In general, R^2 values between 0.7 and 1 elicits model significance (Chaganti et al., 2012). The developed SSF model displayed an R² value of 0.91, indicating that 91% of the variation in the output variable (CA concentration) could be attributed to the input variables. The polynomial model Eq. (5) was generated for the CA concentration and describes the input parameters in relation to the response variable.

CA concentration (g/L) = 8.36 - 1.99A - 0.080B - 0.42C + 0.58AB + 0.42AC - 0.096BC + 0.61A² + 1.09B² - 7.82C² (5)

where, A, B and C correspond to acetone concentration (v/v), ammonium nitrate concentration (w/v) and temperature (°C), respectively.

3.3. Effects of the input parameters on the citric acid concentration

The CA concentrations with corresponding input parameters are portrayed in Table 1. The SSF model displayed a maximum CA concentration of 14.408 g/L for run 2 (1% acetone, 0.5% ammonium nitrate and 35 °C). The median fermentation conditions (3% acetone, 2.75% ammonium nitrate incubated at 35 °C) exhibited similar CA concentration values within the range of 5.571 g/L to 9.413 g/L (run 4, run 5, run 11 run 12 and run 13). The lowest CA concentration recorded (0.640 g/L) was achieved with 5% acetone, 2.75% ammonium nitrate and incubated 30 °C for run 3. The RSM graphs that show the collaborative effects of the different input parameters and the resultant CA concentration is illustrated in Fig. 1A-C. The interactive effect of ammonium nitrate concentration and acetone concentration on the output CA concentration, while temperature was kept at its center point (35 °C) is presented in Fig. 1A. Maximum CA concentration (12.63 g/L) was observed at a minimum acetone (1%) and ammonium nitrate concentration (0.5 %). Further increase in the acetone concentration from 1 to 5%, while ammonium nitrate concentration was maintained at 0.5% led to a decline in the CA concentration from 12.63 to 7.55 g/L. Simultaneous increments in the ammonium nitrate concentration (5%) and acetone concentration (5%) resulted in a decreased CA concentration of 10 g/L. The high CA concentration observed at a low ammonium nitrate concentration may be as a result of reduced vegetative growth of A. brasiliensis. On the other hand, the lower CA observed at higher nitrogen concentrations may be due to the shift in the cell metabolism towards fungal growth instead of product formation (Show et al., 2015). Aspergilli can utilize a variety of nitrogen sources such as ammonium, peptone and nitrate.

The fungus converts nitrate to ammonium, which is then incorporated into the amino acids glutamate and glutamine, that act as nitrogen donors within the TCA cycle (Krappmann and

Braus, 2005). Furthermore, during CA fermentation, nitrogen limitation is a key requirement to prevent the accumulation of oxalic acid, since this compound has been known to lead to lower CA production. While desorbents such as acetone have shown to enhance fungal growth and CA production, elevated concentrations can be toxic to the cells and may inhibit the fungal cellular metabolic functioning, thus resulting in a lower CA concentration. A similar result was obtained in an earlier study by Roukas and Harvey (1988) when methanol was used as a desorbent. The authors achieved a maximum CA concentration of 4.83 g/L from beet molasses at a methanol concentration of 2%, while concentrations beyond this value led to a decrease in CA production and concluded that methanol's inductive effect was caused by its inhibitory effect on metal ions. Similarly, the study by Sarkar and Das (2017) varied ammonium nitrate (0.1 to 0.5%) and desorbent concentration (0 to 5%) on pineapple waste and obtained an optimal CA concentration of 44.1 g/L by using 15% sucrose, 0.25% ammonium nitrate, 4% methanol and 7 days of incubation at 30 °C.

The impact of temperature and acetone concentration on the CA concentration while the ammonium nitrate concentration was maintained at its median value (2.75%) is represented in Fig. 1B. An increase in temperature (30 to 35 °C) coupled with the lowest acetone concentration (1%) led to an increase in the CA concentration from 4.08 to 10.94 g/L. Further increments in the acetone concentration and temperature > 1% and > 35 °C, respectively resulted in a steep decrease in the CA concentration from 10.94 to 2.1 g/L. Temperature has a dual effect on SSF processes due to its influence on both the enzymatic and microbial metabolic processes. With regards to enzymatic saccharification, temperature plays a key role on the enzymatic activity and saccharification efficiency for sugar release (EI-Imam and Du, 2014). Temperature within the range of 45 and 50 °C has been shown to facilitate effective cellulase enzymatic saccharification, however the combined SSF system necessitates a median temperature that stimulates enzyme and microbial activity. Nevertheless, high-performing

cellulase cocktail enzymes such as Cellic CTec2 has been specifically developed to withstand adverse environmental conditions and sustain its activity (Champreda et al., 2019). On the other hand, the temperature parameter also has a profound effect on the microbial growth of fungi. Aspergillus sp. are mesophilic microorganisms that exhibit wide growth ranges between 25 to 45 °C (Aboyeji et al., 2020). Hence, to ensure optimal enzyme function and microbial growth, the temperature employed in the SSF process ranged between 30 and 40 °C. In Fig 1B, the highest CA concentration (10.94 g/L) was observed at a temperature of 35 °C. Likewise, Zhang et al. (2018) examined the influence of temperature on the growth of bacteria and fungi and determined that the growth activity increased at 35 °C. A high concentration of CA was observed when a low concentration of desorbent was employed (1%). In the same vein, Moyer (1953) studied the effects of ammonium chloride and methanol concentration for the synthesis of CA from corn meal and concluded that a low desorbent concentration (2%) enhances the citrate synthase activity while decreasing the aconitase activity during CA fermentation. Citrate synthase is a central enzyme involved in sugar oxidation and is also responsible for catalyzing the TCA cycle, whereas aconitase catalyzes the isomerization of citrate to isocitrate within the TCA cycle (Papagianni, 2007, Show et al., 2015). While desorbents improve the activity of enzymes involved in the TCA cycle, they are inhibitory above the threshold levels (> 5%) and could potentially lead to the degradation of these key enzymes and inhibit the growth of microorganisms by drastically altering the pH of the medium. Similarly, Maddox et al. (1986) utilized a low desorbent concentration (1%) and observed a maximum CA concentration of 25 g/L using galactose as a substrate. Furthermore, in the absence of methanol, very little CA was produced (less than 1 g/L). Kareem et al. (2010) evaluated the effect of sucrose, ammonium nitrate and methanol on CA production using fermentation medium consisting of pineapple waste and achieved optimum CA (60.61 g/L) under 15% sucrose, 0.25% ammonium nitrate and 2% methanol when fermented for 5 days at 30 °C. The aforementioned author also noted

that further increases in the methanol concentration (> 3%) led to a diminished fungal growth and lower CA production (Kareem et al., 2010). The pairwise interaction of temperature and ammonium nitrate concentration on the CA concentration when the acetone concentration was retained at its midpoint value (3%) is demonstrated in Fig. 1C. A simultaneous increase in temperature (30 to 35 °C) and ammonium nitrate concentration (0.5 to 5%) resulted in a minor increase in CA concentration from 8.65 to 9.33 g/L. Similar to the interactive effect of temperature and acetone concentration as shown in Fig. 1B, it is interesting to note that increases in the temperature parameter above 35 °C, led to a drastic decline in CA concentration from the optimal value (9.65 g/L) to approximately 2 g/L. Several factors influence fungal growth; however, temperature has shown to have a drastic effect that could potentially dictate the output and quality of the desired end-product. Temperature also influences the growth of hyphae and the activity of extracellular enzymes associated with the TCA cycle. At high temperatures (> 35 °C), hyphal deterioration may occur and result in a critical reduction of CA (Zhang et al., 2018). Nitrogen concentration has been found to have a strong influence on both cellular metabolism and CA production. Owing to the necessity of employing low-cost substrates in order to reduce CA manufacturing costs and make it more ecologically sustainable, glucose derived from pretreated banana pseudostem makes it a valuable carbon source. By increasing the initial glucose concentration, the specific rate of CA production rises (Papagianni et al., 1999), whereas concentrated ammonium nitrate inhibits the growth and metabolism of fungi (Veverka et al., 2007). Veverka et al. (2007) investigated the sensitivity of fungal growth to two different nitrogen sources (urea and ammonium nitrate). It was concluded that ammonium nitrate (0.06 M) had the most notable effect on the growth of A. niger with an increase of 1.5-fold greater in comparison to urea (0.19 M). The type of nitrogen source utilized is important due to specific properties that affect the fungal growth and synthesis of CA (Show et al., 2015). For example, the absorption of ammonium by germinating

spores release protons, which lowers the pH of the fermentation medium and results in an improvement in the CA production.

The low pH encountered during the production phase (~2) lowers the risk for contamination by unwanted microorganisms and impedes the development of undesirable organic acids such as oxalic and gluconic acids, thereby impacting on downstream processing by making product recovery easier.

3.4. Validation of the RSM model

Validation of the developed SSF model was carried out by utilizing the optimum process conditions (1% acetone, 0.5% ammonium nitrate and 35 °C). At these setpoints, the model predicted an optimal CA concentration of 12.712 g/L. Experimental validation of these parameters displayed a slightly higher CA concentration of 13.703 g/L, translating to a difference of 7.23% when compared to the models predicted value and was considered negligible. This indicates high accuracy and precision of the developed RSM model with a strong reliability for prediction. The validation parameters established by the RSM model were applied to two other fermentation processes consisting of dairy wastewater in place of fresh water (SSF_{DWW}) and SDE medium adapted by substituting glucose with BP (SSF_{SDEmodified}). A comparative assessment between these three SSF processes are detailed under Sections 3.5 and 3.6.

3.5. Comparison of the citric acid production to previous reports

The results obtained from the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} bioprocesses were compared to earlier reports on CA production using dairy waste and lignocellulosic biomass (See supplementary material). The SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes achieved CA concentrations of 12.967, 13.095 and 13.991 g/L respectively. The present study revealed higher CA concentrations compared to Aboyeji et al. (2020) (4.36 g/L), Ghanbartabar et al. (2016) (4.45 g/L), and Campanhol et al. (2019) (0.73 g/L) when *A. niger* was utilized for the fermentation of sweet potato peel waste, cheese whey biomass and sugarcane bagasse, respectively. Conversely, Odu et al. (2020), Sarkar and Das (2017) and Ozdal et al. (2019) investigated banana peel, pineapple waste and sugar beet molasse, respectively, for the generation of CA and observed higher CA concentrations when compared to the present study.



Fig. 1. Three-dimensional response surface graphs showing the interactive effects of the (A) Ammonium nitrate concentration (%) and acetone concentration (%), (B) Temperature (°C)

and acetone concentration (%), and (C) Temperature (°C) and ammonium nitrate concentration (%) on the citric acid concentration.

The variations in this study to previously reported literature may be due to different conditions applied that include: the microbial species (inoculum size, strain), physical parameters (temperature, agitation, pH), the concentration and type of enzyme, desorbent type and concentration, nitrogen source and concentration, substrate and solid loading, in addition to the supplementation of chemical additives. Despite the high concentrations of CA derived from previous reports, the exploitation of fresh water consumption, expensive substrates, chemical supplementation and/or media components render these fermentation processes non-viable and costly for potential industrial scale implementation.

3.6. A. brasiliensis growth kinetics for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes The kinetics of microbial growth and CA production for the SSF bioprocesses (SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified}) were determined by utilizing the logistic and modified Gompertz models, respectively. The lag phase, often known as the microorganism's adaption phase occurs when the microbe begins to acclimatize to changes in the new medium environment and for sufficient glucose to be produced by saccharifying enzymes to support the growth of the microorganism (Asaduzzaman, 2007). The lag phase was observed during the first 12 h of fermentation. In addition, the microorganism is metabolically active and synthesizes the required enzymes and metabolites for exponential cell growth (Asaduzzaman, 2007). During the lag phase, the activation of signaling pathways and transcriptional changes occur, leading to lipopolysaccharide biosynthesis and respiration that is required for multiplication and differentiation of the cells (Hamill et al., 2020). A minor lag phase for conidia germination might last several hours, however temperature and stressors can extend this timeframe to days (Hamill et al., 2020). The lag phase of A. brasiliensis has been shown to be within the range of 12 to 24 h in a previous study, where it was cultivated in media containing glucose (Ho and Hood, 2014). Similar lag phase times (5 to 10 h) has been associated with the commonly used and closely related A. niger for CA production (Baei et al., 2008). The exponential phase or logarithmic phase transpires when the microbial culture undergoes asexual reproduction at a constant rate through simple cell division, in which one cell separates forming two identical daughter cells after undergoing nuclear division (Ellena et al., 2020, Asaduzzaman, 2007). The exponential phase was recorded between 12 and 84 h for all three SSF processes. In general, the exponential phase is initiated by a particular cell concentration that increases metabolism and initiation of DNA replication (Asaduzzaman, 2007). The enzymatic conversion of cellulose to glucose released by cellulase action occurs almost instantaneously during SSF processes and is utilized by the microbial culture (in this case A. brasiliensis) within the TCA cycle for the generation of CA. A high glucose utilization was observed for the SSF_{optimizedFW} (50%), SSF_{DWW} (44%) and SSF_{SDEmodified} (57%) processes (Fig. 2B) and this corresponded to an increase in the fungal growth (1 - 2.5 g/L) (Fig. 2A). The biomass concentration began to stabilize after 84 h for all three SSF bioprocesses. The decline in cell growth above 84 h can be attributed to the decline in Cellic CTec2 enzyme activity that results in a lower saccharification efficiency and thus lower glucose release for the fermentative microorganism. Furthermore, the functioning of enzymes occurs best within a certain designated temperature (45-50 °C) and pH (4.5-5.5) range (Fenila and Shastri, 2016). Therefore, sub-optimal conditions for enzyme functioning under a low temperature (35 °C) and pH (~3.5) required for the initial production of CA, causes a ripple effect on the SSF process, since the enzyme loses its activity and ability to bind to an active site on the substrate. Aspergillus brasiliensis growth data obtained from the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes were inserted into the logistic model and displayed high R² values of 0.997, 0.998 and 0.997, respectively. All

three bioprocesses demonstrated the same maximum specific growth rate (μ_{max}) of 0.05 h⁻¹. The maximum specific growth rate is an important kinetic parameter due to its impact on product yield, formation and productivity (Srimachai et al., 2015, Pramanik et al., 2019). The low μ_{max} values observed may be accounted for by the depletion of nutrients such as glucose or nitrogen for cellular maintenance within the fermentation media. The concentration and availability of nitrogen influences the activity and growth efficiencies of fungi. When nitrogen is scarce within the media, fungi divert their metabolism towards generating more energy for the production of extra-cellular enzymes such as hydrolase and oxidase to sequestrate nitrogen from the surrounding environment, resulting in lower growth efficiencies (Tudzynski, 2014). The maximum (X_{max}) and initial (X₀) biomass cell concentrations are essential indicators for the synthesis of biomass and formation of products (Baei et al., 2008). Kinetic data obtained for the SSF_{optimizedFw}, SSF_{Dww} and SSF_{SDEmodified} processes gave X₀ concentration values of 0.19 g/L, 0.15 g/L and 0.21 g/L, respectively. Similar to the X₀, the highest X_{max} was realized for the SSFsDEmodified (2.91 g/L) process followed by the SSFoptimizedFw (2.79 g/L) and SSFDww (2.74 g/L) processes. Extracellular enzymes produced during fermentation are important for the dissolution and absorption of external substances that contribute to the growth of Aspergillus. The enzymes secreted during growth are related to the lignocellulosic substrate, while temperature fluctuations influence extracellular enzyme activity (Zhang et al., 2018). The marginally lower X_{max} values reported for the SSF_{optimizedFW} and SSF_{DWW} processes in comparison to the SSF_{SDEmodified} may be attributed to variations in the media composition (nutrients) or metabolic dynamics. The citric acid cycle begins when acetyl CoA transfers its acetyl group to oxaloacetate to create citrate. In the citric acid cycle, carbon dioxide is released in tandem with the reduction of NAD⁺ to NADH. However, increased levels of carbon dioxide can be detrimental to the growth of fungal biomass and reduce the concentration of citrate produced (Show et al., 2015). Although peptone and ammonium nitrate are sources of nitrogen,

they vary in the sense that the former is organic, while the latter is inorganic. Furthermore, peptone contains growth factors. A preference for the organic source may indicate a requirement for an amino acid or vitamin present in the organic source. Nevertheless, the addition of amino acids and vitamins to a nutrient medium containing inorganic nitrogen is deemed expensive (Clarke, 2013). Baei et al. (2008) evaluated the potential of *A. niger* for the production of CA from apple pomace under process conditions of 0.1 g/L potassium ferrocyanide, 1.5% (w/v) calcium triphosphate, 2.5 g/L ammonium sulfate and 3 mL methanol, and recorded a lag phase of 5 to 10 h in addition to a μ_{max} , X₀ and X_{max} of 0.015 h⁻¹, 0.04 g/L and 22 g/L, respectively. Discrepancies detected in growth kinetic data between the present study and Baei et al. (2008) may be credited to several aspects such as substrate type, microbial strain, nutrient availability, heat and mass transfer, aeration, agitation, pH, solid and enzyme loading.

3.7. Citric acid kinetics for the SSF optimizedFW, SSF DWW and SSF SDEmodified processes

CA production during the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes, were elucidated over time (Fig. 2C). Similar to an increase in biomass concentration, the exponential CA production occurred from 12 h to 84 h, with a plateau observed from 84 h onwards, indicating the initiation of the stationary phase. A lack of essential resources, such as carbohydrates generally initiates the stationary phase of microbial development. In all three bioprocesses, it was observed that CA increased from the exponential phase and continued into the stationary phase. Citric acid increased from approximately 0.5 to 13 g/L for the SSF_{optimizedFW} process and from 0.6 to 13.1 g/L for the SSF_{DWW} system, while the SSF_{SDEmodified} bioprocess observed the highest CA increase from 0.58 to 14 g/L within the 0 h to 96 h time period (Fig. 2C). Citric acid data obtained for the three SSF processes coincided with the exponential phase of the *A*. *brasiliensis* cell growth (Fig. 2A) and a high glucose utilization (9.77 to 56.79%) (Fig. 2B).

During the exponential phase that occurs within 12 h to 84 h, sufficient time is available for the enzyme to saccharify the BP substrate to release enough glucose to sustain the metabolism of the fungal strain, leading to primary metabolite production of CA. The cellulase-based Cellic CTec2 utilized for enzymatic saccharification is a cocktail of enzymes that act collaboratively on cellulose thus converting it to glucose. Cellic CTec2 exhibits high saccharification efficiencies (β -glucosidase activity) and is less vulnerable to glucose inhibition (Champreda et al., 2019). Furthermore, the buildup of glucose in the fermentation medium enhances the activity of glucose oxidase resulting in improved CA concentrations. The decline in A. brasiliensis growth observed after 84 h, overlaps with a lower glucose utilization for the SSF_{optimizedFW} (36.2 to 18.8%), SSF_{DWW} (24.1 to 19.3%) and SSF_{SDEmodified} (44.5 to 18.9%) processes (Fig. 2B). The experimental CA data obtained for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} experiments fitted the modified Gompertz model exhibiting high R² values of 0.991, 0.993, and 0.992, respectively. The shortest lag time (tL) was obtained for the SSF_{SDEmodified} (5.11 h) process followed by the SSF_{DWW} (9.80 h) and SSF_{optimizedFW} (16.47 h) systems. The SSF_{SDEmodified} process consisted of peptone as the nitrogen source, thus the shorter t_L achieved can be attributed to the incorporation of amino acids and vitamins compared to the SSF_{optimizedFW} and SSF_{DWW} systems that utilized ammonium nitrate. Ammonium nitrate has been shown to be approximately 25% cheaper than peptone, however the latter incorporates growth factors that is fundamental for rapid growth of Aspergillus sp. and CA metabolism (Clarke, 2013). Furthermore, nitrogen contributes to a change in pH, therefore allowing the microorganism to grow and adapt to the fermentation medium (Tudzynski, 2014). Nitrogen is exploited by Aspergillus for the generation of amino acids such as glutamine and glutamate that act as nitrogen donors within the TCA cycle (Krappmann and Braus, 2005). On the other hand, the addition of the acetone desorbent in the SSF_{optimizedFW}, and SSF_{DWW} processes, may increase cell permeability to citrate thus increasing CA production via repression of the 2oxoglutarate dehydrogenase enzyme compared to the SSFsDEmodified system (Maddox et al., 1986). The availability of initial carbohydrates (sugar polymers) in solution may also enhance the cellulase enzyme activity at the onset of fermentation for the accumulation of monomeric sugars (glucose). The glucose utilization in the first 12 h of fermentation for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes corresponded to 28.3%, 13.7% and 30%, respectively. Therefore, the lag phase during CA fermentation is dependent on nutrient and sugar availability. Interestingly, the lag times observed in this study were shorter when compared to data recorded in the previous reports by Dienye et al. (2018) and Odu et al. (2020) that observed lag times of 24 h for CA production. Maximum CA production rates (rp.m) were reported for the SSF_{optimizedFw} and SSF_{Dww} (0.17 g/L/h) processes, followed by a slightly lower r_{p.m} for the SSF_{SDEmodified} (0.16 g/L/h) process. These r_{p.m} values indicate that 0.17 g/L and 0.16 g/L of CA can be generated by A. brasiliensis per hour. The rp.m observed for the present SSF processes may be linked to growth factors such as nutrients or trace elements. Trace elements (manganese, zinc and iron) have significant effects on the production of CA. Manganese is important for cell functioning, sporulation and for the generation of low-molecular weight secondary metabolites that provide protection from predation, environmental stress, communication, competition for resources and toxicity against other microorganisms (Show et al., 2015), whereas zinc and iron are required for growth and maintenance of the cell. Show et al. (2015) revealed that manganese deficiency negatively influences the anabolism of A. niger, resulting in a high intracellular ammonium concentration. Conversely, when sufficient nutrients are available, Aspergillus sp. germinate by switching from isotropic to polarized growth, which impacts their stress resistance and regulatory pathways involved in germination (Baltussen et al., 2020). Sugar molecules that trigger fungal germination include glucose, mannose and xylose. A carbon source coupled with either inorganic phosphate, magnesium sulphate or inorganic nitrogen are the minimum requirements for the germination of A. niger (Ijadpanahsaravi et al., 2021). A striking characteristic of industrial wastes such as DWW, is the presence of key nutrients such as Mn (2.56 μ g/L), Mg (< 0.63 mg/L), Fe (91 μ g/L), Zn (21 µg/L), P (6.96 mg/L) and N (36 mg/L) (David et al., 2022), that may have positively influenced the maximum CA production rate within the SSF_{DWW} process. Another key kinetic parameter is the maximum potential CA concentration (Pm) that provides knowledge on the maximum CA that can be produced using these SSF strategies. The SSF_{optimizedFW} process displayed the highest maximum potential CA concentration (Pm) (29.31) g/L, whereas the SSF_{DWW} and SSF_{SDEmodified} processes exhibited lower P_m values of 18.87 and 21.50 g/L, respectively. High glucose utilization and CA concentrations corresponded with the Xmax data obtained for the SSF_{optimizedFW} and SSF_{SDEmodified} processes. The lowest P_m value (18.87 g/L) attained from the SSF_{DWW} process may be linked to the lowest X₀ concentration of 0.15 g/L. During CA fermentation, Aspergillus obtain their energy through adenosine triphosphate (ATP) via the TCA (the CA) cycle. The addition of certain rate limiting nutrients such as nitrogen and phosphorus from DWW or a rise in pH may have a detrimental effect on the SSF process by inhibiting the activity of the proton pump in the plasma membrane (H⁺ATPase), thereby reducing the amount of ATP produced. Also, at a high pH, a change in the rate of glucose uptake occurs, thereby resulting in a decreased biomass concentration and organic acid production (Vrabl et al., 2019). Even with the drawbacks of the developed SSF systems in the present study, these findings may contribute to a more economical and robust operation within lignocellulosic bioprocesses. This is mainly due to the incorporation of DWW as a substitute for fresh water usage, exploitation of banana pseudostem as an alternative to glucose, utilization of ammonium nitrate as a cost-efficient alternative to conventional peptone and by employing acetone with several beneficial properties during SSF induced CA production. The integration of these key elements could facilitate the biotransformation of waste to valuable bioproducts. A synergy between the academic, industrial and commercial sectors to develop large scale waste-based processes could potentially alleviate the negative environmental effects and economical impacts, which is in line with global sustainable development goals to manage the water, energy and food nexus, and may bring forth the realization of a "waste-to-wealth" circular bioeconomy.



Fig. 2. *Aspergillus brasiliensis* ATCC16404 (A) biomass concentration, (B) glucose utilization and (C) CA concentration, for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes.

4. Conclusion

This research presents an innovative optimization of the simultaneous saccharification and fermentation (SSF) bioprocess for citric acid (CA) production. Optimization gave a maximum CA concentration of 14.408 g/L. Afterwards, the kinetics of cell growth and CA production were determined. The SSF_{SDEmodified} system resulted in the highest maximum potential CA concentration (P_m) (13.991 g/L) contrasted to the SSF_{DWW} (P_m= 13.095 g/L) and SSF_{optimizedFW} (P_m= 12.967 g/L) bioprocesses. The findings from this research accelerates the development of waste-based bioprocesses, that could reduce the economical and environmental burden incurred by industries, while it is aligned with the water, energy and food nexus towards a circular bioeconomy.

Acknowledgements

This work is established on research funded partially by the National Research Foundation (NRF) of South Africa (Grant number 122341). Opinions conveyed and conclusions derived, are of the authors and not ascribed to the NRF.

Appendix A. Supplementary data

Supplementary data associated with this article can be found, in the online version.

Credit authorship contribution statement

Milesh Laltha: Conceptualization, Methodology, Software, Validation, Formal analysis, Investigation, Data curation, Writing - original draft, Writing - review & editing, Visualization, Project administration, Funding acquisition. **Y. Sewsynker-Sukai:** Conceptualization, Methodology, Software, Validation, Writing – review & editing, Project administration, Resources, Supervision. **E.B. Gueguim Kana:** Methodology, Software, Validation, Writing – review & editing, Resources, Supervision.

References

- Aboyeji, O., Oloke, J., Arinkoola, A., Oke, M., Ishola, M.J. 2020. Optimization of media components and fermentation conditions for CA production from sweet potato peel starch hydrolysate by Aspergillus niger. Scientific African, 10, e00554.
- Asaduzzaman, M. 2007. Standardization of yeast growth curves from several curves with different initial sizes. Department of Mathematical Sciences. Chalmers University of Technology and Goteborg University. Sweden.
- Ayeni, A., Daramola, M.O., Taiwo, O., Olanrewaju, O.I., Oyekunle, D.T., Sekoai, P.T., Elehinafe, F.B. 2019. Production of CA from the Fermentation of Pineapple Waste by. The Open Chemical Engineering Journal, 13.
- 4. Baei, M.S., Mahmoudi, M., Yunesi, H. 2008. A kinetic model for CA production from apple pomac by Aspergillus niger. African Journal of Biotechnology, 7.
- Baltussen, T.J., Zoll, J., Verweij, P.E., Melchers, W.J. 2020. Molecular mechanisms of conidial germination in Aspergillus spp. Microbiology and Molecular Biology Reviews, 84, e00049-19.
- 6. Behera, B.C., Mishra, R., Mohapatra, S. 2021. Microbial CA: Production, properties, application, and future perspectives. Food Frontiers, 2, 62-76.

- Campanhol, B.S., Silveira, G.C., Castro, M.C., Ceccato-Antonini, S.R., Bastos, R.G. 2019. Effect of the nutrient solution in the microbial production of CA from sugarcane bagasse and vinasse. Biocatalysis and Agricultural Biotechnology, 19, 101147.
- 8. Chaganti, S.R., Kim, D.-H., Lalman, J.A., Shewa, W.A. 2012. Statistical optimization of factors affecting biohydrogen production from xylose fermentation using inhibited mixed anaerobic cultures. International journal of hydrogen energy, 37, 11710-11718.
- Champreda, V., Mhuantong, W., Lekakarn, H., Bunterngsook, B., Kanokratana, P., Zhao, X.-Q., Zhang, F., Inoue, H., Fujii, T., Eurwilaichitr, L. 2019. Designing cellulolytic enzyme systems for biorefinery: From nature to application. Journal of bioscience and bioengineering, 128(6), 637-654.
- 10. Clarke, K.G. 2013. An introductory engineering and life science approach, Bioprocess Engineering, 1, 1-266.
- 11. David, A., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2022. Co-valorization of corn cobs and dairy wastewater for simultaneous saccharification and lactic acid production: Process optimization and kinetic assessment. Bioresource Technology, 348, 126815.
- Dienye, B.N., Ahaotu, I., Agwa, O.K., Odu, N.N. 2018. Citric acid production potential of Aspergillus niger using Chrysophyllum albidum peel. Advances in Bioscience and Biotechnology, 9, 190-203.
- El-Imam, A. A., Du, C. 2014. Fermentative itaconic acid production. J Biodivers Biopros Dev, 1, 1-8.
- Ellena, V., Sauer, M., Steiger, M.G. 2020. The fungal sexual revolution continues: discovery of sexual development in members of the genus Aspergillus and its consequences. Fungal Biology and Biotechnology, 7, 1-7.
- 15. Fenila, F., Shastri, Y. 2016. Optimal control of enzymatic hydrolysis of lignocellulosic biomass. Resource-Efficient Technologies, 2, 96-104.

- 16. Gaitonde, V.N., Manjaiah, M., Maradi, S., Karnik, S.R., Petkar, P.M., Paulo Davim, J. 2017. Multiresponse optimization in wire electric discharge machining (WEDM) of HCHCr steel by integrating response surface methodology (RSM) with differential evolution (DE). Computational Methods and Production Engineering. Woodhead Publishing, 199-221.
- Germec, M., Turhan, I., Karhan, M., Demirci, A.J.a.S. 2019. Kinetic Modeling and Technoeconomic Feasibility of Ethanol Production From Carob Extract Based Medium in Biofilm Reactor. 9, 2121.
- Ghanbartabar, S.A., Najafpour, G.D., Mohammadi, M. 2016. Comparative studies on CA production from cheese why by submerged and immobilized Aspergillus niger. Pakistan Journal of Biotechnology, 13, 79-85.
- Gogoi, M., Biswas, T., Biswal, P., Saha, T., Modak, A., Gantayet, L.M., Nath, R., Mukherjee,
 I., Thakur, A.R., Sudarshan, M. 2021. A novel strategy for microbial conversion of dairy wastewater into biofertilizer. Journal of Cleaner Production, 293, 126051.
- 20. Hamill, P.G., Stevenson, A., Mcmullan, P.E., Williams, J.P., Lewis, A.D., Stevenson, K.E., Farnsworth, K.D., Khroustalyova, G., Takemoto, J.Y., Quinn, J.P. 2020. Microbial lag phase can be indicative of, or independent from, cellular stress. Sci. Rep., 10, 1-20.
- 21. Ho, H. L., Hood, J. S. 2014. Optimisation of medium formulation and growth conditions for xylanase production by Aspergillus brasiliensis in submerged fermentation (SmF). J Biodivers Biopros Dev, 1, 2376-0214.
- Ijadpanahsaravi, M., Punt, M., Wösten, H.A., Teertstra, W.R. 2021. Minimal nutrient requirements for induction of germination of Aspergillus niger conidia. Fungal Biol., 125, 231-238.
- 23. Islam, M., Asad, M., Hossain, M., Paul, S., Sujan, S. 2019. Bioethanol production from banana pseudostem by using separate and cocultures of cellulase enzyme with Saccharomyces cerevisiae. Journal of Environmental Science Technology, 12, 157-163.

- 24. Kareem, S., Akpan, I., Alebiowu, O. 2010. Production of CA by Aspergillus niger using pineapple waste. Malaysian Journal of Microbiology, 6, 161-165.
- 25. Kaur, N. 2021. Different treatment techniques of dairy wastewater. Groundwater for Sustainable Development, 14, 100640.
- 26. Krappmann, S., Braus, G. 2005. Nitrogen metabolism of Aspergillus and its role in pathogenicity. Med. Mycol., 43, 31-40.
- 27. Laltha, M., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2022. Innovative microwave-assisted iodized table salt or paper wastewater pretreatments for enhanced sugar recovery from banana pseudostem. Biomass Conversion and Biorefinery, 1-15.
- 28. Maddox, I.S., Hossain, M., Brooks, J.D. 1986. The effect of methanol on CA production from galactose by Aspergillus niger. Applied Microbiology and Biotechnology, 23, 203-205.
- 29. Moodley, P., Sewsynker-Sukai, Y., Gueguim Kana, E.B. 2020. Progress in the development of alkali and metal salt catalysed lignocellulosic pretreatment regimes: Potential for bioethanol production. Bioresource Technology, 310, 123372.
- 30. Moyer, A.J. 1953. Effect of Alcohols on the mycological production of CA in surface and submerged culture: I. Nature of the Alcohol Effect. Appl. Microbiol., 1, 1-7.
- 31. Odu, N., Uzah, G., Akani, N. 2020. Optimization of CA Production by Aspergillus niger and Candida tropicalis for Solid State Fermentation Using Banana Peel Substrate. Journal of Life and Bio Sciences Research, 1, 51-60.
- 32. Ozdal, M., Kurbanoglu, E.B. 2019. Citric acid production by Aspergillus niger from agroindustrial by-products: Molasses and chicken feather peptone. Waste and Biomass Valorization, 10, 631-640.
- 33. Papagianni, M. 2007. Advances in CA fermentation by Aspergillus niger: biochemical aspects, membrane transport and modeling. Biotechnol. Adv., 25, 244-263.

- 34. Papagianni, M., Mattey, M., Kristiansen, B. 1999. The influence of glucose concentration on CA production and morphology of Aspergillus niger in batch and culture. Enzyme and Microbial Technology, 25, 710-717.
- 35. Pramanik, S.K., Suja, F.B., Porhemmat, M., Pramanik, B.K. 2019. Performance and Kinetic Model of a Single-Stage Anaerobic Digestion System Operated at Different Successive Operating Stages for the Treatment of Food Waste. Processes, 7, 600.
- 36. Puchalska, P., Crawford, P.A. 2017. Multi-dimensional roles of ketone bodies in fuel metabolism, signaling, and therapeutics. Cell Metab., 25, 262-284.
- 37. Roukas, T., Harvey, L. 1988. The effect of pH on production of citric and gluconic acid from beet molasses using continuous culture. Biotechnology Letters, 10, 289-294.
- 38. Sarkar, D., Das, K. 2017. Optimization Of CA Production From Aspergillus Niger Using Pineapple Waste As Feedstock In Submerged Fermentation. World J. Pharm. Res, 6, 810-818.
- 39. Sekoai, P.T., Ayeni, A.O., Daramola, M.O. 2018. Parametric optimization of CA production from apple pomace and corn steep liquor by a wild type strain of Aspergillus niger: A Response surface methodology approach. International Journal of Engineering Research in Africa, Trans Tech Publ, 98-113.
- 40. Sekoai, P.T., Ouma, C.N.M., Du Preez, S.P., Modisha, P., Engelbrecht, N., Bessarabov, D.G., Ghimire. 2019. Application of nanoparticles in biofuels: an overview. Fuel, 237, 380-397.
- 41. Show, P.L., Oladele, K.O., Siew, Q.Y., Aziz Zakry, F.A., Lan, J.C.-W., Ling, T.C. 2015. Overview of CA production from Aspergillus niger. Frontiers in Life Science, 8, 271-283.
- 42. Slavov, A.K. 2017. Dairy wastewaters–general characteristics and treatment possibilities–a review. Food Technology and Biotechnology, 55, 14-28.
- 43. Srimachai, T., Nuithitikul, K., O-Thong, S., Kongjan, P., Panpong, K. 2015. Optimization and Kinetic Modeling of Ethanol Production from Oil Palm Frond Juice in Batch Fermentation. Energy Procedia, 79, 111-118.

- 44. Tudzynski, B. 2014. Nitrogen regulation of fungal secondary metabolism in fungi. Frontiers in microbiology, 5, 656.
- 45. Varga, J., Kocsube, S., Toth, B., Frisvad, J.C., Perrone, G., Susca, A., Meijer, M., Samson, R.A. 2007. Aspergillus brasiliensis sp. nov., a biseriate black Aspergillus species with world-wide distribution. International Journal of Systematic and Evolutionary Microbiology, 57, 1925-1932.
- 46. Veverka, K., Štolcová, J., Růžek, P. 2007. Sensitivity of fungi to urea, ammonium nitrate and their quimolar solution UAN. Plant Protection Science, 43, 157.
- 47. Vrabl, P., Schinagl, C.W., Artmann, D.J., Heiss, B., Burgstaller, W. 2019. Fungal growth in batch culture–what we could benefit if we start looking closer. Frontiers in microbiology, 2391.
- 48. Yadav, V.G., Yadav, G.D., Patankar, S.C. 2020. The production of fuels and chemicals in the new world: critical analysis of the choice between crude oil and biomass vis-à-vis sustainability and the environment. Clean technologies and environmental policy, 22, 1757-1774.
- 49. Zhang, X., Zhang, B., Miao, R., Zhou, J., Ye, L., Jia, D., Peng, W., Yan, L., Zhang, X., Tan,
 W. 2018. Influence of Temperature on the Bacterial Community in Substrate and Extracellular Enzyme Activity of Auricularia cornea. Mycobiology, 46, 224-235.

Chapter 4 - Supplementary material

Screening	Chemical constituent	Input parameters	Citric acid concentration (g/L)
method			
Desorbent	Acetone	$3\% (v/v)^{a}$, 10% $(w/v)^{c}$, 10% $(v/v)^{d}$, 35 °C ^e , 150 rpm ^f ,	10.886
		144 h ^g	
	Chloroform	3% $(v/v)^{a}$, 10% $(w/v)^{c}$, 10% $(v/v)^{d}$, 35 °C ^e , 150 rpm ^f ,	7.044
		144 h ^g	
	Propan-2-ol	3% $(v/v)^a$, 10% $(w/v)^c$, 10% $(v/v)^d$, 35 °C ^e , 150 rpm ^f ,	10.245
		144 h ^g	
Nitrogen source	Ammonium nitrate	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	12.807
		144 h ^g	
	Corn steep liquor	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	8.324
		144 h ^g	
	Peptone	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	9.605
		144 h ^g	
	Yeast extract	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	7.684
		144 h ^g	
Nitrogen source	Propan-2-ol Ammonium nitrate Corn steep liquor Peptone Yeast extract	144 h ^g 3% (v/v) ^a , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f , 144 h ^g 2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f , 144 h ^g 2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f , 144 h ^g 2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f , 144 h ^g 2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f , 144 h ^g 2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f , 144 h ^g	10.245 12.807 8.324 9.605 7.684

Table S1. Preliminary screening of various desorbents and nitrogen sources to determine their effectiveness for citric acid production

Footnote: ^a=desorbent concentration, ^b=nitrogen concentration, ^c=substrate solid loading, ^d=inoculum size, ^e=temperature, ^f=agitation speed,

^g=time.

Run	Acetone (v/v)	Ammonium nitrate (w/v)	Temperature (°C)	Citric acid concentration (g/L)	
			-	Megazyme assay	Acid-base titration
1	3	0.50	30	1.615	1.857
2	1	0.50	35	13.219	14.408
3	5	2.75	30	0.631	0.640
4	3	2.75	35	8.842	9.093
5	3	2.75	35	9.090	9.413
6	3	0.50	40	1.048	1.153
7	5	2.75	40	0.639	0.704
8	1	2.75	30	2.177	2.433
9	3	5.00	40	1.090	1.217
10	5	0.50	35	5.920	6.275
11	3	2.75	35	8.206	8.324
12	3	2.75	35	5.341	5.571
13	3	2.75	35	8.950	9.413
14	5	5.00	35	6.412	6.852
15	3	5.00	30	2.110	2.305
16	1	5.00	35	12.467	12.679
17	1	2.75	40	0.546	0.832

Table S2. Citric acid concentration determined for the SSF process using the Megazyme assay kit

Source	Sum of Squares	Degrees of freedom (<i>df</i>)	Mean Square	F-value	p-value (probability > F)
Model	294.34	9	32.7	7.89	0.0063 Significant
A-Acetone	31.52	1	31.52	7.61	0.0282
B-Ammonium nitrate	0.051	1	0.051	0.012	0.9146
C-Temperature	1.39	1	1.39	0.33	0.5812
AB	1.33	1	1.33	0.32	0.5889
AC	0.69	1	0.69	0.17	0.6948
BC	0.037	1	0.037	8.9×10 ⁻³	0.9275
A^2	1.54	1	1.54	0.37	0.5611
B^2	4.96	1	4.96	1.20	0.3101
C^2	257.17	1	257.17	62.07	0.0001
Residual	29.00	7	4.14		
Lack of fit	18.47	3	6.16	2.34	0.2150 Not significant
Pure error	10.53	4	2.63		
Cor total	323.34	16			

Footnote: A = Acetone (v/v), B = Ammonium nitrate (w/v) and C = Temperature (°C).

Substrate	Fermentative	Optimal process parameter conditions	Citric acid	References
	microorganism		concentration (g/L)	
Banana pseudostem	Aspergillus brasiliensis	10% $(v/v)^{a}$, 10% $(w/v)^{b}$, 1% (v/v) acetone ^c , 0.5%	12.967	This study
(SSF _{optimizedFW})	ATCC16404	(w/v) ammonium nitrate ^d , 35 °C ^e , 150 rpm ^f , 96		
		h ^g		
Banana pseudostem	Aspergillus brasiliensis	10% $(v/v)^a$, 10% $(w/v)^b$, 1% (v/v) acetone ^c , 0.5%	13.095	This study
(SSF _{Dww})	ATCC16404	(w/v) ammonium nitrate ^d , 35 °C ^e , 150 rpm ^f , 96		
		h ^g		
Banana pseudostem	Aspergillus brasiliensis	$10\% (v/v)^a$, $10\% (w/v)^b$, $1\% (w/v)$ peptone ^d , 35	13.991	This study
(SSFsDEmodified)	ATCC16404	°C ^e , 150 rpm ^f , 96 h ^g		
Dairy wastewater	Aspergillus niger	150 mL medium mixture ^{a,b} , 30 °C ^e , 200 rpm ^f , 10	4.50	Kim et al. (2002)
	ATCC9142	d ^g , pH 3 ^h		
Banana peel	Aspergillus niger	10 mL ^a , 50 g ^b , 3% (v/v) methanol ^c , 0.5% (w/v)	97.60	Odu et al. (2020)
		ammonium chloride ^d , 30 °C ^e , 10 d ^g , pH 4 ^h , 2%		
		(w/v) zinc ⁱ		
Pineapple waste	Aspergillus niger	$6 \text{ x } 10^6 \text{ suspension}^a$, 30 g^b , $4\% \text{ (v/v) methanol}^c$,	44.10	Sarkar and Das
		0.25% w/v ammonium nitrate ^d , 30 °C ^e , 7 d ^g , 15%		(2017)
		(w/v) sucrose ⁱ		

Table S4. Comparative assessment of citric acid derived from lignocellulosic and dairy wastes utilizing Aspergillus sp.

Table S4. Continued

Substrate	Fermentative	Optimal process parameter conditions	Citric acid	References
	microorganism		concentration (g/L)	
Sweet potato peel	Aspergillus niger	5 d, 6 mm agar spores ^a , 90 mL ^b , 3% (v/v)	4.36	Aboyeji et al.
waste		methanol ^c , 1.25% (w/v) ammonium nitrate ^d , 25		(2020)
		°C ^e , 7 d ^g , pH 6.5 ^h		
Sugar beet molasse	Aspergillus niger EB12	1 mL ^a , 150 g/L ^b , 4g/L Chicken feather peptone ^d ,	68.80	Ozdal et al.
		30 °C ^e , 200 rpm ^f , 168 h ^g , pH 6 ^h , 0.15 g/L		(2019)
		K ₂ HPO ₄ ⁱ		
Cheese whey	Aspergillus niger	50 g/L lactose ^b , 0.5 g/L Ammonium iron (II)	4.45	Ghanbartabar et
biomass	PTCC5010	sulfate ^d , 30 °C ^e , 130 rpm ^f , 6 d ^g , pH 4 ^h , 0.1 g/L		al. (2016)
		MgSO4 ⁱ		
Apple pomace	Aspergillus niger	1 mL ^a , 33.81 g/L ^b , 2.05% (v/v) methanol ^c , 42.5	68.26	Sekoai et al.
		g/L of corn steep liquor ^d , 33 $^{\circ}$ C ^e , 33 h ^g , pH 4.54 ^h		(2018)
Sugarcane bagasse	Aspergillus niger	4% (v/v) ethanol ^c , 30 °C ^e , 144 h ^g , 1% (w/v)	0.73	Campanhol et al.
	CCT4355	sucrose ⁱ		(2019)

 $\overline{Footnote.} \ ^{a}=\text{inoculum size, } ^{b}=\text{substrate amount, } ^{c}=\text{desorbent concentration, } ^{d}=\text{nitrogen concentration, } ^{e}=\text{temperature, } ^{f}=\text{agitation speed, } ^{g}=\text{time, } ^{g$

^h=pH, ⁱ= chemical additive.

Kinetic parameters	ic parameters Fermentation process		
	SSF _{optimizedFW}	SSF _{DWW}	SSF _{SDEmodified}
μ_{max} (h ⁻¹)	0.05	0.05	0.05
X _o (g/L)	0.19	0.15	0.21
X _{max} (g/L)	2.79	2.74	2.91
$P_m (g/L)$	29.31	18.87	21.50
$r_{p.m}\left(g/L/h ight)$	0.17	0.17	0.16
t _L (h)	16.47	9.80	5.11

Table S5. Kinetic parameters of the microbial growth and citric acid production for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes

Footnote: μ_{max} = maximum specific growth rate, X_0 = initial cell concentration, X_{max} = maximum cell concentration, P_m = maximum potential

citric acid concentration, $r_{p,m}$ = maximum citric acid production rate, t_L =lag time.

Equations for computing CA concentration according to Ayeni et al. (2019):

Moles NaOH required = vol of NaOH required (mL) × conc. of NaOH (moles/L) × 10^{-3} (1)

The amount in moles of citric acid, $C_3H_5O(COOH)_3$, in the titrated sample was determined according to Eq. (2).

Moles C₃H₅O(COOH)₃ =
$$\frac{\text{moles NaOH × 1 mol C3H5O(COOH)3}}{3 \text{ moles NaOH}}$$
(2)

The concentration (molarity) of citric acid produced was calculated by Eq. (3).

Concentration C₃H₅O(COOH)₃ =
$$\frac{\text{moles } C_3H_5O(\text{COOH})_3}{\text{volume } C_3H_5O(\text{COOH})_3}$$
 (3)

The concentration of $C_3H_5O(COOH)_3$ g/L was determined by Eq. (4) and (5).

Molar mass =
$$\frac{Concentration (g/L)}{Concentration (mol/L)}$$
 (4)

Concentration of citric acid (g/L) = Concentration in mol/L × Molar mass (g/mol) (5)

CHAPTER 5

Conclusions and recommendations for further research

5.1. Conclusions

Lignocellulosic bioprocesses are currently challenged by high capital expenditure for water, energy and lab-based chemicals. Apart from these limitations, current bioconversion processes result in low product yields and has significant environmental consequences. Exploring lowcost lignocellulosic bioprocesses for the production of bioproducts will promote a sustainable circular bioeconomy, while overcoming these bottlenecks. Therefore, this study explored the valorization of banana pseudostem (BP) for citric acid (CA) production by developing novel pretreatment strategies and thereafter optimizing the simultaneous saccharification and fermentation (SSF) process to potentially alleviate the abovementioned bottlenecks. Important findings and their significance are summarized as follows:

5.1.1. The present study modelled and optimized two novel methods namely (1) microwaveassisted-iodized table salt (M-ITS) and (2) microwave-assisted-paper wastewater (M-PWW) strategy for the enhancement of sugar recovery from banana pseudostem. Optimization of pretreatment techniques revealed a slightly higher reducing sugar (0.515 g/g) and glucose yield (0.433 g/g) for the M-ITS strategy (2.48% ITS, 800 W and 10 min) compared to the M-PWW method (30% solid loading, 800 W and 8 min) (reducing sugar = 0.498 g/g and glucose = 0.413 g/g). Although the M-PWW_{optimized} regime displayed a slightly lower RSY (3.3%) and GY (4.62%) compared to the M-ITS_{optimized} strategy, it has been concluded to be a more effective approach, since it is waste-based, requires fewer pretreatment cycles and that generates three times the quantity of pretreated substrate required for subsequent fermentation. As a result, the M-PWW_{optimized} method was selected for simultaneous saccharification and fermentation (SSF) process optimization. The suitability of the developed and optimized pretreatment models were assessed and validated on various lignocellulosic substrates. These findings contribute to eliminating expensive lab-based chemicals and/or fresh water usage, while generating additional revenue.

5.1.2. This study modelled and optimized various fermentation input parameters that comprised of ammonium nitrate concentration, acetone concentration and temperature in the SSF process for optimal citric acid production from pretreated BP. The optimized SSF conditions (0.5% ammonium nitrate, 1% acetone and 35 °C) gave a maximum CA concentration of 14.408 g/L. Thereafter, kinetic assessment of Aspergillus brasiliensis growth and CA production were determined for the optimum conditions using fresh water (SSF_{optimizedFW}) or dairy wastewater (SSF_{DWW}) and compared to Sabouraud Dextrose Emmon's medium modified with BP (SSF_{SDEmodified}). Kinetic assessment revealed the same maximum specific growth rates (μ_{max}) $(0.05 h^{-1})$ for all three bioprocesses. The SSF_{SDEmodified} system resulted in the highest maximum potential CA concentration (Pm) (13.991 g/L) compared to the SSF_{DWW} (Pm= 13.095 g/L) and SSF_{optimizedFW} (P_m= 12.967 g/L) bioprocesses. Therefore, optimization of a waste-based media formulation approach for CA production can be channelled towards an industrial-scale trajectory. Furthermore, these findings facilitate the implementation of waste-based lignocellulosic bioprocesses for microbial-based value-added products while eradicating costly pretreatment chemicals, fermentation medium constituents and fresh water resources, in line with the food, energy and water nexus towards achieving a circular bioeconomy.

5.2. Recommendations for future studies

Based on the concluding findings from the present study, the following recommendations can be drawn for future research on waste-based citric acid production utilizing banana pseudostem:

5.2.1. In addition to the recycling of expensive chemicals and waste, it may be beneficial to investigate the recyclability of the removed lignin. Lignin is an aromatic rich polymer that is generated as a major by-product in the commercial biorefinery industry. Lignin is usually incinerated in boilers to produce energy and heat for functioning of the biorefinery. Lignin may also be transformed into high value-added chemicals that include vanillin, composite resins, carbon fibers and adhesives, thus reducing the overall biorefinery cost.

5.2.2. Microbial biomass derived from bacteria, yeasts, filamentous fungi or microalgae represents a promising alternative to conventional sources of food and feed. Microorganisms are a beneficial source of vitamin and protein. Protein from the fermentative microorganisms may be converted to amino acids and further used to produce chemicals and precursors for biomaterials.

5.2.3. Following fermentation and extraction of the organic acid, a large volume of effluent is generated known as hydrolysate, representing a challenge for downstream processes. This accumulated waste can be utilized as a feedstock for the generation of valuable biofuels such as biohydrogen and bioethanol. Furthermore, complete valorization of the effluent will contribute to a zero-waste-producing biorefinery. This will significantly reduce costs that are associated with the production, downstream and extraction processes.

5.2.4. Exploiting the existing natural biodiversity of microorganisms through mutagenesis and recombinant DNA technology will promote genetic diversification. Additionaly, this will enhance the industrial feasibility of citric acid production and significantly reduce the expenditure associated with bioprocessing.

5.2.5. Conducting technoeconomic analysis of citric acid production incorporating waste-based strategies will provide valuable data for reducing costs at several steps while contributing to knowledge and investment opportunities.

Appendix A

Industrial Crops & Products 174 (2021) 114166



Contents lists available at ScienceDirect

Industrial Crops & Products

journal homepage: www.elsevier.com/locate/indcrop



Development of microwave-assisted alkaline pretreatment methods for enhanced sugar recovery from bamboo and corn cobs: Process optimization, chemical recyclability and kinetics of bioethanol production

Milesh Laltha^a, Y. Sewsynker-Sukai^b, Gueguim Kana E.B.^{a,*}

^a University of KwaZulu-Natal, School of Life Sciences, Pietermaritzburg, South Africa
^b University of Fort Hare, Fort Hare Institute of Technology, Private Bag X1314, Alice, 5700, South Africa

ARTICLE INFO

Keywords: Alkaline pretreatment Optimization Recyclability Bioethanol production

ABSTRACT

The present study optimized two microwave-assisted alkaline (Na₃PO₄.12H₂O and NaOH) pretreatment methods for the enhancement of sugar recovery from different lignocellulosic substrates (bamboo and corn cobs) with varying structural compositions. Subsequently, the chemical recyclability of the optimized alkaline pretreatments were evaluated on corn cobs. Furthermore, kinetics of microbial cell growth and bioethanol production in simultaneous saccharification and fermentation (SSF) processes were assessed. The optimized Na₃PO₄.12H₂O pretreatment conditions resulted in the maximum reducing sugar yield (0.512 g/g) from corn cobs. Interestingly, the recyclability studies demonstrated that the NaOH-based black liquor resulted in a slightly higher average sugar yield (0.37 g/g) compared to the Na₃PO₄.12H₂O (0.31 g/g) over three pretreatment cycles. Kinetics showed a maximum specific growth rate and maximum potential bioethanol concentration of 0.175 h⁻¹ and 16.928 g/L respectively for the optimized NaOH pretreated corn cobs. The findings from this study provide significant knowledge enhancements pertaining to the use of gold standard, well-known NaOH methods, in addition to the emerging Na₃PO₄.12H₂O alkaline catalyst for the pretreatment of lignocellulosic substrates with varying structural compositions. Moreover, recyclability of Na₃PO₄.12H₂O and NaOH black liquor demonstrated the potential to reduce costs associated with the application of chemicals and fresh water in pretreatment systems. Kinetics of cell growth and bioethanol production elucidates noteworthy indices such as the maximum specific growth rate of the cells, lag time and maximum bioethanol production for lignocellulosic process development and scale-up procedures.

1. Introduction

The global energy demand is rapidly escalating due to a sharp increase in the population growth rate (Tagliapietra, 2019). Additionally, the incineration of fossil fuels such as oil, gas and coal contribute to the greenhouse effect and inevitably global warming (Kumari and Singh, 2018). Due to the depletion of fossil fuels and its negative effect on the environment, the search for alternative energy resources that are sustainable, cost-efficient and reduce the carbon footprint is imperative (Kumari and Singh, 2018). Lignocellulosic biomass (LCB) represent excellent feedstocks for the production of biofuels and value-added compounds, since they are cost-effective, abundant and renewable (Sewsynker-Sukai and Gueguim Kana, 2018a).

These substrates (LCB) primarily consist of cellulose (35-50 %),

hemicellulose (15–35 %) and lignin (20–25 %) structures (Gu et al., 2019). Examples of LCB include wheat straw (Geng et al., 2014), sugarcane bagasse (Jugwanth et al., 2019), corn cobs (Sewsynker-Sukai and Gueguim Kana, 2018b) and bamboo (Kuttiraja et al., 2013). Bamboo has garnered interest as a promising feedstock for the production of bioethanol, owing to its perennial nature, rapid growth, low management requirements and tolerance to extreme climatic conditions (Chaowana, 2013). Bamboo comprises 30–47 % cellulose, 16–28 % hemicellulose and 14–28 % lignin (Chaowana, 2013; Kuttiraja et al., 2013). Alternatively, corn cobs consist of 38–45 % cellulose, 26–35 % hemicellulose and 8–19 % lignin and represent another promising feedstock for bioethanol production (Kim, 2018). Corn cobs make up about 30 % of maize agricultural wastes with an estimated global annual production of approximately 500 million metric tons (Czajkowski et al., 2019).

* Corresponding author. E-mail address: kanag@ukzn.ac.za (G.K. E.B.).

https://doi.org/10.1016/j.indcrop.2021.114166

Received 28 May 2021; Received in revised form 8 October 2021; Accepted 10 October 2021 0926-6690/© 2021 Elsevier B.V. All rights reserved.
Moreover, the low lignin content present in corn cobs makes it a desirable feedstock for biofuel production (Sewsynker-Sukai and Gueguim Kana, 2018b).

Even with the immense benefits of LCB, its major constituents (cellulose and hemicellulose) are cemented together by layers of recalcitrant lignin moieties that hamper the enzymatic release of fermentable sugars (Gu et al., 2019). Therefore, the conversion of LCB to biofuels and value-added compounds necessitates the disruption of the steric hindrance of lignin and hemicellulose by use of pretreatment regimes (Yu et al., 2019). Pretreatment alters the cell wall structure and chemical composition of LCB (Gu et al., 2019). This in turn increases cellulose accessibility to cellulolytic enzymes, thus enhancing enzymatic hydrolysis (Gu et al., 2019). The pretreated LCB is enzymatically hydrolyzed to fermentable sugars such as glucose, which may be used in microbial fermentation processes (Yu et al., 2019; Vergara et al., 2019). Several methods of lignocellulosic pretreatment have been developed over the years. These include microwave, acid, alkali, inorganic salts, ammonia fibre expansion (AFEX) and steam explosion, among others (Yu et al., 2019). However, none of the aforementioned techniques have suitably addressed the challenges related to the high cost and consumption of energy (Qing et al., 2016b; Vergara et al., 2019). Significant efforts are still being made to develop energy-efficient and environmental friendly pretreatment strategies that produce high sugar yields and are cost-effective (Qing et al., 2016b).

Several previous studies have demonstrated the high efficiency of strong alkaline pretreatments such as sodium hydroxide (NaOH), sodium ethoxide (C₂H₅ONa), calcium hydroxide (CaOH₂) and potassium hydroxide (KOH) (Gong et al., 2019). Nevertheless, the high costs of alkaline treatments limit its industrial application (Qing et al., 2016b). Recently, alkalic salts have garnered interest due to their recyclability, cost-effectiveness and high sugar yield. Sodium phosphate dodecahy-drate (Na₃PO₄.12H₂O), sodium sulphide (Na₂S), and sodium carbonate (Na₂CO₃) are some examples of alkalic salts that have recently appeared, since these mimic strong alkaline catalysts such as NaOH, but provide a cheaper alternative (Gu et al., 2019). The main pretreatment mechanism of alkaline reagents occurs as a result of the cleavage of ester and glycosidic bonds in the cell wall (Qing et al., 2016b). This leads to the alteration and degradation of lignin, in addition to the swelling of cellulose and hemicellulose moieties (Qing et al., 2016b).

Additionally, the chemical hydrolysate that is generated following alkaline pretreatments is commonly referred to as black liquor. The black liquor possesses a high pH (13–14) (Zhu and Theliander, 2015), chemical oxygen demand (COD) and biochemical oxygen demand (BOD) that has shown to be detrimental to the environment (Wang et al., 2010). Interestingly, the alkaline characteristic of black liquor allows it to be recycled and involves the use of spent liquor for successive pre-treatments (Vergara et al., 2019). The recycling of black liquor obtained from the fractionation of LCB minimizes fresh water consumption, chemical costs and improves environmental sustainability (Vergara et al., 2019).

Microwave heating process (MHP) has emerged as an efficient alternative to conventional steam-assisted heating for the pretreatment of LCB (Aguilar-Reynosa et al., 2017a). MHP uses microwave energy to vibrate and heat molecules causing them to collide and result in the formation of thermal hot spots. Intensive heat and vibration rupture the components of LCB causing the rearrangement of crystalline cellulose. Advantages of MHP include its faster heat transfer within a short time, high product yields, environmentally benign nature, energy efficiency and cost-effectiveness (Aguilar-Reynosa et al., 2017a).

The comparative optimization of microwave-assisted alkalic salt $(Na_3PO_4.12H_2O)$ or a strong base (NaOH) on different lignocellulosic substrates (corn cobs and bamboo) will provide significant insights for the enhancement of sugar recovery. In addition, the potential recyclability of these alkaline solutions will enhance the process economics of the pretreatment for industrial scale implementation. The sugar that is generated from effective lignocellulosic pretreatments can be

channelled towards biofuel production or high value commodities. In the recent time, bioethanol has received considerable attention due to several beneficial properties. These may include its high evaporation enthalpy, high-octane number and flammability for combustion generating heat (Waqas et al., 2016). The aforementioned properties of bioethanol also enables its combination with derivatives of hydrocarbons (Waqas et al., 2016). Various fermentation process types exist for bioethanol production. These include separate hydrolysis and fermentation (SHF), simultaneous saccharification and fermentation (SSF) and SSF with a prehydrolysis stage (PSSF).

However, SSF strategies are becoming an attractive approach for bioethanol production (Gonçalves et al., 2016), due to the lower production cost, increased ethanol concentration and ethanol conversion rates by eradicating separate, lengthy hydrolysis steps (Sewsynker-Sukai and Gueguim Kana, 2018b; Gonçalves et al., 2016). The SSF process type also reduces the feedback inhibition of enzymes and the risk for contamination (Mithra et al., 2018). Kinetic modelling approaches may be used for SSF bioprocess development, optimization, scale-up and process control. These models help to reduce undesirable by-products and improve the product yield (Sewsynker-Sukai and Gueguim Kana, 2018a). Some examples of modelling approaches include the logistic and modified Gompertz models. The logistic model describes microbial cell growth whereas the modified Gompertz model predicts maximum bioethanol formation (Sewsynker-Sukai and Gueguim Kana, 2018b).

Therefore, the objectives of the present study were to: (1) develop and assess two different microwave-assisted alkaline-catalyzed pretreatments (Na₃PO₄.12H₂O and NaOH) on bamboo and corn cobs, (2) determine the interactive effects of alkali concentration, microwave power intensity and pretreatment time on the reducing sugar yield, (3) evaluate the potential recyclability of the Na₃PO₄.12H₂O and NaOH black liquor hydrolysate, and (4) assess the kinetics of yeast cell growth and bioethanol production on the pretreated substrate using the logistic and modified Gompertz models respectively.

2. Materials and methods

2.1. Materials

The bamboo and corn cob substrates utilized in this study were obtained from Cedara (Hilton, South Africa) (29°32'25.8"S 30°16'49.8"E) and the Ukulinga research farm (Pietermaritzburg, South Africa) (29°39'45.6"S 30°24'17.9"E) respectively. The substrates were oven dried at 60 °C for 2 days and were thereafter milled to a particle size between 1–2 mm. The powdered bamboo and corn cob substrates were stored in airtight containers at room temperature. The chemicals used in this study were purchased from Merck, South Africa. The enzyme used for saccharification was Cellic CTec 2 (160 FPU/mL) and was kindly provided by Novozymes (Novozymes A/S, Denmark).

2.2. Preliminary screening

Preliminary screening was carried out to determine the influence of a strong base (NaOH) and an alkalic salt (Na₃PO₄.12H₂O) on the reducing sugar yield from the bamboo and corn cob substrates. A total working volume of 250 mL was used for the preliminary screening. Bamboo or corn cobs with a solid loading of 10 % (w/v) were submerged in 12.50 % (w/v) of either NaOH or Na₃PO₄.12H₂O alkali solutions and were heated at a power intensity of 500 W for 7.5 min. After the pretreatment process, the substrates were washed rigorously with distilled water and filtered using a domestic sieve (<1 mm). The washed pretreated substrates were thereafter oven dried at 55 °C overnight prior to the enzymatic hydrolysis stage. Control experiments that included: (1) untreated (native) corn cobs or bamboo and, (2) corn cobs or bamboo pretreated and heated with distilled water only (devoid of NaOH or Na₃PO₄.12H₂O) were performed.

2.3. Modelling and optimization of pretreatment parameters

The response surface methodology (RSM) model (Box-Behnken design) was used to develop four pretreatment models consisting of 17 experimental runs each. The pretreatment input parameters consisted of alkali concentration (NaOH or Na3PO4.12H2O) (5-20 %, w/v), microwave power intensity (100-900 W) and microwave pretreatment time (5-10 min) while the output parameter was the reducing sugar yield (g/ g). The input parameter ranges (Table 1) were selected following the preliminary screening in addition to an extensive literature survey (Qing et al., 2016a, b, Boonsombuti et al., 2013). Therefore, four RSM pretreatment models were designated as follows: (1) bamboo pretreated with NaOH (NaOH_{BB}), (2) bamboo pretreated with Na₃PO₄.12H₂O (Na₃PO₄.12H₂O_{BB}), (3) corn cobs pretreated with NaOH (NaOH_{CC}), and (4) corn cobs pretreated with $Na_3PO_4.12H_2O_{BB}$ ($Na_3PO_4.12H_2O_{CC}$). Experimental reducing sugar yields obtained after the enzymatic saccharification stage were used to develop the polynomial models using Design Expert software (Stat-Ease Inc., USA). The general form of the RSM polynomial model equation is shown in Eq. (1):

$$Y = \beta_0 + \beta_1 x_1 + \beta_2 x_2 + \beta_3 x_3 + \beta_4 x_4 + \beta_{11} x_1^2 + \beta_{22} x_2^2 + \beta_{33} x_3^2 + \beta_{44} x_4^2 + \beta_{12} x_1 x_2 + \beta_{13} x_1 x_3 + \beta_{14} x_1 x_4 + \beta_{23} x_2 x_3 + \beta_{24} x_2 x_4 + \beta_{34} x_3 x_4$$
(1)

where Y is the reducing sugar yield response, β_0 is the free or offset term. $\beta_1 x_1$ to $\beta_4 x_4$ represents the linear coefficients, $\beta_{11} x_1^2$ to $\beta_{44} x_4^2$ are quadratic coefficients and $\beta_{12} x_1 x_2$ to $\beta_{34} x_3 x_4$ are the interaction coefficients. The significance of the model is assessed by the Analysis of Variance (ANOVA). Process optimization was carried out by solving the polynomial equation using the method of Myers et al. (1995).

2.4. Enzymatic hydrolysis

The alkaline pretreated substrates (corn cobs or bamboo) were submerged in 10 mL of sodium citrate buffer (0.05 M, pH 4.8). In addition, standard enzyme loading (10 FPU/g) and solid loading (10 %, w/v) were used. The enzymatic hydrolysis stage was carried out at a temperature and agitation of 50 °C and 120 rpm respectively, for 72 h in an orbital shaker incubator (FMH Instruments, South Africa). Subsequently, the hydrolysate samples were centrifuged at 10 000 rpm for 5 min (Biofuge pico, Heraeus) and analyzed for reducing sugar yield (g/g) using the 3,5dinitrosalicyclic acid (DNS) method established by Miller (1959).

2.5. Validation of the developed pretreatment models

Validation pretreatment experiments were carried out using the optimized input conditions as specified by each developed model (NaOH_{BB}, Na₃PO₄.12H₂O_{BB}, NaOH_{CC} and Na₃PO₄.12H₂O_{CC}) for the particular substrate. The chemical pretreatment and enzymatic hydrolysis reactions were performed as previously stated in Sections 2.3 and 2.4 above.

2.6. Potential recyclability of the NaOH and Na₃PO₄.12H₂O black liquor

The black liquor generated after the optimized NaOH and $Na_3PO_4.12H_2O$ (validation experiments) using corn cobs were collected separately and used for the recyclability assessment. The spent black

Table 1

Table 1				
Factors and levels er	mployed in the	e three-factor	Box-Behnken	design.

InputsUnits	Coded values				
		-1	0	+1	
A: Alkaline concentration	% (w/v)	5.00	12.50	20.00	
B: Power intensity	W	100	500	900	
C: Pretreatment time	min	5	7.5	10	

liquor hydrolysate (NaOH or Na₃PO₄.12H₂O) was used to pretreat corn cobs with the same heating parameters used for the validation experiments in order to determine the effectiveness of recycling the used chemicals. Recycling of the spent NaOH and Na₃PO₄.12H₂O was carried out over several cycles, until the liquor became highly viscous, turbid and concentrated with soluble solids due to evaporation during the pretreatment process. Thus, rendering the black liquor hydrolysates ineffective for further substrate penetration. The pH of the NaOH or Na₃PO₄.12H₂O spent liquor following each pretreatment cycle was recorded. Enzymatic hydrolysis was carried out on the NaOH and Na₃PO₄.12H₂O spent liquor pretreated corn cobs as previously outlined.

2.7. Ethanol fermentation bioprocess

2.7.1. Development of yeast inoculum

Angel yeast cells were grown on yeast peptone dextrose (YPD) solid media containing 10 g/L yeast extract, 20 g/L glucose, 20 g/L peptone and 20 g/L agar and incubated at 30 °C for 24 h. After incubation, a colony of yeast was inoculated into YPD broth with a composition of 10 g/L yeast extract, 20 g/L glucose, 20 g/L peptone. The inoculated media was incubated at 30 °C and 120 rpm for a period of 18 h, until the exponential growth phase was reached. This culture was then used as the starter culture for the subsequent SSF processes. The Angel yeast cells were quantified by means of a cell counting chamber (Neubauer, Germany).

2.7.2. Simultaneous saccharification and fermentation (SSF) process

Two sets of SSF bioethanol processes were carried out on the optimized NaOH and Na₃PO₄.12H₂O pretreated corn cobs and were designated (1) SSF process with NaOH pretreated corn cobs (NaOH_{CC(SSF)}) and (2) SSF process with Na₃PO₄.12H₂O pretreated corn cobs (Na₃PO₄.12H₂O $_{CC(SSF)}$). The SSF processes (50 mL) contained 10 % (w/ v) pretreated corn cobs, 40 mL sodium citrate buffer, 5 mL fermentation nutrients (10 g/L peptone, 5 g/L yeast extract), Cellic CTec2 enzyme (10 FPU/g) and 10 % (v/v) yeast inoculum. Control experiments (50 mL) were performed in parallel and consisted of 10 % (w/v) pretreated corn cobs, 45 mL sodium citrate buffer, 5 mL fermentation nutrients (10 g/L peptone, 5 g/L yeast extract) and Cellic CTec2 enzyme (10 FPU/g). The controls (uninoculated) were used to estimate the initial glucose concentration while the inoculated SSF experiments were regarded as the final glucose concentration. The initial and final glucose concentrations were thereafter used to compute the glucose utilization. All the SSF and control experiments were incubated at 35 °C and 120 rpm until the glucose was completely consumed and the ethanol concentration did not increase further. Liquid samples (1 mL) for cell biomass concentration and glucose utilization were extracted in 2 h intervals.

2.8. Analytical methods

2.8.1. Scanning electron microscopy (SEM)

The bamboo and corn cobs substrates (untreated and optimally pretreated) were visually analyzed using scanning electron microscopy (SEM) (ZEISS EVO LS 15). The dried samples were placed on carbon tape and sputter coated with gold for conductivity (Quorum Q150 R ES). The surface morphological changes of the lignocellulosic biomass samples were examined using conventional SEM at a magnification of 500 X (ZEISS EVO LS 15).

2.8.2. Fourier transform infrared spectroscopy (FTIR)

Fourier Transform Infrared (FTIR) spectroscopy was used to determine the chemical structure and functional group changes of the untreated and optimally pretreated substrates (corn cobs and bamboo). The FTIR was carried out using a Perkin Elmer 100 system (Waltham, MA USA). The substrate samples were each mixed with spectroscopic grade potassium bromide (KBr) and pressed into uniform discs. The samples were then scanned and recorded from 400 to 3800 cm⁻¹.

(2)

2.8.3. Sugar analysis

Reducing sugar in the preliminary screening, experimental design runs, validation and recyclability processes were analyzed using the DNS method (Miller, 1959). The glucose content for the validation, recyclability and ethanol experiments were determined using Megazyme GOPOD assay kits (Megazyme, Ireland). Glucose utilization (%) was computed using Equation (2) (Srimachai et al., 2015).

$$P = P_m \cdot exp\left\{-exp\left[\frac{r_{p,m} \cdot exp(1)}{P_m}\right] \cdot (t_L - t) + 1\right\}$$
(5)

The modified Gompertz model Equation (6), denotes P, P_m , $r_{p,m}$, t_L and t as the bioethanol concentration (g/L), potential maximum bioethanol concentration (g/L), maximum bioethanol production rate (g/L/h), lag time (h), and the fermentation time (h) respectively.

Glucose utilization (%) =
$$\frac{\text{Initial glucose concentration } (g/L) - \text{Final glucose concentration } (g/L)}{\text{Initial glucose concentration } (g/L)} \times 100$$

Where the initial glucose concentration (g/L) is the glucose obtained from the control flask reactor (uninoculated), and the final glucose concentration (g/L) is the glucose obtained from the experimental SSF flask at the same time point.

2.8.4. Biomass concentration

The Angel yeast biomass concentration was quantified by using the dry cell weight (g/L) as a function of the cell count (cells/mL). Angel yeast cells were grown for 16 h in YPD broth consisting of 10 g/L yeast extract, 20 g/L glucose, 20 g/L peptone and was thereafter diluted accordingly (1, 1/2, 1/4, 1/8 and 1/16). A total of 40 mL of the varying dilutions (1, 1/2, 1/4, 1/8 and 1/16) were centrifuged at 5000 rpm for 10 min. The supernatant obtained after centrifugation was discarded and the remaining biomass pellet was dried at 60 °C until a constant mass was obtained. The dry cell weights (or biomass concentration in g/L) and corresponding total cell counts (cells/mL) for each dilution were used to develop the standard calibration curve. Biomass concentrations (g/L) for the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes were then extrapolated from the standard curve by substituting the yeast cell count.

2.8.5. Ethanol concentration

The ethanol concentration was measured every 2 h in the gas phase of the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes using an ethanol vapour sensor (ETH-BTA, Vernier Software and Technology, Beaverton, OR, USA) as previously described by Jugwanth et al. (2019).

2.9. Computation of the kinetic coefficients

2.9.1. The logistic model

The differential formula of the logistic model shown in Eq. (3) was integrated to form Eq. (4). Eq. (4) accounts for the stationary and exponential phases of growth. The logistic model Eq. (4), relates the biomass (X) to the initial cell conc. (X₀), max specific growth rate (μ_{max}) and max cell concentration (X_{max}) at a certain time (t) during the stationary and exponential growth phases of yeast cells.

$$\frac{dX}{dt} = \mu_{max} \left(1 - \frac{X}{X_{max}} \right) X \tag{3}$$

$$X = \frac{X_0 \exp(\mu_{max}t)}{1 - \left[\left(\frac{X_0}{X_{max}} \right) (1 - exp(\mu_{max}t)) \right]}$$
(4)

2.9.2. The modified Gompertz model

Bioethanol data obtained for the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC} (SSF) processes were fitted to the modified Gompertz model using the least squares method (CurveExpert V1.5.5). The kinetic parameters such as maximum bioethanol production rate (r $_{p,m}$), lag time (t_L), and the maximum potential bioethanol concentration (P_m) were determined using Eq. (5).

3. Results and discussion

3.1. Preliminary screening

The preliminary Na₃PO₄.12H₂O and NaOH pretreatments with 12.50 % (w/v) on bamboo resulted in a reducing sugar yield of 0.11 g/g and 0.25 g/g respectively. However, higher reducing sugar yields of 0.37 g/g (Na₃PO₄.12H₂O) and 0.38 g/g (NaOH) were obtained from the corn cob substrate. The preliminary screening using Na₃PO₄.12H₂O and NaOH pretreatments on bamboo and corn cobs indicated the release of fermentable sugars and these processes were subsequently optimized. Control pretreatments (without Na₃PO₄.12H₂O and NaOH) were performed with the native (untreated) and distilled water pretreated substrates (bamboo and corn cobs) and very low yields of fermentable sugars were observed of (0.05 g/g and 0.08 g/g respectively). Similarly, the distilled water pretreated substrates resulted in low sugar yields of 0.06 g/g and 0.09 g/g from bamboo and corn cobs respectively.

3.2. RSM model development and significance

The pretreatment experimental reducing sugar data were used to develop the polynomial model Eqs. (6)–(9) (Table 2). The Analysis of Variance (ANOVA) was used to assess the fitness of the RSM models (Table 3). The high F-values (>4) and low p-values (<0.05) indicated that all four pretreatment models were significant. A p-value <0.05 indicates a significant model (Qing et al., 2016b) and that there is a five percent chance that a 'Model F-value' could occur due to noise (Nasirpour and Mousavi, 2018).

The significance of the pretreatment input parameters and their

Table 2

RSM polynomial model equations relating the input parameters to the output reducing sugar yields for the microwave-assisted alkaline pretreatments.

Model	Equation	Equation number
$Na_3PO_4.12H_2O_{BB}$	$\begin{array}{l} - 0.215 + 0.012A + 3.13 \times 10^{-4}B + 0.034C + \\ 2.54 \times 10^{-6}AB - 2.53 \times 10^{-5}AC + 1.41 \times \\ 10^{-6}BC - 4.82 \times 10^{-4}A^2 - 3.02 \times 10^{-7}B^2 - \\ 2.28 \times 10^{-3}C^2 \end{array}$	(6)
NaOH _{BB}	$\begin{array}{l} - \ 0.056 \ + \ 0.028A \ + \ 1.95 \ \times \ 10^{-4}B \ - \ 5.52 \ \times \\ 10^{-3}C \ - \ 5.17 \ \times \ 10^{-6}AB \ - \ 6.10 \ \times \ 10^{-4}AC \ + \\ 2.96 \ \times \ 10^{-6}BC \ - \ 6.05 \ \times \ 10^{-4}A^2 \ - \ 1.01 \ \times \\ 10^{-7}B^2 \ + \ 8.12 \ \times \ 10^{-4}C^2 \end{array}$	(7)
$Na_3PO_4.12H_2O_{CC}$	$\begin{array}{l} + \ 0.463 - 6.98 \times 10^{-3}A + 7.39 \times 10^{-4}B - \\ 0.089C + 2.73 \times 10^{-5}AB + 3.45 \times 10^{-3}AC + \\ 4.48 \times 10^{-5}BC - 8.69 \times 10^{-4}A^2 - 1.05 \times \\ 10^{-6}B^2 + 6.41 \times 10^{-4}C^2 \end{array}$	(8)
NaOH _{CC}	$\begin{array}{l} + \ 0.474 + 8.22 \times 10^{-3}A + 1.80 \times 10^{-4}B - \\ 0.035C + 4.61 \times 10^{-7}AB + 4.71 \times 10^{-4}AC - \\ 2.07 \times 10^{-5}BC - 4.53 \times 10^{-4}A^2 - 1.15 \times \\ 10^{-7}B^2 + 2.16 \times 10^{-3}C^2 \end{array}$	(9)

Footnote: A = Alkaline concentration, B = Power intensity, C = Pretreatment time.

Table 3

Analysis of variance (ANOVA) of the developed pretreatment models.

•				-		
Model	Sum of squares	df	Mean squares	F- value	p-value	\mathbb{R}^2
$Na_3PO_4.12H_2O_{BB}$	0.020	9	2.209×10^{-3}	4.49	0.0302	0.8523
NaOH _{BB}	0.027	9	2.990×10^{-3}	31.64	< 0.0001	0.9760
Na ₃ PO ₄ .12H ₂ O _{CC}	0.42	9	0.046	21.58	0.0003	0.9652
NaOH _{CC}	0.019	9	$\begin{array}{c} 2.143 \times \\ 10^{-3} \end{array}$	4.21	0.0357	0.8439

Footnote: df = degrees of freedom, F-value = Fisher-Snedecor distribution value, P-value = probability value, R^2 =coefficient of determination.

effect on the reducing sugar yields was evaluated by using the corresponding p-values. From all the investigated parameters, the effect of microwave power intensity (0.0303) was shown to be the most significant for the Na₃PO₄.12H₂O_{BB} model whereas both alkaline concentration (<0.0001) and power intensity (0.0006) were significant for the $NaOH_{BB}$ model. For the $Na_3PO_4.12H_2O_{CC}$ model, the power intensity (<0.0001) was shown to be the most significant parameter followed by the alkaline concentration (p = 0.0016). Conversely, the NaOH_{CC} model showed that the power intensity (0.0038) had the most significant effect on the reducing sugar yield followed by pretreatment time (0.0468). Microwave power intensity displayed the greatest effect on the reducing sugar yield for all four pretreatment models. The significance of microwave power intensity can be attributed to the generation of thermal heat zones that accelerate the collision of ions, thus raising the temperature at shorter times with a high heating efficiency during lignocellulosic pretreatment (Sewsynker-Sukai and Gueguim Kana, 2018a, Alvira et al., 2010). Similarly, the influence of pretreatment time may be ascribed to an increase in the temperature, thus enhancing chemical changes in the lignocellulosic matrix (Lin et al., 2015). However, long pretreatment times speed up evaporation that can adversely influence dielectric properties and reduce pretreatment impact (Aguilar-Reynosa et al., 2017a). On the other hand, the significant impact of alkali concentration (Na₃PO₄.12H₂O and NaOH) can be accounted for by the cleavage of ester and glycosidic bonds, altering the structure of lignin, causing cellulose swelling, and partial depolymerization of hemicellulose (Brodeur et al., 2011; Qing et al., 2016b).

To display the developed pretreatment models' reliability, the coefficient of determination (R^2) value was used as a statistical index. R^2 values between 0.70 and 1 represent a model with high accuracy (Jugwanth et al., 2019). High R^2 values were achieved for the Na₃PO₄.12H₂O_{BB} (0.85), NaOH_{BB} (0.98), Na₃PO₄.12H₂O_{CC} (0.97) and NaOH_{CC} (0.84) models. This indicated that the developed models could account for 85 % (Na₃PO₄.12H₂O_{BB}), 98 % (NaOH_{BB}), 97 % (Na₃PO₄.12H₂O_{CC}) and 84 % (NaOH_{CC}) of the variation in the observed data. Therefore, all four pretreatment models displayed a good fit regarding the relationship between the pretreatment input parameters and the reducing sugar yield.

3.3. Effect of the pretreatment input parameters on the reducing sugar yield

Table 1 shows the reducing sugar yields (g/g) obtained for each experimental run from all four models. The Na₃PO₄.12H₂O_{BB} model displayed a maximum reducing sugar yield of 0.116 g/g under the median pretreatment conditions of 12.50 % Na₃PO₄.12H₂O, 500 W and 7.5 min (run 13). The other median pretreatment runs (12.50 % Na₃PO₄.12H₂O, 500 W, 7.5 min) displayed sugar yields of 0.095 g/g (run 1), 0.099 g/g (run 2), 0.107 g/g (run 3) and 0.059 g/g (run 5) for the Na₃PO₄.12H₂O_{BB} model. Conversely, the lowest reducing sugar yield (Na₃PO₄.12H₂O_{BB} = 0.001 g/g) was obtained with 20 % Na₃PO₄.12H₂O, 100 W and 7.5 min (run 4). On the other hand, the NaOH_{BB} model displayed a maximum reducing sugar yield of 0.220 g/g

with 20 % NaOH, heated at 500 W for 5 min (run 17). The lowest reducing sugar yield (NaOH_{BB} = 0.061 g/g) was obtained with 5.00 % NaOH, 100 W and 7.5 min (run 14). Alternatively, the NaOH_{BB} gave reducing sugar yields of 0.200 g/g (run 1), 0.195 g/g (run 2), 0.200 g/g (run 3), 0.195 g/g (run 5) and 0.208 g/g (run 13) under the median pretreatment conditions (12.50 % NaOH, 500 W, 7.5 min).

For the Na₃PO₄.12H₂O_{CC} model, a maximum reducing sugar yield of 0.463 g/g was observed with 20 % Na₃PO₄.12H₂O, heated at 900 W for 7.5 min (run 12). The lowest reducing sugar yield (0.001 g/g) was obtained for run 9 (12.50 % Na₃PO₄.12H₂O, 100 W and 10 min). The median pretreatment conditions for the Na₃PO₄.12H₂O_{CC} model (12.50 % Na₃PO₄.12H₂O, 500 W, 7.5 min) achieved sugar yields of 0.421 g/g (run 1), 0.359 g/g (run 2), 0.403 g/g (run 3), 0.397 g/g (run 5) and 0.312 g/g (run 13). Conversely, the NaOH_{CC} model displayed a maximum reducing sugar yield of 0.419 g/g for run 2 (12.50 % NaOH, 500 W, 7.5 min) and run 9 (12.50 % NaOH, 100 W, 10 min). The NaOH_{CC} demonstrated the lowest reducing sugar yield (0.290 g/g) for run 8 (5.00 % NaOH, 900 W and 7.5 min). The median pretreatment conditions for the aforementioned model (12.50 % NaOH, 500 W, 7.5 min) resulted in reducing sugar yields of 0.395 g/g (run 1), 0.419 g/g (run 2), 0.362 g/g (run 3), 0.406 g/g (run 5) and 0.372 g/g (run 13).

The interactions between the pretreatment input parameters and the reducing sugar yields are shown in Fig. 1A-H. Fig. 1A illustrates the interactive effects of power intensity and Na₃PO₄.12H₂O concentration on the reducing sugar yield (g/g), while the pretreatment time was maintained at its median ($Na_3PO_4.12H_2O_{BB}$). Simultaneous increases in the Na₃PO₄.12H₂O concentration (5-13 %) and microwave power intensity (100-500 W) increased the reducing sugar yield from 0.001 to 0.11 g/g from bamboo. During microwave heating, polarization influences the dipole-dipole molecules to align. The alignment of these molecules in the radiation field causes rapid heating of the hydrolysate enhancing the solubility of the substrate (Aguilar-Reynosa et al., 2017a). Furthermore, microwave heat accumulates within the material resulting in an improved heat transfer and generates a high-energy yield (Aguilar-Reynosa et al., 2017a). Correspondingly, the increase in sugar yield can be ascribed to the alkaline pretreatment. During pretreatment, alkaline chemicals solubilize amorphous regions (hemicellulose and lignin) resulting in the saponification of intermolecular ester bonds (Kim et al., 2016). Further increases in the Na₃PO₄.12H₂O concentration (>13 %) and power intensity (>500 W) resulted in a decrease in the reducing sugar yield (0.11-0.01 g/g) (Fig. 1A). The decrease in reducing sugars can be due to the elevated temperatures and alkaline conditions that may result in the formation of inhibitory compounds, which are toxic to cellulases, thus reducing the enzyme activity during hydrolysis reactions (Wang et al., 2010). Likewise, Qing et al. (2016b) observed no further increase in sugar yields beyond 11 % Na₃PO₄.12H₂O concentration using bamboo.

The influence of pretreatment time and power intensity on the reducing sugar yield whilst the Na₃PO₄.12H₂O concentration was the median value is depicted in Fig. 1B (Na₃PO₄.12H₂O_{BB}). A simultaneous increment in the pretreatment time (5-7 min) and power intensity (100–600 W) increased the reducing sugar yield from 0.01 to 0.10 g/g using bamboo. Microwave irradiation has a fast heat transfer, reduces time by 10-fold and decreases energy consumption (Aguilar-Reynosa et al., 2017a). The high heating efficiency of microwave irradiation can account for the improvement om sugar yield even at lower power intensities and pretreatment times. Alternatively, longer times (7.5-10 min) and a low power intensity (100 W) led to reducing sugar yields <0.014 g/g. Shorter times are generally preferred with higher power intensities during microwave pretreatment since the latter dictates the heating efficiency (Aguilar-Reynosa et al., 2017a). This may be attributed to homogenous heating of the hydrolysate resulting in a higher throughput and bioconversion rate at higher microwave power intensities (Maurya et al., 2015). Moreover, high temperatures due to the elevated power intensities, rapidly generate pressure within the absorbent biopolymers resulting in their rapid expansion (Aguilar-Reynosa



Fig. 1. Response surface graphs illustrating the interactive effects of the various input pretreatment parameters on the output reducing sugar yields: (A) power intensity and $Na_3PO_4.12H_2O$ concentration ($Na_3PO_4.12H_2O_{BB}$); (B) pretreatment time and power intensity ($Na_3PO_4.12H_2O_{BB}$); (C) power intensity and NaOH concentration ($NaOH_{BB}$); (D) pretreatment time and NaOH concentration ($NaOH_{BB}$); (E) power intensity and $Na_3PO_4.12H_2O$ concentration ($Na_3PO_4.12H_2O_{CC}$); (F) pretreatment time and NaOH concentration ($Na_3PO_4.12H_2O$ concentration ($Na_3PO_4.12H_2O_{CC}$); (G) pretreatment time and NaOH concentration ($NaOH_{CC}$); (H) pretreatment time and power intensity ($NaOH_{CC}$).

Table 4

Experimental	l runs with	the correspondin	g sugar	yields	obtained	for t	he alkaline	pretreatments of	n bar	iboo and	l corn	cobs.

Input parameters				Output				
		Denne internetien (MD)	Protocologica (min)	Reducing sugar yield from different substrates (g/g)				
Kun	Aikaine catalyst concentration (%)	Power Intensity (W)	Pretreatment time (min)	NaOH _{BB}	$Na_3PO_4.12H_2O_{BB}$	NaOH _{CC}	Na ₃ PO ₄ .12H ₂ O _{CC}	
1	12.50	500	7.50	0.200	0.095	0.395	0.421	
2	12.50	500	7.50	0.195	0.099	0.419	0.359	
3	12.50	500	7.50	0.200	0.107	0.362	0.403	
4	20.00	100	7.50	0.196	0.001	0.401	0.038	
5	12.50	500	7.50	0.195	0.059	0.406	0.397	
6	12.50	100	5.00	0.168	0.002	0.410	0.108	
7	5.00	500	5.00	0.124	0.031	0.416	0.385	
8	5.00	900	7.50	0.134	0.023	0.290	0.121	
9	12.50	100	10.00	0.169	0.002	0.419	0.001	
10	5.00	500	10.00	0.145	0.034	0.354	0.136	
11	12.50	900	10.00	0.215	0.066	0.320	0.410	
12	20.00	900	7.50	0.207	0.040	0.325	0.463	
13	12.50	500	7.50	0.208	0.116	0.372	0.312	
14	5.00	100	7.50	0.061	0.014	0.371	0.023	
15	20.00	500	10.00	0.195	0.076	0.359	0.412	
16	12.50	900	5.00	0.202	0.060	0.394	0.337	
17	20.00	500	5.00	0.220	0.074	0.386	0.401	

Footnote: g/g = gram reducing sugar/ gram dry weight substrate.

et al., 2017a) (Table 4).

In addition, minimum generation of inhibitors occur at shorter pretreatment times (Maurya et al., 2015). A similar trend was observed by Sewsynker-Sukai and Gueguim Kana (2018a) whereby an increase in power intensity above 700 W did not significantly influence the reducing sugar yield.

Fig. 1C depicts the interaction between power intensity and NaOH concentration on the reducing sugar yield from bamboo while pretreatment time was held at the centre point (NaOH_{BB}). Simultaneous increases in the NaOH concentration from 5 to 20 % and power intensity from 100 to 900 W enhanced the reducing sugar yield (0.061 g/g–0.220 g/g). During pretreatment, NaOH disassociates into sodium (Na⁺) and hydroxide (OH⁻) ions. An increase in OH⁻ ions increases the rate of hydrolysis resulting in a high sugar yield (Kim et al., 2016). Furthermore, sodium hydroxide cleaves the ether and ester bonds between lignin and hemicellulose and C—C bonds within lignin molecules (Kim et al., 2016). In addition, a high microwave power enhances collisions between anions and cations allowing for fast heating of the intracellular molecules, causing breakages of hydrogen bonds within the structural chain (Aguilar-Reynosa et al., 2017a).

On the other hand, Fig. 1D shows the influence of pretreatment time and NaOH concentration on the reducing sugar yield while power intensity was the median value (NaOHBB). Increases in the reducing sugar yield (0.110-0.220 g/g) were observed with a fixed pretreatment time of 5 min and an increase in the NaOH concentration (5–20 %) using bamboo. High alkaline concentrations enhance the degradation of lignocellulosic structures (Qing et al., 2016b). However, an increase in pretreatment time (>5 min) reduced the sugar yields (0.23-0.20 g/g). At longer pretreatment times, abundant heat is generated within the alkaline solution, resulting in a high rate of evaporation (Aguilar-Reynosa et al., 2017a). Dehydrated components within the reaction mixture cannot contend with granule expansion, leading to stress between the biopolymers inevitably causing their structures to collapse and rupture (Aguilar-Reynosa et al., 2017a) and may account for the lower sugar observed at longer pretreatment times. The study by Sewsynker-Sukai and Gueguim Kana (2018a) assessed the effects of pretreatment time and alkali salt concentration on corn cobs and recorded a sugar yield of 0.30 to 0.75 g/g when both parameters were simultaneously increased.

The interaction between power intensity and Na₃PO₄.12H₂O concentration on the reducing sugar yield from corn cobs, while pretreatment time was maintained at its midpoint is illustrated in Fig. 1E (Na₃PO₄.12H₂O_{CC}). Simultaneous increments in the Na₃PO₄.12H₂O concentration from 5 to 20 % and power intensity from 100 to 900 W

increased the reducing sugar yield (0.001 g/g–0.49 g/g). Alkalic salt has recently been used to enhance enzymatic hydrolysis and sugar yields due to its low corrosivity, ability to alter the lignin structure and increase the substrate porosity (Qing et al., 2016b). In addition, high microwave irradiation creates a magnetic field that facilitates the movement of ions back and forth (oscillation) enhancing substrate penetration (Aguilar-Reynosa et al., 2017a). Similarly, Qing et al. (2016b) reported that higher Na₃PO₄.12H₂O concentration and pretreatment times resulted in a higher reducing sugar yield using bamboo as a substrate.

Fig. 1F relates the interactive effects of pretreatment time and $Na_3PO_4.12H_2O$ concentration on the reducing sugar yield while power intensity was held at the centre point ($Na_3PO_4.12H_2O_{CC}$). The reducing sugar yield increased from 0.15 to 0.45 g/g using corn cobs with simultaneous increases in the pretreatment time (5–10 min) and $Na_3PO_4.12H_2O$ concentration (5–20 %). The high reducing sugar yield can be attributed to the highly concentrated alkaline salt which exhibits a high pH. The high alkaline pH is important for delignification by facilitating the oxidation of ethylene double bonds in the side chains of lignin moieties thus enhancing the exposure of cellulose and hemicellulose structures to enzymes (Geng et al., 2014). Furthermore, an increase in exposure time, facilitates the accumulation of heat within the solution, resulting in the effective disruption of lignin (Pu et al., 2013).

The influence of pretreatment time and NaOH concentration while power intensity was kept at the median value is depicted in Fig. 1G (NaOH_{CC}). Pretreatment times ranging from 5 to 6.25 min and NaOH concentrations from 16.25 to 20 % gave the maximum reducing sugar yields of 0.395 to 0.40 g/g using corn cobs. Similar to the effect of Na₃PO₄.12H₂O (Fig. 1F), a high alkaline concentration and pH for the NaOH aids the delignification of the lignocellulosic biomass due to fragmentation of recalcitrant lignin moieties. However, a further increase in the pretreatment time beyond 6.25 min drastically reduced the sugar yield from 0.40 to 0.345 g/g. Reduction in the reducing sugar yields at long exposure times (>6.25 min) may be attributed to the evaporation of water. Water is an important factor during pretreatment, since the moisture content determines the dielectric properties of the material (Aguilar-Reynosa et al., 2017a). Furthermore, water facilitates heating and determines the depth of penetration of the irradiation energy (Aguilar-Reynosa et al., 2017a). The study by Jia-Jia et al. (2017) pretreated bamboo with a power intensity of 440 W for 23 min and observed a decrease in the sugar yield from 0.76 to 0.432 g/g beyond 440 W.

Fig. 1H illustrates the interactive effects between pretreatment time

and power intensity on the reducing sugar yield from corn cobs whilst the NaOH concentration was held at its midpoint (NaOH_{CC}). A slight increase in the reducing sugar yield from 0.40 to 0.42 g/g was observed with increases in the pretreatment time from 5 to 10 min and a power intensity of 100 W. Similarly, Lin et al. (2015) studied the interactive effects of NaOH and microwave duration for the pretreatment of water hyacinth and found that an increase in time from 5–10 min increased the reducing sugar yield (0.013–0.028 g/g). Moreover, the present study displayed that power intensity beyond 100 W with NaOH led to a reduction in the sugar yield (0.40–0.35 g/g). High temperatures achieved at increased microwave power intensities coupled with strong alkaline concentrations may negatively affect the substrate by damaging lignocellulosic structures such as cellulose (Wang et al., 2010). As a result, low glucose yields are attained after enzymatic saccharification due to a decrease in cellulose recovery (Wang et al., 2010).

3.4. Experimental validation of the developed models

The four developed pretreatment models were validated and optimized to obtain a maximum reducing sugar yield (Table 5). Under optimal conditions, the Na₃PO₄.12H₂O_{BB,} NaOH_{BB,} Na₃PO₄.12H₂O_{CC,} and $NaOH_{CC}$ models predicted reducing sugar yields of 0.099, 0.220, 0.492 and 0.421 g/g respectively. Experimental sugar yields of 0.086, 0.225, 0.512 and 0.420 were obtained for the Na₃PO₄.12H₂O_{BB.} NaOH_{BB}, Na₃PO₄.12H₂O_{CC}, and NaOH_{CC} optimized pretreatments respectively. Slight deviations of 13 % (Na_3PO_4.12H_2O_{BB}), 2.22 %(NaOH_{BB}), 3.91 % (Na₃PO₄.12H₂O_{CC}), and 0.24 % (NaOH_{CC}) were recorded between the predicted and observed reducing sugar yields, illustrating the capability of the developed pretreatment models. In addition, the glucose yields achieved for the optimized pretreated experiments were 0.025 (Na₃PO₄.12H₂O_{BB}), 0.123 (NaOH_{BB}), 0.171 (Na₃PO₄.12H₂O_{CC}) and 0.294 g/g (NaOH_{CC}). Furthermore, the native substrate samples resulted in low glucose yields of 0.050 g/g (untreated corn cobs) and 0.022 g/g (untreated bamboo). The reducing sugar yields obtained in this study were compared to other pretreatment reports on bamboo and corn cobs.

The reducing sugar yields obtained in the present study was compared to previous alkaline pretreatment studies on corn cobs and bamboo. The study by Qing et al. (2016b) observed an 11-fold higher sugar yield obtained compared to the present optimized Na₃PO₄.12-H₂O_{BB} Likewise, Kuttiraja et al. (2013) recorded a 54 % higher sugar yield in comparison with the current optimized NaOH pretreatment for bamboo Similarly, Sewsynker-Sukai and Gueguim Kana (2018a) and (Boonsombuti et al., 2013) also achieved higher sugar yields from corn cobs. The higher sugar observed for the previous studies can be ascribed to the difference in lignocellulosic composition and pretreatment conditions. For instance, Kuttiraja et al. (2013), Boonsombuti et al. (2013) and Qing et al. (2016a) employed pretreatment times of 30 min, 30 min and 120 min respectively, that was significantly longer than the present study. Generally, a shorter pretreatment time is preferred since it reduces processes costs. Moreover, the studies by Kuttiraja et al. (2013) and Qing et al. (2016a) utilized steam and oven heating respectively for pretreatment. These heating mechanisms are known to be less efficient than microwave heating, thus the use of 20-fold (Qing et al., 2016a) and 5-fold (Kuttiraja et al., 2013) longer pretreatment times.

Furthermore, the present study displayed a significantly higher

reducing sugar yield for corn cobs compared to bamboo. Variations in the sugar yields obtained for these substrates may be attributed to differences in the lignocellulosic composition, which has shown to impact the efficiency of enzymatic hydrolysis (Suseela, 2019). The present optimization study demonstrated that Na₃PO₄.12H₂O exhibited a higher pretreatment efficiency than NaOH for the pretreatment of corn cobs with a 21.9 % higher sugar yield. This provides valuable knowledge since Na₃PO₄.12H₂O could potentially be a suitable replacement for expensive NaOH techniques in the near future (Qing et al., 2016a).

3.5. Effect of the alkaline pretreatments on the lignocellulosic structure

3.5.1. Compositional analysis

Lignocellulosic structures such as cellulose and hemicellulose polymers are attached to lignin through covalent and hydrogen bonding, thus making these structures highly recalcitrant to both chemical and biological degradation (Zabed et al., 2016). Therefore, unravelling of these structures to release cellulose and hemicellulose while removing lignin is imperative (Nasirpour and Mousavi, 2018). The compositional analysis data was determined for the untreated and optimally pretreated bamboo and corn cob substrates. The native (untreated) bamboo consisted of 54.33 % cellulose, 14.82 % hemicellulose and 25.64 % lignin. Optimally pretreated bamboo with NaOH contained 68.11 % cellulose, 11.57 % hemicellulose and 18.92 % lignin. Thus, a 25.36 % improvement in the cellulose content, 21.93 % hemicellulose solubilization and 26.21 % delignification were observed. Similarly, bamboo optimally pretreated with Na₃PO₄.12H₂O contained 67.78 % cellulose, 12.89 % hemicellulose and 19.22 % lignin. Therefore, a 24.76 % enhancement in the cellulose content, 13.02 % hemicellulose solubilization and 25.04 % delignification were achieved for the optimized Na₃PO₄.12H₂O bamboo sample. Strong alkali chemicals such as NaOH and Na3PO4.12H2O break the glycosidic and ester bonds in the cell wall matrix, therefore causing cellulose swelling and decrystallization, altering the structure of lignin, and partial solvation of hemicellulose (Brodeur et al., 2011). More specifically, NaOH targets specific bonds such as the hydrolysis of the ester bonds between the ferulic acid and hemicellulose causing lignin degradation and lignocellulose particle swelling. The study by Yang et al. (2019) pretreated bamboo with NaOH and achieved a high cellulose and xylan conversion rate of 38.2 % and 39.8 % respectively. On the other hand, alkalic salts such as $Na_3PO_4.12H_2O$ contain nucleophilic species that facilitate the cleavage of phenolic *β*-aryl ether bonds of lignin resulting in delignification (Gu et al., 2019). Correspondingly, the weak bases enhance the removal of acetyl groups from xylan structures, which improves cellulose digestibility during enzymatic hydrolysis (Gu et al., 2019). Likewise, Qing et al. (2016b) pretreated bamboo shoot shell using Na₃PO₄.12H₂O combined with H₂O₂ and observed an increase in the cellulose content, hemicellulose removal and delignification of 86.1 %, 37.8 % and 85.1 % respectively.

The native (untreated) corn cobs consisted of 40.31 % cellulose, 43.50 % hemicellulose and 9.31 % lignin while the optimized NaOH pretreated corn cobs contained 72.48 % cellulose, 12.73 % hemicellulose and 10.13 % lignin. Therefore, a cellulose improvement and hemicellulose solubilization of 79.81 % and 70.74 % respectively was recorded. You et al. (2019) developed an ultrasound-NaOH pretreatment on corn cobs and achieved a high cellulose improvement and lignin conversion of 47 % and 60 % respectively. Alternatively, the

Table	5
-------	---

Validation of optimized	conditions for	the alkaline	pretreatment of	n bamboo and	corn cobs.
1			1		

Model	Input parameters	Reducing sugar yield (g/g)			
	Alkaline catalyst concentration (%)	Power intensity (W)	Pretreatment time (min)	Predicted	Observed
Na ₃ PO ₄ .12H ₂ O _{BB}	14	600	7	0.099	0.086 ± 0.001
NaOH _{BB}	18	600	6	0.220	0.225 ± 0.0005
Na ₃ PO ₄ .12H ₂ O _{CC}	18	600	8	0.492	0.512 ± 0.0025
NaOH _{CC}	12	500	5	0.421	0.420 ± 0.0065

optimally pretreated Na₃PO₄.12H₂O corn cobs consisted of 56.08 % cellulose, 30.74 % hemicellulose and 9.91 % lignin. This translated to a cellulose improvement and hemicellulose solubilization of 39.12 % and 29.33 % respectively. A previous study by Oing et al. (2016a) recorded improvements of 74.26 % in the cellulose recovery, 36.24 % hemicellulose removal and 62.20 % delignification of corn stover after using Na₃PO₄.12H₂O based pretreatments. Nevertheless, the present study recorded an increase in the lignin content of 8.81 % and 6.44 % for both the NaOH and Na₃PO₄.12H₂O pretreated corn cobs respectively. The increase in lignin content can be attributed to the formation of Klason lignin. Klason lignin (pseudo-lignin), is a product formed from cellulose and hemicellulose when biomass is pretreated with an acid or base at extreme thermal conditions (Shinde et al., 2018). Moreover, the present study only displayed pseudo-lignin formation from corn cobs whereas the bamboo demonstrated efficient delignification. The monomeric units that make up lignin vary in the cell walls of different lignocellulosic biomasses and can account for the pseudo-lignin formation in the corn cobs without its occurrence for the bamboo substrate. Therefore, the proportion of lignin monomers is an important factor deciding its susceptibility to degradation during chemical pretreatment (Suseela, 2019).

3.5.2. Visual observations using scanning electron microscopy (SEM)

The untreated and optimally pretreated bamboo and corn cobs surfaces were visually analyzed using SEM. The native (untreated) substrates exhibited a smooth and continuous surface area whereas the optimized pretreated substrates demonstrated the disintegration of the lignocellulosic structures. The substrates optimally pretreated with NaOH displayed a high degree of damage to its surface area compared to Na₃PO₄.12H₂O pretreatments. During pretreatment, hydrogen bonds are destroyed resulting in the leakage of lignin moieties (Pu et al., 2013). An increase in porosity and surface area enhances the availability of cellulose and hemicellulose to enzymes during enzymatic hydrolysis (Nasirpour and Mousavi, 2018; Qing et al., 2016a). These results were consistent with previous studies on corn stover (Qing et al., 2016a), bamboo (Qing et al., 2016b), corn cobs (Boonsombuti et al., 2013) and bamboo (Kuttiraja et al., 2013) pretreated with Na₃PO₄.12H₂O (Qing et al., 2016b) and NaOH (Boonsombuti et al., 2013; Kuttiraja et al., 2013). The corn cob substrate displayed a more disorganized morphology characterized by the exposure of fibres and loosening of the fibrous network. This illustrated that the corn cobs were prone to a higher level of fragmentation compared to the bamboo substrate. The high concentration of lignin that is present within plant cell walls hold the fibres of polysaccharides tightly packed together contributing to its recalcitrant nature (Pu et al., 2013) and may account for the more in tact surface structure of the bamboo substrate.

3.5.3. Fourier transform infrared (FTIR) analysis

FTIR spectral analysis was performed and showed the chemical structural changes of cellulose, hemicellulose and lignin in the wavelength range of 400 cm^{-1} to 3800 cm^{-1} . The absorption band peaks were shown to be stronger for the optimized pretreated samples as compared to the native samples for both corn cobs and bamboo. All the pretreated samples displayed a similar banding pattern. For instance, the absorptive band peaks at 900 cm⁻¹ corresponds to the breakage of the β -1,4 glycosidic linkages in cellulose. The band peak at 1034 cm⁻¹ is ascribed to the aromatic C-H and C-O deformation (Boukir et al., 2019). The absorption peak at 1166 cm⁻¹ represents the vibrations of C-O-C associated with the pyranose ring skeleton and C-O-C asymmetric stretching in cellulose and hemicellulose (Boukir et al., 2019). Two absorption bands were detected at 1371 and 1431 cm⁻¹ and occur in the range between 1300-1500 cm⁻¹ illustrating the C-H-bending vibrations of cellulose (Li et al., 2018). Conversely, the band peaks observed at 1656 cm^{-1} demonstrate in-plane C=C aromatic vibration in the aromatic ring within lignin (Li et al., 2018). Additionally, C-H stretching and O-H stretching vibrations of intramolecular hydrogen within

cellulose were indicated by absorptive peaks at 2901 cm⁻¹ and 3323 cm⁻¹, respectively (Li et al., 2018). Previous reports on strong alkali (NaOH) and alkalic salt (Na₃PO₄.12H₂O) pretreatments on bamboo (Kuttiraja et al., 2013; Qing et al., 2016b) and corn cobs (Boonsombuti et al., 2013; Sewsynker-Sukai and Gueguim Kana, 2018a) respectively displayed similar findings.

3.6. Recyclability of spent alkaline pretreatment liquid

The black liquor obtained after the optimized NaOH and Na₃PO₄.12H₂O pretreatments were collected and subsequently used for the recyclability studies on corn cobs (Fig. 2A-B). Pretreatment cycle 0 corresponds to the optimized experiments that were performed with the original NaOH and Na3PO4.12H2O chemicals and resulted in reducing sugar yields of 0.42 g/g and 0.512 g/g respectively. The first, second and third pretreatment cycles displayed reducing sugar yields of 0.437 g/g, 0.352 g/g and 0.333 g/g for the NaOH black liquor recyclability. Alternatively, the Na₃PO₄.12H₂O black liquor demonstrated sugar yields of 0.425 g/g, 0.302 g/g and 0.191 g/g for the first, second and third pretreatment cycles. The variations observed in the sugar yields between the different pretreatment cycles for the NaOH and Na₃PO_{4.}12H₂O black liquor recyclability may be linked with the chemical changes that occur in the hydrolysate after each consecutive cycle. For instance, the reducing sugar yield increased in the first cycle (4.05 %) whereas a higher glucose yield (10.33 %) was recorded for the second cycle using the NaOH black liquor. Volatile compounds such as phenolics, aliphatic acids, amines and amides are generally released when ether and covalent bonds in lignin macromolecules are fragmented during alkaline pretreatment (Rorke et al., 2016; Sewsynker-Sukai and Gueguim Kana, 2018a). The improved reducing sugar and glucose yields after recycling the black liquor could be explained by the accumulation of these volatile compounds that may assist with hemicellulose solubilization. The studies by Rorke et al. (2016) and Sewsynker-Sukai and Gueguim Kana (2018a) detected similar volatile compounds when sorghum leaves and corn cobs respectively, were pretreated with microwave-assisted alkaline catalysts.

The first, second and third pretreatment cycles for the NaOH black liquor revealed pH values of 12.84, 12.84 and 12.64 respectively. On the other hand, the Na₃PO_{4.}12H₂O black liquor recyclability illustrated a sharp decrease in the pH values for the first (pH 11.94), second (pH 11.30) and third (pH 10.83) pretreatment cycles. Alkaline pH has shown to be critical for delignification by facilitating the oxidation of ethylene double bonds in the side chains of lignin phenylpropane units (Geng et al., 2014). The pH change observed for each consecutive recyclability cycle may be attributed to the formation of inhibitor compounds such as formic acid derived from furan derivatives (furfural), carboxylic acids produced from the acetyl groups in hemicelluloses, and lignin-derived phenolics (Kim, 2018). These inhibitors may be acidic and cause a reduction in the pH (Kim, 2018). The study by Mudassar et al. (2014) observed a decrease in pH during second stage pretreatment of sawdust suggesting that alkali was consumed by carboxylic acids and carbon dioxide in the reaction mixture. Likewise, Qing et al. (2016b) also observed a decline in the pH after recyclability of alkalic salt black liquor on bamboo. Mudassar et al. (2014) reported that alkaline liquor exhibits a high performance on lignin removal, enhancing the digestibility of cellulose and hemicellulose.

The present study revealed that both NaOH and Na₃PO₄.12H₂O black liquor could be recycled up to three times. Although NaOH produced a lower reducing sugar yield compared to Na₃PO₄.12H₂O on corn cobs, it released a higher glucose yield before and after recycling. The higher reducing sugar yield may be due to the specificity of the DNS method. DNS is an aromatic compound that is used to detect the presence of free carbonyl groups of reducing sugars (monosaccharides and disaccharides) and is used to quantify α -amylase activity (Keharom et al., 2016). However, the detection range of the DNS method is limited to a specific concentration, and degree of polymerization of the



Fig. 2. Chemical recyclability of (A) NaOH and (B) Na₃PO₄.12H₂O for the pretreatment of corn cobs.

carbohydrates (Rivers et al., 1984). Another downfall is that by-products produced as a result of pretreatment processes may interfere with the aromatic compound by supplying reactive reducing groups (aldehydes and ketones) resulting in significant quantitative errors (Rivers et al., 1984). The aforementioned limitations of the DNS method may have led to the overestimation of reducing sugar yields, therefore displaying higher sugar compared to the glucose yields. Even though the DNS method displays these limitations for sugar quantification, it has been extensively used and is a well-established protocol for sugar analysis (Gonçalves et al., 2010).

3.7. Kinetic assessment

3.7.1. Angel yeast cell growth using the logistic model

The growth of Angel yeast over time during the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes are illustrated in Fig. 3A. The lag phase of yeast growth was observed during the first 2 h for both SSF processes. During the lag phase, yeast cells are metabolically active and synthesize proteins needed for growth within the fermentation medium (Ginovart et al., 2018). The NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes demonstrated an exponential increase in the Angel yeast cells between 2–18 h. During SSF processes, the enzymatic conversion of cellulose to glucose occurs at a rapid rate. A high glucose utilization was observed by the Angel yeast cells (>70 %) and corresponded with the exponential phase (Fig. 3B). The decline in cell growth above 18 h for the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes can be attributed to the depletion of sugars, a decrease in enzymatic activity, or glucose-induced repression of respiration also known as the Crabtree-effect (De Deken, 1966).

The Angel yeast growth data obtained from the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes fitted the logistic model with high correlation coefficient (R²) values of 0.973 and 0.988 respectively. The observed kinetic parameters for the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC}(SSF) processes were compared to previous studies. The NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC}(SSF) processes demonstrated maximum specific growth rate (μ_{max}) values of 0 0.175 h⁻¹ and 0.049 h⁻¹ respectively. Previous kinetic studies on sugarcane bagasse (Jugwanth et al., 2019), potato peel waste (Chohan et al., 2020), and corn cobs (Sewsynker-Sukai and Gueguim Kana, 2018b) produced similar μ_{max} values of 0.150 h⁻¹, 0.195 h⁻¹ and 0.216 h⁻¹ respectively. The initial (X₀) and maximum (X_{max}) biomass cell concentrations are useful indicators for biomass synthesis and product formation. Kinetic data obtained for the NaOH_{CC(SSF)} and

Na₃PO₄.12H₂O_{CC(SSF)} processes gave X₀ values of 0.548 g/L and 0.501 g/L respectively. Also X_{max} values of 1.817 g/L and 5.579 g/L were obtained for the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes respectively. The study by Sewsynker-Sukai and Gueguim Kana (2018b) recorded X₀ and X_{max} values of 0.556 g/L and 3.650 g/L respectively from corn cobs pretreated with Na₃PO₄.12H₂O followed by H₂SO₄. Likewise, Jugwanth et al. (2019) observed X₀ and X_{max} values of 0.160 g/L and 2.580 g/L respectively from sequential ZnCl₂-NaOH pretreated sugarcane bagasse. Variables such as yeast strain, substrate type, composition and pretreatment, in addition to fermentation conditions such as pH, temperature, solid and enzyme loading may contribute to the discrepancies observed between the different studies.

3.7.2. Kinetic modelling of bioethanol production using the modified Gompertz model

Bioethanol formation during the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC} (SSF) processes was recorded (Fig. 3C). Exponential production of bioethanol occurred for both processes from 0 h to 18 h. For the Na₃PO₄.12H₂O_{CC(SSF)} and NaOH_{CC(SSF)} processes, bioethanol increased from 0 to 6.05 g/L and 0-16.71 g/L respectively, with the latter being the highest bioethanol produced in this study after 18 h. Additionally, the highest ethanol produced from both processes after 18 h coincides with the exponential phase of yeast cell growth (Fig. 3A) and a high glucose utilization (>95 %) (Fig. 3B). During the exponential phase, a sufficient amount of glucose was produced and consumed in order to sustain Angel yeast metabolism, resulting in the formation of primary metabolites such as bioethanol. Similar to the low ethanol concentration obtained for the Na₃PO₄.12H₂O_{CC(SSF)} process, Qing et al. (2016b) produced a maximum ethanol concentration of 8.6 g/L from corn stover pretreated with Na₃PO₄.12H₂O. The lower ethanol concentration observed for the Na₃PO₄.12H₂O_{CC(SSF)} process compared to the NaOH_{CC} (SSF) system may be ascribed to the reduced glucose generated from the alkalic salt. After 18 h, the ethanol concentration began to decline for both processes, which corresponds to the stationary phase of growth.

Bioethanol production data obtained from the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes were used to fit the modified Gompertz model with high correlation coefficient (R²) values of 0.96 and 0.933 respectively. The observed bioethanol kinetic parameters in this study were compared to previous studies. A maximum bioethanol concentration (P_m) of 16.928 g/L and 5.286 g/L were obtained for the NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes respectively. Similarly,



Fig. 3. Angel yeast biomass growth (A), glucose utilisation (B), and bioethanol production (C) during the SSF processes using the optimized pretreated corn cobs.

Aguilar-Reynosa et al. (2017b) comparatively investigated a conduction-convection heating and microwave heating processing pretreatment for bioethanol production using corn stover. The aforementioned authors obtained a maximum bioethanol concentration of 33.8 g/L and a 92 % conversion rate using microwave heating. Previous kinetic studies on acid-pretreated potato peel waste (Chohan et al., 2020), acid-pretreated sorghum leaves (Rorke and Gueguim Kana, 2017), and salt-alkali pretreated sugarcane bagasse (Jugwanth et al., 2019) produced similar P_m values of 15.48 g/L, 17.15 g/L, and 3.12 g/L respectively. The NaOH_{CC(SSF)} and Na₃PO₄.12H₂O_{CC(SSF)} processes demonstrated maximum bioethanol production rate (r _{p,m}) values of 1.246 g/L/h and 1.364 g/L/h respectively. This indicated that 1.246 g/L

and 1.364 g/L of ethanol could be produced by Angel yeast every 60 min. The studies by Sewsynker-Sukai and Gueguim Kana (2018b) and Chohan et al. (2020) observed similar maximum bioethanol production rates of 2.39 g/L/h and 1.513 g/L/h from corn cobs and potato peel waste respectively. The lag time (t_L) for bioethanol formation was the shortest (0 h) as compared to other studies. Therefore, a shorter time was required for the yeast cells to adapt to the fermentation medium resulting in the production of bioethanol at the onset of the SSF process. Likewise, the studies by Wang et al. (2013) and Jugwanth et al. (2019) reported short lag times of 0 h and 0.97 h respectively. On the other hand, Rorke and Gueguim Kana (2017) and Chohan et al. (2020) encountered extensively longer lag times of 6.31 h and 4.658 h respectively. The kinetic data from the present study demonstrates major knowledge enhancements for the potential of lignocellulosic bioethanol production using SSF processes.

4. Conclusion

From the developed pretreatments on different lignocellulosic substrates, the optimized Na₃PO₄.12H₂O pretreatment on corn cobs resulted in the highest reducing sugar yield (0.512 g/g). On the other hand, recyclability studies showed a 16 % higher average sugar yield for the NaOH-based black liquor compared to the Na₃PO₄.12H₂O hydrolysate over three pretreatment cycles. Interestingly, kinetic data revealed a 72 % and 69 % higher maximum specific growth rate and maximum potential bioethanol concentration respectively for the bioprocess using NaOH pretreated corn cobs. The findings from this study provides valuable insights on newly emerging alkaline chemicals such as Na₃PO₄.12H₂O compared to the well-known NaOH for lignocellulosic pretreatment, hydrolysate recyclability and bioethanol production. While Na₃PO₄.12H₂O represents a cheaper alternative to the gold standard NaOH method, a long trajectory towards achieving competitive lignocellulosic bioprocess yields is foreseeable. Nevertheless, recyclability of the hydrolysate from both chemicals demonstrated the potential to minimize pretreatment costs associated with the usage of chemicals and fresh water in lignocellulosic biorefineries.

CRediT authorship contribution statement

Milesh Laltha: Conceptualization, Methodology, Software, Validation, Formal analysis, Investigation, Data curation, Writing - original draft, Writing - review & editing, Visualization, Project administration, Funding acquisition. Y. Sewsynker-Sukai: Conceptualization, Methodology, Software, Validation, Investigation, Data curation, Writing review & editing, Visualization, Project administration, Resources, Supervision, Funding acquisition. Gueguim Kana E.B.: Software, Validation, Data curation, Writing - review & editing, Resources, Supervision.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

The authors thankfully acknowledge the National Research Foundation (NRF) of South Africa (Grant numbers 122341 and 129556) for financially supporting this research. Opinions expressed and conclusions arrived at, are those of the author and are not necessarily attributed to the NRF.

Appendix A. Supplementary data

Supplementary material related to this article can be found, in the online version, at doi:https://doi.org/10.1016/j.indcrop.2021.114166.

References

- Aguilar-Reynosa, A., Romaní, A., Ma Rodríguez-Jasso, R., Aguilar, C.N., Garrote, G., Ruiz, H.A., 2017a. Microwave heating processing as alternative of pretreatment in econd-generation biorefinery: an overview. Energy Convers. Manage. 136, 50-65.
- Aguilar-Revnosa, A., Romaní, A., Rodríguez-Jasso, R.M., Aguilar, C.N., Garrote, G., Ruiz, H.A., 2017b. Comparison of microwave and conduction-convection heating autohydrolysis pretreatment for bioethanol production. Bioresour. Technol. 243, 273-283
- Boonsombuti, A., Luengnaruemitchai, A., Wongkasemjit, S., 2013. Enhancement of enzymatic hydrolysis of corncob by microwave-assisted alkali pretreatment and its effect in morphology. Cellulose 20, 1957–1966.
- Boukir, A., Fellak, S., Doumenq, P., 2019. Structural characterization of Argania spinosa Moroccan wooden artifacts during natural degradation progress using infrared spectroscopy (ATR-FTIR) and X-Ray diffraction (XRD). Heliyon 5, 1-9.
- Brodeur, G., Yau, E., Badal, K., Collier, J., Ramachandran, K.B., Ramakrishnan, S., 2011. Chemical and physicochemical pretreatment of lignocellulosic biomass: a review. Enzyme Res. 2011, 1–17.
- Chaowana, P., 2013. Bamboo: an alternative raw material for wood and wood-based composites. J. Mater. Sci. Res. 2, 90-102.
- Chohan, N.A., Aruwajoye, G.S., Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2020. Valorisation of potato peel wastes for bioethanol production using simultaneous saccharification and fermentation: process optimization and kinetic assessment. Renew. Energy 146, 1031–1040.
- Czajkowski, Ł., Wojcieszak, D., Olek, W., Przybył, J., 2019. Thermal properties of fractions of corn stover. Constr. Build. Mater. 210, 709-712.
- De Deken, R.H., 1966. The crabtree effect: a regulatory system in yeast. J. Gen. Microbiol. 44, 149-156.
- Geng, W., Huang, T., Jin, Y., Song, J., Chang, H., Jameel, H., 2014. Comparison of sodium carbonate-oxygen and sodium hydroxide-oxygen pretreatments on the chemical composition and enzymatic saccharification of wheat straw. Bioresour. Technol. 161, 63-68.
- Ginovart, M., Carbó, R., Blanco, M., Portell, X., 2018. Digital image analysis of yeast single cells growing in two different oxygen concentrations to analyze the population growth and to assist individual-based modeling. Front. Microbiol. 8, 1-21. Gonçalves, C., Rodriguez-Jasso, R.M., Gomes, N., Teixeira, J.A., Belo, I., 2010.
- Adaptation of dinitrosalicylic acid method to microtiter plates. Anal. Methods 2, 2046-2048.
- Gonçalves, F.A., Ruiz, H.A., Silvino dos Santos, E., Teixeira, J.A., de Macedo, G.R., 2016. Bioethanol production by Saccharomyces cerevisiae, Pichia stipitis and Zymomonas mobilis from delignified coconut fibre mature and lignin extraction according to biorefinery concept. Renew. Energy 94, 353-365.
- Gong, Y., Fu, Z., Liu, M., Dai, Y., Lin, J., Liu, Z., 2019. Combined alkali pretreatment for enhanced enzymatic saccharification of sugarcane leaf. Bioresour. Technol. Rep. 7, 1-6.
- Gu, T., Wang, B., Zhang, Z., Wang, Z., Chong, G., Ma, C., Tang, Y.J., He, Y., 2019. Sequential pretreatment of bamboo shoot shell and biosynthesis of ethyl (R)-4 chloro-3-hydroxybutanoate in aqueous-butyl acetate media. Process. Biochem. 80, 112-118.
- Jia-Jia, F., Shen, S., Jun-Lu, D., Chen, T., Xiao-Ying, D., Hong-Bo, W., Wei-Dong, G., 2017. Microwave heating a potential pretreating method for bamboo fiber extraction. Therm. Sci. 21, 1695-1699.
- Jugwanth, Y., Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2019. Valorization of sugarcane bagasse for bioethanol production through simultaneous saccharification and fermentation: optimization and kinetic studies. Fuel 1, 1-10.
- Keharom, S., Mahachai, R., Chanthai, S., 2016. The optimization study of α -amylase activity based on central composite design-response surface methodology by dinitrosalicylic acid method. Int. Food Res. J. 23, 10-17.
- Kim, D.J.M., 2018. Physico-chemical conversion of lignocellulose: inhibitor effects and detoxification strategies: a mini review. Molecules 23, 1-21.
- Kim, J.S., Lee, Y.Y., Kim, T.H., 2016. A review on alkaline pretreatment technology for bioconversion of lignocellulosic biomass. Bioresour. Technol. 199, 42-48.
- Kumari, D., Singh, R., 2018. Pretreatment of lignocellulosic wastes for biofuel production: a critical review. Renew. Sustain. Energy Rev. 90, 877-891.
- Kuttiraja, M., Sindhu, R., Varghese, P.E., Sandhya, S.V., Binod, P., Vani, S., Pandey, A., Sukumaran, R.K., 2013. Bioethanol production from bamboo (Dendrocalamus sp.) process waste. Biomass Bioenergy 59, 142–150. Li, X., Wei, Y., Xu, J., Xu, N., He, Y., 2018. Quantitative visualization of lignocellulose
- components in transverse sections of moso bamboo based on FTIR macro- and microspectroscopy coupled with chemometrics. Biotechnol. Biofuels 11, 263.
- Lin, R., Cheng, J., Song, W., Ding, L., Xie, B., Zhou, J., Cen, K., 2015. Characterisation of water hyacinth with microwave-heated alkali pretreatment for enhanced enzymatic digestibility and hydrogen/methane fermentation. Bioresour. Technol. 182, 1-7.
- Maurva, D.P., Singla, A., Negi, S., 2015, An overview of key pretreatment processes for biological conversion of lignocellulosic biomass to bioethanol. 3 Biotech 5, 597-609.

- Miller, G.L., 1959. Use of dinitrosalicylic acid reagent for determination of reducing sugar. Anal. Chem. 31, 426-428.
- Mithra, M.G., Jeeva, M.L., Sajeev, M.S., Padmaja, G., 2018. Comparison of ethanol yield from pretreated lignocellulo-starch biomass under fed-batch SHF or SSF mode Heliyon 4, 1-31.
- Mudassar, H., Melin, K., Koskinen, J., 2014. Production of carboxylic acids from alkaline pretreatment byproduct of softwood. Cell. Chem. Technol. 48, 835-842.
- Myers, R.H., Montgomery, D.C., Anderson-Cook, C.M., 1995. Response Surface Methodology: Process and Product Optimization Using Designed Experiments, 4. Wiley, pp. 1-15.
- Nasirpour, N., Mousavi, S.M., 2018. RSM based optimization of PEG assisted ionic liquid pretreatment of sugarcane bagasse for enhanced bioethanol production: effect of process parameters. Biomass Bioenergy 116, 89-98.
- Pu, Y., Hu, F., Huang, F., Davison, B.H., Ragauskas, A.J., 2013. Assessing the molecular structure basis for biomass recalcitrance during dilute acid and hydrothermal pretreatments. Biotechnol. Biofuels 6, 15.
- Qing, Q., Zhou, L., Guo, Q., Huang, M., He, Y., Wang, L., Zhang, Y., 2016a. A combined sodium phosphate and sodium sulfide pretreatment for enhanced enzymatic digestibility and delignification of corn stover. Bioresour. Technol. 218, 209-216.
- Qing, Q., Zhou, L., Huang, M., Guo, Q., He, Y., Wang, L., Zhang, Y., 2016b. Improving enzymatic saccharification of bamboo shoot shell by alkalic salt pretreatment with H2O2. Bioresour. Technol. 201, 230-236.
- Rivers, D.B., Gracheck, S.J., Woodford, L.C., Emert, G.H., 1984. Limitations of the DNS assay for reducing sugars from saccharified lignocellulosics. Biotechnol. Bioeng. 26, 800-802.
- Rorke, D., Gueguim Kana, E.B., 2017. Kinetics of bioethanol production from waste sorghum leaves using Saccharomyces cerevisiae BY4743. Fermentation 3, 1–10.
- Rorke, D., Suinyuy, T., Gueguim Kana, E.B., 2016. Microwave-assisted chemical pretreatment of waste sorghum leaves: process optimization and development of an intelligent model for determination of volatile compound fractions. Bioresour. Technol. 224, 590-600.
- Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2018a. Microwave-assisted alkalic salt pretreatment of corn cob wastes: process optimization for improved sugar recovery. Ind. Crops Prod. 125, 284–292.
- Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2018b. Simultaneous saccharification and bioethanol production from corn cobs: process optimization and kinetic studies. Bioresour. Technol. 262, 32–41.
- Shinde, S.D., Meng, X., Kumar, R., Ragauskas, A.J., 2018. Recent advances in understanding the pseudo-lignin formation in a lignocellulosic biorefinery. Green Chem. 20, 2192–2205.
- Srimachai, T., Nuithitikul, K., O-Thong, S., Kongjan, P., Panpong, K., 2015. Optimization and kinetic modeling of ethanol production from Oil Palm Frond Juice in batch fermentation. Energy Procedia 79, 111–118. Suseela, V., 2019. Potential Roles of Plant Biochemistry in Mediating Ecosystem
- Responses to Warming and Drought, 4. Academic Press, pp. 103-124.
- Tagliapietra, S., 2019. The impact of the global energy transition on MENA oil and gas producers. Energy Strategy Rev. 26, 1-6.
- Vergara, P., García-Ochoa, F., Ladero, M., Gutiérrez, S., Villar, J.C., 2019. Liquor re-use strategy in lignocellulosic biomass fractionation with ethanol-water mixtures. Bioresour. Technol. 280, 396-403.
- Wang, Z., Keshwani, D.R., Redding, A.P., Cheng, J.J., 2010. Sodium hydroxide pretreatment and enzymatic hydrolysis of coastal Bermuda grass. Bioresour Technol. 101, 3583–3585.
- Wang, L., Luo, Z., Shahbazi, A., 2013. Optimization of simultaneous saccharification and fermentation for the production of ethanol from sweet sorghum (Sorghum bicolor) bagasse using response surface methodology. Ind. Crops Prod. 42, 280-291.
- Waqas, M., Naser, N., Sarathy, M., Morganti, K., Al-Qurashi, K., Johansson, B., 2016. Blending octane number of ethanol in HCCI, SI and CI combustion modes. SAE Int. 9, 660-682
- Yang, H., Shi, Z., Xu, G., Qin, Y., Deng, J., Yang, J., 2019. Bioethanol production from bamboo with alkali-catalyzed liquid hot water pretreatment. Bioresour. Technol. 274, 261-266.
- You, Z., Pan, S.Y., Sun, N., Kim, H., Chiang, P.C., 2019. Enhanced corn-stover fermentation for biogas production by NaOH pretreatment with CaO additive and ultrasound. Appl. Sci. 9, 1-12.
- Yu, J., Xu, Z., Liu, L., Chen, S., Wang, S., Jin, M., 2019. Process integration for ethanol production from corn and corn stover as mixed substrates. Bioresour. Technol. 279, . 10–16.
- Zabed, H., Sahu, J.N., Boyce, A.N., Faruq, G., 2016. Fuel ethanol production from lignocellulosic biomass: an overview on feedstocks and technological approaches. Renew. Sustain. Energy Rev. 66, 751-774.
- Zhu, W., Theliander, H., 2015. Precipitation of lignin from softwood black liquor: an investigation of the equilibrium and molecular properties of lignin, J. BioResour, 1, 1696-1714.





Fig. S1. SEM micrographs (X500): (A) untreated bamboo; (B) optimally pretreated bamboo (NaOH); (C) optimally pretreated bamboo (Na₃PO₄.12H₂O); (D) untreated corn cobs; (E) optimally pretreated corn cobs (NaOH); (F) optimally pretreated corn cobs (Na₃PO₄.12H₂O).



Fig. S2. FTIR spectra of untreated and the optimized NaOH and Na₃PO₄.12H₂O pretreated bamboo and corn cobs.

Table S1. Analysis of variance of the developed model of bamboo pretreated with

Na3PO4.12H2OBE

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.020	9	2.209×10 ⁻³	4.49	0.0302 Significant
A-Alkaline concentration	9.997×10 ⁻⁴	1	9.997×10 ⁻⁴	2.03	0.1971
B-Power intensity	3.608×10 ⁻³	1	3.608×10 ⁻³	7.33	0.0303
C-Pretreatment time	1.503×10 ⁻⁵	1	1.503×10 ⁻⁵	0.031	0.8662
AB	2.317×10 ⁻⁴	1	2.317×10 ⁻⁴	0.47	0.5147
AC	9.022×10 ⁻⁷	1	9.022×10 ⁻⁷	1.833×10 ⁻³	0.9670
BC	7.983×10 ⁻⁶	1	7.983×10 ⁻⁶	0.016	0.9022
A^2	3.099×10 ⁻³	1	3.099×10 ⁻³	6.30	0.0404
B^2	9.858×10 ⁻³	1	9.858×10 ⁻³	20.03	0.0029
C^2	8.531×10 ⁻⁴	1	8.531×10 ⁻⁴	1.73	0.2294
Lack-of-fit	1.563×10 ⁻³	3	5.211×10 ⁻⁴	1.11	0.4438 not significant

Footnote: df=degrees of freedom, F-value=Fisher-Snedecor distribution value, P-

value=probability value.

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.027	9	2.990×10 ⁻³	31.64	< 0.0001 Significant
A-Alkaline concentration	0.016	1	0.016	166.52	< 0.0001
B-Power intensity	3.382×10 ⁻³	1	3.382×10 ⁻³	35.79	0.0006
C-Pretreatment time	1.312×10 ⁻⁵	1	1.312×10 ⁻⁵	0.14	0.7205
AB	9.622×10 ⁻⁴	1	9.622×10 ⁻⁴	10.18	0.0152
AC	5.230×10 ⁻⁴	1	5.230×10 ⁻⁴	5.53	0.0509
BC	3.499×10 ⁻⁵	1	3.499×10 ⁻⁵	0.37	0.5620
A^2	4.874×10 ⁻³	1	4.874×10 ⁻³	51.59	0.0002
B^2	1.101×10 ⁻³	1	1.101×10 ⁻³	11.65	0.0112
C^2	1.084×10 ⁻⁴	1	1.084×10 ⁻⁴	1.15	0.3196
Lack-of-fit	5.536×10 ⁻⁴	3	1.845×10 ⁻⁴	6.85	0.0470 significant

Table S2. Analysis of variance of the developed model of bamboo pretreated with NaOHBB

Footnote: df=degrees of freedom, F-value=Fisher-Snedecor distribution value, P-

value=probability value.

Table S3. Analysis of variance of the developed model of corn cobs pretreated with Na₃PO₄.12H₂O_{CC}

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.42	9	0.046	21.58	0.0003 Significant
A-Alkaline concentration	0.053	1	0.053	24.68	0.0016
B-Power intensity	0.17	1	0.17	78.63	< 0.0001
C-Pretreatment time	9.282×10 ⁻³	1	9.282×10 ⁻³	4.33	0.0759
AB	0.027	1	0.027	12.50	0.0095
AC	0.017	1	0.017	7.83	0.0266
BC	8.030×10 ⁻³	1	8.030×10 ⁻³	3.75	0.0941
A^2	0.010	1	0.010	4.70	0.0669
B ²	0.12	1	0.12	55.77	0.0001
C^2	6.760×10 ⁻⁵	1	6.760×10 ⁻⁵	0.032	0.8640
Lack-of-fit	7.370×10 ⁻³	3	2.457×10 ⁻³	1.29	0.3926 not significant

Footnote: df=degrees of freedom, F-value=Fisher-Snedecor distribution value, P-value=probability value.

Source	Sum of	df	Mean	F-value	P-value
	squares		square		(Probability > F)
Model	0.019	9	2.143×10 ⁻³	4.21	0.0357 Significant
A-Alkaline concentration	2.040×10 ⁻⁴	1	2.040×10 ⁻⁴	0.40	0.5470
B-Power intensity	9.197×10 ⁻³	1	9.197×10 ⁻³	18.05	0.0038
C-Pretreatment time	2.961×10 ⁻³	1	2.961×10 ⁻³	5.81	0.0468
AB	7.647×10 ⁻⁶	1	7.647×10 ⁻⁶	0.015	0.9059
AC	3.124×10 ⁻⁴	1	3.124×10 ⁻⁴	0.61	0.4593
BC	1.721×10 ⁻³	1	1.721×10 ⁻³	3.38	0.1087
A^2	2.729×10 ⁻³	1	2.729×10 ⁻³	5.36	0.0539
B^2	1.420×10 ⁻³	1	1.420×10 ⁻³	2.79	0.1390
C^2	7.669×10 ⁻⁴	1	7.669×10 ⁻⁴	1.50	0.2596
Lack-of-fit	1.321×10 ⁻³	3	4.404×10 ⁻⁴	0.78	0.5617 not significant

Table S4. Analysis of variance of the developed model of corn cobs pretreated with $NaOH_{CC}$

Footnote: df=degrees of freedom, F-value=Fisher-Snedecor distribution value, P-

value=probability value.

Table S5. Comparison of reducing sugar yields from bamboo and corn cobs reported for various studies with present study

Heating	Feedstock	Pretreatment conditions	RSY	References
mechanism			(g/g)	
Microwave	Bamboo	14% Na ₃ PO ₄ .12H ₂ O, 10% SL, 600 W for 7 min	0.086	This study
Microwave	Bamboo	18% NaOH, 10% SL, 600 W for 6 min	0.225	This study
Microwave	Corn cobs	18% Na ₃ PO ₄ .12H ₂ O, 10% SL, 600 W for 8 min	0.512	This study
Microwave	Corn cobs	12% NaOH, 10% SL, 500 W for 5 min	0.420	This study
Microwave	Corn cobs	11.55% Na ₃ PO ₄ .12H ₂ O, 10% SL, 700W for 6	0.760	Sewsynker-Sukai and
		min		Gueguim Kana (2018a)
Microwave	Corn cobs	2% NaOH, 6.7% SL, 100 °C for 30 min	0.680	Boonsombuti et al. (2013)
Steam	Bamboo	2% NaOH, 2% SL, 180 °C, 150 rpm and	0.764	Kuttiraja et al. (2013)
		30 min		
Oven	Bamboo	9% Na ₃ PO ₄ .12H ₂ O and 0.3 g/g H ₂ O ₂ , SL of	0.971	Qing et al. (2016b)
		1/20 (w/w) at 80 °C for 120 min		

Footnote: SL = Solid loading, RSY = Reducing sugar yield.

Sample	Cellulose (%)	Hemicellulose (%)	Lignin (%)
Native _{BB}	54.33	14.82	25.64
NaOH _{BB}	68.11	11.57	18.92
Na ₃ PO ₄ .12H ₂ O _{BB}	67.78	12.89	19.22
Nativecc	40.31	43.50	9.31
NaOHcc	72.48	12.73	10.13
Na ₃ PO ₄ .12H ₂ Occ	56.08	30.74	9.91

Table S6. Structural composition of the control (untreated) and optimally pretreated

lignocellulosic substrates

Footnote: $_{BB}$ = Bamboo, $_{CC}$ = Corn cobs.

Table S7. Comparison of kinetic coefficients obtained from the logistic model to previous

studies

Substrate	μ_{max} (h ⁻¹)	$X_0(g/L)$	X _{max}	Reference
			(g/L)	
Corn cobs	0.175	0.548	1.817	This study (NaOH _{CC(SSF)})
Corn cobs	0.049	0.501	5.579	This study (Na ₃ PO ₄ .12H ₂ O _{CC(SSF)})
Corn cobs	0.216	0.556	3.650	Sewsynker-Sukai and Gueguim Kana (2018b)
Potato peel waste	0.195	0.441	2.428	Chohan <i>et al.</i> (2020)
Sugarcane bagasse	0.150	0.160	2.580	Jugwanth et al. (2019)

Footnote: μ_{max} = maximum specific growth rate; X_0 = initial cell concentration; X_{max} =

maximum cell concentration.

Table S8. Comparison of the ethanol kinetic coefficients obtained using the modified

Substrate	P _m (g/L)	$r_{p,m}(g/L/h)$	t _L (h)	Reference
Corn cobs	16.928	1.246	0	This study (NaOH _{CC(SSF)})
Corn cobs	5.286	1.364	0	This study (Na ₃ PO ₄ .12H ₂ O _{CC(SSF)})
Corn cobs	42.24	2.39	1.98	Sewsynker-Sukai and Gueguim Kana (2018b)
Potato peel waste	15.475	1.513	4.658	Chohan <i>et al.</i> (2020)
Sorghum leaves	17.15	0.52	6.31	Rorke and Gueguim Kana, (2017)
Sugarcane bagasse	3.12	0.29	0.97	Jugwanth et al. (2019)
Sorghum bagasse	16.64	0.51	0	Wang <i>et al.</i> (2013)

Gompertz model to previous studies

Footnote: P_m = maximum potential bioethanol concentration, $r_{p,m}$ = maximum bioethanol production rate; t_L = lag time.

Chapter 4 of the thesis has been published during the examination period in the journal Bioresource Technology (BITE) entitled: Simultaneous saccharification and citric acid production from paper wastewater pretreated banana pseudostem: Optimization of fermentation medium formulation and kinetic assessment.

DOI: https://doi.org/10.1016/j.biortech.2022.127700

The published paper and supplementary material are presented in the following pages.

Contents lists available at ScienceDirect

Bioresource Technology

journal homepage: www.elsevier.com/locate/biortech



Simultaneous saccharification and citric acid production from paper wastewater pretreated banana pseudostem: Optimization of fermentation medium formulation and kinetic assessment

Milesh Laltha^a, Y. Sewsynker-Sukai^b, E.B. Gueguim Kana^{a,*}

^a University of KwaZulu-Natal, School of Life Sciences, Pietermaritzburg, South Africa ^b University of Fort Hare, Fort Hare Institute of Technology, Private Bag X1314, Alice 5700, South Africa

HIGHLIGHTS

GRAPHICAL ABSTRACT

- First report on the optimization of citric acid (CA) using banana pseudostem.
- The optimized process gave a maximum CA concentration of 14.408 g/L.
- Kinetics were determined on fresh water. wastewater and standard medium.
- The wastewater-based bioprocess demon strated only a 9.11% lower CA concen tration.
- Findings facilitate the development of waste-based lignocellulosic bioprocesses.

Aedium optimization Citric acid production fr Kraft paper and pulp (desorbent, nitrogen fresh water, dairy ewater pretreatment wastewater or standard urce, temperature) and Banana pseudostem Kinetic assessment media

ARTICLE INFO

Keywords: Citric acid Dairy wastewater Banana pseudostem Simultaneous saccharification and fermentation

ABSTRACT

This study optimized the simultaneous saccharification and citric acid (CA) production from banana pseudostem (BP). Thereafter, kinetic assessment of Aspergillus brasiliensis growth and CA production were determined for the optimum conditions using fresh water (SSF $_{optimizedFW}$) or dairy wastewater (SSF $_{DWW}$) and compared to Sabouraud Dextrose Emmon's medium modified with BP (SSF_{SDEmodified}). The optimized conditions gave a CA concentration of 14.408 g/L. Kinetic assessment revealed the same maximum specific growth rates (μ_{max}) (0.05 h⁻¹) for all three bioprocesses, while the $SSF_{SDEmodified}$ process resulted in the highest maximum potential CA concentration (P_m) (13.991 g/L) in comparison to the SSF_{DWW} ($P_m = 13.095$ g/L) and SSF_{optimizedFW} ($P_m = 12.967$ g/ L) systems. Findings from this study facilitates the implementation of waste-based lignocellulosic bioprocesses that may eradicate the use of expensive pretreatment chemicals, fermentation medium constituents, and resources, in keeping with the water, energy and food nexus towards developing a circular bioeconomy.

* Corresponding author.

E-mail address: kanag@ukzn.ac.za (E.B. Gueguim Kana).

https://doi.org/10.1016/j.biortech.2022.127700

Received 17 June 2022; Received in revised form 21 July 2022; Accepted 22 July 2022 Available online 25 July 2022 0960-8524/© 2022 Elsevier Ltd. All rights reserved.





1. Introduction

The bioconversion of renewable lignocellulosic biomass to valuable products such as citric acid (CA) has become a research hotspot towards achieving sustainable global economic development (Yadav et al., 2020). CA is a valuable commercial bioproduct with an existing market and widespread use in the chemical, food, pharmaceutical and cosmetic industries (Sekoai et al., 2018). Approximately 80 % of the world's CA is generated mainly by Aspergillus niger. This microbe has typically been used, due to its exclusive physiological characteristics for industrial CA fermentation. A. niger shares a phylogenetic relationship with Aspergillus brasiliensis, (Varga et al., 2007), however, A. brasiliensis produces various other unique compounds that justify its separation into a separate species from A. niger (Varga et al., 2007), while still demonstrating high potential for CA production via the citric acid cycle (TCA). Around 2 million tons of CA is generated annually and its rate of demand is expected to rise by 5 % each year (Ozdal and Kurbanoglu, 2019, Sekoai et al., 2018). The CA market is constantly under pressure due to expensive substrates and low production yields (Sekoai et al., 2018). Additionally, the cost of CA production via chemical synthesis is approximately ~\$10 USD per kg, whereas for fungal fermentation it has been estimated to be ~ 50 % lower (Ozdal and Kurbanoglu, 2019). The global market for CA is expected to surpass \$3.2 billion USD by 2023 (Behera et al., 2021). More importantly, the CA global demand has surpassed the natural supply, therefore the incorporation of lignocellulosic-based bioprocesses that are novel, low-cost, efficient, and environmentally friendly are urgently required. Lignocellulosic biomass is a naturally occurring complex composite that originates from plant dry matter (Moodley et al., 2020), and has emerged as an ideal feedstock for generating renewable bioproducts such as CA due to its abundance and cost-effectiveness. In addition, LCB is considered a waste and therefore does not pose any risk to socio-economic concerns such as deforestation, food security and water shortages (Sekoai et al., 2019). CA has been derived from lignocellulosic substrates that include banana peels (Odu et al., 2020), sweet potato peels (Aboyeji et al., 2020), and sugarcane bagasse (Campanhol et al., 2019). Banana pseudostem with a composition of cellulose (42.35 %), hemicellulose (22.63 %) and lignin (15.36 %) represents a novel substrate for generating CA (Islam et al., 2019). However, the development of lignocellulosic CA production is prone to several drawbacks that include lack of suitable feedstocks, expensive chemicals and/or nutrients as well as energy intensive processes that often result in low product yields. Therefore, overcoming the aforementioned bottlenecks is imperative to enhance lignocellulosic CA production. For one, finding an alternative to fresh water within lignocellulosic bioprocesses could substantially reduce the burden placed on this finite resource and costs incurred. Taking this into consideration, untreated wastewater from the Kraft and dairy industries represents a sustainable and cost-effective option for the replacement of fresh water and/or chemicals within lignocellulosic bioprocessing systems. The emergence of waste-based technologies for lignocellulosic CA production combined with simultaneous saccharification and fermentation (SSF) processes could reduce the time, expenditure for operation, energy input, risk of contamination and enhance the overall productivity and product yields. Fermentation medium constituents that are required to satisfy the nutritional needs of microbes are costly. The microbial growth of Aspergillus sp. typically occurs in the Sabouraud Dextrose Emmon's modification (SDE) media, which constitutes peptone and glucose that function as nitrogen and carbon sources, respectively. While glucose may be supplied by the LCB, peptone is an expensive constituent estimated at \$151 USD per kg, contributing majorly to the entire bioprocessing cost. Therefore, the investigation of a cost-effective nitrogen source is imperative. Ammonium nitrate has an estimated cost of \$108 USD per kg and may be employed as a cheaper alternative to peptone, translating to a cost reduction of ~ 28 %. Odu et al. (2020), Sarkar and Das (2017) and Sekoai et al. (2018) previously utilized ammonium-based salts (ammonium nitrate or ammonium sulfate) in the

range of 0.1 to 8 % w/v and demonstrated its adequacy as a nitrogen source in CA production. Other sources of nitrogen for CA production include urea, corn steep liquor (CSL) and yeast extract that are also considered significantly cheaper than peptone (Sekoai et al., 2018). Moreover, several studies have reported on the improvement of CA yields by the addition of desorbents such as methanol and acetone in the range of 1 to 4 % (Sarkar and Das, 2017, Odu et al., 2020). Desorbents have shown to enhance CA production by improving the cell permeability (Sekoai et al., 2018). In addition to these nutritional components, temperature influences the cellulase-based enzyme and fungal culture in lignocellulosic CA production. Fungi such as A. brasiliensis have shown to thrive under various temperatures ranging from 25 to 45 °C (Aboyeji et al., 2020), while cellulase-based enzymes require temperatures between 45 and 50 °C for optimal functioning. Exploring a suitable, intermediate temperature that provide optimal conditions for the enzyme and microbial culture functioning is crucial in lignocellulosic bioprocessing. Optimizing input parameters that comprise of temperature, nitrogen and desorbent concentration may prove valuable for CA production. The response surface methodology (RSM) is a mathematical model that may be used for the optimization process and determines the interactive effects of input parameters and their corresponding output (Gaitonde et al., 2017). Apart from bioprocess optimization, kinetic modelling provides an in-depth understanding of the metabolic nature of a fermentation process (Germec et al., 2019) and plays an essential role in predicting, monitoring, optimizing and simulating the performance of a process (Pramanik et al., 2019). The logistic model is used to describe the kinetics of microbial biomass growth to the rate-limiting substrate over a certain time period (Germec et al., 2019). Conversely, the modified Gompertz model estimates the maximum production rate, maximum concentration and the lag phase (Srimachai et al., 2015). Although SDE media has shown to be effective for the cultivation of Aspergillus sp., this medium consists of expensive constituents and uses fresh water, which is a finite resource. Ideally, the application of industrial wastes as a water replacement in SSF bioprocesses could alleviate water security. Interestingly, the dairy industry produces 1.575 imes10¹² L of wastewater that may be harnessed as a substitute for fresh water (Slavov, 2017). Dairy wastewater undergoes expensive treatment processes to reduce contaminants before it is expelled into the environment. From an industrial perspective, these treatments are considered an economical burden, due to the long processing times, high costs and energy. Adding to this, is the release of greenhouse gas emissions that has severe negative implications on the environment (Gogoi et al., 2021). Therefore, using DWW in lignocellulosic bioprocesses also presents opportunities for commercial industries regarding disposal and treatment costs. Another striking characteristic of DWW is the presence of trace amounts of carbohydrates, soluble organic compounds, protein and nitrogen that could contribute to promoting the growth of the microbial culture (Kaur, 2021). Even though several studies have reported on the production of CA from different lignocellulosic substrates (Odu et al., 2020, Sarkar and Das, 2017, Aboyeji et al., 2020), there has been a scarcity of information on the effects of temperature, ammonium nitrate and acetone on CA production using BP. Furthermore, the existing knowledge gap on kinetic data of A. brasiliensis growth and CA production with fresh water, DWW and standard SDE media has piqued interest. Therefore, the present study aimed to model and optimize the SSF process using the RSM for CA production from pretreated BP. For the RSM optimization, the effect of acetone concentration, ammonium nitrate concentration and temperature on the CA production was determined. Subsequently, the modified Gompertz and logistic models were employed to evaluate the kinetics of A. brasiliensis growth and CA production respectively. For the kinetic experiments, three bioprocesses consisting of the: (1) optimized SSF process with fresh water (SSF_{opti-} mizedFW), (2) SSF_{optimizedFW} process conditions while substituting dairy wastewater in place of fresh water (SSF_{DWW}), and (3) SDE medium adapted by substituting glucose with BP (SSF_{SDEmodified}) were assessed.

2. Materials and methods

2.1. Materials

The banana pseudostem substrate used in this study was obtained from Bakerville Heights (Pietermaritzburg, South Africa) (29.5418° S, 30.4077° E). The substrate was dried at 75 °C for 48 h in an oven (Scientific series 2000, South Africa) to reduce the remaining moisture content and was thereafter milled to a particle size that ranged from 1 to 2 mm. The chemical reagents utilized were acquired from Merck (South Africa), where suitable. The Cellic CTec2 enzyme (160 FPU/mL) harnessed for the SSF processes was supplied by Novozymes (Novozymes A/ S, Denmark). The paper wastewater (pH \sim 8) together with the safety data sheet containing its chemical composition (Laltha et al., 2022) was provided by a Kraft paper and pulp mill (Mondi, Richards Bay, South Africa). The dairy wastewater employed was obtained from a regional dairy (Fairfield, Howick, South Africa) and its composition has been reported by David et al. (2022).

2.2. Revival and microbial development

The fungal strain employed in this study, Aspergillus brasiliensis ATCC16404 was obtained from ATCC® type cultures originating from Microbiologics Incorporated. The Microbiologics KWIK-STIK pack contained a protocol, ampoule of hydrating fluid, qualitative lyophilized microorganism pellet and an inoculating swab. The hydrating fluid was applied to the lyophilized pellet and aseptically swabbed onto SDE (Sabouraud Dextrose Emmon's) agar by using the three-way streak plating technique to facilitate colony isolation. The SDE agar medium formulation comprised of 20 g/L glucose, 20 g/L agar and 10 g/L peptone autoclaved at 121 °C for 15 min. The fungal strain was then incubated at 30 °C for 7 days. Preparation of a fungal spore suspension was carried out by the addition of 10 mL of sterilized distilled water that contained non-ionic surfactant (0.1 % v/v Tween 80) by covering the fungal growth and using a spatula to scrape the spores into 100 mL of SDE broth (20 g/L glucose and 10 g/L peptone) incubated at 30 °C and 150 rpm for 7 days.

2.3. Short-term and long-term storage of culture

To ensure continuous longevity of the microorganism, the fungal strain was subcultured every 7 days by transferring a piece of inoculated agar containing mature fungal hyphae and spores onto SDE agar, which was thereafter subjected to incubation at 30 °C for 7 days. The previous growth plate was stored at 4 °C in the refrigerator. For long-term storage, *A. brasiliensis* was inoculated in glycerol stocks (10 %, 20 % and 50 % v/v) and agar slants (double-strength medium) and stored at -80 °C and 4 °C, respectively.

2.4. Pretreatment of banana pseudostem

The powdered BP substrate was subjected to pretreatment using a previously optimized microwave-assisted paper wastewater method (Laltha et al., 2022). The BP substrate at a 30 % (w/v) solid loading was submerged in 100 mL of paper wastewater and exposed to microwave heating at 800 W for 8 min. The pretreated banana pseudostem was filtered using a domestic sieve (<1 mm) and washed several times with deionized water, oven-dried at 55 °C overnight prior to being used in the SSF bioprocess. The lignocellulosic composition (cellulose, hemicellulose, and lignin) of the untreated (native) and optimally pretreated banana pseudostem was analyzed using the detergent fiber technique described by Van Soest and McQueen, 1973. The composition of the optimally pretreated banana pseudostem was previously determined and consisted of cellulose (17.06 %), hemicellulose (13.04 %) and lignin (5.9 %) (Laltha et al., 2022).

2.5. Simultaneous saccharification and fermentation (SSF) process

2.5.1. Inoculum development

A. brasiliensis was grown on SDE agar as previously described in section 2.2. The grown suspension was diluted to a final spore concentration of 1×10^6 spores/mL. A Neubauer counting chamber (Neubauer, Germany) was used to determine the spore concentration.

2.5.2. Sodium citrate buffer preparation

Sodium citrate buffer (0.05 M) was made up by combining 0.1 M citric acid and 0.1 M sodium citrate in distilled water. The pH was adjusted to 4.8 prior to autoclaving at 121 $^{\circ}$ C for 15 min.

2.5.3. Preliminary screening

Preliminary screening of three desorbents (3 % v/v) (acetone, chloroform and propan-2-ol) and four nitrogen sources (2 % w/v) (corn steep liquor, ammonium nitrate, yeast extract and peptone) were performed to determine their individual effects on CA production. The SSF parameter ranges were designated based on an extensive literature review of related lignocellulosic fermentation studies (Sarkar and Das, 2017, Odu et al., 2020). The pretreated BP with a solid loading of 10 %w/v, an inoculum size of 10 % (v/v), and 10 FPU/g enzyme loading was added to an Erlenmeyer flask and brought up to a final volume of 50 mL using 0.05 M sodium citrate buffer. The desorbents (3 % v/v) (acetone, chloroform or propan-2-ol) and nitrogen sources (2 % w/v) (corn steep liquor, ammonium nitrate, yeast extract or peptone) were also added accordingly (See supplementary material). The SSF flasks constituted a working volume of 50 mL, incubated at 35 °C and 150 rpm (Ho and Hood, 2014) for a period of 144 h. Samples (1 mL) were aseptically collected every 24 h and centrifuged at 10 000 rpm for 5 min prior to CA quantification using an acid-base titration method (Ayeni et al., 2019).

2.5.4. RSM experimental design of the SSF process

The response surface methodology (RSM) model (Box-Behnken design) (Design Expert 7.0, Stat Ease Inc, USA) was employed for the optimization. The experimental design consisted of seventeen (17) SSF processes. The fermentation input parameters comprised of desorbent (acetone) concentration (1–5 % v/v), nitrogen (ammonium nitrate) concentration (0.5–5 % w/v) and temperature (30–40 °C), while the output was CA concentration (g/L). The SSF input parameters with its ranges were chosen based on an extensive literature survey (Sarkar and Das, 2017, Sekoai et al., 2018, Odu et al., 2020) in combination with data curated from the screening of preliminary experiments. The resultant experimental CA concentration data fitted the RSM polynomial equations to assess the combined input parameter effects for the enhancement and generation of CA.

2.5.5. Citric acid production

The CA experiments were performed in 100 mL Erlenmeyer flasks comprising of a 50 mL working volume. A solid loading of 10 % (w/v) of paper wastewater pretreated BP was added to each flask. In addition, 10 FPU/g Cellic CTec2 enzyme, 0.05 M sodium citrate buffer, and 10 % v/v inoculum containing 1×10^6 spores/mL *A. brasiliensis* were aseptically added. The respective amounts of acetone (1–5 % v/v) and ammonium nitrate (0.5–5 % w/v) were added to each fermentation flask and incubated at the desired temperature (30–40 °C) as specified in the RSM experimental design (Table 1). The SSF processes were agitated at 150 rpm for a period of 96 h. Using aseptic techniques, the aliquots (1 mL) were sampled from the SSF flasks every 12-hour and stored at -20 °C for further analysis.

2.6. Validation of the optimized SSF bioprocess and kinetic modelling

The variable conditions for optimal CA generation were determined and carried out to validate the developed RSM model. The SSF reaction working volume of 50 mL contained: 10 % (w/v) pretreated BP, 1 % (v/

Table 1

Citric acid concentration for the SSF process on inputs of acetone, ammonium nitrate concentration and temperature.

	Input varia	bles		Output
Run	Acetone (v/v)	Ammonium nitrate (w/v)	Temperature (°C)	Citric acid concentration (g/ L)
1	3	0.50	30	1.857 ± 0.121
2	1	0.50	35	14.408 ± 0.594
3	5	2.75	30	0.640 ± 0.005
4	3	2.75	35	9.093 ± 0.126
5	3	2.75	35	9.413 ± 0.162
6	3	0.50	40	1.153 ± 0.052
7	5	2.75	40	0.704 ± 0.033
8	1	2.75	30	2.433 ± 0.128
9	3	5.00	40	1.217 ± 0.063
10	5	0.50	35	6.275 ± 0.178
11	3	2.75	35	8.324 ± 0.059
12	3	2.75	35	5.571 ± 0.115
13	3	2.75	35	9.413 ± 0.231
14	5	5.00	35	6.852 ± 0.220
15	3	5.00	30	2.305 ± 0.098
16	1	5.00	35	12.679 ± 0.106
17	1	2.75	40	$\textbf{0.832} \pm \textbf{0.143}$

v) acetone, 0.5 % (w/v) ammonium nitrate, with an inoculum size of 10 % (v/v) and citrate buffer. This process was designated as SSF_{optimizedFW}. In addition, the SSF_{optimizedFW} process was substituted with dairy wastewater in place of fresh water (SSF_{DWW}). The SSF_{DWW} process contained: 10 % (w/v) pretreated BP, 1 % (v/v) acetone, 0.5 % (w/v) ammonium nitrate, with an inoculum size of 10 % (v/v) and citrate buffer formulated using DWW. The DWW formulated citrate buffer was the same as described in section 2.5.2., with the substitution of DWW instead of fresh water. For comparison, the standard SDE medium was modified by substituting glucose with BP (SSF_{SDEmodified}) and consisted of 10 % (w/v) pretreated BP, 1 % (w/v) peptone, with an inoculum size of 10 % (v/v) and citrate buffer. Each SSF process was therefore characterized as follows: (1) optimized SSF process with fresh water

method employed by Ayeni et al. (2019) and thereafter verified by the Megazyme citric acid assay kit, product code (K-CITR) (©Megazyme, Wicklow, Ireland) (See supplementary material). In the acid-base titration analysis, the SSF samples collected from each time point were first centrifuged for 5 min at 10 000 rpm. Subsequently, 100 µL of the supernatant fluid was transferred to a flask containing phenolphthalein indicator (2 to 3 drops). A glass burette was rinsed with distilled water and then filled with the 0.01 M NaOH titrant. The burette was opened to allow few drops of the NaOH (base) titrant to react with the sample (acid) and shaken continuously to observe a change in colour (from colourless to light pink). Excess titrant was washed of the sides of the flask with distilled water. The endpoint of the reaction was determined when the colour change had occurred. Following this, the volume of NaOH utilized was read directly from the burette and substituted into the formula described by Ayeni et al. (2019) to ascertain the amount of CA present. The corresponding number of moles of alkali NaOH required to neutralize CA, C3H5O(COOH)3, was then determined based on the volume and concentration of NaOH solution utilized during the acidbase titration (See supplementary material) (Ayeni et al., 2019). For verification of the acid-base titration method described above, the CA concentration was measured by means of the Megazyme citric acid assay kit, product code (K-CITR) (©Megazyme, Wicklow, Ireland) corresponding to the specific instructions in the manufacturers protocol. The Megazyme citric acid assay kit is an enzyme-based analytical method that is specific. The CA content was analyzed spectrophotometrically using a SpectraMax ABS microplate reader (Molecular Devices, California, USA).

2.7.2. Glucose concentration

The glucose concentration of the control (uninoculated) and experimental kinetic SSF experiments were analyzed using the Megazymé, glucose kit (K-GLUC) (©Megazyme, Wicklow, Ireland) and protocols detailed by the manufacturer. To determine the glucose utilization (%), the final glucose concentration (inoculated) was subtracted from the initial glucose concentration as per Eq. (1).

$$Glucose \text{ utilization } (\%) = \frac{Initial glucose \ concentration \ (g/L) - Final glucose \ concentration \ (g/L)}{Initial glucose \ concentration \ (g/L)} \times 100$$
(1)

 $(SSF_{optimizedFW})$, (2) $SSF_{optimizedFW}$ process conditions while substituting dairy wastewater in place of fresh water (SSF_{DWW}), and (3) SDE medium modified by substituting glucose with BP ($SSF_{SDEmodified}$). All SSF experiments were incubated at 150 rpm, 35 °C for 96 h. The experimental controls were performed parallel to the $SSF_{optimizedFW}$, SSF_{DWW} and $SSF_{SDEmodified}$ and constituted the same conditions with the exception of inoculum, which was omitted. The aforementioned uninoculated controls were used as a reference to determine the initial sugar concentration at a certain time period corresponding to the test SSF experiments ($SSF_{optimizedFW}$, SSF_{DWW} and $SSF_{SDEmodified}$). All of the experiments were carried out in duplicate. Aseptically, sample aliquots (1 mL) were removed from the experimental reaction flask and its equivalent control every 12 h for 96 h. The samples were thereafter analyzed for CA, glucose concentrations and biomass as explained below.

2.7. Analytical methods

2.7.1. Citric acid concentration

SSF processes are dynamic, biological systems and by-products generated during the fermentation may sometimes cause interference when using methods encompassing titration. In the present study, CA concentration was primarily analyzed using the acid-base titration where the final glucose concentration (g/L) is derived from the experimental SSF flask while the initial glucose concentration (g/L) is attained from the uninoculated control flask at the same moment in time.

2.7.3. Biomass concentration

The biomass concentration of *A. brasiliensis* was analyzed using the dry cell weight (g/L) as a function of the total spore count (spores/mL). The fungal isolate was grown for 7 days in SDE broth to reach exponential phase (10^6 to 10^7 spores/mL) and diluted accordingly (Vrabl et al., 2019). A 15 mL volume per diluted sample (1, 1/2, 1/4, 1/8, 1/16 and 1/32) was centrifuged for 5 min at 10 000 rpm. All dilutions were performed in duplicate. Succeeding centrifugation involved discarding of the supernatant and drying of the biomass pellet at 60 °C until the mass was constant. The standard curve was developed by plotting the total spore counts (spores/mL) to the corresponding dry cell weights (biomass concentration in g/L) for each dilution. The spore counts (spores/mL) attained for the kinetic experiments were computed in the equation and extrapolated to ascertain the biomass concentration (g/L).

2.8. Kinetic modelling

2.8.1. The logistic model for A. brasiliensis growth

The logistic model's differential formula, stated in Eq. (2), was integrated to obtain Eq (3). Equation (3) considers the exponential and stationary growth phases. The logistic model Eq. (3) relates biomass (X) to the initial cell concentration (X₀), maximum cell concentration (X_{max}) and maximum specific growth rate (μ_{max}) during the exponential and stationary growth stages of *A. brasiliensis*.

$$\frac{dX}{dt} = \mu_{max} \left(1 - \frac{X}{X_{max}} \right) X \tag{2}$$

$$X = \frac{X_0 \exp(\mu_{max}t)}{1 - \left[\left(\frac{X_0}{X_{max}} \right) (1 - \exp(\mu_{max}t)) \right]}$$
(3)

2.8.2. The modified Gompertz model for citric acid production

Using the least squares approach, the CA data from the SSF processes (SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified}) were fitted to the modified Gompertz model(CurveExpert V1.5.5, MyBiosource, Inc., USA). As indicated in Eq. (4), the modified Gompertz model depicts the link between the CA concentration (P), maximum CA production rate ($r_{p,m}$), potential maximum CA concentration (P_m) and lag time (t_L) from the start of fermentation to the exponential CA generation.

$$P = P_m . exp\left\{-exp\left[\frac{r_{p,m} . exp(1)}{P_m}\right] . (t_L - t) + 1\right\}$$
(4)

3. Results and discussion

3.1. Preliminary screening

Preliminary screening data revealed that the desorbent that gave the highest concentration of 10.886 g/L CA was acetone, followed by propan-2-ol (10.245 g/L) and chloroform (7.044 g/L) (See supplementary material). For the nitrogen source, ammonium nitrate resulted in the highest CA concentration (12.807 g/L), while peptone, corn steep liquor and yeast extract resulted in CA concentrations of 9.605 g/L, 8.324 g/L and 7.684 g/L respectively. Therefore, acetone and ammonium nitrate were selected as the desorbent and nitrogen source respectively, for the SSF optimization.

3.2. Development of the RSM model

To assess the fitness of the developed RSM model, Analysis of Variance (ANOVA) was used (See supplementary material). P-values < 0.05 indicate parameter and model significance, whereas high F-values demonstrate that the response trends could be explained using the regression equations (Chaganti et al., 2012). The model had a low Pvalue of 0.0063, which was < 0.05, indicating model significance. In addition, a high F-value (7.89) was attained for the SSF model, thus illustrating statistical significance of the model. Of the three key variables that were investigated, acetone concentration (p = 0.0282) demonstrated the most significant effect on the CA concentration compared to temperature (p = 0.5812) and ammonium nitrate concentration (p = 0.9146). Acetone serves as the desorbent and enhances cell permeability of the microorganism for nutrient sequestration and may also repress the enzyme 2-oxoglutarate dehydrogenase, thus increasing CA concentration (Sekoai et al., 2018, Maddox et al., 1986). This desorbent (acetone) is also a ketone that plays a vital role as an alternative metabolic fuel source for eukaryotes (Puchalska and Crawford, 2017). On the other hand, temperature is a key parameter for the optimum functioning of cellulolytic enzymes for the effective saccharification of cellulose to glucose that will be consumed by the microbe. Additionally, temperature also plays a significant role for regulating the internal temperature, growth, secondary metabolite production and

activity of fungi such as *A. brasiliensis* (Tudzynski, 2014). The presence of nitrogen such as ammonium nitrate in the fermentation medium is essential for fungal growth since it facilitates cell adaptation to different niches and survival of adverse environmental conditions such as nutrient limitation (Tudzynski, 2014). The coefficient of determination (\mathbb{R}^2) value is a statistic that measures the proportion of variation in the output values that can be attributed to the input variables. In general, \mathbb{R}^2 values between 0.7 and 1 elicits model significance (Chaganti et al., 2012). The developed SSF model displayed an \mathbb{R}^2 value of 0.91, indicating that 91 % of the variation in the output variable. The polynomial model Eq. (5) was generated for the CA concentration and describes the input parameters in relation to the response variable.

CA concentration (g/L) =
$$8.36 - 1.99A - 0.080B - 0.42C + 0.58AB$$

+ $0.42AC - 0.096BC + 0.61A^2 + 1.09B^2$
- $7.82C^2$

where, A, B and C correspond to acetone concentration (v/v), ammonium nitrate concentration (w/v) and temperature (°C), respectively.

3.3. Effects of the input parameters on the citric acid concentration

The CA concentrations with corresponding input parameters are portrayed in Table 1. The SSF model displayed a maximum CA concentration of 14.408 g/L for run 2 (1 % acetone, 0.5 % ammonium nitrate and 35 °C). The median fermentation conditions (3 % acetone, 2.75 % ammonium nitrate incubated at 35 °C) exhibited similar CA concentration values within the range of 5.571 g/L to 9.413 g/L (run 4, run 5, run 11 run 12 and run 13). The lowest CA concentration recorded (0.640 g/L) was achieved with 5 % acetone, 2.75 % ammonium nitrate and incubated 30 °C for run 3. The RSM graphs that show the collaborative effects of the different input parameters and the resultant CA concentration is illustrated in Fig. 1A-C. The interactive effect of ammonium nitrate concentration and acetone concentration on the output CA concentration, while temperature was kept at its center point (35 °C) is presented in Fig. 1A. Maximum CA concentration (12.63 g/L) was observed at a minimum acetone (1 %) and ammonium nitrate concentration (0.5 %). Further increase in the acetone concentration from 1 to 5 %, while ammonium nitrate concentration was maintained at 0.5 % led to a decline in the CA concentration from 12.63 to 7.55 g/L. Simultaneous increments in the ammonium nitrate concentration (5 %) and acetone concentration (5 %) resulted in a decreased CA concentration of 10 g/L. The high CA concentration observed at a low ammonium nitrate concentration may be as a result of reduced vegetative growth of A. brasiliensis. On the other hand, the lower CA observed at higher nitrogen concentrations may be due to the shift in the cell metabolism towards fungal growth instead of product formation (Show et al., 2015). Aspergilli can utilize a variety of nitrogen sources such as ammonium, peptone and nitrate. The fungus converts nitrate to ammonium, which is then incorporated into the amino acids glutamate and glutamine, that act as nitrogen donors within the TCA cycle (Krappmann and Braus, 2005). Furthermore, during CA fermentation, nitrogen limitation is a key requirement to prevent the accumulation of oxalic acid, since this compound has been known to lead to lower CA production. While desorbents such as acetone have shown to enhance fungal growth and CA production, elevated concentrations can be toxic to the cells and may inhibit the fungal cellular metabolic functioning, thus resulting in a lower CA concentration. A similar result was obtained in an earlier study by Roukas and Harvey (1988) when methanol was used as a desorbent. The authors achieved a maximum CA concentration of 4.83 g/L from beet molasses at a methanol concentration of 2 %, while concentrations beyond this value led to a decrease in CA production and concluded that methanol's inductive effect was caused by its inhibitory effect on metal ions. Similarly, the study by Sarkar and Das (2017)



Fig. 1. Three-dimensional response surface graphs showing the interactive effects of the (A) Ammonium nitrate concentration (%) and acetone concentration (%), (B) Temperature (°C) and acetone concentration (%), and (C) Temperature (°C) and ammonium nitrate concentration (%) on the citric acid concentration.

varied ammonium nitrate (0.1 to 0.5 %) and desorbent concentration (0 to 5 %) on pineapple waste and obtained an optimal CA concentration of 44.1 g/L by using 15 % sucrose, 0.25 % ammonium nitrate, 4 % methanol and 7 days of incubation at 30 $^\circ$ C.

The impact of temperature and acetone concentration on the CA

concentration while the ammonium nitrate concentration was maintained at its median value (2.75 %) is represented in Fig. 1B. An increase in temperature (30 to 35 °C) coupled with the lowest acetone concentration (1 %) led to an increase in the CA concentration from 4.08 to 10.94 g/L. Further increments in the acetone concentration and temperature > 1 % and > 35 $^\circ C$ respectively resulted in a steep decrease in the CA concentration from 10.94 to 2.1 g/L. Temperature has a dual effect on SSF processes due to its influence on both the enzymatic and microbial metabolic processes. With regards to enzymatic saccharification, temperature plays a key role on the enzymatic activity and saccharification efficiency for sugar release (El-Imam and Du, 2014). Temperature within the range of 45 and 50 °C has been shown to facilitate effective cellulase enzymatic saccharification, however the combined SSF system necessitates a median temperature that stimulates enzyme and microbial activity. Nevertheless, high-performing cellulase cocktail enzymes such as Cellic CTec2 has been specifically developed to withstand adverse environmental conditions and sustain its activity (Champreda et al., 2019). On the other hand, the temperature parameter also has a profound effect on the microbial growth of fungi. Aspergillus sp. are mesophilic microorganisms that exhibit wide growth ranges between 25 and 45 °C (Aboyeji et al., 2020). Hence, to ensure optimal enzyme function and microbial growth, the temperature employed in the SSF process ranged between 30 and 40 °C. In Fig. 1B, the highest CA concentration (10.94 g/L) was observed at a temperature of 35 °C. Likewise, Zhang et al. (2018) examined the influence of temperature on the growth of bacteria and fungi and determined that the growth activity increased at 35 °C. A high concentration of CA was observed when a low concentration of desorbent was employed (1 %). In the same vein, Moyer (1953) studied the effects of ammonium chloride and methanol concentration for the synthesis of CA from corn meal and concluded that a low desorbent concentration (2 %) enhances the citrate synthase activity while decreasing the aconitase activity during CA fermentation. Citrate synthase is a central enzyme involved in sugar oxidation and is also responsible for catalyzing the TCA cycle, whereas aconitase catalyzes the isomerization of citrate to isocitrate within the TCA cycle (Papagianni, 2007, Show et al., 2015). While desorbents improve the activity of enzymes involved in the TCA cycle, they are inhibitory above the threshold levels (>5%) and could potentially lead to the degradation of these key enzymes and inhibit the growth of microorganisms by drastically altering the pH of the medium. Maddox et al. (1986) observed a similar observation with a low desorbent concentration (1%) and observed a maximum CA concentration of 25 g/L using galactose as a substrate. Furthermore, in the absence of methanol, very little CA was produced (<1 g/L). Kareem et al. (2010) evaluated the effect of sucrose, ammonium nitrate and methanol on CA production using fermentation medium consisting of pineapple waste and achieved optimum CA (60.61 g/L) under 15 % sucrose, 0.25 % ammonium nitrate and 2 % methanol when fermented for 5 days at 30 $^\circ\text{C}.$ The aforementioned author also noted that further increases in the methanol concentration (>3%) led to a diminished fungal growth and lower CA production (Kareem et al., 2010). The pairwise interaction of temperature and ammonium nitrate concentration on the CA concentration when the acetone concentration was retained at its midpoint value (3 %) is demonstrated in Fig. 1C. A simultaneous increase in temperature (30 to 35 °C) and ammonium nitrate concentration (0.5 to 5 %) resulted in a minor increase in CA concentration from 8.65 to 9.33 g/L. Similar to the interactive effect of temperature and acetone concentration as shown in Fig. 1B, it is interesting to note that increases in the temperature parameter above 35 °C, led to a drastic decline in CA concentration from the optimal value (9.65 g/L) to approximately 2 g/L. Several factors influence fungal growth; however, temperature has shown to have a drastic effect that could potentially dictate the output and quality of the desired end-product. Temperature also influences the growth of hyphae and the activity of extracellular enzymes associated with the TCA cycle. At high temperatures (>35 $^\circ\text{C}$), hyphal deterioration may occur and result in a critical reduction of CA (Zhang et al., 2018). Nitrogen

concentration has been found to have a strong influence on both cellular metabolism and CA production. Owing to the necessity of employing low-cost substrates in order to reduce CA manufacturing costs and make it more ecologically sustainable, glucose derived from pretreated banana pseudostem makes it a valuable carbon source. By increasing the initial glucose concentration, the specific rate of CA production rises (Papagianni et al., 1999), whereas concentrated ammonium nitrate inhibits the growth and metabolism of fungi (Veverka et al., 2007). Veverka et al. (2007) investigated the sensitivity of fungal growth to two different nitrogen sources (urea and ammonium nitrate). It was concluded that ammonium nitrate (0.06 M) had the most notable effect on the growth of A. niger with an increase of 1.5-fold greater in comparison to urea (0.19 M). The type of nitrogen source utilized is important due to specific properties that affect the fungal growth and synthesis of CA (Show et al., 2015). For example, the absorption of ammonium by germinating spores release protons, which lowers the pH of the fermentation medium and results in an improvement in the CA production.

The low pH encountered during the production phase (\sim 2) lowers the risk for contamination by unwanted microorganisms and impedes the development of undesirable organic acids such as oxalic and gluconic acids, thereby impacting on downstream processing by making product recovery easier.

3.4. Validation of the RSM model

Validation of the developed SSF model was carried out by utilizing the optimum process conditions (1 % acetone, 0.5 % ammonium nitrate and 35 °C). At these setpoints, the model predicted an optimal CA concentration of 12.712 g/L. Experimental validation of these parameters displayed a slightly higher CA concentration of 13.703 g/L, translating to a difference of 7.23 % when compared to the models predicted value and was considered negligible. This indicates high accuracy and precision of the developed RSM model with a strong reliability for prediction. The validation parameters established by the RSM model were applied to two other fermentation processes consisting of dairy wastewater in place of fresh water (SSF_{DEWW}) and SDE medium adapted by substituting glucose with BP (SSF_{SDEmodified}). A comparative assessment between these three SSF processes are detailed under Sections 3.5 and 3.6.

3.5. Comparison of the citric acid production to previous reports

The results obtained from the SSFoptimizedFW, SSFDWW and SSFSDEmodified bioprocesses were compared to earlier reports on CA production using dairy waste and lignocellulosic biomass (See supplementary material). The SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes achieved CA concentrations of 12.967, 13.095 and 13.991 g/L respectively. The present study revealed comparable CA concentrations compared to Aboyeji et al. (2020) (4.36 g/L), Ghanbartabar et al. (2016) (4.45 g/L), and Campanhol et al. (2019) (0.73 g/L) when A. niger was utilized for the fermentation of sweet potato peel waste, cheese whey biomass and sugarcane bagasse respectively. Conversely, Odu et al. (2020), Sarkar and Das (2017) and Ozdal and Kurbanoglu (2019) investigated banana peel, pineapple waste and sugar beet molasse respectively, for the generation of CA and observed higher CA concentrations when compared to the present study. The variations in this study to previously reported literature may be due to different conditions applied that include: the microbial species (inoculum size, strain), physical parameters (temperature, agitation, pH), the concentration and type of enzyme, desorbent type and concentration, nitrogen source and concentration, substrate and solid loading, in addition to the supplementation of chemical additives. Despite the comparable concentrations of CA derived from previous reports, the exploitation of fresh water consumption, expensive substrates, chemical supplementation and/or media components render these fermentation processes non-viable and costly for potential

3.6. A. brasiliensis growth kinetics for the $SSF_{optimizedFW}$, SSF_{DWW} and $SSF_{SDEmodified}$ processes

The kinetics of microbial growth and CA production for the SSF bioprocesses (SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified}) were determined by utilizing the logistic and modified Gompertz models respectively. The lag phase, often known as the microorganism's adaption phase occurs when the microbe begins to acclimatize to changes in the new medium environment and for sufficient glucose to be produced by saccharifying enzymes to support the growth of the microorganism (Asaduzzaman, 2007). The lag phase was observed during the first 12 h of fermentation. In addition, the microorganism is metabolically active and synthesizes the required enzymes and metabolites for exponential cell growth (Asaduzzaman, 2007). During the lag phase, the activation of signaling pathways and transcriptional changes occur, leading to lipopolysaccharide biosynthesis and respiration that is required for multiplication and differentiation of the cells (Hamill et al., 2020). A minor lag phase for conidia germination might last several hours, however temperature and stressors can extend this timeframe to days (Hamill et al., 2020). The lag phase of A. brasiliensis has been shown to be within the range of 12 to 24 h in a previous study, where it was cultivated in media containing glucose (Ho and Hood, 2014). Similar lag phase times (5 to 10 h) has been associated with the commonly used and closely related A. niger for CA production (Baei et al., 2008). The exponential phase or logarithmic phase transpires when the microbial culture undergoes asexual reproduction at a constant rate through simple cell division, in which one cell separates forming two identical daughter cells after undergoing nuclear division (Ellena et al., 2020, Asaduzzaman, 2007). The exponential phase was recorded between 12 and 84 h for all three SSF processes. In general, the exponential phase is initiated by a particular cell concentration that increases metabolism and initiation of DNA replication (Asaduzzaman, 2007). The enzymatic conversion of cellulose to glucose released by cellulase action occurs almost instantaneously during SSF processes and is utilized by the microbial culture (in this case A. brasiliensis) within the TCA cycle for the generation of CA. A high glucose utilization (GU) (Fig. 2B) and glucose consumption (GC) (See supplementary material) was observed for the $SSF_{optimizedFW}$ (GU = 50 %, GC = 9 g/L), SSF_{DWW} (GU = 44 %, GC = 8 g/ L) and $SSF_{SDEmodified}$ (GU = 57 %, GC = 14 g/L) processes and this corresponded to an increase in the fungal growth (1-2.5 g/L) (Fig. 2A). The biomass concentration began to stabilize after 84 h for all three SSF bioprocesses. The decline in cell growth above 84 h can be attributed to the decline in Cellic CTec2 enzyme activity that results in a lower saccharification efficiency and thus lower glucose release for the fermentative microorganism. Furthermore, the functioning of enzymes occurs best within a certain designated temperature (45-50 °C) and pH (4.5-5.5) range (Fenila and Shastri, 2016). Therefore, sub-optimal conditions for enzyme functioning under a low temperature (35 °C) and pH (\sim 3.5) required for the initial production of CA, causes a ripple effect on the SSF process, since the enzyme loses its activity and ability to bind to an active site on the substrate. Aspergillus brasiliensis growth data obtained from the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes were inserted into the logistic model and displayed high R² values of 0.997, 0.998 and 0.997 respectively. All three bioprocesses demonstrated the same maximum specific growth rate (μ_{max}) of 0.05 h⁻¹. The maximum specific growth rate is an important kinetic parameter due to its impact on product yield, formation and productivity (Srimachai et al., 2015, Pramanik et al., 2019). The low μ_{max} values observed may be accounted for by the depletion of nutrients such as glucose or nitrogen for cellular maintenance within the fermentation media. The concentration and availability of nitrogen influences the activity and growth efficiencies of fungi. When nitrogen is scarce within the media, fungi divert their metabolism towards generating more energy for the production of extra-cellular enzymes such as hydrolase and oxidase to



Fig. 2. Aspergillus brasiliensis ATCC16404 biomass concentration (A), glucose utilization (B) and CA concentration (C), for the $SSF_{optimizedFW}$, SSF_{DWW} and $SSF_{SDEmodified}$ processes.

sequestrate nitrogen from the surrounding environment, resulting in lower growth efficiencies (Tudzynski, 2014). The maximum (X_{max}) and initial (X₀) biomass cell concentrations are essential indicators for the synthesis of biomass and formation of products (Baei et al., 2008). Kinetic data obtained for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes gave X₀ concentration values of 0.19 g/L, 0.15 g/L and 0.21 g/ L respectively. Similar to the X₀, the highest X_{max} was realized for the SSF_{SDEmodified} (2.91 g/L) process followed by the SSF_{optimizedFW} (2.79 g/ L) and SSF_{DWW} (2.74 g/L) processes. Extracellular enzymes produced during fermentation are important for the dissolution and absorption of external substances that contribute to the growth of Aspergillus. The enzymes secreted during growth are related to the lignocellulosic substrate, while temperature fluctuations influence extracellular enzyme activity (Zhang et al., 2018). The marginally lower Xmax values reported for the SSF_{optimizedFW} and SSF_{DWW} processes in comparison to the SSF_{SDEmodified} may be attributed to variations in the media composition (nutrients) or metabolic dynamics. The citric acid cycle begins when acetyl CoA transfers its acetyl group to oxaloacetate to create citrate. In the citric acid cycle, carbon dioxide is released in tandem with the reduction of NAD⁺ to NADH. However, increased levels of carbon dioxide can be detrimental to the growth of fungal biomass and reduce the concentration of citrate produced (Show et al., 2015). Peptone is an organic source that contains growth factors such as amino acid or vitamins, while ammonium nitrate is inorganic. However, the addition of amino acids and vitamins to a nutrient medium containing inorganic nitrogen is deemed expensive (Clarke, 2013). Baei et al. (2008) evaluated the potential of A. niger for the production of CA from apple pomace under process conditions of 0.1 g/L potassium ferrocyanide, 1.5 % (w/v) calcium triphosphate, 2.5 g/L ammonium sulfate and 3 mL methanol and recorded a lag phase of 5 to 10 h in addition to a μ_{max}, X_0 and X_{max} of $0.015\ h^{-1},\,0.04\ g/L$ and 22 g/L, respectively. Discrepancies detected in growth kinetic data between the present study and Baei et al. (2008) may be ascribed to variations in substrate type, microbial strain, nutrient availability, heat and mass transfer, aeration, agitation, pH, solid and enzyme loading.

3.7. Citric acid production kinetics for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes

CA production during the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes, were elucidated over time (Fig. 2C). Similar to an increase in biomass concentration (Fig. 2A), a steady increase in the CA production occurred from 0 to 96 h (Fig. 2C), no further increments in CA were recorded from 96 h (data not shown), indicating the initiation of the end of the fermentation process. A lack of essential resources, such as carbohydrates generally initiates the stationary phase of microbial growth. In all three bioprocesses, it was observed that CA increased from the exponential phase of microbial growth and continued into the stationary phase. Citric acid increased from approximately 0.5 to 13 g/L for the SSF_{optimizedFW} process and from 0.6 to 13.1 g/L for the SSF_{DWW} system, while the SSF_{SDEmodified} bioprocess observed the highest CA increase from 0.58 to 14 g/L (Fig. 2C). Citric acid data obtained for the three SSF processes coincided with the exponential phase of A. brasiliensis cell growth (Fig. 2A). The highest glucose utilization (GU) and glucose consumption (GC) values for the SSF_{optimizedFW} (GU = 50 %, GC = 9 g/L at 48 h), SSF_{DWW} (GU = 44 %, GC = 8 g/L at 72 h) and $SSF_{SDEmodified}$ (GU = 57 %, GC = 14 g/L at 60 h) processes were observed from 48 to 72 h (Fig. 2B). The period before maximum CA production (0 h to 36 h) allows for adequate time, that is imperative for the enzyme to saccharify the BP substrate to release sufficient glucose that is required to sustain the metabolism of the fungal strain, leading to primary metabolite production of CA. The cellulase-based Cellic CTec2 utilized for the SSF process is a cocktail of enzymes that act collaboratively on cellulose thus converting it to glucose and exhibits high saccharification efficiencies coupled with a reduced vulnerability to glucose inhibition (Champreda et al., 2019). The decline in A. brasiliensis growth observed after 84 h, overlaps with a lower glucose utilization and glucose consumption for the SSF_{optimizedFW} (GU = 36.2 to 18.8 %; GC = 4.5 to 2.7 g/L), SSF_{DWW} (GU = 24.1 to 19.3 %; GC = 4.2 to 2.8 g/L) and SSF_{SDEmodified} (GU = 44.5 to 18.9 %; GC = 5.4 to 2.2 g/L) processes (Fig. 2B). The experimental CA data obtained for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} experiments fitted the modified Gompertz model exhibiting high R² values of 0.991, 0.993, and 0.992 respectively. The shortest lag time (t_L) was obtained for the $SSF_{SDEmodified}$ (5.11 h) process followed by the SSF_{DWW} (9.80 h) and SSF_{optimizedFW} (16.47 h) systems. The SSF_{SDEmodified} process consisted of peptone as the nitrogen source, thus the shorter t_L achieved can be attributed to the incorporation of amino acids and vitamins compared to the $\text{SSF}_{\text{optimizedFW}}$ and SSF_{DWW} systems that utilized ammonium nitrate. Ammonium nitrate has been shown to be approximately 25 % cheaper than peptone, however the latter incorporates growth factors that is fundamental for rapid growth of Aspergillus sp. and CA metabolism (Clarke, 2013). Nitrogen is exploited by Aspergillus for the generation of amino acids such as glutamine and glutamate that act as nitrogen donors within the TCA cycle (Krappmann and Braus, 2005). On the other hand, the addition of the desorbent in the SSF_{opti-} $_{\rm mizedFW}\!,$ and SSF_{\rm DWW} processes, may increase cell permeability to citrate thus increasing CA production via repression of the 2-oxoglutarate dehydrogenase enzyme compared to the SSF_{SDEmodified} system (Maddox et al., 1986). The availability of initial carbohydrates (sugar polymers) in solution may also enhance the cellulase enzyme activity at the onset of fermentation for the accumulation of monomeric sugars (glucose). The glucose utilization in the first 12 h of fermentation for the SSF_{opti-} $_{mizedFW}\!\!\!\!$, SSF_DWW and SSF_SDEmodified processes corresponded to 28.3 %, 13.7 % and 30 % respectively. Therefore, the lag phase during CA fermentation is dependent on nutrient and sugar availability. Interestingly, the lag times observed in this study were shorter when compared to data recorded in the previous reports by Dienye et al. (2018) and Odu et al. (2020) that both observed lag times of 24 h for CA production. Maximum CA production rates $(r_{p,m})$ were reported for the $\ensuremath{\mathsf{SSF}}_{optimi}$ zedFW (0.17 g/L/h) and SSF_{DWW} (0.17 g/L/h) processes, followed by a slightly lower $r_{p.m}$ for the SSF_{SDEmodified} (0.16 g/L/h) process. The $r_{p.m}$ observed for the present SSF processes may be linked to growth factors such as nutrients or trace elements. Trace elements (manganese, zinc and iron) have significant effects on the production of CA. Manganese is important for cell functioning, sporulation and for the generation of lowmolecular weight secondary metabolites that provide protection from predation, environmental stress, communication, competition for resources and toxicity against other microorganisms (Show et al., 2015), whereas zinc and iron are required for growth and maintenance of the cell. Show et al. (2015) revealed that manganese deficiency negatively influences the anabolism of A. niger, resulting in a high intracellular ammonium concentration. Conversely, when sufficient nutrients are available, Aspergillus sp. germinate by switching from isotropic to polarized growth, which impacts their stress resistance and regulatory pathways involved in germination (Baltussen et al., 2020). Sugar molecules that trigger fungal germination include glucose, mannose and xylose. A carbon source coupled with either inorganic phosphate, magnesium sulphate or inorganic nitrogen are the minimum requirements for the germination of A. niger (Ijadpanahsaravi et al., 2021). A striking characteristic of industrial wastes such as DWW, is the presence of key nutrients such as Mn (2.56 μ g/L), Mg (<0.63 mg/L), Fe (91 µg/L), Zn (21 µg/L), P (6.96 mg/L) and N (36 mg/L) (David et al., 2022), that may have positively influenced the maximum CA production rate within the SSF_{DWW} process. Another key kinetic parameter is the maximum potential CA concentration (P_m) that provides knowledge on the maximum CA that can be produced using these SSF strategies. The SSF_{optimizedFW} process displayed the highest maximum potential CA concentration (P_m) (29.31) g/L, whereas the SSF_{DWW} and SSF_{SDEmodified} processes exhibited lower P_m values of 18.87 and 21.50 g/L respectively. 5778501400048000High glucose utilization and CA concentrations corresponded with the X_{max} data obtained for the $SSF_{\text{optimizedFW}}$ and SSF_{SDEmodified} processes. The lowest P_m value (18.87 g/L) attained from the SSF_{DWW} process may be linked to the lowest X₀ concentration of 0.15 g/L. During CA fermentation, Aspergillus obtain their energy through adenosine triphosphate (ATP) via the TCA (the CA) cycle. The addition of certain rate limiting nutrients such as nitrogen and phosphorus from DWW or a rise in pH may have a detrimental effect on the SSF process by inhibiting the activity of the proton pump in the plasma membrane (H⁺ATPase), thereby reducing the amount of ATP produced. Also, at a high pH, a change in the rate of glucose uptake occurs, thereby resulting in a decreased biomass concentration and organic acid production (Vrabl et al., 2019). Even with the drawbacks of the developed

SSF systems in the present study, these findings may contribute to a more economical and robust operation within lignocellulosic bioprocesses. This is mainly due to the incorporation of DWW as a substitute for fresh water usage, exploitation of banana pseudostem as an alternative to glucose, utilization of ammonium nitrate as a costefficient alternative to conventional peptone and by employing acetone with several beneficial properties during SSF induced CA production. The integration of these key elements could facilitate the biotransformation of waste to valuable bioproducts. A synergy between the academic, industrial and commercial sectors to develop large scale waste-based processes could potentially alleviate the negative environmental effects and economical impacts, which is in line with global sustainable development goals to manage the water, energy and food nexus, and may bring forth the realization of a "waste-to-wealth" circular bioeconomy.

4. Conclusion

This research presents the optimization of citric acid (CA) production using simultaneous saccharification and fermentation (SSF). Optimization gave a maximum CA concentration of 14.408 g/L. Subsequently, the kinetics of cell growth and CA production were determined. The SSF_{SDEmodified} system resulted in the highest maximum potential CA concentration (P_m) (13.991 g/L) contrasted to the SSF_{DWW} ($P_m = 13.095$ g/L) and SSF_{optimizedFW} ($P_m = 12.967$ g/L) bioprocesses. The findings from this research accelerates the development of waste-based bioprocesses that could reduce the economical and environmental burden sustained by commercial industries, while still aligned with the waterenergy-food nexus towards a circular bioeconomy.

CRediT authorship contribution statement

Milesh Laltha: Conceptualization, Methodology, Software, Validation, Formal analysis, Investigation, Data curation, Writing – original draft, Writing – review & editing, Visualization, Project administration, Funding acquisition. Y. Sewsynker-Sukai: Conceptualization, Methodology, Software, Validation, Writing – review & editing, Project administration, Resources, Supervision, Funding acquisition. E.B. Gueguim Kana: Methodology, Software, Validation, Writing – review & editing, Resources, Supervision.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

Acknowledgements

This work is established on research funded partially by the National Research Foundation (NRF) of South Africa (Grant number 122341). Opinions conveyed and conclusions derived, are of the authors and not ascribed to the NRF.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.biortech.2022.127700.

M. Laltha et al.

References

- Aboyeji, O., Oloke, J., Arinkoola, A., Oke, M., Ishola, M.J., 2020. Optimization of media components and fermentation conditions for CA production from sweet potato peel starch hydrolysate by Aspergillus niger. Scientific African 10, e00554.
- Asaduzzaman, M., 2007. Standardization of yeast growth curves from several curves with different initial sizes. Department of Mathematical Sciences. Chalmers University of Technology and Goteborg University, Sweden.
- Ayeni, A., Daramola, M.O., Taiwo, O., Olanrewaju, O.I., Oyekunle, D.T., Sekoai, P.T., Elehinafe, F.B., 2019. Production of CA from the fermentation of pineapple waste by. Open Chem. Eng. J. 13.
- Baei, M.S., Mahmoudi, M., Yunesi, H., 2008. A kinetic model for CA production from apple pomac by Aspergillus niger. Afr. J. Biotechnol. 7.
 Baltussen, T.J., Zoll, J., Verweij, P.E., Melchers, W.J., 2020. Molecular mechanisms of
- Baltussen, T.J., Zoll, J., Verweij, P.E., Melchers, W.J., 2020. Molecular mechanisms of conidial germination in Aspergillus spp. Microbiol. Mol. Biol. Rev. 84, e00049–e00119.
- Behera, B.C., Mishra, R., Mohapatra, S., 2021. Microbial CA: Production, properties, application, and future perspectives. Food Frontiers 2, 62–76.
- Campanhol, B.S., Silveira, G.C., Castro, M.C., Ceccato-Antonini, S.R., Bastos, R.G., 2019. Effect of the nutrient solution in the microbial production of CA from sugarcane bagasse and vinasse. Biocatalysis and Agricultural Biotechnology 19, 101147.
- Chaganti, S.R., Kim, D.-H., Lalman, J.A., Shewa, W.A., 2012. Statistical optimization of factors affecting biohydrogen production from xylose fermentation using inhibited mixed anaerobic cultures. Int. J. Hydrogen Energy 37, 11710–11718.
- Champreda, V., Mhuantong, W., Lekakarn, H., Bunterngsook, B., Kanokratana, P., Zhao, X.-Q., Zhang, F., Inoue, H., Fujii, T., Eurwilaichitr, L., 2019. Designing cellulolytic enzyme systems for biorefinery: From nature to application. J. Biosci. Bioeng. 128 (6), 637–654.
- Clarke, K.G., 2013. An introductory engineering and life science approach. Bioprocess. Eng. 1, 1–266.
- David, A., Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2022. Co-valorization of corn cobs and dairy wastewater for simultaneous saccharification and lactic acid production: Process optimization and kinetic assessment. Bioresour. Technol. 348, 126815.
- Dienye, B.N., Ahaotu, I., Agwa, O.K., Odu, N.N., 2018. Citric acid production potential of Aspergillus niger using Chrysophyllum albidum peel. Advances in Bioscience and Biotechnology 9, 190–203.
- El-Imam, A.A., Du, C., 2014. Fermentative itaconic acid production. J. Biodivers. Biopros. Dev 1, 1–8.
- Ellena, V., Sauer, M., Steiger, M.G., 2020. The fungal sexual revolution continues: discovery of sexual development in members of the genus Aspergillus and its consequences. Fungal Biology and Biotechnology 7, 1–7.
- Fenila, F., Shastri, Y., 2016. Optimal control of enzymatic hydrolysis of lignocellulosic biomass. Resour.-Effic. Technol. 2, 96–104.Gaitonde, V.N., Manjaiah, M., Maradi, S., Karnik, S.R., Petkar, P.M., Paulo Davim, J.,
- Gaitonde, V.N., Manjaiah, M., Maradi, S., Karnik, S.R., Petkar, P.M., Paulo Davim, J., 2017. In: MultiResponse Optimization in Wire Electric Discharge Machining (WEDM) of HCHCr Steel by Integrating Response Surface Methodology (RSM). Woodhead Publishing, pp. 199–221.
- Germec, M., Turhan, I., Karhan, M., Demirci, A.J.a.S., 2019. Kinetic Modeling and Techno-economic Feasibility of Ethanol Production From Carob Extract Based Medium in Biofilm Reactor. 9, 2121.
- Ghanbartabar, S.A., Najafpour, G.D., Mohammadi, M., 2016. Comparative studies on CA production from cheese why by submerged and immobilized Aspergillus niger. Pakistan J. Biotech. 13, 79–85.
- Gogoi, M., Biswas, T., Biswal, P., Saha, T., Modak, A., Gantayet, L.M., Nath, R., Mukherjee, I., Thakur, A.R., Sudarshan, M., 2021. A novel strategy for microbial conversion of dairy wastewater into biofertilizer. J. Cleaner Prod. 293, 126051.
- Hamill, P.G., Stevenson, A., Mcmullan, P.E., Williams, J.P., Lewis, A.D., Stevenson, K.E., Farnsworth, K.D., Khroustalyova, G., Takemoto, J.Y., Quinn, J.P., 2020. Microbial lag phase can be indicative of, or independent from, cellular stress. Sci. Rep. 10, 1–20.
- Ho, H. L., Hood, J. S. 2014. Optimisation of medium formulation and growth conditions for xylanase production by Aspergillus brasiliensis in submerged fermentation (SmF). J Biodivers Biopros Dev, 1, 2376-0214.
- Ijadpanahsaravi, M., Punt, M., Wösten, H.A., Teertstra, W.R., 2021. Minimal nutrient requirements for induction of germination of Aspergillus niger conidia. Fungal Biol. 125, 231–238.
- Islam, M., Asad, M., Hossain, M., Paul, S., Sujan, S., 2019. Bioethanol production from banana pseudostem by using separate and cocultures of cellulase enzyme with Saccharomyces cerevisiae. J. Environ. Sci. Tech. 12, 157–163.

- Kareem, S., Akpan, I., Alebiowu, O., 2010. Production of CA by Aspergillus niger using pineapple waste. Malaysian Journal of Microbiology 6, 161–165.
- Kaur, N., 2021. Different treatment techniques of dairy wastewater. Groundwater Sustainable Dev. 14, 100640.
- Krappmann, S., Braus, G., 2005. Nitrogen metabolism of Aspergillus and its role in pathogenicity. Med. Mycol. 43, 31–40.
- Laltha, M., Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2022. Innovative microwaveassisted iodized table salt or paper wastewater pretreatments for enhanced sugar recovery from banana pseudostem. Biomass Convers. Biorefin. 1–15.
- Maddox, I.S., Hossain, M., Brooks, J.D., 1986. The effect of methanol on CA production from galactose by Aspergillus niger. Appl. Microbiol. Biotechnol. 23, 203–205. Moodley, P., Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2020. Progress in the
- Moodley, P., Sewsynker-Sukai, Y., Gueguim Kana, E.B., 2020. Progress in the development of alkali and metal salt catalysed lignocellulosic pretreatment regimes: Potential for bioethanol production. Bioresour. Technol. 310, 123372.
- Moyer, A.J., 1953. Effect of Alcohols on the mycological production of CA in surface and submerged culture: I. Nature of the Alcohol Effect. Appl. Microbiol. 1, 1–7.
 Odu, N., Uzah, G., Akani, N., 2020. Optimization of CA Production by Aspergillus niger
- Odu, N., Uzah, G., Akani, N., 2020. Optimization of CA Production by Aspergillus niger and Candida tropicalis for Solid State Fermentation Using Banana Peel Substrate. Journal of Life and Bio Sciences Research 1, 51–60.
- Ozdal, M., Kurbanoglu, E.B., 2019. Citric acid production by Aspergillus niger from agroindustrial by-products: Molasses and chicken feather peptone. Waste Biomass Valorization 10, 631–640.
- Papagianni, M., 2007. Advances in CA fermentation by Aspergillus niger: biochemical aspects, membrane transport and modeling. Biotechnol. Adv. 25, 244–263.Papagianni, M., Mattey, M., Kristiansen, B., 1999. The influence of glucose concentration
- Papagianni, M., Mattey, M., Kristiansen, B., 1999. The influence of glucose concentration on CA production and morphology of Aspergillus niger in batch and culture. Enzyme Microb. Technol. 25, 710–717.
- Pramanik, S.K., Suja, F.B., Porhemmat, M., Pramanik, B.K., 2019. Performance and Kinetic Model of a Single-Stage Anaerobic Digestion System Operated at Different Successive Operating Stages for the Treatment of Food Waste. Processes 7, 600.
- Puchalska, P., Crawford, P.A., 2017. Multi-dimensional roles of ketone bodies in fuel metabolism, signaling, and therapeutics. Cell Metab. 25, 262–284.
- Roukas, T., Harvey, L., 1988. The effect of pH on production of citric and gluconic acid: from beet molasses using continuous culture. Biotechnol. Lett. 10, 289–294.
- Sarkar, D., Das, K., 2017. Optimization Of CA Production From Aspergillus Niger Using Pineapple Waste As Feedstock In Submerged Fermentation. World J. Pharm. Res 6, 810–818.
- Sekoai, P.T., Ayeni, A.O., Daramola, M.O., 2018. Parametric optimization of CA production from apple pomace and corn steep liquor by a wild type strain of Aspergillus niger: A Response surface methodology approach. International Journal of Engineering Research in Africa. Trans Tech Publ 98–113.
- Sekoai, P.T., Ouma, C.N.M., Du Preez, S.P., Modisha, P., Engelbrecht, N., Bessarabov, D. G., Ghimire., 2019. Application of nanoparticles in biofuels: an overview. Fuel 237, 380–397.
- Show, P.L., Oladele, K.O., Siew, Q.Y., Aziz Zakry, F.A., Lan, J.-C.-W., Ling, T.C., 2015. Overview of CA production from Aspergillus niger. Frontiers in Life Science 8, 271–283.
- Slavov, A.K., 2017. Dairy wastewaters-general characteristics and treatment
- possibilities-a review. Food Technology and Biotechnology 55, 14–28. Srimachai, T., Nuithitikul, K., O-Thong, S., Kongjan, P., Panpong, K., 2015. Optimization and Kinetic Modeling of Ethanol Production from Oil Palm Frond Juice in Batch Fermentation. Energy Procedia 79, 111–118.
- Tudzynski, B., 2014. Nitrogen regulation of fungal secondary metabolism in fungi. Front. Microbiol. 5, 656.
- Van Soest, P.J., McQueen, R.W., 1973. The chemistry and estimation of fibre. Proceeding of the Nutrition Society 32, 123–130.
- Varga, J., Kocsube, S., Toth, B., Frisvad, J.C., Perrone, G., Susca, A., Meijer, M., Samson, R.A., 2007. Aspergillus brasiliensis sp. nov., a biseriate black Aspergillus species with world-wide distribution. Int. J. Syst. Evol. Microbiol. 57, 1925–1932.
- Veverka, K., Štolcová, J., Růžek, P., 2007. Sensitivity of fungi to urea, ammonium nitrate and their quimolar solution UAN. Plant Prot. Sci. 43, 157.
- Vrabl, P., Schinagl, C.W., Artmann, D.J., Heiss, B., Burgstaller, W., 2019. Fungal growth in batch culture–what we could benefit if we start looking closer. Front. Microbiol. 2391.
- Yadav, V.G., Yadav, G.D., Patankar, S.C., 2020. The production of fuels and chemicals in the new world: critical analysis of the choice between crude oil and biomass vis-à-vis sustainability and the environment. Clean Technol. Environ. Policy 22, 1757–1774.
- Zhang, X., Zhang, B., Miao, R., Zhou, J., Ye, L., Jia, D., Peng, W., Yan, L., Zhang, X., Tan, W., 2018. Influence of Temperature on the Bacterial Community in Substrate and Extracellular Enzyme Activity of Auricularia cornea. Mycobiology 46, 224–235.

Supplementary material

Table S1. Preliminary screening of various desorbents and nitrogen sources to determine their effectiveness for citric acid production.

Screening	Chemical constituent	Input parameters	Citric acid concentration (g/L)
method			
Desorbent	Acetone	$3\% (v/v)^a$, 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	10.886
		144 hr ^g	
	Chloroform	3% $(v/v)^a$, 10% $(w/v)^c$, 10% $(v/v)^d$, 35 °C ^e , 150 rpm ^f ,	7.044
		144 hr ^g	
	Propan-2-ol	3% $(v/v)^{a}$, 10% $(w/v)^{c}$, 10% $(v/v)^{d}$, 35 °C ^e , 150 rpm ^f ,	10.245
		144 hr ^g	
Nitrogen source	Ammonium nitrate	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	12.807
		144 hr ^g	
	Corn steep liquor	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	8.324
		144 hr ^g	
	Peptone	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	9.605
		144 hr ^g	
	Yeast extract	2% (w/v) ^b , 10% (w/v) ^c , 10% (v/v) ^d , 35 °C ^e , 150 rpm ^f ,	7.684
		144 hr ^g	

Footnote: ^a=desorbent concentration, ^b=nitrogen concentration, ^c=substrate solid loading, ^d=inoculum size, ^e=temperature, ^f=agitation speed,

^g=time.

Run	Acetone (v/v)	Ammonium nitrate (w/v)	Temperature (°C)	Citric acid concentration (g/L)	
			-	Megazyme assay	Acid-base titration
1	3	0.50	30	1.615	1.857
2	1	0.50	35	13.219	14.408
3	5	2.75	30	0.631	0.640
4	3	2.75	35	8.842	9.093
5	3	2.75	35	9.090	9.413
6	3	0.50	40	1.048	1.153
7	5	2.75	40	0.639	0.704
8	1	2.75	30	2.177	2.433
9	3	5.00	40	1.090	1.217
10	5	0.50	35	5.920	6.275
11	3	2.75	35	8.206	8.324
12	3	2.75	35	5.341	5.571
13	3	2.75	35	8.950	9.413
14	5	5.00	35	6.412	6.852
15	3	5.00	30	2.110	2.305
16	1	5.00	35	12.467	12.679
17	1	2.75	40	0.546	0.832

Table S2. Citric acid concentration determined for the SSF process using the Megazyme assay kit.

Source	Sum of Squares	Degrees of freedom (<i>df</i>)	Mean Square	F-value	p-value (probability > F)
Model	294.34	9	32.7	7.89	0.0063 Significant
A-Acetone	31.52	1	31.52	7.61	0.0282
B-Ammonium nitrate	0.051	1	0.051	0.012	0.9146
C-Temperature	1.39	1	1.39	0.33	0.5812
AB	1.33	1	1.33	0.32	0.5889
AC	0.69	1	0.69	0.17	0.6948
BC	0.037	1	0.037	8.9×10 ⁻³	0.9275
A^2	1.54	1	1.54	0.37	0.5611
B^2	4.96	1	4.96	1.20	0.3101
C^2	257.17	1	257.17	62.07	0.0001
Residual	29.00	7	4.14		
Lack of fit	18.47	3	6.16	2.34	0.2150 Not significant
Pure error	10.53	4	2.63		
Cor total	323.34	16			

Table S3. Analysis of variance (ANOVA) of the developed SSF model for citric acid production.

 $\overline{Footnote: A = Acetone (v/v), B = Ammonium nitrate (w/v) and C = Temperature (°C).}$

Substrate	Fermentative microorganism	Optimal process parameter conditions	Citric acid concentration (g/L)	References
Banana pseudostem	Aspergillus brasiliensis	$10\% (v/v)^a$, $10\% (w/v)^b$, $1\% (v/v)$ acetone ^c , 0.5%	12.967	This study
(SSF _{optimizedFW})	ATCC16404	(w/v) ammonium nitrate ^d , 35 °C ^e , 150 rpm ^f , 96		
		hr ^g		
Banana pseudostem	Aspergillus brasiliensis	10% $(v/v)^a$, 10% $(w/v)^b$, 1% (v/v) acetone ^c , 0.5%	13.095	This study
(SSF _{Dww})	ATCC16404	(w/v) ammonium nitrate ^d , 35 °C ^e , 150 rpm ^f , 96		
		hr ^g		
Banana pseudostem	Aspergillus brasiliensis	$10\% (v/v)^a$, $10\% (w/v)^b$, $1\% (w/v)$ peptone ^d , 35	13.991	This study
(SSF _{SDEmodified})	ATCC16404	°C ^e , 150 rpm ^f , 96 hr ^g		
Dairy wastewater	Aspergillus niger	150 mL medium mixture ^{a,b} , 30 °C ^e , 200 rpm ^f , 10	4.50	Kim et al. (2002)
	ATCC9142	d ^g , pH 3 ^h		
Banana peel	Aspergillus niger	10 mL ^a , 50 g ^b , 3% (v/v) methanol ^c , 0.5% (w/v)	97.60	Odu et al. (2020)
		ammonium chloride ^d , 30 °C ^e , 10 d ^g , pH 4 ^h , 2%		
		(w/v) zinc ⁱ		
Pineapple waste	Aspergillus niger	$6 \text{ x } 10^6 \text{ suspension}^a$, 30 g^b , $4\% \text{ (v/v) methanol}^c$,	44.10	Sarkar and Das
		0.25% w/v ammonium nitrate ^d , 30 °C ^e , 7 d ^g , 15%		(2017)
		(w/v) sucrose ⁱ		

Table S4. Comparative assessment of citric acid derived from lignocellulosic and dairy wastes utilizing Aspergillus sp.

Substrate	Fermentative	Optimal process parameter conditions	Citric acid	References
	microorganism		concentration (g/L)	
Sweet potato peel	Aspergillus niger	5 d, 6 mm agar spores ^a , 90 mL ^b , 3% (v/v)	4.36	Aboyeji et al.
waste		methanol ^c , 1.25% (w/v) ammonium nitrate ^d , 25		(2020)
		°C°, 7 d ^g , pH 6.5 ^h		
Sugar beet molasse	Aspergillus niger EB12	1 mL ^a , 150 g/L ^b , 4g/L Chicken feather peptone ^d ,	68.80	Ozdal et al.
		30 °Ce, 200 rpmf, 168 hrg, pH 6h, 0.15 g/L		(2019)
		K ₂ HPO ₄ ⁱ		
Cheese whey	Aspergillus niger	50 g/L lactose ^b , 0.5 g/L Ammonium iron (II)	4.45	Ghanbartabar et
biomass	PTCC5010	sulfate ^d , 30 °C ^e , 130 rpm ^f , 6 d ^g , pH 4 ^h , 0.1 g/L		al. (2016)
		MgSO4 ⁱ		
Apple pomace	Aspergillus niger	1 mL ^a , 33.81 g/L ^b , 2.05% (v/v) methanol ^c , 42.5	68.26	Sekoai et al.
		g/L of corn steep liquor ^d , 33 °C ^e , 33 hr ^g , pH 4.54 ^h		(2018)
Sugarcane bagasse	Aspergillus niger	4% (v/v) ethanol ^c , 30 °C ^e , 144 hr ^g , 1% (w/v)	0.73	Campanhol et al.
	CCT4355	sucrose ⁱ		(2019)

 $\overline{Footnote.} \ ^{a}=\text{inoculum size, } ^{b}=\text{substrate amount, } ^{c}=\text{desorbent concentration, } ^{d}=\text{nitrogen concentration, } ^{e}=\text{temperature, } ^{f}=\text{agitation speed, } ^{g}=\text{time, } ^{g}=\text{time,$

^h=pH, ⁱ= chemical additive.
Kinetic parameters	Fermentation process		
	SSF _{optimizedFW}	SSF _{DWW}	SSF _{SDEmodified}
μ_{max} (h ⁻¹)	0.05	0.05	0.05
X _o (g/L)	0.19	0.15	0.21
X _{max} (g/L)	2.79	2.74	2.91
$P_m (g/L)$	29.31	18.87	21.50
$r_{p.m}(g/L/h)$	0.17	0.17	0.16
t _L (h)	16.47	9.80	5.11

Table S5. Kinetic parameters of the microbial growth and citric acid production for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes.

Footnote: μ_{max} = maximum specific growth rate, X_0 = initial cell concentration, X_{max} = maximum cell concentration, P_m = maximum potential

citric acid concentration, $r_{p,m}$ = maximum citric acid production rate, t_L =lag time.

Moles NaOH required = vol of NaOH required (mL) × conc. of NaOH (moles/L) × 10^{-3} (1)

The amount in moles of citric acid, $C_3H_5O(COOH)_3$, in the titrated sample was determined according to Eq. (2).

Moles C₃H₅O(COOH)₃ =
$$\frac{\text{moles NaOH × 1 mol C_3H_5O(COOH)_3}}{3 \text{ moles NaOH}}$$
 (2)

The concentration (molarity) of citric acid produced was calculated by Eq. (3).

Concentration C₃H₅O(COOH)₃ =
$$\frac{\text{moles } C_3H_5O(\text{COOH})_3}{\text{volume } C_3H_5O(\text{COOH})_3}$$
 (3)

The concentration of $C_3H_5O(COOH)_3$ g/L was determined by Eq. (4) and (5).

$$Molar mass = \frac{Concentration (g/L)}{Concentration (mol/L)}$$
(4)

Concentration of citric acid
$$(g/L) = Concentration in mol/L \times Molar mass (g/mol)$$
 (5)



Fig. S1. Aspergillus brasiliensis ATCC16404 glucose consumption over time for the SSF_{optimizedFW}, SSF_{DWW} and SSF_{SDEmodified} processes.